Feature Patterns

Their Sources and Status in Grammar and Reconstruction

A thesis submitted for the degree of Doctor of Philosophy in the University of Cambridge

by

Brett Miller Trinity College

25 May 2012

Joint supervisors: James Clackson, Jesus College Bert Vaux, King's College

© 2012 Brett Miller. All rights reserved.

Abstract

This work investigates feature co-occurrence trends, with special focus on the phonology of the larynx and its interaction with supralaryngeal articulation. My examination of the phonological behavior and phonetic realization of various features, especially [voi], [spr], [cst], and [son], probes the causes behind a wide range of cross-linguistic trends in segment inventory structure, along with the representation of several sound types including breathy-voiced stops and implosives. The most general insight from the roughly eighty patterns investigated is that their specific content can be plausibly derived from non-cognitive factors. This poses a major duplication problem if the same patterns are attributed to innate cognitive stipulations.

I address this problem with a critical review of recent trends in grammar design, arguing for a theory where as much as possible grammatical structure is learned rather than innate and where as much as possible optimization occurs during acquisition rather than during mature speech. After exploring the logical relationships among several major components of existing grammar design theories, I recommend integrating generation with evaluation and configuring grammatical directions in a representational format. I also review research on the interface between linguistics and cognitive science and argue that my proposal is well suited for implementation through distributed, continuous, and highly parallel processing systems.

The main feature co-occurrence insights in the earlier chapters also facilitate a reanalysis of one of the oldest unsolved problems in historical linguistics: the evolution of the Indo-European stop series. In the process of developing a novel solution, I show how the use of features can increase the efficiency and comprehensiveness of reconstruction.

By highlighting the extent to which feature co-occurrence patterns can be derived from independently evident non-cognitive factors, this work contributes to a growing body of research focused on removing innate/learnable duplication problems. While most of the grammatical details of sound systems seem best removed from the innate endowment, I do not conclude that phonetic content and phonological cognition never interact directly. In each of the three main dimensions of this work – developing phonological profiles of specific sound types, exploring implications for grammar design, and historical problem solving – I find evidence that cognition maintains extensive awareness of the phonetic values of the variables it manipulates. This thesis is the result of my own work, neither previously nor currently submitted in whole or part for any other qualification at any other university. It includes nothing which is the outcome of work done in collaboration except where specifically indicated in the text.

Length: 76,774 words excluding bibliography

Table of Contents

| Abstract | 2 |
|--|----|
| Abbreviations | 11 |
| Acknowledgments | 13 |
| 1 Introduction | 14 |
| 1.1 Introduction | 14 |
| 1.2 Features | 14 |
| 1.3 Grammar and sound change | |
| 1.4 Chapter summaries | |
| 1.5 Feature definitions | 21 |
| 2 Laryngeal Phonetics | 24 |
| 2.1 Introduction | |
| 2.2 The larynx | 24 |
| 2.3 Laryngeal articulations of voiceless unaspirated stops | |
| 2.3.1 Utterance-initial prephonation. | |
| 2.3.2 Silent abduction after voiced segments | |
| 2.3.3 Proposed roles of vocal fold tensing | |
| 2.3.3.1 Devoicing | |
| 2.3.3.2 Glottal widening and pitch raising | |
| 2.3.4 Voiceless unaspirated stops: conclusion | |
| 2.4 Laryngeal constriction and whisper | |
| 2.5 Breathy voice | |
| 2.5.1 Introduction | |
| 2.5.2 Whispery and strict breathy voice: articulation | 41 |
| 2.5.3 Voiced aspirates | |
| 2.5.4 Breathy voice: cues | 44 |
| 2.5.5 Slack voice | |
| | |

| 2.5.5.1 General | 46 |
|---|--|
| 2.5.5.2 Voiceless slack stops occurring initially and medially | 49 |
| 2.5.5.2.1 Javanese | 49 |
| 2.5.5.2.2 Xhosa | 50 |
| 2.5.5.2.3 Zulu | |
| 2.5.5.2.4 An Eastern Armenian speech variety | 60 |
| 2.5.5.2.5 Conclusion | 61 |
| 2.5.5.3 Voiceless slack stops with voiced medial allophones | 61 |
| 2.5.5.3.1 Korean | 61 |
| 2.5.5.3.2 Wu Chinese | 66 |
| 2.5.5.4 Voiceless slack stops and voiced stops in phonetic contrast | 68 |
| 2.5.6 Conclusion | 69 |
| 2.6. Articulatory strategies for maintaining closure voicing | 70 |
| 2.6.1 General | 70 |
| 2.6.2 Non-explosion | 75 |
| 2.7 Conclusion | 78 |
| | |
| | |
| 3 Laryngeal Phonology | 79 |
| 3 Laryngeal Phonology 3.1 Introduction | 79 79 |
| 3 Laryngeal Phonology3.1 Introduction3.2 Economy and enhancement | 79 79 79 |
| 3 Laryngeal Phonology 3.1 Introduction 3.2 Economy and enhancement | 79 79 79 79 |
| 3 Laryngeal Phonology 3.1 Introduction | 79 79 79 79 81 |
| 3 Laryngeal Phonology | 79 79 79 79 81 82 |
| 3 Laryngeal Phonology | 79 79 79 81 82 89 |
| 3 Laryngeal Phonology | 79 79 79 81 82 89 89 |
| 3 Laryngeal Phonology | 79 79 79 81 82 82 89 89 91 |
| 3 Laryngeal Phonology | 79 79 79 79 81 82 82 89 91 92 |
| 3 Laryngeal Phonology | 79 79 79 81 82 89 89 91 92 93 |
| 3 Laryngeal Phonology | 79 79 79 81 82 89 91 91 92 93 94 |
| 3 Laryngeal Phonology | 79 79 79 79 79 82 82 82 82 89 91 92 91 92 93 94 94 |
| 3 Laryngeal Phonology | 79 |

| 3.2.6.4 Kelabit and Lun Dayeh | 101 |
|---|-----|
| 3.2.6.5 Conclusion | 104 |
| 3.2.7 Diachronic ejective voicing | 105 |
| 3.3 Features | 107 |
| 3.3.1 Introduction | 107 |
| 3.3.2 Phonetically natural and unnatural features | 108 |
| 3.3.3 Feature acquisition | 112 |
| 3.3.4 Universal constraints on feature-correlate mapping | 115 |
| 3.3.5 Modeling phonetic variation | 125 |
| 3.3.6 Representing aspiration | 126 |
| 3.3.6.1 [spread glottis] | 126 |
| 3.3.6.2 [heightened subglottal pressure] | 127 |
| 3.3.6.3 VOT and timing in phonological representations | 128 |
| 3.3.6.4 Auditory definitions: intensity, duration, and turbulent noise | 131 |
| 3.3.6.5 Broader issues | 133 |
| 3.3.6.6 Conclusion | 135 |
| 3.3.7 Representing voice | 135 |
| 3.3.7.1 Articulatory or acoustic [voice] | 135 |
| 3.3.7.2 Abstract low-frequency [voice]; slack and stiff stops | 136 |
| 3.3.7.3 [stiff], [slack], and larynx height | 138 |
| 3.3.7.4 Split treatments: Sonorant/Laryngeal Voice and underspecification | 142 |
| 3.3.7.5 Conclusion | 146 |
| 3.3.8 Feature ambivalence and sonorance | 146 |
| 3.3.9 Laryngeal ambivalence in Germanic | 151 |
| 3.4. Beyond Feature Geometry trees: larynx-stricture interactions | 153 |
| 3.5 Comparison of enhancement with partial positional neutralization | 156 |
| 3.6 Conclusion | 158 |
| 4 Feature Co-Occurrence Trends | 161 |
| 4.1. Introduction | 161 |
| 4.2. Feature co-occurrence trends | 161 |
| 4.3. Explanations | 164 |

| 4.3.1 Place | 164 |
|--|-----|
| 4.3.1.1 Coronals | |
| 4.3.1.2 Rounding | 167 |
| 4.3.1.3 Liquids and nasals | 167 |
| 4.3.1.4 Uvulars, laryngeals, and clicks | 168 |
| 4.3.2 The larynx and aerodynamics | 170 |
| 4.3.2.1 General | 170 |
| 4.3.2.2 Nasals and liquids | 171 |
| 4.3.2.3 Fricatives | 173 |
| 4.3.3 Vowels | 174 |
| 4.3.4 Interim reduction | 176 |
| 4.4. Implications | |
| 4.4.1 Introduction | 178 |
| 4.4.2 Why sound change is not teleological | 178 |
| 4.4.3 Why phonology is not dysfunctional | |
| 4.5. Conclusion | |
| 5 Grammar | |
| 5.1 Introduction | |
| 5.2 Grammar | |
| 5.2.1 Definition | |
| 5.2.2 Development | |
| 5.3 Optimality Theory and its classic schema | |
| 5.4 Necessary parallelism? | |
| 5.4.1 Imdlawn Tashlhiyt Berber | |
| 5.4.2 Tongan | |
| 5.3.3 Lardil | |
| 5.4.4 Southern Paiute, Yidin, and Axininca Campa | |
| 5.4.5 Southeastern Tepehuan | |
| 5.4.6 Review | |
| 5.5 The architecture of phonology | |
| 5.5.1 Grammars differ in content, not just its permutation | |

| 5.5.2 Users optimize grammars, not just words (and not typologies) | 204 |
|---|-----|
| 5.5.3 Grammars don't handle the rich base | 207 |
| 5.6 Generalized partial representations | 208 |
| 5.6.1 Introduction | 208 |
| 5.6.2 Illustrations: epenthesis and spirantization | 211 |
| 5.6.3 Underspecification and multi-modal cognition | 215 |
| 5.6.4 Feeding | 221 |
| 5.6.5 Counterfeeding | 223 |
| 5.6.6 Bleeding and counterbleeding | 226 |
| 5.6.7 Do something only/except when | 229 |
| 5.6.8 Summary | 231 |
| 5.7 Output-drivenness | 233 |
| | |
| 6 Reconstruction | 236 |
| 6.1. Introduction | 236 |
| 6.2. The Proto-Indo-European stop system | 237 |
| 6.2.1 Comparative evidence | 237 |
| 6.2.2 A new proposal | 239 |
| 6.2.3 Place of articulation issues | 239 |
| 6.3. Diachronic outcomes | 240 |
| 6.4. Comparison with other reconstructions | 243 |
| 6.4.1 Other kinds of voiced stops | 243 |
| 6.4.2 Ejectives | 244 |
| 6.4.3 Fricatives | 246 |
| 6.4.4 Conclusion | 247 |
| 6.5. The labial "gap" | 247 |
| 6.5.1 Critical typology of reconstructions | 247 |
| 6.5.1.1 PIE had a $/D^{(<)}/$ series with the bilabial unusually rare or absent | 248 |
| 6.5.1.2 PIE had a bilabial gap, but it was natural | 249 |
| 6.5.1.3 Traditional PIE /b/ merged with some other segment(s) before PIE. | 250 |
| 6.5.2 Merger of traditional PIE /b/ with /b ^ĥ / after PIE | 252 |
| 6.5.2.1 Introduction | 252 |

| 6.5.2.2 The simpler cases | 253 |
|---|-----|
| 6.5.2.3 Armenian and Germanic | 253 |
| 6.5.2.4 Italic | 257 |
| 6.5.2.5 Indic | 258 |
| 6.5.2.6 Greek | 259 |
| 6.5.2.7 Conclusion | 262 |
| 6.6. Alleged voiceless aspirate etyma | 263 |
| 6.7. Conclusion | 264 |
| 7 Conclusion | 267 |
| Bibliography | 268 |
| Figures | |
| 2.1 Glottal width continuum | 24 |
| 2.2 Cricoid rotation during larynx lowering | 48 |
| Zulu spectrograms | |
| 2.3 bhabha [baːba] 'trap' | 56 |
| 2.4 <i>baba</i> [ba:ba] 'father (direct address)' | 56 |
| 2.5, 2.6 <i>idada</i> [i ^l dada] 'duck' | 57 |
| 2.7, 2.8 ukuguga [uguˈġuġa] 'to grow old' | 58 |
| Hindi spectrograms | |
| 2.9 ata [a:ta] 'he comes' | 59 |
| 2.10 mata [ma:ta] 'mother' | 59 |
| 2.11 Korean inter-speaker variation in stop-conditioned vowel phonation | 62 |
| 2.12 Korean coronal stop closure durations | 65 |
| 3.1 Phonetic geometry (partial) | 118 |
| 3.2 Suggested maximum number of features assignable to the correlates shown | 120 |
| 3.3 Association of some correlates to some features in Hindi stops | 122 |
| 3.4 Example of phonetic correlate assignment to [slack], [spr], and [voi] | 123 |

Tables

| 1.1 Some features with working definitions | 22 |
|--|-----|
| 1.2 Features and definitions (continued) | 23 |
| 2.1 Voiceless slack stop types | 47 |
| 2.2 Phonetic differences between Korean initial fortis and lenis stops | 66 |
| 3.1 Stop inventories with enhanced voice series | 92 |
| 3.2 Preponderantly voiced stop inventories | 94 |
| 4 Constraint set adapted from Calabrese (1992) | 163 |
| 6.1 Comparative evidence for the PIE stops | 237 |
| 6.2 Diachronic trends | 241 |

Abbreviations

| Alb. | Albanian |
|-------------------|--|
| Arm. | Armenian |
| BS | Balto-Slavic |
| С | Celtic |
| cf. | compare with |
| f. | and the next section |
| G | Greek |
| Gm. | Germanic |
| Hitt. | Hittite |
| IE | Indo-European |
| In. | Indic |
| Ir. | Iranian |
| It. | Italic |
| L | Latin |
| Myc. | Mycenaean Greek |
| n. | note |
| n.d. | no date |
| OT | Optimality Theory |
| p. | page |
| p.c. | personal communication |
| PIE | Proto-Indo-European |
| pp. | pages |
| Sk. | Sanskrit |
| TH | Tiberian Hebrew |
| Toch. | Tocharian |
| VOT | voice onset time |
| # | word boundary |
| \$ | syllable boundary |
| * | penalized; non-occurring; reconstructed |
| Ť | rejected (constraint) |
| > | becomes (in sound changes) |
| < | comes from (in sound changes); less than (in rankings); non-explosive |
| | (when superscripted) |
| \rightarrow | maps to |
| \leftarrow | maps from |
| \leftrightarrow | maps to/from (no directional restriction expressed) |
| \sim | alternates with |
| : | contrasts with |
| () | optional elements (in a string) (other uses are defined locally in the text) |

| [] | phonetic (if enclosing segmental symbols); feature (if enclosing feature |
|-----|--|
| | abbreviations); phonological output forms not necessarily containing |
| | only contrastive information |
| // | phonemic, contrastive |
| { } | set |

Acknowledgments

Navigating the interfaces between phonology and phonetics, synchrony and history, theoretical universals and the deepest, darkest details of Armenian would have been impossible without the combined guidance of experts in several disciplines. James Clackson, Bert Vaux, and my examiners Francis Nolan and Jane Stuart-Smith provided the most intense and rewarding linguistic conversations and feedback I have ever had.

Jeff Mielke generously read and discussed the sections where I quote him most. Steve Parker and two anonymous reviewers helped me transform a chapter for early publication, and Patricia Shaw offered unexpected encouragement in the process. Robert Blust, Charles Barrack, Matthew Gordon, Johanna Nichols, and Philomen Probert graciously provided feedback on other draft sections. Other linguists who answered questions or helped me access material include Luca Busetto, Marc Brunelle, François Dell, Didier Demolin, Adamantios Gafos, William Idsardi, Máté Ittzés, Gregory Iverson, Ranko Matasović, Joyce Mathangwane, Pramod Pandey, Joe Pittayaporn, Rachid Ridouane, Catherine Ringen, David Rood, Joseph Salmons, Bridget Samuels, Rebecca Scarborough, Paul Sidwell, Kenneth Stevens, Rupert Thompson, and Michael Weiss. I am grateful for all their help.

Susan Prince instigated this project as a conference paper. The excellent hosts of GLIEP I at Cambridge later edited the proceedings for the University Press. The College of the Holy and Undivided Trinity provided funding and an unparalleled living environment for the next three years. Since then I have owed particular thanks to David Rood and Lise Menn for ongoing help and inspiration, and to Norlin Library for special access to resources.

Among friends I owe Jesse Bangs for nearly two decades of brilliance and inventive insight in all things linguistic including several discussions of this project, and my brother Jason for troubleshooting and rebuilding my laptop. The David, Daniel, and Joe Rice families, Jerusha and Kenny Pretty On Top, Derek Schreinert, Susie Siders, Josiah Bigelow, Gary and Marvia Robbins, Chuck and Kay Rice, the Buford King family, Dean Ness, Phil Klotzbach, Dustin Rapp, John Young, Robert Patterson, Alex Ievins, Drew and Stacy Frazer, Brian and Heather Hanneman, Josh Dillard, Aaron and Jody Britton, Alec Battles, Cole McGinnis, and my coworkers at the Erie Rec Center enriched my life beyond the page.

My parents Brad and Jeanne Miller have proven indispensible friends and helpers. I honor my master Jesus Christ, First and Last, Breath of Life, for holding everything together and for being the most faithful.

1 Introduction

1.1 Introduction

This chapter begins with some basic terminology about features, grammar, and sound change. Summaries of later chapters follow.

The thesis of this work is that the particular properties of phonological features and their combinatorial behaviors emerge through language acquisition and sound change thanks to the way non-cognitive factors such as acoustic physics and vocal tract physiology interact with general cognitive abilities that may also be relevant to nonlinguistic learning activities. Stipulating either features or their co-occurrence patterns as part of innate human linguistic knowledge is therefore redundant. Major implications for theories of grammar design and sound change are explored with attention to four topics: how features come to exist in grammars, how grammars manipulate features, how information about the features of prehistoric languages can be inferred from attested descendant languages, and how important such information is for understanding prehistoric phonologies.

1.2 Features

Phonetics is the study of sounds in general; phonology studies the way sounds are organized in particular languages. Fairly standard, traditional introductions to these respective topics include Ladefoged and Maddieson (1996) and Kenstowicz (1994). New discoveries and proposals are constantly emerging, and many of them are discussed in this work.

One of the main insights of modern phonology is that distinguishable sounds or *segments* can be analyzed into grammatically significant components called *features*. Features cross-classify segments, defining *contrasts* between different segments and defining *classes* of segments based on shared traits, such as participation in a grammatically significant sound pattern. Because features play these two roles, it is natural that many linguists have also used features to describe the differences and similarities among languages.

In order to signal differences and similarities between segments in speech, features must relate to aspects of sound that speakers can produce and listeners can hear. This makes it important to study the articulatory and acoustic correlates of features (also known as *gestures* and *cues* respectively). Yet features themselves are not simply actions of the speaker's vocal tract or aspects of the sound wave traveling through the air to the listener. In order to carry meaning, features must also exist in the minds of language users. Linguists are consequently interested in features as elements of the user's mental *representation* of speech. There is ongoing debate about how these elements are translated into articulation during speech production and back out of the acoustic signal by the auditory component during speech processing; for a recent overview see Clements and Hallé (2010).

Studying the role of features in defining phonological classes is important for understanding how speakers transform mentally stored representations of parts of words into seamless spoken utterances. For example, the most common English plural marker is realized as *s* after the segments $p \ t \ k \ f \ \theta$ and with a *z* sound in other environments. The productivity of this pattern has been repeatedly proven by experiments like the *wug* test, where young children shown pictures of an imaginary animal called a *wug* were consistently able to infer that the plural was pronounced *wug*[*z*], not **wug*[s] (Berko 1958). Rather than describing this pattern by listing all the segments separately, we can describe it more simply with the insight that the plural marker ends up being voiceless after voiceless sounds and voiced elsewhere: in other words, we can distinguish these classes of sounds with a single feature [voice]. While that change may not seem to simplify this particular example much, the power of features is more dramatically evident in some polysynthetic languages, where speakers more often find themselves forming very long words (equivalent to whole English sentences) which they may not have heard before and in which there are much more complex relationships among the sound patterns (e.g. Blackfoot, Menomini, Wichita: §5.7). Speakers also betray some of their own abstract generalizations about the sound patterns of their language in how they adapt loanwords from other languages (Pierrehumbert 2001: 138-139).

Turning to the second major function of features, we say that features are *contrastive* when a change in feature can change the meaning of a word. For example, the words *met* and *bet* can be distinguished by noting that the segment *m* has the feature [+nasal] while *b* does not. Seminal discussions of this topic include Jakobson, Fant, and Halle (1954) and Jakobson and Halle (1956). One of the related issues they address is when the distribution of one feature can be predicted from that of another. The predictable relationship can be either complementary or redundant; two examples will illustrate the difference.

In English, the segment p is typically followed by a puff of air at the beginning of a stressed syllable unless preceded by s. If we represent this puff of air with a feature – say, [+spread glottis] or [+spr] for short (§3.3.6) – then in English, we can predict that sounds like p are [+spr] at the beginning of a stressed syllable unless preceded by s. For such sounds in those environments, the plus and minus values of [±spr] are *allophonic*, meaning that they represent variant sounds found in predictable, complementary distribution. To illustrate the second type of predictable relationship between features: b typically has air pressure buildup in the mouth behind the closed lips, while m does not because the air escapes through the nose. If we were to represent lack of significant air pressure buildup with a feature – say, [+sonorant] (§3.3.8) – then we would find that in English, [+nasal] sounds are also [+sonorant], while [-sonorant] sounds are also [-nasal]. In such cases, we might say that only one of the features encodes differences in meaning,

while the other is *redundant*. When a redundant feature appears to make the contrast expressed by another feature more perceptible, the redundant feature is sometimes called an *enhancement*, following Stevens and Keyser (1989, 2010).

There is ongoing debate about how to decide which member of a cooperative feature pair is the redundant one, and when to leave predictable features out of a mental representation of speech (underspecification). These debates relate to a broader one about how abstract features are in the minds of ordinary language users. In the examples above, we identified features directly with articulatory configurations that control airflow in particular ways (nasality, oral air pressure buildup, glottal widening). How are such details stored in the mind? When a speaker is retrieving feature representations of words from memory and beginning to manipulate them in order to plan an utterance, are the features just abstract mental elements – which only get translated into specific phonetic content later, by a subsequent and separate *module* of the speaker's grammar – or are language users ever more aware of phonetic content, and can this awareness influence other areas of linguistic knowledge during language development (including historical sound change)? In turn, the debate on feature abstractness relates directly to the question of how to compare languages: how similar must a sound in one language be to a sound in another language in order for them to be defined with the same feature? Two major discussions of these topics can be found in Chomsky and Halle (1968: 295-298) and Ladefoged (1999).

Another question closely related to the ones just mentioned is whether features are innate. A fairly small, innate set of features universally shared by healthy human beings might be quite abstract; at the other extreme, if features are not innate but *emerge* by induction as the learner analyzes patterns heard during language acquisition, then some features may be much more closely tied to phonetics, and different speakers of the same language might even induce features with different definitions (Brunelle 2010, Mielke 2008a). At several points in this work I argue for the emergent view of features, as well as a not discretely modular view of the relationship between mental feature representations and phonetic awareness. This is supported by recent work in psycholinguistics (e.g. Menn 2010, Vihman et al. 2009), research on phonetic differences among speakers of the same language (e.g. Brunelle 2005, 2010, Wayland and Jongman 2003), and other topics that will be introduced later. Only some of my arguments depend on emergent features, though, and I continue discussing alternative implications of emergent and innate approaches throughout the thesis.

1.3 Grammar and sound change

A grammar can be defined as a person's systematic knowledge needed for communicating in a language by converting mental information to physical speech and vice versa (cf. Ussishkin 2011). Sound patterns that a person must understand in order to use a language are part of the grammar. Exactly how features are used in grammar depends on one's theory of grammar design. One of the dominant early paradigms was that speakers perform a series of operations called *rules* on an input representation in a certain order, and the output is then pronounced. The rules are often stated in terms of features; a seminal example of such work is Chomsky and Halle's (1968) analysis of English. More recently, linguists have experimented with groups of *constraints* that weed out incorrect but theoretically possible output forms for a given linguistic input. Again, the constraints are often expressed using features. The best-known pioneering example of this approach is Prince and Smolensky's (1993) Optimality Theory. Recent comparisons of rule- and constraint-based approaches include Odden (2011) and Vaux and Nevins (2008).

Sound change refers to how the systematic sound patterns of a language change over time (for introductions see Bermúdez-Otero 2007 and Fox 1995). Because of this systematic aspect, models of sound change are inextricably linked to ideas about how grammars are structured and how they are learned. Based on the idea that grammars encode sound patterns through series of rules, it was popular for a time to model at least some sound changes as the reordering, addition, or deletion of rules (Picard 1994: 33-38, van der Hulst 1980). On the alternative supposition that grammars encode at least some sound patterns by ranking innate constraints, it makes sense to model at least some sound changes as changes in constraint ranking (Holt 2003, McMahon 2000) or constraint activation (Calabrese 1992, 1995).

A third approach to sound change, promoted in this work, is emergent: the detailed contents of grammars are induced afresh by each learner, and sound change occurs when learners conventionalize novel analyses of the data they observe (e.g. Blevins 2004, Ohala 1981b, 1983). Hale (2007: 33-35) compares this approach to earlier models via an analogy of grammar as a rock: sound change is not like a rock rolling down a hill, but rather like a sequence of people observing the rock left by the previous person and trying to set beside it a rock which looks more or less the same. Misperception and novel generalizations thus play a central role in why learners induce different grammatical structure from their predecessors (Blevins 2004, Mielke 2008a, Ohala 1989, Stuart-Smith 2004). Learners' understanding of their languages remains malleable into adolescence and beyond (Kerswill 1996, Sankoff and Blondeau 2007).

1.4 Chapter summaries

The larynx plays an important role in many sound patterns. Chapter 2 reviews evidence on several laryngeal gestures and cues, focusing on the roles of vocal fold abduction and tensing in voiceless unaspirated stops, strategies for sustaining voice during stop closure, and different kinds of breathy voice.

Chapter 3 uses this information to develop an emergent understanding of laryngeal features and the kinds of contrastive and redundant behaviors they show, with special attention to enhancements of voiced stops and the characteristics of stop inventories in which more than half of the contrastive series are voiced. A well-known distributional restriction on breathy-voiced stops ("Jakobson's universal") is reviewed and reinterpreted in light of a principle of feature economy interacting with enhancement. Relationships between economy, enhancement, and contrast neutralization are briefly examined. More broadly, several arguments are synthesized to support the view that features emerge through the learning process rather than being innate.

In this view, learners induce features by identifying groups of cues as patterning together in a functionally significant way for the communication of meaning. Learners often diverge in the details of which cues they identify for this purpose and which gestures they identify as appropriate for producing those bundles of cues; this results in phonetic variation among speakers of the same language. Physiology and physics create natural relationships among phonetic correlates; these relationships ("phonetic geometry") influence cross-linguistic trends in feaure-correlate mapping and limit the maximum number of features that a speaker can distribute to a given set of cues.

Chapter 4 shifts focus from features to cross-linguistic trends in the combinatorial behavior of features, arguing from over seventy examples that feature co-occurrence trends can be plausibly explained as emerging through sound change, whether features themselves are learned or innate. While respecting arguments that sound change is not goal-oriented, I argue that many conceivable patterns are rare because they are needlessly complex and that attrition of existing omissible complexity plays a role in sound change due to articulatory negligence, especially in less careful speech styles.

Chapter 5 applies the findings of chs. 3-4 to the question of how grammars are designed, arguing that features and feature co-occurrence trends, as well as evidence from psycholinguistics and language acquisition, support a theory of grammar design in which grammars differ in their contents, not just in how the contents are organized. A framework for modeling grammar design is proposed in which grammatical information about sound patterns is encoded in partial representations which are combined with lexically stored representations of idiosyncratic information about particular linguistic expressions to create a total representation of what speakers need to know in order to utter those expressions. Such a total representation or parts of it may then be remembered

for various lengths of time without being reassembled every time the expression is repeated.

Chapter 6 applies the foregoing insights about laryngeal phonetics, features, feature co-occurrence trends, and grammar design to the topic of historical reconstructive methodology with a detailed case study of the laryngeal stop contrasts in Proto-Indo-European. My critique of past proposals and new possibilities illustrates the importance of reconstructing the phonological categories of prehistoric languages in accordance with natural physical relationships among phonetic correlates. These relationships constrain the ways in which successive generations reanalyze a language's sound system and thus function as guideposts for the linguist trying to retrace those analytical steps – the paths of sound change.

1.5 Feature definitions

Features mentioned repeatedly in this thesis are listed in Tables 1.1 and 1.2 with working definitions couched in terms of phonetic correlates. The features and most of the definitions reflect fairly standard views (e.g. Hall 2007). Exceptions and relevant controversies will be handled as they arise; most of those discussions are indexed in the tables.

As a final comment on terminology, features are notationally binary throughout this work rather than privative; e.g. the opposite values of [voi] are written [+voi] and [-voi] rather than [voi] and nothing. This is simply to facilitate clarity and implies nothing about the phonological activity or inactivity of features or about default vs. nondefault values. Generalizations on those topics will be identified separately. For a more integrated discussion of gradient and heterogeneous diagnostics of feature activity see Mielke (forthcoming a). Most of the features discussed in much detail here have correlates that can be plausibly viewed as either present or absent, making binary feature treatment relatively straightforward. Some features have been defined with more than two values for phonetically obvious reasons, like vowel height (Clements and Hume 1995), or accused of having more than two values for abstract phonological reasons such as archiphonemic underspecification (Samuels 2009a: 81, 95-98); most of those topics are not addressed here (see Vaux 2010).

| Feature | Abbr. | Definition | |
|-----------------------|---------|---|--|
| [consonantal] | [cons] | active supralaryngeal constriction greater than that of vowels and (usually) glides (§3.3.8) | |
| [sonorant] | [son] | relatively low oral air pressure (") | |
| [continuant] | [cont] | (conventionally oral) egressive airflow (") | |
| [voice] | [voi] | periodic vocal fold movement producing pitch (§3.3.7) | |
| [spread glottis] | [spr] | glottal turbulent noise (§3.3.6.4) | |
| [constricted glottis] | [cst] | true and false vocal fold adduction and often narrowing of the laryngeal vestibule (§2.4, 2.6.2) | |
| [nasal] | [nas] | more lowered velum and nasal resonance than in oral sounds (§2.3.1, 4.3.2.2) | |
| [lateral] | [lat] | airflow out of the mouth over lowered tongue sides but not over the tongue center | |
| [rhotic] | [rhot] | lingual trills, taps, or flaps; glides with similar articulatory vectors (§4.3.1.3) | |
| [strident] | [strid] | frication with heightened turbulence (§4.3.2.3) | |
| [labial] | [lab] | actively articulated with one or both lips | |
| [round] | [rd] | adducting and protruding the lips (§4.3.1.2) | |
| [coronal] | [cor] | articulated with roughly the front half of the tongue (§4.3.1.1) | |
| [anterior] | [ant] | the further forward of two options (") | |
| [distributed] | [dist] | a more plane- than point-like articulation (") | |
| [dorsal] | [dor] | articulated with roughly the back half of the tongue | |

Table 1.1 Some features with working definitions

Several other proposed features have definitions that are either relatively clear from their names, or too complex or debatable to be listed here:

| Table 1.2 Features and definitions (continued) | | | | | | |
|--|------------------|----------------------|--|--|--|--|
| Feature | Abbr. | see | | | | |
| [advanced/retracted tongue root] | [ATR] | §2.4, 4.3.1.4, 4.3.3 | | | | |
| [tense] | [tns] | §3.3.6.4 | | | | |
| [raised/lowered larynx] | [RL/LL] | §3.3.7.3 | | | | |
| [stiff/slack vocal folds] | [stiff] | ", §3.3.7.2 | | | | |
| [high], [low], [back] | [hi], [lo], [bk] | §4.3.3 | | | | |

Table 1.2 Features and definitions (continued)

2 Laryngeal Phonetics

2.1 Introduction

This chapter explores the acoustic and aerodynamic effects of the interaction between laryngeal articulation and stricture (supraglottal constriction), with special focus on voiceless unaspirated stops, different kinds of breathy voice, and active voicing in stops. The findings of this chapter help illustrate how phonological categories – the grammatical sound structures of particular languages – are built within a universal, gradient space of phonetic possibilities determined by physics and physiology, such that asymmetries within the universal phonetic space extensively influence the patterning of features within the phonologies of particular languages.

2.2 The larynx

The larynx can make a variety of sounds. One way of organizing these sounds is to arrange them in a continuum of glottal width, where the overall size of the glottis (the opening in the larynx) increases in one direction and decreases in the other (see Gordon and Ladefoged 2001, Henton, Ladefoged, and Maddieson 1992, Ladefoged 2001, Ladefoged and Maddieson 1996):

| widest | ← | | | | \longrightarrow | narrowest |
|-----------|---------|-------|-------|-------|-------------------|-----------|
| wide-open | breathy | slack | modal | stiff | creaky | closed |

Figure 2.1 Glottal width continuum.

The sources just cited refer to the wide-open and closed states at the two ends of the continuum as voiceless while the five states between them are described as having voice (defined next). This is accurate for many but not all sound types: roughly the same degree of glottal width is compatible with both voice and voicelessness depending on stricture (\S 2.3.2), both the slack and stiff configurations are reported in voiceless stops (\S 2.5.5), and the glottal width range compatible with modal voice is only accompanied by actual voice depending on aerodynamic factors (\S 2.6).

Modal voice, the middle state in this continuum, can be thought of in most languages as a person's normal speaking voice; studies of various languages have found voiced sounds occurring more frequently in ordinary speech than voiceless sounds do (Catford 1977: 107, Vaissière 1997: 115-116). The standard theory of voice production is called the myoelastic-aerodynamic theory of voicing and can be summarized in the following way (see Hirose 2010: 140-141, Hoole 2004, Lisker and Abramson 1964: 415-416, Shadle 2010: 56, Stuart-Smith 2004: 167). The vocal folds are first adducted into close enough proximity that pulmonic egressive air pressure builds up behind them. Air is forced to speed up as it passes between the vocal folds, due to the Venturi effect (which correlates a constriction in a passage with accelerated movement of a fluid or gas forced into the passage). At the same time, the accumulated air pressure behind the vocal folds causes them to begin separating, increasing the volume of air moving at an accelerated speed through the widening glottis. This rapid escape of air reduces air pressure between the vocal folds by the Bernoulli effect (which correlates accelerated movement with reduced pressure). When the air pressure has decreased enough, the greater air pressure around as opposed to between the vocal folds along with the natural tension of the vocal folds themselves forces them back together again. Then the cycle repeats.

This rapid, quasi-regular emission of small bursts of air from the glottis due to the opening-closing cycle of the vocal folds is heard as voice, a relatively homogeneous sound with pitch. Mechanical oscillations are also possible with other vocal tract constrictions, including trilling of the lips, tongue tip, uvula, aryepiglottic folds, and

ventricular folds (also called the false vocal folds, located within the larynx slightly above the vocal folds) (Edmondson et al. 2007, Shadle 2010: 56). These oscillations are sufficiently slower than voice that they are not perceived as a single sound (Brosnahan and Malmberg 1976: 54); compare the frequency of most trills at 20-35 Hz with the frequency range for voice, from around 40 Hz for creaky voice to 1000 Hz for a soprano or child's voice (Shadle 1999: 50).

The range of glottal widths we have just seen is controlled by multiple muscles acting in concert (DiCanio 2008: 180-187, Hirose 2010: 139-140, Hunter and Titze 2007). The vocal folds are attached to a point at the front of the throat below the epiglottis; their posterior ends, toward the back of the throat, are attached to two movable cartilages called the arytenoids. The primary muscle that abducts, or widens, the glottis is the posterior cricoarytenoid (PCA), which attaches to the arytenoids and rotates them apart when it contracts. Cricothyroid (CT) contraction can also slightly increase the size of the glottal opening by elongating it (§2.3.3.2). Another set of muscles is able to adduct the vocal folds, bringing them together and narrowing the glottis. The interarytenoid (INT) draws the arytenoids together, the lateral cricoarytenoid (LCA) rotates the arytenoids to compress the vocal folds medially (cf. Gobl and Ní Chasaide 2010: 396), and the thyroarytenoid (TA) draws the arytenoids forward to compress the body of the vocal folds by shortening them. This last action by itself relaxes the covering of the vocal folds, but the cover can be separately stiffened by the vocalis (VOC) muscle, which is inside the TA and can be contracted separately. (See Berry 2001 on ways in which the body/cover two-mass model of the vocal folds appears to be a categorical simplification of a more gradient reality.) All these muscles are classified as intrinsic laryngeal muscles, meaning that they do not extend beyond the larynx. Extrinsic laryngeal muscles, which include the suprahyoid and infrahyoid, connect the larynx to other structures in the throat and play a role in raising and lowering the larynx (e.g. Harris 1981: 21-22, Ohala 1972).

Glottal widening and narrowing are opposite gestures. PCA and INT in particular are said to be reciprocally active and mutually antagonistic, meaning that their activities

directly conflict; each is normally active only to the extent that the other is not (Hirose 2010: 142, Hirose and Gay 1973, Sawashima 1997: 71). This norm is not always maintained. Stuttering typically involves "simultaneous, presumably antagonistic, abductor-adductor activity" along with increased LCA activity (Freeman and Ushijima 1974: 110-111; 1975, 1976). Further, Löfqvist and Yoshioka (1979: 117, 120) found that while PCA triggered abduction during voiceless obstruent clusters as expected, the waning phase of these gestures (the phase of return to an adducted position) did not always correlate with greater INT activity, suggesting that PCA and INT activity are not always inversely correlated even in healthy speech.

Activity of other intrinsic laryngeal muscles is often lower in consonants than in vowels. In a Hindi speaker (Kagaya and Hirose 1975: 32-36), VOC began decreasing around 30-40 ms before stop closure and then increased again after stop release; the stop types ranked in ascending order of total VOC decrease were voiced unaspirated < voiced aspirated < voiceless unaspirated < voiceless aspirated. Various other studies have simply noted general VOC suppression in consonants (Hoole 2006a: 23-28, cf. Hoole and Honda 2011: 133-134). In American English speakers, Hirose (1971a: 115) found more INT activity for vowels than for voiced consonants, and VOC and LCA were active in vowels but mostly suppressed in consonants. Jaiswal (2011: 4) summarizes a lengthy array of studies that found TA and LCA activity largely suppressed for consonants but increasing in preparation for following vowels, with TA reaching higher levels more quickly after voiceless than voiced consonants. Controlling the degree of VOC and LCA suppression also appears important for realizing stop contrasts in Korean (§2.5.5.3.1).

The following sections give a more detailed look at the laryngeal articulations seen in various types of stops. For descriptions of invasive articulatory measurement techniques such as electromyography (EMG), fiberoptic laryngoscopy (fiberscopic video), and photoelectroglottography (PEG), with lists of additional experiments using these methods, see Dixit (1989), Harris (1981), Hirose (1971c), Honda (2008), Hoole (2006a), Titze (1979).

2.3 Laryngeal articulations of voiceless unaspirated stops

2.3.1 Utterance-initial prephonation

Voiceless unaspirated stops are widely treated as a default (simplest or "unmarked") stop category on the grounds that other kinds of stops are phonologically defined by additional laryngeal specifications (see Vaux and Samuels 2005, Hall 2007: 317). These additional specifications are required to produce voice throughout closure, laryngeal constriction, aspiration, and so on (§2.6, 3.2.7, 3.3.6), while voiceless unaspirated stops are regarded as "plain" in the sense that they eschew all those properties by default. This does not necessarily mean that voiceless unaspirated stops lack systematic phonetic laryngeal activity, however.

Is there a neutral or least active laryngeal state in speech? Trubetzkoy (Blevins 2004: 74) identified quiet respiration or ordinary breathing as a kind of default from which phonological complexity could be measured (cf. Hirose 1997: 128). Respiration is not properly a speech state at all, however. Chomsky and Halle (1968: 300) write:

In most X-ray motion pictures of speech, it can readily be observed that just prior to speaking the subject positions his vocal tract in a certain characteristic manner. We shall call this configuration the "neutral position" and shall describe some of the ways in which it differs from the configuration of the vocal tract during quiet breathing. In the latter state the velum is lowered, thereby allowing air to pass through the nose; in the neutral position, on the other hand, the velum is raised...

After describing oral and pulmonic differences between this neutral position and ordinary respiration, Chomsky and Halle then focus on the larynx:

During quiet breathing, the vocal cords must be held widely apart since practically no sound is emitted. On the other hand, there is good reason to believe that prior to speaking the subject normally narrows his glottis and positions his vocal cords so that in the neutral position they will vibrate... in response to the normal, unimpeded air flow.

Somewhat similarly Ladefoged and Maddieson (1996: 50) write, "The physiological position for modal voice can be regarded as one in which the arytenoid cartilages are in a neutral position for speech, neither pulled apart nor pushed together (Stevens 1988). The vocal folds would be very slightly apart, if there were no air flow."

The laryngeal configuration that Chomsky and Halle describe as the "neutral position" preparatory to speech closely resembles what a more recent stream of research has described occurring during utterance-initial voiceless unaspirated stops in various languages. In this context, this laryngeal state has been called "prephonation."

Our observations of Thai, Cantonese, Pame, Tibetan, Yi, and Bai (cf. Edmondson et al., 2000) confirm that at stop onset, the arytenoids are adducted as for voice, but the space between the vocal folds remains parted. We propose the term 'prephonation' to refer to this state of the glottis accompanying voiceless unaspirated oral stops. Our research has also shown that this configuration precedes an initial vowel in modal voice when [the vowel is] not preceded by a glottal stop. (Esling and Harris 2004: 1, cf. Edmondson, Esling, Harris, and Wei 2004: 54)

Laryngeal observations of the state of the glottis during the articulatory stricture phase of initial unaspirated oral stops and affricates by Harris (1999) indicate a state of the glottis... referred to as 'prephonation.' Prephonation, as distinct from other voiceless states of the glottis such as breath, is made with the arytenoid cartilages adducted as for modal voice, but with the vocal folds forming a narrowed convex-convex opening medially in the glottis during which there is presumably insufficient subglottal air pressure to initiate airflow through the partially open glottis throughout the closure phase of oral articulatory stricture. (Esling and Harris 2005: 357, cf. Esling and Harris 2003, Esling 2006: 130 with film clip 6 in the online version).

The "initial" environment in which prephonation has been observed appears to be utterance-initial. Harris (2001: 7) refers to "the isolative style data" elicited for Standard Thai. Esling and Harris (2005: 357) also report observing prephonation immediately before a modally voiced initial vowel "that is not heavily stressed and does not have a glottal stop before it" in Thai and North American English: presumably this vowel is post-pausal, since otherwise it could not be preceded by the silent laryngeal configuration which they describe as prephonation. The same configuration was fiberoptically observed by Sawashima (1970: 187) just before an utterance-initial English [1]: "the arytenoids are closed, while a narrow spindle-shaped opening is seen along the membranous portion of the glottis."

2.3.2 Silent abduction after voiced segments

Voiceless unaspirated stops that are not in utterance-initial position have been observed with laryngeal configurations different from prephonation. Fiberoptic and EMG investigations of a Hindi speaker by Kagaya and Hirose (1975: 43) found that intervocalic voiceless unaspirated stops and voiced aspirated stops had an abduction gesture of roughly comparable duration, peaking at roughly comparable glottal widths. The duration of abduction in voiceless aspirated stops was typically longer, and peak glottal width was roughly twice as great. The voiceless unaspirated and voiced aspirated stops differed in the timing of their abducted phase: it mostly coincided with closure for the voiceless unaspirated stops but occurred mostly after release for the voiced aspirates, which had vocal fold vibration during closure. All the stops were measured in two environments: between vowels as the onset of a stressed syllable in a nonsense word, and as the onset of a stressed syllable at the beginning of a nonsense word immediately preceded by a vowel-final word. Abduction was found to be slightly greater for the wordinitial tokens, and full adduction for modal voice was often not resumed until 20-40 ms or occasionally even longer after stop release (Kagaya and Hirose 1975: 29-31).

The measure of how long voice is delayed past consonant release is called Voice Onset Time or VOT. In Japanese, where /p/ and /t/ average around 20-30 ms VOT while

/k/ averages 40-56 ms VOT (Itoh et al. 1979: 126-127, Itoh et al. 1980, Riney et al. 2007), varying degrees of vocal fold abduction have been observed in post-vocalic utterance-medial position, with the word-initial stops tending to have higher peak glottal width (Saito 1992: 35). In American English, Sawashima (1970) laryngoscopically observed abduction during the closure of voiceless unaspirated stops, usually beginning slightly before stop closure, with vocal fold vibration ceasing at or often slightly after the moment of stop closure; the arytenoids returned to an adducted position and voicing resumed around the moment of stop release (versus considerably later, in voiceless aspirated stops). Nuancing such findings, a set of EMG, transillumination, articulography, aerodynamic, and acoustic measurements by Lisker and Baer (1984: 133-134) found PCA contraction and INT suppression in intervocalic voiceless unaspirated stops to tokens in American English, but sometimes without any noticeable glottal opening.

While the data we have seen so far unfortunately do not include experimental verification of prephonated voiceless unaspirated stops occurring utterance-initially and moderately abducted ones occurring intervocalically in the same language, an aerodynamic motive has been proposed which appears to explain both types of articulation. Kagaya and Hirose (1975: 42-43) suggest that moderate abduction in voiceless unaspirated stops helps turn off the voicing of the previous sound to ensure that the stop is voiceless. Use of abduction for active devoicing was also suggested by Lindqvist (1972: 3). Without a mechanism for turning off voicing, voice would be expected to continue part-way into the closure of the stop; oral air pressure buildup would cause voicing to fail, but not immediately, and vocal fold inertia might prolong vibration somewhat. (There is debate about how quickly voicing would cease under these conditions: §2.6.) In an utterance-initial voiceless unaspirated stop, however, there is obviously no voicing to turn off. Kagaya and Hirose's explanation for moderate abduction in Hindi intervocalic voiceless unaspirated stops thus readily expands to account for why such abduction has been found absent in utterance-initial voiceless unaspirated stops in various other languages.

If this idea is correct, then abduction can serve two distinct acoustic purposes in a language with an aspiration contrast. In voiceless unaspirated stops, abduction can occur intervocalically to turn off voicing, but in that usage it has no other auditorily significant result. In voiceless aspirated stops, on the other hand, a large portion of the abducted phase is timed to occur after stop release, generating turbulent airflow heard as aspiration (§3.3.6.4). The laryngeal contour seen in Hindi voiceless unaspirated stops may thus be a product of strategies that have to do with the surrounding segments rather than being determined by the stop's own phonological specifications in isolation: first PCA activity turning off the voicing of the previous segment, then activation of the adductors at roughly the midpoint of the stop in order to reach the configuration for voicing in the following vowel.

On adduction being initiated before voicing occurs, compare Löfqvist and Yoshioka's (1979: 119) finding that adduction begins just before stop release in Swedish voiceless aspirates. They observe that "no aerodynamic factors could be responsible for the initiation of the gesture" at that point in time and refer instead to the "continuous" or "ballistic" nature of glottal widening and narrowing gestures throughout speech (pp. 105-107). Similarly Löfqvist (1980: 484) writes, "A general feature of laryngeal articulation for voiceless obstruents that emerges from all these investigations is that the vocal folds seem to be constantly moving in what can be described as a single ballistic opening and closing gesture." In light of this and our discussion so far, it may be that there is no laryngeally inactive state in speech. Instead, the laryngeal default involves arytenoid adduction, in the sense of both a starting point (if Chomsky and Halle 1968 are correct about the "neutral position") and also frequency (at least in languages where modally voiced segments are the most frequent type: $\S2.2$). Various gestural strategies then provide options for moving away from this neutral position (cf. Steriade 1995: 152-153) and subsequently back to it, as in the abduction-adduction sequence in Hindi and English voiceless unaspirated stops. The timing of this continuous sequence of glottal width

changes can be plausibly attributed to the auditory goals of controlling voice and aspiration (§3.3.6.4, 3.3.7.1).

Further evidence supports the idea that moderate abduction serves as an active strategy for producing silent closure in contrastively voiceless unaspirated stops after voiced sounds. First, the gesture is hard to explain otherwise. Oral air pressure continues to build during stop closure; if glottal width were passively determined by the stop's pressure contour, then we would expect glottal width to continue increasing until stop release, instead of peaking around the middle of the closure as in Kagaya and Hirose's (1975) data (Nolan, p.c.). On the other hand, if glottal width were controlled independently of distinctions in audible voice and aspiration, it is unclear how else the glottal width differences would be perceived and what role such an independent glottal width parameter would play in the phonological organization of languages with a contrastive voiceless unaspirated stop series. Finally, PCA activity is not the only gesture which appears to function as a strategy for turning off voicing in unaspirated stops. Other gestures are found occurring at the same time or instead, as described next.

2.3.3 Proposed roles of vocal fold tensing

2.3.3.1 Devoicing

The cricothyroid muscle (CT) has frequently been found active in voiceless stops. CT contraction lengthens the vocal folds, increasing their tension (Hirose 2010: 139). VOC and LCA activity also affect vocal fold tension, as we will see later (§2.4, 2.5.5.3.1), but here we will focus on CT.

Kagaya and Hirose (1975: 38, 43) report that unlike their study, Dixit (1975, cf. 1989) found no significant glottal opening in Hindi voiceless unaspirated stops; instead he found relatively consistent CT activity in both unaspirated and aspirated voiceless stops. This led him to conclude that CT rather than PCA activity played a key role in

defining voicelessness in stops. A similar conclusion was reached by Dixit and MacNeilage (1981) and, in English and Dutch, Löfqvist et al. (1989). In Kagaya and Hirose's (1975: 34-35) Hindi speaker, however, CT activity moderately peaked at the onset of closure for the voiceless unaspirated stops, but was less active for the voiceless aspirates; it also showed considerable activity during the latter part of closure for voiced unaspirated /b/ and /b^f/. Kagaya and Hirose (1975: 38) remain open to possible interspeaker variation in the use of devoicing gestures.

Hirose (1971b: 162) found that CT activity did not correlate with the voicelessvoiced distinction in Japanese and concluded that (generally) reciprocal PCA and INT activity were probably the main gestures articulating the distinction. Jaiswal (2011: 2) found increased CT activity in voiceless consonants (including unaspirated stops) in VC transitions for only half of the eight American English subjects who participated in a study that included EMG recordings of CT and TA muscle activity, audio signals, and video nasendoscopy. For all but one subject, however, CT activity was similar or lower in CV transitions with a voiceless consonant than with a voiced one. CT activity showed no statistically significant differences correlating with aspiration.

2.3.3.2 Glottal widening and pitch raising

That correlation is precisely what Hoole (2006a: 109-115) found in a transillumination study of German speakers, however: rather than peaking around stop closure as would be expected in a devoicing strategy, the timing of CT activity was correlated with abduction, suggesting that for these speakers, CT contraction helps widen the glottis without being primarily aimed at devoicing. Hoole explains the role of CT contraction in glottal widening by noting that the glottal opening is triangular, with the base at the arytenoids: the area of the triangle can be enlarged by abducting the

arytenoids, but stretching the vocal folds longitudinally increases the triangle's height and thus can also help enlarge the glottal area.

Besides assisting in devoicing and in glottal enlargement, another and more consistently reported function of CT is to serve as the primary control of pitch (see Baer et al. 1976, Gobl and Ní Chasaide 2010: 396, Hirose 2010: 147, Hoole and Honda 2011: 149). However, pitch is also affect by other mechanisms, some of which do not seem entirely subject to speaker control though they may be actively enhanced (Hoole 2006a, Hoole and Honda 2011: 136, 151). Voiceless stop contrasts correlating largely or mainly with the pitch of the following vowel occur in various languages (§2.5.4, 3.2.4, 3.3.7.3), but most of these studies rely on auditory impressions or acoustic measurements, hindering identification of the laryngeal muscles involved.

Shyrock (1995) argues from a detailed acoustic comparison of Musey (Chadic) with several other languages that the primary articulatory correlate distinguishing the two contrastive stop series in Musey may be vocal fold length as controlled by CT contraction, which would be responsible for one of the series exhibiting both higher pitch and very slightly greater VOT. The latter correlate would be due to slight enlargment of the glottis caused by CT contraction.

2.3.4 Voiceless unaspirated stops: conclusion

We have seen that various laryngeal muscles do not correlate with acoustic cues in a one-to-one fashion but rather synergize with effects on voicing, aspiration, and the pitch of the following vowel. We have also seen significant variation both across languages and among speakers of the same language in terms of what gestures they use to produce voiceless unaspirated stops. This suggests that phonological distinctions involving the laryngeal configurations of stops may be better defined in auditory rather than articulatory terms for languages with clear contrastive stop voice and aspiration (§3.3.63.3.9). We turn now from voiceless unaspirated stops to examine a series of other laryngeal configurations.

2.4 Laryngeal constriction and whisper

Once the vocal folds have been adducted for modal voice, reducing the size of the glottis even further requires the activity of other muscles besides the vocal fold adductors. Some of these other muscles are located higher in the larynx and in the epiglottal area above the larynx. The boundaries between the larynx, epiglottis, and pharynx are fairly ill-defined, but we will use the term "laryngeal constriction" in reference to any muscle activities which exert a significant squeezing effect on the larynx to narrow the glottis beyond the configuration appropriate for modal voice, other things being equal.

One situation where other things are not equal is when laryngeal constriction combines with vocal fold abduction to create hoarse whisper. The existence of such an articulation shows that laryngeal constriction and vocal fold abduction are not mutually antagonistic, even though there are other sounds in which laryngeal constriction does synergize with vocal fold adduction in opposition to abduction. We will examine that synergy first, before looking at the combination of abduction with laryngeal constriction in whisper.

During full glottal closure as seen in the glottal stop and in swallowing, an EMG study of English (Freeman and Ushijima 1974) found activity of both VOC and LCA. PCA was suppressed in this environment but showed a slight burst of activity coinciding with reopening of the glottis for modal voice. Other EMG studies have also found TA activity during glottal closure, along with supraglottal laryngeal constrictions (Hirose 2010: 147). Laryngoscopic work by Esling et al. (2005: 386-387) across an unspecified variety of languages found a range of laryngeal constrictions in the production of glottal stops. The minimum observed was "slight partial adduction of the ventricular folds", but
more often they also saw moderate constriction of higher muscles causing "moderate narrowing of the laryngeal vestibule".

The laryngeal vestibule is defined as the supraglottal cavity of the larynx that extends from the ventricle of the larynx upwards through the aditus laryngis [into the pharyngeal cavity]... it is formed anteriorly by the epiglottis, posteriorly by the apices of the arytenoid cartilages, and laterally by the aryepiglottic folds with the prominent cuneiform cartilages as hinge points. The aryepiglottic folds extend from the sides of the epiglottis to the apices of the arytenoid cartilages. A slight retraction of the tongue and epiglottis also appears to be a component of the constriction of the whole laryngeal vestibule. (Esling et al. 2005: 386-387, cf. Lindqvist 1972: 3, Moisik and Esling 2011, Moisik et al. 2011)

While laryngeal constriction and release typically work together with adductor and abductor activity during the closure and release of a glottal stop, this type of cooperation is not physiologically necessary. As already noted, glottal abduction and laryngeal constriction actually combine in the production of whisper (Hirose 2010: 146):

In whisper, there is arytenoid separation at the vocal process with an adduction of the false vocal folds taking place with a decrease in the size of the anterior-posterior dimension of the laryngeal cavity. For this type of laryngeal adjustment, PCA continues to be active and the thyropharyngeal activation is also observed most likely for realization of supraglottal constriction... This particular gesture for whispering is considered to contribute to the prevention of the vocal fold vibration... as well as to facilitate the generation of turbulent noise in the laryngeal cavity.

This passage describes voiceless whisper. A similar articulatory configuration is possible with voicing, producing whispery voice. Whisper with both voiceless and voiced phases occurs contrastively upon release in some stops in Zhenhai Wu (Chinese: Rose 1989). Zulu depressor stops involve contraction of the laryngeal vestibule for some speakers (Traill et al. 1987: 265) and have also often been impressionistically described as triggering breathy voice on following vowels (see Chen and Downing 2011, Traill et al. 1987); whispery voice will be analyzed as a kind of breathy voice below (§2.5.2).

Whispery voiced nasals in Rwanda have been found to involve mild pharyngeal pressure increase during oral closure that takes longer to occur than in stops and suggests active reduction of the pharyngeal cavity (Demolin and Delvaux 2001: 2). Articulatory and aerodynamic data on whispery voice thus supports Hirose's (2010) definition of whisper as involving laryngeal constriction and some compression of the pharyngeal cavity, in addition to vocal fold abduction.

The boundaries are still being explored between this type of laryngeal constriction and other articulations involving the aryepiglottic folds and sometimes larynx raising or lowering, which are found in various sounds including stiff or "tense" voice, pharyngeals, and epiglottals (Bellem 2005, 2007, 2009, Edmondson 2000: 3, 2004, 2006, Edmondson and Esling 2006, Edmondson et al. 2001, 2005: 383, Esling 2002c, Esling and Harris 2004, Esling et al. 2005: 396-403, Moisik, Czaykowska-Higgins, and Esling 2011, Moisik and Esling 2011, Moisik, Esling, and Crevier-Buchman 2009, Rose 1989: 236-241, Shahin 2011). As a brief introduction, Esling (2000, 2002a,b) provides fiberoptic visuals of several of these sound types with phonetic comparison and some examples of the complex phonological interactions that can occur among them. Transitioning back in the opposite direction – decreasing articulatory constriction – we find whispery voice shading into breathy voice both acoustically and articulatorily, so they are treated together in the next section.

Two other comments on whisper are in order. The first regards a universal-feature proposal known as "Dimensional Theory," in which laryngeal features are grouped in dimensions or pairs where each pair is correlated with reciprocally antagonistic muscles and the general principle is that only one member of a given dimension can be active in a given speech sound (Avery and Idsardi 2001: 44; Ahn and Iverson 2003: 5). The mutually exclusive antagonism which this model posits between glottal spreading and laryngeal constriction appears to contradict the physiology of whisper. Second, we will see later how the range of laryngeally unconstricted glottal widths that can occur contrastively in a sound is partly dependent on stricture (§3.4). That relationship can only be identified if

the reciprocal relationship between the abductor and adductor muscles is distinguished from laryngeal constriction, because segments can be laryngeally constricted regardless of their stricture. Thus, recognizing that laryngeal constriction is distinct from abductoradductor reciprocity is important for elucidating phonological patterns.

2.5 Breathy voice

2.5.1 Introduction

"Breathy voice" is an expression used with varying degrees of precision for sounds that include both voicing and a range of other acoustic cues such as glottal turbulent noise, lowered fundamental frequency, an amplified first harmonic, and various acoustic configurations involving increased amplitude in the lower harmonics and decreased amplitude in the higher ones (spectral tilt). These cues tend to co-occur, but there is great variation of detail among languages and even individual speakers.¹

On the articulatory side, "breathy voice" in the broadest sense may include active abduction of the vocal folds with high medial compression and perhaps some supraglottal laryngeal constriction, causing a small chink at the posterior end of the glottis while the vocal folds are fairly closely adducted along the anterior glottis. The more precise name for this configuration is "whispery voice," while "breathy voice" in a stricter sense has moderate vocal fold abduction without these other gestures, allowing the vocal folds to flap more loosely without ever fully closing. Larynx lowering and concomitant vocal fold slackening with little or no active abduction can also generate many of the cues associated with breathy voice in the broad sense, though resulting glottal turbulent noise is expected to be less salient; this third configuration is termed "slack voice."

¹ Formants are bundles of amplified harmonics, which in turn are resonant frequencies (overtones) that are multiples of the fundamental frequency of a sound. The fundamental frequency of a complex periodic wave is the frequency of the slowest wave component and is thus also the rate at which the total pattern of the wave repeats; the first harmonic is equivalent to the fundamental frequency since it is the fundamental frequency multiplied by one (Johnson 2003: 9, 138, cf. Stevens 1998, 1999).

Articulatorily, then, whispery voice shades into strict breathy voice in proportion to a decrease in active gestures that compress or constrict the larynx, while strict breathy voice shades into slack voice with a decrease in active glottal widening.

The correspondence between these articulatory strategies on one hand and the acoustic cues just mentioned on the other are very imperfectly understood, partly because most descriptions of the relevant languages are either acoustic or impressionistic auditory ones; a limited number of fiberoptic observations exist, but their significance is somewhat unclear pending direct measurements of muscle activity by methods like EMG. The complex gradient relationships among the relevant gestures and cues not only challenge linguistic description, however; they also appear to elicit diverse analyses by individual learners during acquisition, leading to extensive fine variations among speakers of individual languages, as we will see.

The next few sections compare the articulations of whispery and strict breathy voice, then summarize relevant acoustic literature on breathy voice in the broader sense. This sets the stage for the treatment of slack voice (§2.5.5).

2.5.2 Whispery and strict breathy voice: articulation

According to Gobl and Ní Chasaide (2010: 400), whispery voice involves (1) some arytenoid separation, abducting the posterior, cartilaginous part of the vocal folds, but also (2) moderate longitudinal tension and (3) moderate to high medial compression ensuring vibration of the more anterior, ligamental part of the vocal folds. Given the foregoing discussion, these three conditions presumably require PCA, TA, and LCA contraction respectively, though Gobl and Ní Chasaide's description does not mention the muscles by name:

Whispery voice is characterized by low adductive tension, moderate to high medial compression, and moderate longitudinal tension. As a consequence, there is a

triangular opening of the cartilaginous glottis, whose size varies with the degree of medial compression. In weak whisper the medial compression is moderate and the opening may include a part of the ligamental glottis as well as the cartilaginous. Whisper with increasingly higher intensity has increasingly higher medial compression and smaller glottal opening, until only the cartilaginous glottis is open. Laryngeal vibration is assumed to be confined to that portion of the ligamental glottis which is adducted, and the whispery component to the triangular opening between the arytenoids. It is very inefficient and there is a considerable degree of audible aspiration noise.

Whispery voice occurs at the release of the voiced aspirates in some Indic languages (Stuart-Smith 2004: 169). It is also reported occurring as aspiration on a voiced stop in White Hmong (Fulop and Golston 2008), but previous acoustic work found the situation somewhat complex, with closure voicing failing just before release in a way that has prompted impressionistic descriptions of a glottalized catch (Jarkey 1987: 57-60; further data in Kehrein and Golston 2004: 339). Gobl and Ní Chasaide's description of whispery voice does not suggest any supraglottal laryngeal constriction, but given its similarity to Hirose's description of voiceless whisper, it seems hard to exclude the possibility of such constriction in whispery voice as well (presumably shading into epiglottal frication as aryepigglotic valve activity increases).

In the direction of decreased constriction, Gobl and Ní Chasaide (2010: 400) describe whispery voice as shading into breathy voice the more freely the vocal folds are allowed to vibrate and the more the noise component at the posterior glottal opening is reduced. Such reduction occurs as decreased medial compression (due to LCA suppression: Hirose et al. 1972: 189) and decreased longitudinal tension allow the vocal folds to move more loosely, to the point where they no longer fully adduct even when they approach each other most closely.

The auditory similarities and gradient phonetic differences between breathy and whispery voice have received inconsistent treatment in earlier phonetics literature (see Fulop and Golston 2008: 8, Stuart-Smith 2004: 164-165). Both voice qualities have been found in different speakers' realization of the voiced aspirates in Indic language (Stuart-

Smith 2004: 169). To distinguish breathy voice in the more precise sense from the broader usage which may include whispery voice and also slack voice, we will continue referring to the more precise category as strict breathy voice.

2.5.3 Voiced aspirates

Voiced aspirate stops measured in Hindi and Maithili have shown a typical pattern involving modal voicing during the first part of stop closure, followed at some point (often quite close to the stop release) by a glottal widening phase that peaks after release before adducting back to modal voice. Investigations agreeing on this pattern include PEG work by Dixit (1989), fiberoptic and EMG work by Kagaya and Hirose (1975), and other experiments discussed in Stuart-Smith (2004: 168). Breathy voice is heard during this abduction phase. As we have seen (§2.3.2), this abduction is roughly comparable in duration and magnitude with that of voiceless unaspirated stops and roughly half that seen in voiceless aspirates, though it can range from 70 to 130 ms (Stuart-Smith 2004: 168, 174).

Vocal fold vibrations typically last throughout the closure and aspiration of voiced aspirate stops, straight into the modal portion of the following vowel (Ladefoged and Maddieson 1996: 59, Stuart-Smith 2004: 167). Abduction beginning before stop release may contribute to a decrease in voicing amplitude sometimes seen toward the end of closure in Hindi voiced aspirates as compared to the unaspirated voiced stops (e.g. Ladefoged and Maddieson 1996: 59). At least in the case of strict breathy voice, the slackness of the vocal folds due to lack of medial compression may cause them to have a higher open quotient than in modal voice even apart from active abduction (see Stuart-Smith 2004: 168). Open quotient, or the proportion of time during voing that the vocal folds are apart rather than fully adducted, also increases as air pressure builds up during stop closure (Bickley and Stevens 1986). It has been shown instrumentally that pitch

lowering tends to be significantly greater with breathy-voiced than with modally voiced stops (Hombert, Ohala, and Ewen 1973: 47, Mikuteit 2009: 48-60, Kagaya and Hirose 1975: 36-37, Schiefer 1986), and both increased open quotient and larynx lowering would be natural articulatory contributors to this (see Jessen and Roux 2002: 27, Stuart-Smith 2004: 167).

The modal vowels before voiced aspirates tend to be slightly lengthened (see Keating 1984a: 293, Ladefoged and Maddieson 1996: 60). Some studies have also found the closure of voiced aspirates shorter than those of voiced unaspirated stops (Jessen 2002: 154, 175, Kagaya and Hirose 1975: 37), which in turn are often shorter than those of voiceless unaspirated stops (§2.6).

2.5.4 Breathy voice: cues

The actions of particular muscles are important for differentiating whispery voice, strict breathy voice, and slack voice. In practice, much of the evidence from relevant languages is impressionistic or acoustic, forcing us to deal with descriptions that refer to breathy voice in a broader sense which may include various degrees of whispery or slack voice.

Wayland and Jongman (2003) describe several techniques that have been developed for studying breathy-voice cues, most of which involve measuring the amplitude difference between various pairs of harmonics. Indicators reported for breathy voice in various languages include lowered fundamental frequency (F0) due to the slower vibration of the slackened vocal folds, an amplified first hamonic (H1), reduced amplitude of the first formant (F1), additive noise from increased glottal turbulence which can mask or replace the third and higher formants, and other changes in spectral tilt involving amplified lower harmonics and reduced amplitude of higher harmonics compared to the spectral slope of modal voice. Wayland and Jongman (2003: 186) note that since glottal additive noise amplifies higher frequencies while breathy spectral tilt decreases them, "the spectral balance of breathy signals is affected by two separate phenomena acting in opposite directions," making it hard to measure breathy spectral tilt unless the periodic and noisy components are distinguished. Further, there are multiple ways to try to quantify the effect of spectral tilt distributed across multiple harmonics (Blankenship 2002, Brunelle 2005: 163-165, Gordon 2001, Jessen and Roux 2002: 7-8, Park 2002).

Lowered F1 frequency is also reported in breathy vowels in some languages, though not in others (Thurgood 2004: 280); this has been linked to larynx lowering, which lengthens the supraglottal vocal tract (Brunelle 2010: 8-10; §3.3.7.3). Wholly or partly breathy-voiced vowels may also be longer than modal vowels (Thurgood 2004: 280, Wayland and Jongman 2003: 187), possibly in some cases to compensate auditorily for their reduced amplitude or the masking effect of additive noise on upper formants (see Gordon 1998).

In addition to cross-linguistic variation in the degree of breathiness detected by these various criteria, different studies have found striking dialectal and inter-speaker variation, suggesting a range of fine articulatory adjustments. An acoustic study of Eastern Cham dialects (Austronesian: Brunelle 2009: 14) provides a dramatic example of dialect variation in the realization of the register distinction (a contrast between two sets of cues realized on vowels but conditioned by a contrast in the preceding stops; see also Brunelle 2009, Wayland and Jongman 2002: 101-105):

[I]t seems that the initial portion of low register vowels is very breathy in Kompong Chhnang Cham, whereas their high register counterparts are modal. Both registers then seem to become modal before ending on a breathier voice quality. Most Eastern Cham speakers follow the same general trend, but the amount of overlap between their registers is much greater. Finally, Châu Đốc Cham register seems to be "ballistic". The low register starts breathy and then become modal (or even tense), while the high register starts modal and then becomes breathy. It is difficult to isolate a clear pattern at the end of the vowel, but it suffices to say that the voice quality distinction seems lost by then. Overall, the salience of voice quality distinctions seems too low [in these dialects] to be used alone for contrasting registers, but it could possibly reinforce a contrast in other phonetic properties.

Chanthaburi Khmer illustrates inter-speaker variation (Wayland and Jongman 2003: 197-198). For the female speakers, the register contrast was between breathy and modal voice, while the males contrasted modal voice with a phonation (voice quality) involving steeper positive spectral tilt due largely to an enhanced second harmonic (H2). Fairly steep positive spectral tilt is associated with creaky voice and can be visualized as rotated counter-clockwise from the moderately rising tilt of modal voice, while the tilt of breathy voice can be visualized as rotated in the opposite direction; however, creaky voice also typically involves slowed vocal fold vibrations which may produce a visibly choppy effect in spectrograms (see Gordon 2001). Wayland's impression of the male Chanthaburi Khmer speakers' higher (steeper) register was not one of creaky voice, and he suggests calling it "tense" instead (cf. §2.5.6). A breathy vs. modal register contrast has similarly shifted to modal vs. steeper spectral tilt in various other languages (see Brunelle 2005: 147).

In a second variation within Chanthaburi Khmer, two of the three female speakers produced their breathy vowels with a slightly higher F0 than their modal vowels, which is unusual (Wayland and Jongman 2003: 187, 194-195). One possible articulatory cause that could be investigated is whether the type of breathy voice used by these two speakers is whispery voice with relatively high medial compression and little larynx lowering. As we already saw (§2.5.2), lower medial compression creates a longer posterior glottal opening and leaves the vocal folds free to vibrate more loosely along more of their length, which tends to slow down their vibration and thus, like larynx lowering, tends to lower pitch. We turn now to examine the interaction between larynx lowering and another type of breathy voice: slack voice.

2.5.5 Slack voice

2.5.5.1 General

Ladefoged and Maddieson (1996: 63-66) describe slack voice as having a slightly larger glottis and higher airflow than modal voice, but still considerably less than strict breathy voice. Amplified lower harmonics, decreased spectral tilt, F1 lowering, and increased VOT are reported after slack stops in Javanese, while F0 lowering and F2 differences are less consistent (Brunelle 2010: 8). Similar sounds are reported in Wu Chinese dialects (Cao and Maddieson 1992). In both languages, Ladefoged and Maddieson refer to slack voiced stops. Since the stops themselves are admittedly voiceless during closure, however, with their distinctive cues being realized on following vowels, we will call them slack stops instead, with the implication that the slack quality automatically carries over into the voice of the following vowel but without implying that voice is an active strategy specified by the stop.

Below we will examine two distributional patterns of voiceless slack stops – that is, stops with entirely or mostly voiceless closure and cues of the slack quality being heard on the following vowel. In the first pattern, the slack stops occur initially and medially, while in the second pattern they occur only initially, with voiced medial allophones that are not consistently associated with cues of slack voice on the following vowel. The languages in which we will examine these patterns are listed in Table 2.1. The possibility of voiced slack stops (stops with closure voicing that trigger the cues of slack voice on a following vowel) is considered later (§3.3.7.2).

Table 2.1 Voiceless slack stop types.

| Voiceless | Voiceless initial with voiced medial allophones | |
|--------------------------|---|--|
| Javanese | Korean | |
| Xhosa, Zulu | Wu Chinese | |
| Eastern Armenian variety | Various Armenian dialects | |

Besides acoustic and fiberoptic investigations of slack voice in Javanese and Wu Chinese, impressionistic descriptions of the relevant Armenian dialects repeatedly mention breathy voice, murmur, glottal noise, and the like as auditorily salient properties (§2.5.5.4). Similar impressionistic descriptions of Xhosa and Zulu have been qualified by later acoustic studies that found considerable inter-speaker variation (§2.5.5.2). Several acoustic measures of breathy voice based on harmonic amplitude do not correlate reliably with glottal turbulent noise (Wayland and Jongman 2003: 184-185), indicating the difficulty in measuring glottal turbulence, let alone inferring glottal width, from acoustic data alone. This contrasts with research on Hindi-type voiced aspirates, where glottal widening and salient turbulence are consistently reported (e.g. Ladefoged and Maddieson 1996: 57-64, Stuart-Smith 2004: 164-166).

Several writers have linked the acoustic cues of slack voice to existing hypotheses about larynx lowering and vocal fold slackening (Brunelle 2005, 2010: 10-11, Jessen and Roux 2002, Thurgood 2008: 20-21, Traill et al. 1985, Wayland and Jongman 2003). The anterior convexity of the spine seems to play an especially interesting role: as the larynx lowers, the spinal curve and the natural range of the cricoid joint cause the front part of the cricoid cartilage to slide in a downward posterior direction in advance of the thyroid cartilage above it while the back part of the cricoid rotates forward (fig. 2.2). The arytenoids are attached to the back of the cricoid cartilage, and the vocal folds stretch from these to the interior surface of the front of the thyroid cartilage. The sliding or rotation of the cricoid thus rocks the arytenoids forward, which naturally decreases the longitudinal tension of the vocal folds, causing them to vibrate more slowly and lowering the pitch (Honda 2008: 12). This presumably increases the open quotient and thus the rate of airflow. Larynx lowering presumably also increases subglottal pressure, and this along with additional vertical factors may increase open quotient as well (Brunelle 2010: 11). The consequent airflow increase then causes turbulence heard as breathiness. Increased open quotient also correlates with decreased spectral tilt, though it seems possible to mitigate this by fine muscular adjustments to vocal fold tension independent of larynx height (Wayland and Jongman 2003: 187-188). Similar adjustments (e.g. of TA or CT) could also presumably assist or hinder the effect of larynx lowering on pitch.



Figure 2.2. Cricoid rotation during larynx lowering shown at right. The thick, dark line inside the thyroid cartilage represents the axis of the vocal folds, which shorten as the cricoid rotates in the direction of the arrow. The vocal folds (ligaments) are depicted more clearly with the arytenoids at left. In both diagrams the anterior direction faces left. Adapted from Honda (2008: 9, 12).

This picture of slack stops again differs from that of Hindi-type voiced aspirates, where closure voicing and turbulence during the offset (the period immediately following stop release) can be explained by the sequential activity of the adductors and abductors and where the degree of larynx lowering needed to explain pitch lowering is unclear (since other factors may contribute to slower vocal fold movement and thus lower pitch: §2.5.3). Based on such considerations, Jessen and Roux (2002: 38-43) suggest that slack stops share a different articulatory strategy from voiced aspirates like those of Hindi, with slack stops relying on larynx lowering which contributes to vocal fold slackening, while voiced aspirates rely (at least inter alia) on closure voicing with an abduction gesture peaking after stop release.

This in turn furnishes an explanation for why slack stops can have voiceless closure while the higher-airflow aspiration occurring after Hindi voiced aspirates has not been clearly documented at the release of voiceless stops (though it often occurs briefly at the tail end of voiceless aspiration as the abducted vocal folds return to the position for modal voice: Jessen and Roux 2002: 8). When abduction is phased so that it creates significant glottal turbulence after the release of a voiceless stop, the resulting aspiration is largely voiceless, presumably because voice initiation requires a narrower glottis and a higher transglottal pressure drop than voice maintenance does (see Hirose and Niimi 1991: 388, Kagaya and Hirose 1975: 39, Kingston et al. 2009: 10). Thus, a strategy relying largely on abduction to generate voiced turbulent airflow upon stop release might require voiced closure. Larynx lowering and vocal fold slackening, however, can increase offset airflow and generate some degree of turbulence without either devoicing the offset on one hand or requiring voiced closure on the other. If production strategy differences of this sort really do distinguish voiced aspirates of the Hindi type from slack stops, then that distinction might motivate different phonological feature specifications for them as well (§3.3.7.2).

2.5.5.2 Voiceless slack stops occurring initially and medially

2.5.5.2.1 Javanese

As already noted, fiberoptic and acoustic studies have found voiceless slack stops initially and postvocalically in Javanese (Austronesian), typically with slightly wider

glottis, longer (though still short) VOT, and breathier spectral tilt than after the other contrastive type of stop; some speakers also have larynx lowering in the slack stops, conditioning lower F0 and F1 on the following vowel (Brunelle 2010: 8, Ladefoged and Maddieson 1996: 63-66, Thurgood 2004). The slack stops are normally voiced in postnasal position (Brunelle 2010: 1). The other contrastive stop series is referred to as tense (Brunelle 2010) or as "stiff voiced," with a smaller glottis possibly related to VOC contraction; this series too is normally voiceless during closure, and is associated with higher pitch and increased (more positive) spectral tilt on the following vowel (see Ladefoged and Maddieson 1996: 55-56, 63-64; §2.5.6).

2.5.5.2.2 Xhosa

Jessen and Roux (2002) present acoustic measurements of the depressor stops which lower the pitch of a following vowel in Xhosa (Niger-Congo: Bantu). They studied stem-initial position followed by a stressed vowel, with and without a preceding prefix, and in all cases with a vowel-final preceding word. They concluded that these stops can also be classified as slack, with zero or negligible closure voicing comparable to that of the contrastive voiceless aspirates and ejectives. The depressor stops had VOT ranging from 11-17 ms (Jessen 2002: 162) and were followed by breathy voice (measured by the difference between H1 and H2) for five out of eight speakers. For four of those five, this breathy voice was greater than that seen at the tail end of voiceless aspiration. All but one or two of the speakers also had breathy voice after the depressors comparable to that at the tail end of voiceless aspiration as measured by H1-A3 (Jessen and Roux 2002: 20-24; A3 is the most amplified harmonic in F3, the third formant). Further, besides lowering the pitch of the following vowel, the depressors lowered F1 throughout most of the following vowel for four speakers except in postnasal position, for one speaker only in postnasal position, and for one speaker only early in the vowel, with additional speaker variation in the magnitude of lowering (Jessen and Roux 2002: 18-19).

The depressors were normally voiced in postnasal position (Jessen and Roux 2002: 2), where they had "very short closure durations...usually below 30 ms", which were "not systematically longer" than the amount of closure voicing found in any of the three contrastive voiceless stop series in postvocalic position (Jessen and Roux 2002: 4, cf. Jessen 2002: 167). The slack stops of Xhosa contrast with voiceless aspirates and ejectives that often lose their ejection medially except in careful speech; there is also a voiced bilabial implosive. (On variously conditioned loss of ejection in other Nguni languages see Fallon 2002: 287-289.)

After hypothesizing that larynx lowering triggers vocal fold slackening which generates the acoustic cues found in Xhosa depressors, Jessen and Roux (2002) suggest that this larynx lowering is extensive enough to prevent voicing:

[B]eyond a certain level of vocal fold slackening transglottal airflow becomes so high that it actually works against the maintenance of closure voicing, since it leads to a fast buildup of intraoral air pressure. If such extensive levels of vocal fold slackening are achieved by additional larynx lowering, it is expected that f_0 lowering becomes more extensive as well and achieves the status of a strong perceptual cue... It is also likely that the high levels of transglottal airflow in extensive vocal fold slackening lead to levels of breathy voice that are [more substantial].

Larynx lowering can be relatively rapid and extensive in voiced implosives, where it correlates with voicing crescendo rather than devoicing; implosives also tend to raise the pitch of the following vowel (see Demolin et al. 2002, Ladefoged and Maddieson 1996: 84-85; §2.5.5.2.3, 2.6.2). Given known laryngeal muscle functions (§2.2), these traits might be due to medial compression of the vocal folds helping limit transglottal airflow and maintain vocal fold vibration. CT contraction might also contribute to the higher pitch correlate. This raises the possibility that the Xhosa depressors are voiceless not because they have extreme larynx lowering beyond that normally found in any type of stop with closure voicing, but rather because of less medial compression and CT activity. Since LCA activity is associated with medial compression, it is worth noting that Kagaya and Hirose (1975: 34) found considerable LCA in /b/ as compared to other labial stops in Hindi, as well as moderate CT activity. If the voiceless closures of Xhosa depressor stops depend on LCA and CT suppression rather than on extreme larynx lowering, we would have no need to wonder why breathy voice is not more substantial and consistent across Xhosa speakers.

2.5.5.2.3 Zulu

Zulu is closely related to Xhosa within the Nguni group of Southern Bantu. The phonetic profile of the Zulu depressor consonants that emerges from fiberoptic and acoustic research (Chen and Downing 2011, Traill et al. 1985) is very similar to what we have already seen for Xhosa – except for the claim that "no voicing whatsoever was found during the closure" of postvocalic depressor stops (Traill et al. 1985: 266), an observation questioned below. Zulu depressors have short positive VOT (10-16 ms for labials and coronals, 20-37 ms for velars: Midtlyng 2011: 110). Breathy voice in the following vowel has been reported impressionistically in several studies but was not consistently evident in laryngographic traces (Traill et al. 1987), nor was it detected in Doke's (1926) kymographic waveforms (Jessen and Roux 2002: 6). The exception is fricatives, which Traill et al. (1987) found frequently associated with breathy voice; further, their description of the breathy-voiced depressor transcribed as /h/ suggests whispery voice in particular ("a slight chink in the intercartilaginous glottis" which "disappears on release" pp. 267, 271). They also highlight the fact that pitch depression lowers the tone of the following vowel (high or low) quite considerably: more than the distance between non-depressed high and low tones (p. 270). Finally, the depressors are normally voiced in postnasal position (p. 266).

Traill et al.'s (1987) fiberoptic investigation found that depressor stops and fricatives had "very slightly abducted vocal folds" (p. 266), with "a spindle-shaped glottis...most frequently in the membranous [ligamental] portion of the vocal folds"; the arytenoids when visible "were almost always adducted." This spindle-shaped opening recalls the description of prephonation (§2.3.1) and suggests lack of medial compression. Additionally, depressor stops involved contraction of the laryngeal vestibule (§2.4) in two speakers. The gesture was considerably "more extreme" for one speaker than another and is analyzed as follows (p. 265):

The observable movements involve an anterior movement of the arytenoids and the posterior movement of the tubercle of the epiglottis... Since we know that this adjustment is associated with extra-low pitch, it seems reasonable to assume that it induces an unusual degree of shortening of the vocal chords [*sic*] with a consequent lowering of their rate of vibration.

Traill et al. speculate that in subjects which did not have this gesture, CT relaxation contributed to pitch depression because larynx lowering "was not observed." It is not clear exactly how they gauged larynx lowering other than by imprecise visual observation. Presumably the extreme degree of larynx lowering that Jessen and Roux suggest was responsible for the mostly voiceless closures of Xhosa depressors would have been visually detected if it occurred in Zulu, but possibly lesser degrees would not.

Traill et al. (1987: 266, 271) assert that "no voicing whatsoever was found during the closure" of postvocalic depressor stops in their Zulu speaker; instead "voicelessness coincides with closure" as determined by oral airflow measurements synchronized with laryngographic waveforms of vocal fold pulses. In contrast, the depressor fricatives and postnasal depressor stops had clear closure voicing. Based on this difference, Traill et al. suggest that Zulu postvocalic depressor stops involve an active devoicing gesture, probably one primarily involving the vocal folds since aerodynamic data did not seem "compatible with vocal tract tensing or reduction" (p. 266; also recall the lack of visible

larynx lowering mentioned above). They tentatively suggest that VOC activity is responsible for this devoicing by increasing tension in the cover of the vocal folds in spite of the body of the folds being relaxed and shortened (p. 271). More recent work has confirmed the independence of VOC and TA in controlling the cover and body of the vocal folds (§2.2). A spike in CT activity around the moment of closure might also contribute to devoicing in Zulu depressor stops, just as it appears to do in Hindi voiceless unaspirated stops (§2.3.2).

However, I observed 20-30 ms of closure voicing in depressors produced by a different Zulu speaker. The recording of this speaker is stored in the UCLA Phonetics Lab Archive (UCLA 2007a), and I analyzed it with PRAAT (Boersma and Weenink 2010). Example waveforms and spectrograms are given below (figs. 2.3-2.8). I identified the approximate duration of voicing past closure onset as the portion of the fundamental frequency which continues with noticeably decreased amplitude after other harmonics have failed. These examples of depressor prevoicing are highlighted in the waveforms.

Regarding transcription in these examples, the orthographic system follows the UCLA record, where *bh d g* are depressors, *b* is a non-depressor voiced stop, and *k* is an ejective which often becomes [g] intervocalically (cf. Fallon 2002: 239, 271, Halpert 2010). Figures 2.3 and 2.4 compare the bilabial depressor with the non-depressor *b*. The latter is phonetically transcribed in the UCLA record as an implosive. Traill et al. (1987: 245) note that this traditional classification appears incorrect in their data since this segment lowers the tone of a following vowel in all their subjects (though not nearly as much as the depressors do) and has positive pharyngeal air pressure. This contrasts with reports of progressive increase in voice amplitude and formant clarity in the Xhosa bilabial implosive, which are more consistent with the identification of that sound as an implosive (see Flemming 2004: 259, Jessen 2002: 160-161). Traill et al. conclude that the so-called bilabial implosive in Zulu is an ordinary voiced stop. In line with this, I have provisionally retranscribed the sound as [b] below because the speaker in this recording

pronounced the segment without audible implosion, with relatively low pitch at the onset of the following vowel, and with amplitude comparable to that of the velar stop phonetically transcribed as [g] in the UCLA source. Otherwise, the phonetic transcriptions below essentially follow the UCLA source, which renders the depressor stops as voiced stops with a breathy umlaut. A more accurate alternative is discussed later (§2.5.6).

In all of the following samples of Zulu postvocalic depressor stops, voicing clearly leaks past stop closure for 20-30 ms, or roughly 4-5 glottal pulses. For at least this Zulu speaker, there is no clear evidence that an active devoicing gesture is necessary in the depressor stops. Instead, the slight prevoicing seen below in the postvocalic depressor stops and the full voicing that Traill et al. observed in the postnasal depressor stops and in the depressor fricatives might be due to differences in stricture rather than in laryngeal articulation. If the postnasal depressor stops are commonly as short as those in Xhosa, then they would be fully voiced simply because they are short, and if it has been argued correctly (Park 2002: 70) that voice is aerodynamically easier to maintain in fricatives than in stops, then the Zulu depressor fricatives would be fully voiced for this reason and not because of a laryngeal devoicing strategy uniquely present in postvocalic depressor stops.



Figures 2.3-2.8: Zulu depressors and fully voiced stops (adapted from UCLA 2007a).

Figure 2.3. Zulu *bhabha* [ba:ba] 'trap' shown with partial final vowel. Medial depressor prevoicing highlighted (32 ms).



Figure 2.4. Zulu *baba* [ba:ba] 'father (direct address)' with partial final vowel; both stops have full closure voicing.



Figure 2.5. Zulu *idada* [i^ldada] 'duck'. First stop with adjacent vowels shown; medial depressor prevoicing highlighted (27 ms).



Figure 2.6. Zulu *idada* [i'dada] 'duck'. Second stop with adjacent vowels shown; medial depressor prevoicing highlighted (29 ms).



Figure 2.7. Zulu *ukuguga* [uguˈġuġa] 'to grow old'. Only the first stop has full closure voicing. In spite of the release irregularities, the depressors were not audibly aspirated.



Figure 2.8. Zulu *ukuguga* [ugu'guga] 'to grow old' (close-up). Last stop and adjacent vowels shown; depressor prevoicing highlighted (25 ms).

Figures 2.9-2.10 Hindi intervocalic voiceless unaspirated stops (audio and transcriptions: UCLA 2007b).



Figure 2.9. Hindi ata [a:ta] 'he comes'. VC transition shown; stop prevoicing highlighted (8 ms).



commencement of a medial unstressed syllable," and sometimes simply inaudible (Allen 1950: 188).

2.5.5.2.5 Conclusion

The laryngeal distinctions in the stop inventories of the languages examined above are strikingly similar: in every case the stops described as slack or in similar terms contrast with a voiceless unaspirated series that has some degree of glottal narrowing ranging from vocal fold tensing to actual ejection, and in each language except Javanese there is a further contrastive voiceless aspirate series. Also in all these languages the slack or similarly described stops are normally voiced in postnasal position.

A similar type of three-series stop system also seems to occur word-initially in Wu Chinese, Korean, and Sivas Armenian. At least in Wu and Korean, allophones in some medial positions are voiced and are reported as not conditioning breathiness on the following vowel. (Information on Sivas medial stops was not available.) These languages are treated next.

2.5.5.3 Voiceless slack stops with voiced medial allophones

2.5.5.3.1 Korean

Korean has three contrastive stop series which have elicited a great deal of phonetic attention and phonological debate (e.g. H. Kim et al. 2005, M. Kim and Duanmu 2004). The three series are realized at various places of articulation by both stops and affricates; there is also a contrast between two types of *s*, which Kim et al. (2010) argue are fortis and lenis. We will continue focusing on stops.

One series is aspirated, the other two frequently described as showing a fortis/lenis or tense/lax contrast. All have voiceless closures, except for gradient voicing

of the lenes between vowels that are not devoiced. This medial voicing of the lenis series is consistent within words and optional across word boundaries (Kim and Duanmu 2004: 86-88). Word-initially, the stop series are differentiated by the cues of the following vowel. The lenes condition lower pitch, often with breathy voice covering at least half the vowel (Cho, Jun, and Ladefoged 2002: 220, Garellek 2010, Kim and Duanmu 2004: 64); this breathy quality is more specifically identified as slack voice on the basis of acoustic measurements in Zhou et al. (2010). The fortes trigger cues on the following vowel that resemble those of creaky voice (Cho et al. 2002, Garellek 2010: 71). The phonatory differences conditioned by the three stop series also exhibit considerable inter-speaker variation in small details, though for most of the speakers tested by Cho et al. (2002) the lenes still sounded breathier and the fortes creaker (fig. 2.11).



Figure 2.11. Korean inter-speaker variation in stop-conditioned vowel phonation, showing H1-H2 measured in dB for twelve speakers of (a) Seoul and (b) Cheju dialects. Reproduced from Cho et al. (2002: 208).

In word-initial, utterance-medial Korean stops, fiberoptic and EMG experiments by Hirose et al. (1973) found VOC activity peaking just before release in the fortes, rising to a somewhat lower peak located shortly after release in the aspirates, and rising to a still lower peak occurring after release in the lenes. LCA activity was similar to VOC. The same LCA pattern was seen in some word-medial stops, for which VOC data is not given. Later EMG work nuances the description of the VOC pattern: Hong et al. (1991: 30) found VOC peaking sooner and higher for fortes than lenes, while in the aspirates they found later but faster activation, to a peak comparable to that of the fortes. The VOC peak occurred just before release in the aspirates and perhaps slightly later in the fortes (it's hard to tell since contours after release are not shown). Hong et al. (1991) highlight Hirose et al.'s (1973) claim that steep VOC activation is characteristic of the tense series and disagree, at least for their speaker. Roughly in reverse of the VOC pattern, Hong et al. (1991) found PCA activity highest in the aspirates, moderate in the lenes, and minimal in the fortes. PCA peak in lenis stops was about half that of aspirate stops; PCA peak in the lenis coronal affricate was closer to that of the aspirate counterpart but shorter in duration (with more of a spike shape). Also interesting is that the aspirates /k^h/ and /ts^h/ had peak PCA activity roughly 200 ms and /t^h/ roughly 100 ms before release (Hong et al. 1991: 23-27), since in Hindi voiceless aspirates peak PCA activity has been found much closer to release (Kagaya and Hirose 1975: 30-31).

Hirose et al. (1973) found CT activity roughly constant for all three stop series, suggesting that CT activity does not explain the higher pitch that tends to occur after Korean initial fortis and aspirated stops. Given the role of LCA in decreasing the glottis and the role of VOC in tensing the vocal-fold cover (§2.2), it seems appropriate to classify the Korean fortes as tense. Some feature analyses treat the aspirates as [+tense] too, however (e.g. Kim et al. 2005; cf. Kim and Duanmu 2004), so we will continue using the fortis/lenis terminology although its cross-linguistic uses are disparate and often vague (Butcher 2004, Cser 2003, Holton 2001). Incidentally, Fischer-Jørgensen and Hirose (1974b) and Hirose et al. (1973: 200) also found a trend of increased VOC and LCA activity in the articulation of the Danish prosodic phenomenon known as *stød*. The Danish *stød* has given impressions of glottalization (e.g. Kortlandt 1988b), though Fischer-Jørgensen and Hirose (1974b) found no difference in PCA and INT activity between *stød* and non-*stød* words in a speaker that they tested. Similarly, the Korean fortes have repeatedly been analyzed as [+cst] without clear articulatory support (Kim and Duanmu 2004: 81-82, Lee 2006: 4); Ladefoged and Maddieson (1996: 55-57) prefer

the term "stiff voice", described in terms similar to the "tense" register of some male Chanthaburi Khmer speakers (§2.5.4).

Lisker and Abramson (1964: 397) acoustically measured nearly zero, moderate, and high VOT in Korean fortis, lenis, and aspirated stops respectively; similar findings by Kim (1965) are discussed in Chomsky and Halle (1968: 327). Kim (2005: 289) mentions that fiberoptic and acoustic work by Kagaya (1974) found small, moderate, and large glottal width in the same series. An MRI study of coronals (Kim et al. 2005, Kim 2005: 294-295, 298) found that peak glottal width, glottal height, tongue blade height, duration of stop closure, and duration of peak glottal height increased in the order lenis, aspirated, fortis, suggesting coordinated tongue and larynx movement and simultaneous but independent control of glottal width. Kim (2005: 302) correlates glottal width with the decrease in airflow measured across the same three series respectively by Cho et al. (2002). Further, aspirates and fortes "have more extended palatal contact" than lenes (Kim 2005: 303), and this effect is noticeably greater word-medially than word-initially (Cho and Keating 2001). An electromagnetic articulography experiment (Brunner, Fuchs, and Perrier 2011, cf. Brunner 2005, Brunner et al. 2003) also found that the oral closing gesture in Korean velar stops accelerated faster in the order lenis, aspirated, fortis.

The feature analysis of the three Korean stop series is much contested (see Kim 2005, Kim et al. 2005, Kim and Duanmu 2004). Here we will only mention the proposal in several works (including Ahn and Iverson 2003, Averi and Idsardi 2001 and others mentioned in Kim and Duanmu 2004: 289) that the fortis stops are phonologically geminates. This proposal incurs a number of objections. The fortes are not significantly longer than other stops initially, and aspirates are almost as long as fortes medially (Kim 2005: 296-297). The duration ranking is fortis > aspirated > lenis, with the fortis and aspirated stops very roughly 40 ms shorter initially than intervocalically but maintaining about a 25 ms difference between each other. Since the aspirates in turn vary from about 90 (initial) to 40 (medial) ms longer than the lenes, we see that the aspirates are much closer to the fortes in duration, and that those two series differ by a more nearly constant,

as well as smaller, factor than the factors that differentiate them from the lenes (fig. 2.12). Kim (2005: 298) points out that the geminate analysis of the fortes does not explain either how long the aspirates are or the fact that the fortes are still slightly longer.



Mean measurements of the closure duration of the consonants in the wordmedial /a_a, a_i/ and word-initial contexts /#_a, #_i/ (based on H. Kim 2002)

Figure 2.12. Korean coronal stop closure durations (from Kim 2005: 297).

Regarding the phonetic effects of Korean stops on following vowels, Kim and Duanmu (2004: 60-64) highlight how Korean neutral-speech pitch patterns differ from the standard stop-pitch correlations of tense-high and lax-low: after word-initial aspirates and fortes there is high pitch up through the first two syllables of a word, while after word-initial lenis stops and nasals there is low pitch on the first syllable and high pitch on the second. There is considerable variation in pitch from third syllables onward, but this was not found to depend on the initial consonant. Medial consonants also do not categorically affect pitch patterns. Kim and Duanmu (2004: 64) summarize several phonetic studies on Korean with a list of correlates in fortis ("tense") vs. lenis ("lax") stops, adapted in Table 2.2.

| | Fortis | Lenis |
|----------------------------------|----------|--------|
| Following pitch | higher | lower |
| VOT | shorter | longer |
| Glottal opening | narrower | wider |
| H1-H2 | smaller | larger |
| Intensity | stronger | weaker |
| Airflow at release | lower | higher |
| Oral air pressure before release | lower | higher |

Table 2.2. Phonetic differences between Korean initial fortis and lenis stops (adapted from Kim and Duanmu 2004: 64).

2.5.5.3.2 Wu Chinese

A number of northern and southern Wu dialects of Chinese including the Shanghai dialect have three contrastive stop series: voiceless aspirates, voiceless unaspirated stops, and a series classified as slack by Ladefoged and Maddieson (1996: 64). According to Cao and Maddieson (1992), this third series is voiceless in prestressed and utterance-initial positions, with lowered pitch in the following vowel and breathy voice on the first part of the following vowel. For detecting breathiness, Cao and Maddieson used H2-H1 and A1-H1 (where A1 is the most amplified harmonic in the F1 region, cf. Jessen and Roux 2002). Medially before an unstressed vowel, the same phonological series is voiced during closure. Cao and Maddieson did not find significant breathiness in the vowels following these voiced allophones; they report that most other studies agree on this point, with Ren (1987) an exception. They also made aerodynamic measurements and found higher transglottal airflow in syllables with an initial slack stop than than in syllables with a non-slack voiceless unaspirated initial. From acoustic and transillumination data by Ren, they report minimal or no difference in VOT between the slack stops and the other voiceless unaspirated series, but a wider glottis just before release in the former.

Gao et al. (2011) however found peak glottal width occurring earlier in the slack stop series than in the other voiceless unaspirated series in the Shanghai Wu dialect. They also detected slack voice in onsetless and nasal-onset syllables bearing certain tones, suggesting that in the Shanghai dialect, slackness is synchronically a property associated with some tones rather than with stops alone (further on the tonal restrictions see Chen 2008). Steed (2011) argues from tone sandhi patterns that slack voice is a property of morphemes in the Qingtian Wu dialect (cf. Rose 1994 arguing for both depressor and register components in Wenzhou Wu tone representation). Steed (2011: 1895) also mentions that the non-slack unaspirated stops of Qingtian Wu often have free variation between voiced and voiceless closures, a trait he calls rare for Wu dialects and one that we will see again in Sivas Armenian (§2.5.5.4). This should not be confused with crossdialect variation between slack and voiced stops: Cao and Maddieson (1992) cite reports that in some Wu dialects, the cognates of the slack series have full closure voicing even in prestress position.

Cao and Maddieson (1992) note that tonal distinctions are fully or nearly neutralized in unstressed vowels. Chen and Downing (2011: 260) say that tone is assigned to a "roughly...bisyllabic" domain in the Shanghai dialect as opposed to a polysyllabic domain in Zulu and Xhosa. Chen and Downing (2011) highlight the apparent complementarity in the phonetic realization of the contrast between the slack stops and the other voiceless unaspirated series: utterance-initially and in prestress position, the contrast depends on the phonation and pitch of the following vowel, while medially in unstressed syllable onsets it depends primarily on stop closure voicing. Another Chinese dialect reported with the same three types of initial stop as described for Shanghai Wu is the Lengshuijiang dialect of Hunan province (Zhang 2009). The stop system in Wu dialects like Shanghai has several similarities with Korean (Kim and Duanmu 2004: 89): in both languages, besides a contrastive voiceless aspirate series, there is an initial contrast where fortis stops are voiceless followed by higher pitch and lenis stops are voiceless followed lower pitch and some degree of breathiness, while medially the fortes and lenes are voiceless and voiced without categorical effects on pitch. The lenes differ from the voiced series of languages like French in being initially voiceless and apparently in having more salient pitch-lowering effects. If features are innate, we must ask whether this difference is small enough for the lenis stops of Korean and Wu still to be viewed as phonologically voiced. If features are emergent, the question is moot (§3.3).

2.5.5.4 Voiceless slack stops and voiced stops in phonetic contrast

Khachaturian (1983) mentions several Eastern Armenian dialects that have voiceless stops followed by breathy voice in word-initial position with plain voiced medial allophones. Unfortunately, she does not present the inventories or much other detail.

Free variation between voiced and voiceless realization of non-slack unaspirated stops in Qingtian Wu was already mentioned (§2.5.5.3.2). The allophones with closure voicing are thus in contrast with slack stops in initial position. I refer here only to the phonetic realizations of the contrasting sounds; their supposed phonological specification is a separate topic (§3.3.7.2; for a similar situation with nasalized laterals see §4.3.3.2).

A similar contrast is reported in the Western Armenian dialect of Sivas (or Sebastian; Adjarian 1899, Busetto 2007: 57, Kortlandt 1985: 188, Scripture 1901: 369-370, Vaux 1998b: 8-9). This dialect is described as having three contrastive series of stops: voiceless aspirate, plain voiced, and a series that conditions breathy voice on the following vowel. Adjarian (1899) coordinated measurements of oral egressive airflow and voicing for all three types of stop in word-initial position, detecting vocal fold movement with a capsule externally attached to the larynx. In the plain voiced series he found voice starting 15-80 ms before stop release, while in the breathy series he found substantially higher airflow upon release and voice starting 20-30 ms after stop release, and in the voiceless aspirates VOT was roughly 10-40 ms. These figures in his discussion of the three series (pp. 125-127) reflect the tokens for which he presents data (pp. 120-123). In particular, the latter show roughly 60-80 ms of closure voicing in the plain voiced series for the stops properly speaking, with much lower or negligible closure voicing in the affricates of the same series. Substantial closure voicing variation for individual phonemes within this series is not reported.

Adjarian's description of the breathy series agrees with portrayals of so-called voiced aspirates in Armenian dialects more generally, where these sounds are sometimes called "half-voiced" (Vaux 1997, 1998b) or "voiceless breathy" (Pisowicz 1998) and described as having voiceless closure with audibly breathy offset (Khachaturian 1983, Sievers in Rousselot 1899: 132, Weitenberg 2002: 149) associated with lowered pitch (Allen 1950: 200, Khachaturian 1992: 123). Following the release of the breathy-offset stops in Sivas, Adjarian (1899: 125) says "la masse d'air employée est... considérable, ce qui me donne l'impression... d'un bruit confus dans la gorge". It is unclear whether this turbulence is a result of glottal widening, vocal fold slackening at least partly mechanically triggered by larynx lowering, or some combination (§2.5.5.1).

The Eastern Armenian dialect of Nor Bayazet also appears to have an unpredictable and thus lexically specified distinction between initial voiceless slack and plain voiced stops: in addition to inherited tokens of the former it has borrowed words with initial plain voiced stops from literary Armenian, though there do not seem to be any minimal pairs (Khachaturian 1992). To this situation one may compare an unpredictable place distinction reported as dental vs. alveolar in Mapudungun (or Mapuche, formerly known as Araucanian), a language of debated genetic affiliation spoken in Chile and Argentina: here again minimal pairs do not seem to be documented, and the contrast itself may now be moribund (Smeets 2008, cf. Chomsky and Halle 1968: 312, Salamanca and Quintrileo 2009).

2.5.6 Conclusion

Slack stops are relatively under-documented, with perhaps the most familiar examples being from Javanese and Shanghai Wu Chinese thanks to Ladefoged and Maddieson (1996). We have seen similar stops in several other languages, with a number of intriguing phonological parallels. These include a contrastive voiceless unaspirated series conditioning higher pitch and often associated with variable laryngeal constriction realized as ejective release or as creaky or stiff voice on the following vowel (cf. §2.4); there is also often a contrastive voiceless aspirate series. The slack stops have voiced allophones in postnasal position (except that in Wu and some Eastern Armenian dialects, I did not find information on postnasal realizations); the slack stops have intervocalic voiced allophones in some languages as well. Mutually unpredictable distribution of slack and ordinary voiced stops is rare but reported in Sivas and Nor Bayazet Armenian and in Qingtian Wu Chinese. Phonological implications are discussed in the next chapter.

Voiceless slack stops that condition some degree of breathiness on the following vowel, as reported in Sivas Armenian, in some Xhosa and Zulu speakers, and in Korean (§2.5.5), can be transcribed as [\ddot{p} <u>t</u> <u>k</u>] etc., following a suggestion for Korean by Zhou et al. (2010: 5). Here the breathy diacritic can be interpreted as indicating a laryngeal configuration which triggers higher than modal airflow on the following vowel. The virtue of this transcription as opposed to [$p^{fi} t^{fi} k^{fi}$] etc. (used for Javanese and Wu Chinese by Maddieson 1984) is that the latter can be misunderstood as indicating breathiness with even higher airflow or with a more salient concentration at the beginning of the vowel, such as is found with the aspiration of voiced aspirates. That characterization is inappropriate for slack stops (§2.5.5.1).

We now turn finally to the subject of stop closure voicing.

2.6. Articulatory strategies for maintaining closure voicing

2.6.1 General

Voicing requires not only adducted vocal folds but also a sufficiently higher air pressure below the glottis than above it in order to maintain the transglottal airflow necessary for vocal fold vibration. In stops, all other things being equal, airflow into the closed oral cavity raises air pressure there, leading to transglottal air pressure equilibrium and voicing cessation (Gordon and Applebaum 2006, Hayes 1997, Kenstowicz 1994: 36, Ohala 1989: 177, 1997, Thurgood 2008: 20). Under these conditions, voicing can fail before stop release if the stop is long enough.

Devoicing has a higher incidence and aspiration tends to be longer in stops with more posterior oral place articulations. Both trends have been attributed to the fact that oral air pressure builds up more quickly the smaller the cavity size (or more precisely, the smaller the cavity wall surface area: Keating 1984b: 30). On the correlation of place with voice see Blevins (2004: 12), Hayes (1997), Howe (2003: 73-74), Keating (1984b, 1985: 124), Maddieson (1984: 35-37), Ohala (1989: 177, 1997), Ohala and Riordan (1979); both voice and aspiration: Gordon and Applebaum (2006: 6-9), Helgason and Ringen (2008: 623-624). Helgason and Ringen note that the pressure buildup explanation does not work for the same place correlation which they observed with preaspiration in Swedish, but they suggest that articulator speed may fill this explanatory gap. The aerodynamic argument for voice failing sooner given smaller supralaryngeal cavities has also been extended to findings that females show less voicing leak from a vowel into a following stop than men do (German: Jessen and Ringen 2002: 206, Swedish: Helgason and Ringen 2008: 611; cf. Koenig and Lucero 2008: 1083). Overall, the research appears to support the idea that the duration of closure voicing in stops is limited by place-related

mechanical factors. The influence of such mechanical effects is limited by active strategies that show considerable inter-speaker variation, however (Midtlyng 2011: 114-115).

Clements and Osu (2002: 305–306) describe several attested mechanisms for prolonging stop closure voicing:

These may include increasing subglottal pressure, slackening the vocal folds, or decreasing supraglottal pressure. The latter adjustment can be achieved either by venting the airstream outward through the nasal cavity during part of the occlusion, or by expanding the oral cavity...

The last of these adjustments, oral cavity expansion, is of particular interest... It can be achieved by several different but complementary maneuvers, including larynx lowering, tongue root advancement, relaxation of the soft tissues of the vocal tract walls, raising of the velum, shifting of the oral closure forward to expand cavity size longitudinally, and lowering of the jaw... Most of these mechanisms, including larynx lowering, have been observed in the production of "ordinary" voiced obstruents in better-studied languages such as English and French, often in combination.

Parker (2002: 138-139) has a similar list with further references, mentioning also the frequently shorter duration of voiced stops. Although this last trait has a clear aerodynamic function (Gordon and Applebaum 2006: 5), it has been repeatedly found in various languages that this stop durational difference also trades off with the duration of a preceding vowel in what appears to be an actively controlled fashion (Kingston et al. 2009: 9, Solé 2007; exceptional languages: Keating 1985: 120-124; interaction of duration with F0 in perceived rhythm: Cumming 2010). Wide inter-speaker variation in stop closure duration (e.g. in Kabardian: Gordon and Applebaum 2006: 5-6) also indicates active control of the duration of singleton stops. In a further correlation, Keating (1984b) found evidence that prestress position favors more stop closure voicing than other positions. She relates this to the higher subglottal air pressure associated with
stress (cf. §3.3.6.2). Again, cross-linguistic variation in her results suggest some degree of active control by speakers.

It is not clear how long stop voicing can continue if supported by no other adjustments than passive vocal tract expansion in response to increasing air pressure. Ohala and Riordan (1979: 90) estimate that without such passive expansion, stop voicing would fail in 5-15 ms, but that with it, voicing lasts 50-100 ms depending on oral articulation. Clements and Osu (2002: 305) cite Westbury (1983) as estimating that stop voicing lasts 7-80 ms depending on the tension of the different tissues bounding the vocal tract. Ladefoged and Maddieson (1996: 53) consider 50 ms of voicing "consistent with models of passive cavity expansion." Traill et al. (1987: 266 citing Hutters 1985) extend the range up to 70 ms. These estimates imply that intervocalic stops can be passively voiced throughout closure if closure does not last beyond somewhere in the range of 70-100 ms. Jessen and Ringen (2002: 190) agree that intervocalic stops can be passively voiced, again citing Westbury (1983).

However, Kingston et al. (2009: 11) note that Westbury (1983) detected tongue root advancement and larynx lowering in the intervocalic voiced stops that he recorded, and that Bell-Berti and Hirose (1975) found the soft palate actively raised in postsonorant voiced compared to voiceless stops – another cavity resizing gesture which would help prolong voicing. Active cavity expansion in voiced stops is also documented in several other studies (Koenig and Lucero 2008: 1077). Both Westbury (1983) and Bell-Berti and Hirose (1975) used English data, which is significant because the laryngeal contrast in English stops is not implemented with consistent closure voicing comparable to French or Hindi. Iverson and Salmons (2003b: 49-53) model the stops in English and most other Germanic languages as having an aspiration contrast, with stops only voicing passively to the degree that intersegmental conditions aerodynamically favor this; Jessen and Ringen (2002) apply the same analysis to German with more acoustic detail. Yet neither of these works (nor Ohala and Riordan 1979, Ladefoged and Maddieson 1996: 53, nor Traill et al. 1987: 266) mention measurement of larynx height or tongue root advancement in their estimates of passive voicing duration. Kingston et al. (2009: 9-13) conclude that systematic passive voicing throughout the closure of intervocalic stops in a language has not been confirmed, and suggest that it does not occur. Keating (1985: 124) infers from additional cross-linguistic evidence that "voiceless unaspirated stops are favored, even in intervocalic position, contrary to popular belief".

Nevertheless, Jessen (1998: 306-307) reports the intervocalic unaspirated stops of seven German speakers averaging 36-60 ms of voicing followed by 6-16 ms (13-16 ms for five speakers) between voicing failure and closure release. Some tokens have even less voicing and a longer voiceless portion, and some tokens are short (around 30 ms) and fully voiced (e.g. Jessen and Ringen 2002: 199). It is unclear why voice leak in the range of 30 ms would require active voice sustaining efforts, and the idea that it can occur passively would explain why it seems actively prevented in some languages (e.g. Hindi with its clearly contrastive phonetic voicing §2.3.4, vs. Xhosa, Zulu, and English which favor other interpretations of their laryngeal contrasts §2.5.5.2). In Jessen and Ringen's German data, the relatively short duration of voicing leak into unaspirated intervocalic stops much of the time and for several of the speakers, as well as the trend of voice failing sooner in more posterior stops (Jessen 1998: 322-324, Jessen and Ringen 2002: 205) and sooner in females (Jessen and Ringen 2002: 206), suggests that any active voicing support is very limited. The mechanical factors which plausibly contribute to such place or gender variation can clearly be overridden by speaker intention, as seen in any language where speakers show more complete or consistent closure voicing across such variables.

As mentioned by Jessen (1992: 324), Docherty (1992) concluded that his British English subjects do actively control voicing, given their lack of correlation between voice leak duration and stop place anteriority. Docherty (1992: 120-149) reports a number of other factors indicating stop voicing control in British English. Inter-speaker range in VOT differences by place was considerable, as was VOT range for both stop series; there was significant VOT overlap between the two series in some environments, including VOT overlap between homorganic members of the two series in some speakers – an ambiguity which Docherty interprets as not impeding comprehension due to top-down processing (expectations drawing on other grammatical and lexical information available from the context; cf. Menn 2010). Docherty (1992: 139) concludes that much of this inter-speaker variation is due to the varying implementation strategies of different speakers, since it is unclear how the details could be modeled as purely mechanical.

The question of how long passive voicing can extend into a stop remains unanswered. Since it is independently evident that speakers sometimes control other noncontrastive phonetic properties (e.g. control of V-C duration ratios earlier in this section, active generation of negative oral air pressure in §2.6.2, and enhancement in §3.2.4), it is not clear what relationship we should expect between the degree of control that English or German speakers exert over intervocalic stop voicing, on one hand, and the contrastive specifications of stops in these languages, on the other (§3.3.9).

2.6.2 Non-explosion

Voiced implosives have traditionally been distinguished from ordinary voiced stops with the additional characteristics of laryngeal constriction, negative oral air pressure generated by lowering the constricted larynx during closure, and ingressive airflow upon release; however, a growing body of research reveals that none of these traits is necessary for implosives (Clements and Osu 2002: 302). This challenges theories which specify implosives with [+constricted glottis] (e.g. Halle and Stevens 2002[1971]: 51) or [+lowered larynx] (Avery and Idsardi 2001: 44).

Maddieson (2003: 28) writes that implosives tend to be modally voiced rather than laryngeally constricted in Bantu languages. Ladefoged (2009[1964]: 6-7) found that the velarized bilabial implosive in three dialects of Igbo (Niger-Congo) had negative air pressure only about eight percent of the time, while three quarters of the time it had between one and three quarters of the oral air pressure found in the plain voiced explosive /b/ in similar environments. Along similar lines, Clements and Osu (2002: 316–334) found that the voiced bilabial implosive of Ikwere (Niger-Congo) is typically characterized by the absence of positive oral air pressure buildup rather than by negative oral air pressure or ingressive airflow, and that the sound relies on neither larynx lowering nor laryngeal constriction; instead, it evidently avoids pressure buildup through passive expansion of the oral cavity by incoming air as well as active lip protrusion, jaw lowering, and perhaps tongue root retraction or velarization. Clements and Osu found no evidence for velar stopping and hence did not classify this sound as labial-velar, though labial-velars can possess oral air pressure rarefaction effects like those of implosives (Greenberg 1970: 128-129).

This creates a few challenges. A terminological challenge is evident even in Ladefoged's (2009: 6-7) remarks, calling implosives ingressive but immediately adding that they are usually not ingressive in the languages that he measured because air entering the oral cavity from the lungs during stop closure tends to counteract any oral airpressure rarefying effect from larynx lowering. This is not always the case; for example, most of the implosive tokens measured and reported in Seereer-Siin (Niger-Congo; McLaughlin 2005) have negative oral air pressure. Yet the fact still remains: for implosives, "it is often the case that the airflow through the glottis" is enough to keep oral air pressure "from becoming negative" (Ladefoged and Maddieson 1996: 82), suggesting that in some languages, including those mentioned by Ladefoged (2009) and Ikwere as measured by Clements and Osu (2002), the articulatory goal for implosives is to ensure substantially lower oral air pressure than in contrastive voiced explosive stops – not necessarily negative oral air pressure.

Clements and Osu (2002: 299) subsume both truly imploded stops and stops that simply avoid substantial positive oral air pressure buildup under the umbrella term *nonexplosive* because the two do not seem systematically distinguished in any language. I follow suit, adding a hyphen to the neologism (*non-explosive*). Clements and Osu (2002) further argue that non-explosive stops are not obstruents; one of their reasons for this claim is that like sonorants, non-explosives are typically voiced (a fact also noted by Maddieson 1984: 111-115; further on the correlates of sonorance see §3.3.8).

Since truly imploded stops and other kinds of non-explosives do not seem to contrast in any language, grouping them together seems phonologically useful. The noncontrastive category probably extends beyond non-explosion, though, including preglottalized and creaky-voiced stops regardless of their oral air pressure levels during closure. According to Clements and Osu (2002: 300, 313) and Greenberg (1970: 125), the only known contrast among all these types of stops is between a voiceless (closed-glottis) type and a voiced one; their remaining differences are notably gradient and prone to under-documentation. To the degree that the increased laryngeal constriction in creaky as opposed to modal voice temporarily decreases transglottal airflow, it would seem to be another potential strategy for prolonging closure voicing. Creaky voice and preglottalization are sometimes found co-occurring with larynx lowering and other cavity-expanding gestures in voiced stops (Ladefoged and Maddieson 1996: 55).

Non-explosion itself is not always contrastive. Nyangi (Eastern Sudanic) reportedly contrasts voiceless stops and voiced non-explosives without a separate series of voiced explosive stops (Flemming 2004: 259-260, Maddieson 1984: 113; for further examples in Chinese dialects and Bantu subgroups see Cun 2005: 1, Grégoire 2003: 353). In some languages, voiced explosive stops are reported as a narrowly conditioned allophone of voiced non-explosives whose distribution is less restricted (Greenberg 1970: 134-135). Non-explosion and explosion can even be in gradient variation with each other. Murrinh-Patha (Australia: Butcher 1992) has two contrastive stop series reported as "fortis" and "lenis"; they typically differ in voicing, but they have other differences as well:

Visual observation of the Murrinh-Patha speaker at the time of recording confirmed that the larynx is lowered quite substantially during the *lenis*

articulations, presumably to facilitate the prolongation of glottal pulsing into the stop closure. This is clearly the reason for the low, fluctuating, and sometimes negative intraoral pressure values recorded during these fully voiced stops (in the case of this speaker's initial *lenis* stops, more than half the tokens were fully implosive). (Butcher 1992: 288, emphasis original)

Similarly, "many languages of South America have a full set of voiceless stops (say, /p t c k/) and a contrastively defective series of voiced stops which are often imploded, namely /b d/" (Kaufman 1968: 154, listing Cayapa, Colorado, Aguaruna, Cashinahua, and Huitoto as examples). Prenasalized and preglottalized variants of a voiced stop series are also reported in Oi (Mon-Khmer: Bahnaric; Sidwell 2003).

2.7 Conclusion

This chapter highlighted multiple roles of active vocal fold abduction in voiceless unaspirated stops, whisper (with laryngeal constriction), breathy voice, and release of glottal stops, as well as multiple articulatory strategies for sustaining voicing during stop closure. Some of the latter strategies are also found in slack stops that have entirely or mostly voiceless closures. Phonological implications of the many-to-many relationship between laryngeal gestures and cues are explored in the next chapter.

3 Laryngeal Phonology

3.1 Introduction

This chapter treats several topics in laryngeal phonology, drawing on notions of economy and enhancement to model a variety of cross-linguistic trends in the laryngeal structure of stop inventories without recourse to innate feature co-occurrence stipulations (§3.2). Arguments are marshaled for the emergence of features themselves, for a range of degrees of phonetic naturalness and cross-linguistic probability in features, and for a distinction between features with articulatorily and auditorily organized implementations. Several major proposals about particular laryngeal features are assessed, and sources of feature ambivalence are discussed along with implications for innate vs. emergent feature theories (§3.3). Feature Geometry is addressed similarly, with arguments for it emerging from phonetic sources in a cross-linguistically variable way; asymmetric interactions between the larynx and stricture receive special attention here (§3.4). The overall picture suggests, finally, that enhancement and positional partial neutralization differ from the default type of feature-correlate mapping primarily in the number of phonetic correlates involved (§3.5).

3.2 Economy and enhancement

3.2.1 Introduction

A notion of phonetic economy may explain why the stops in languages without laryngeally contrastive stop series are typically voiceless unaspirated or laryngeally unspecified (§3.2.2) and why positional neutralization of all surface stops or obstruents to voice is unattested (§3.2.3). Economy is violated by phonetically unmotivated complexity but not by phonetically motivated enhancements. Such enhancement occurs when a contrast is realized in some languages with a larger bundle of naturally related phonetic correlates than is found in the realization of a similar contrast in some other languages. The relations may be articulatory, aerodynamic, or acoustic. I present the results of a large typological survey showing that in dozens of languages, enhancements of voiced stop series involve gestures that aerodynamically facilitate voicing (§3.2.4).

Several of the same gestures are found realizing contrasts responsible for preponderantly voiced stop inventories (inventories where more than half the contrastive stop series are voiced). This type of inventory is interesting since voice is generally considered "marked" in stops (§2.3, 2.6); preponderantly voiced stop inventories have even been considered impossible, a claim that has impacted the discussion of how to reconstruct the stop system of Proto-Indo-European (see Stewart 1973: 141, 1993: 4). Also noteworthy is the lack of evidence for aspiration as a contrastive property responsible for preponderantly voiced stop inventories. The diachronic development of both contrastive and non-contrastive breathy offsets on voiced stops appears attested and phonetically explicable, but tends not to develop a clear aspiration-like quality (high airflow localized at the start of the vowel) without closure devoicing (§3.2.5).

This suggests an insight into a famous statement by Roman Jakobson about the restricted distribution of voiced aspirates. The fact that aspiration has no apparent phonetic basis as an enhancement on voiced stops and the fact that aspiration seems unattested as the factor responsible for preponderantly voiced stop inventories may share a common explanation in economy (§3.2.6). The same concept of economy in conjunction with the understanding of voice-supporting strategies in stops also plausibly explains the lack of evidence for diachronic voicing processes specifically targeting ejectives except as a result of oral air pressure decrease triggered at an intermediary preglottalized stage (§3.2.7).

3.2.2 Languages with a single stop series

Maddieson (1984: 27) observes that in all of the languages he surveyed with only a single phonological stop series, the series is voiceless – with a doubtful possible exception from Australia where the series was reported as voiced. Ladefoged and Maddieson (1996: 53) "suspect, however, that... in... most of the Australian languages, the [single series of] stops may be produced with no actual [glottal] opening required, with vibration ceasing due to lack of efforts to sustain it." In other words the stops retain more or less the same adducted-arytenoid position seen in neighboring vowels and in prephonation (§2.3.1), with closure voicing occurring passively during the early part of post-vocalic stops or throughout the stop if it is short enough, as might occur in postnasal position (§2.5.5.2.2). Alternatively, in inter-voiced position, such stops might involve moderate voice-sustaining strategies like cavity-resizing gestures (§2.6) as a way of simplifying the acoustic score by generalizing voice throughout the stop and then resume again for the following sonorant.

These possibilities seem reflected in Clements' (2005b: 19-20) more detailed discussion of the matter. Noting four languages, "all spoken in Australia, in which *all* obstruents are classified as voiced" in the expanded UCLA Phonological Segment Inventory Database (UPSID), he found a different story upon closer examination of individual grammars:

For example, in Mbabaram, according to a more recent description, "the allophony is as follows: (i) stops are normally voiceless in initial position, in final position, and in medial position after y; (ii) they are voiced after a nasal; (iii) they alternate between voiced and voiceless in medial position between vowels, or after l, r, or t" (Dixon 1991: 355).

Similarly Clements found that in Dyirbal, voicing was more common between vowels, while voiced and voiceless stop allophones were free variants word-initially. In Yidiny, stops were mostly voiced but sometimes partly devoiced word-initially, which would be consistent with a higher incidence of pressure-reducing gestures. Clements continues:

Generalizing over Australian languages, Yallop observes: "The plosives of aboriginal languages may be pronounced sometimes as voiced sounds (b, d, etc.) and sometimes as voiceless sounds (p, t, etc.) – but the voiced and voiceless counterparts are either freely interchangeable or in complementary distribution" (Yallop 1982, 56)...

The absence or rarity of languages where the stop inventory is limited to a single series systematically realized with voice throughout most or all of stop closure is paralleled by the absence or rarity of languages where the stop inventory is limited to a single series systematically realized with any laryngeal configuration other than unaspirated with fully or mostly voiceless closure. One possible explanation for this pattern is that other laryngeal configurations require more activity in terms of glottal widening, ejection, or voice-sustaining strategies. This perspective is reflected in the common phonological analysis of voiceless unaspirated stops as default or "unmarked" in various senses (§2.3.1; Haspelmath 2002, 2006, 2008a, Steriade 1995: 152-153 for critical reviews and discussions of the concept of markedness).

3.2.3 Neutralization

The previous section suggested that stops in inter-voiced position might be targeted by moderate voice-sustaining strategies such as cavity-resizing gestures as a way to simplify the acoustic score by generalizing throughout stop closure the voicing that would otherwise passively leak into only part of the closure. Intriguingly, languages that neutralize all stop series to full closure voicing in any position other than inter-voiced seem rare or non-existent. In each case that I have seen, neutralization does not affect all surface stops in the language; there are always at least some segments that surface as voiceless stops in the neutralization environment. Since this topic relates closely to some important recent debates, brief elaboration is in order.

Blevins (2006b) argues that in a number of languages, one or more series of obstruents voice in syllable- or word-final position. Kiparsky (2006) rejects all her examples. A review of the evidence suggests that some stop series may indeed voice in final position in some of the languages, though the details also seem open to alternative interpretations; either way, however, we still find contrastive voiceless stops occurring in syllable- and word-final positions.

Lezgian (Nakh-Dagestanian: Yu 2004) has an alternation between plain voiceless and voiced stops where the voiceless alternant occurs in onsets and the voiced one in codas at the end of a large set of monosyllabic stems (these codas are word-final unless followed by suffixes or by the second element of a compound). The alternating stops contrast with non-alternating voiced stops, voiceless aspirates, and ejectives, all of which are found in both onsets and codas including monosyllabic word-final position. The question is how to represent the alternating pair phonologically. If the onset alternant is taken as basic (as in Blevins 2006b: 150-152 and Yu 2004), then Lezgian has a pattern in which otherwise plain voiceless stops are voiced in codas.

Kiparsky (2006) instead takes the coda allophone as basic and underlyingly geminate, treating the alternation as a case of onset devoicing and degemination. Yet while the coda alternant does appear to be the historically more conservative one, it is not clear whether Lezgian learners would consider it either underlying or geminate. As seen in Yu (2004), its closure duration is about a quarter longer than the duration of its voiceless intervocalic onset alternant, about a third longer than onset non-alternating voiced stops, and about a fifth longer than coda non-alternating voiced stops. Would these length differences provide a sufficient basis for treating the coda alternant as

geminate while treating all the other sounds just mentioned as singletons? Kiparsky notes that onset devoicing occurs in other languages but does not provide examples where voiced or any other kind of geminate stops surface only when not followed by a vowel. In fact, Yu's (2004) historical analysis is that the coda alternants were singletons and that they geminated in onsets (which for independent reasons were generally pretonic), subsequently devoicing and then degeminating in Lezgian.

The closure and voicing duration differences between alternating and nonalternating coda voiced stops - 25 and 34 ms average in the tokens measured - shows that they do not completely neutralize (Yu 2004: 81-83; see §3.5). If Lezgian does not involve categorical neutralization of alternating and non-alternating coda voiced stops, then it is not a counter-example to Kiparsky's claim that such neutralizations are nonexistent. It would still involve final voicing of an underlyingly voiceless or laryngeally unspecified stop series if the onset alternants are taken as basic, but again it is unclear which alternant learners select as basic. Yu (2004: 76-78, 87-88) notes that Lezgian has additional lexically restricted alternations between prevocalic ejectives and word-final voiced stops or aspirates. In at least the former case, the prevocalic ejective alternant can be derived by assimilation with an underlying root-initial ejective, leaving the word-final voiced alternant as basic. The word-final voiced alternants of prevocalic ejectives are virtually identical in closure and voicing duration to those of prevocalic plain voiceless stops (the final alternants of the prevocalic ejectives average 7 ms longer in closure duration and 10 ms shorter in voicing duration than those of the plain voiceless stops in the tokens measured; Yu 2004: 81). At the same time, the restriction of both voicing alternations (with plain voiceless and ejective onsets) to particular sets of monosyllables within Yu's data could mean that neither alternation is synchronically productive; monosyllables with non-alternating voiced stop codas include both obvious loans and less easily explained forms, and Yu does not discuss other factors which might indicate which patterns are currently productive (see Yu 2004: 89-92). This seems to leave open the possibility that the putative phonological neutralization to voice in Lezgian codas is neither productive, nor a neutralization, nor a process with a voiced outcome.

In another controversial case, Somali (East Cushitic), we find two contrastive stop series, one aspirated and the other unaspirated. Unaspirated stops at the bilabial, coronal, and velar places of articulation are medially voiced between voiced segments; except for the lone (coronal) implosive, they are also spirantized between vowels unless the second vowel is a phonetic schwa offglide as described further below (Edmondson, Esling, and Harris 2004, cf. Gabbard 2010: 20-22). Word-initially, unaspirated stops other than the glottal and epiglottal-uvular are described by Edmondson, Esling, and Harris (2004) as partly voiced, with at least the bilabial being entirely voiceless in the speech of one informant. Gabbard (2010) shows 86-115 ms of voicing for the bilabial, coronal non-implosive, and velar unaspirated stops in apparently utterance-initial position (i.e. preceded by a flatline waveform), so perhaps the degree of word-initial voicing varies considerably by speaker.

Gabbard (2010: 7, 10) generalizes that non-uvular voiceless stops are aspirated, but without providing any evidence or argumentation for voice in coda stops. Edmondson, Esling, and Harris (2004: 2-5) go into more detail. At the end of a word (perhaps when not followed by a vowel-initial word, though this is not stated), unaspirated stops other than the glottal stop are followed by a schwa offglide "in careful, overly correct speech", with non-uvular ones being voiced. In the same environment in "conversational style", stops apart from the implosive are voiceless glottalized; the examples are all unaspirated, and it is stated that aspirated stops are not found in codas. Coda unaspirated stops as the first member of a word-medial geminate are also identified as voiceless glottalized. This last point disagrees with Gabbard (2010: 14, 28-29), who transcribes orthographic voiced geminates phonetically as voiced singleton stops, but he provides no experimental evidence on either closure duration or uniformity of closure voicing.

85

On the whole, regardless of how the laryngeal contrast in Somali stops is phonologically specified, it appears that only unaspirated stops occur in codas, that they are voiceless glottalized in ordinary speech, and that in especially careful speech, some underlyingly final stops are voiced with a schwa offglide. The fact that underlyingly final epiglottal-uvular stops are followed by this schwa without being voiced makes it harder to argue that the schwa is merely a result of sustaining underlying voice through stop release. Conceivably, the schwa is an artefact of stop release itself in careful speech, and the more anterior stops have become voiced before it as a result of being phonetically intervocalic and prone to greater degrees of voice leak from a preceding vowel (on the aerodynamic correlation of voice with anteriority see §2.6.1). This would entail that spirantization is more restricted than voicing since it would apply only between underlying vowels. We will not pursue the problem further here.

In addition to the descriptions of final voicing above, we also find reports in Howe (2000: 139) of stop voicing in various pre-voiced positions:

[L]anguages in which obstruents are systematically voiced before tautosyllabic sonorants [include] Mohawk (Iroquoian), in which all and only prevocalic stops are voiced... (e.g. Halle and Clements 1983:59, 121-3). Other examples include Tsimshianic Nisga'a (Tarpent 1987) and Smałgyax (Dunn 1995). As Dunn (1995:I:5) describes: "The letters b, d, dz, g, and g generally occur between vowels and before vowels... The letters k, k, p, and t generally occur at the end of words and in clusters." Yet another example is Amele (spoken in Papua New Guinea) which Roberts (1987) describes as having only voiced stops prevocalically.

However, none of these cases of voicing seem to involve all surface stops that occur in the relevant pre-voiced environments in these languages. Mohawk has initial and medial pre-sonorant voiceless aspirate stops that are phonologically analyzed as clusters (see Beatty 1974: 16-18, Bonvillain 1978: 38, 1985, Dinneen 1990: 72-82, Hall 2010: 7, Hopkins 1987: 453; on the apparently universal lack of contrast between monosegmental voiceless aspirates /T^h/ and tautosyllabic underlying-cluster voiceless aspirates /Th/ see Kehrein 2002, Kehrein and Golston 2004). Among the Tsimshianic languages, Nisgha

(Nisga'a) maintains ejectives before vowels (Brown 2008: 5, 105, 175). Smalgyax also maintains ejectives before vowels, and its pre-vocalic voicing of non-ejective stops is defined as having exceptions, such as *kyooxk* 'grass', *puksk* 'to spit, spitoon', *taagan* 'planking for a boat', *taalsgmts'ooxs* 'stocking' (see Dunn 1995: 4). In the related Gitksan, a rule voicing stops before voiced segments is reported by Hoard (1978) according to Kirchner (1998: 184) and Rigsby and Ingram (1990), but the latter found from careful examination of their fieldwork audio recordings that the rule did not affect ejectives and that Gitksan also has intervocalic phonetic voiceless aspirates from underlying clusters. Brown (2008: 80, 109-114) confirms this impressionistically. Finally, while Roberts transcribes most Amele prevocalic stops as voiced, /t/ and /d/ are listed and exemplified as separate phonemes including between vowels (Roberts 1987, cf. 1991: 128, 144; 2001: 206). More precise phonetic data on some of these languages would be helpful.

If word-initial voicing and word-final voicing of all obstruents or stops in a language are non-existent or even just rare patterns, there would seem to be something disfavored about them. Some languages are reported with general neutralization of coda or word-final stops to other laryngeal configurations besides voiceless unaspirated, but in these cases the neutralization outcomes are still voiceless (aspirated or laryngeally constricted: Iverson and Salmons 2011, Vaux and Samuels 2005). The active laryngeal gestures involved in these outcomes could conceivably be devoicing strategies motivated either by regressive assimilation to the silent respiration following an utterance-final stop, in the case of neutralization to aspiration (see Iverson and Salmons 2011 §2.3), or as part of an anticipatory generalization throughout closure of the voicing failure which naturally occurs in final stops that are unexploded or otherwise long enough to achieve transglottal pressure equilibrium. (See Kingston et al. 2009: 6 on the possible use of coda stop glottalization to turn off voicing; cf. §2.3.3.1.) While such speculations may be permanently inconclusive, they identify plausible environmental triggers for these attested laryngeal neutralizations. Such triggers do not seem available to motivate full

neutralization of all stops to voice in any position other than inter-voiced. Additionally, such triggers do not seem available to motivate the diachronic development of singleseries stop systems predominantly realized with any laryngeal configuration other than voiceless unaspirated (§3.2.2). One conceivable exception to both these generalizations is acoustically motivated voice assimilation, where stops would neutralize to active closure voice in order to become acoustically more similar to adjacent passively voiced sonorants. If this development were phonetically natural, we would expect to see it relatively often producing word-initial or final neutralizations to voice in some languages and all-voiced stop systems in others. As reviewed above, though, those patterns seem very rare or non-existent.

These observations raise the possibility that the same factor opposes both the neutralization of all stops to voice outside of the inter-voiced environment and the development of single-series stop inventories predominantly realized with any laryngeal configuration other than voiceless unaspirated. One candidate for this factor is a principle of economy such as that formulated by Clements (2005b: 1, cf. 2009): languages avoid distinguishing segments with unmotivated accumulations of multiple features at the phonological level. Since we are not presently concerned with the division of labor between phonology and phonetics, we will provisionally extend this principle to the phonetic level. The concept of *unmotivated* combinations at the phonological contrasts are realized is discussed in the next section under the aegis of another concept to which any successful definition of economy must be sensitive: enhancement. (For further investigation of how principles of economy may formally impact phonological representations see Weijer and Hinskens 2004.)

3.2.4 Enhancement

3.2.4.1 Voiceless aspiration

A contrast between generally voiceless aspirated and generally voiced stops, once thought non-existent, is in fact well-attested in Swedish and has been analyzed with the notions of enhancement and dispersion (Helgason and Ringen 2008: 624): that is, instead of differentiating the two series with a single feature such as voice or aspiration, such a system uses both strategies simultaneously. They enhance each other, making the two series perceptually more distinct or dispersed from one another (see Clements 2005b, 2009, Flemming 2003: 14-15, 2004, Keyser and Stevens 1994, Stevens and Keyser 1989, 2010, Vaux and Samuels 2005: 410-13, Weijer 2004. Blevins 2004: 10-11 discusses early dispersion proposals by Trubetzkoy and notes that the topic "can be viewed in either synchronic or diachronic terms".)

Enhancement for perceptual dispersion is widely considered plausible but also clearly needs to be constrained. For example, dispersive enhancement on its own would be satisfied by inventories in which place and laryngeal features enhanced one another – such as a language where only labials were aspirated and only velars were prenasalized – but this would clash with the principle of economy introduced above, and in fact we do not find such inventories (see Clements 2009, Maddieson 1984, Mielke 2008a, Ruhlen 1975). Stevens and Keyser (2010) discuss cases of enhancement where the enhancing element increases a phonetic property which would already be present in the sound without the enhancement, but the examples are not much concerned with laryngeal distinctions in stops. Below, we will investigate to what degree a similar rationale succeeds in explaining attested laryngeal enhancements in stops.

Before we proceed with this task it is worth noting that perceptual clarity is not the only motive which has been suggested for enhancement. Another is variation in strictness of timing. Vaux and Samuels (2005) suggest on the basis of a wide variety of evidence that non-contrastively aspirated voiceless stops are articulatorily and perceptually easier to handle than unaspirated voiceless stops in at least one respect: they require less exact control over the relative timing of stop release and vocal fold adduction (or in Vaux and Samuels' terms, voice onset. Initiation of adduction reverses a previous abduction gesture and can determine VOT.) The idea is that salient aspiration is easily perceived anywhere after a few centiseconds of voice onset delay, so preventing this percept requires deliberately maintaining a VOT below that boundary. This resembles the idea of a phonetic anchor discussed by Midtlyng (2011): in languages with contrastive stop series characterized by high, minimal, and negative VOT, the middle of these three series may exhibit less VOT variation than the others and act as a sort of perceptual benchmark from which the other two series phonetically diverge.

However, Kong et al. (2011) argue that the minimal VOT of Korean fortis stops makes them auditorily simpler because they involve fewer cues (closure rather than closure plus aspiration), and that this is why most Korean children learn to produce the fortes earlier than the aspirates (with high VOT) and lenes (with medium VOT plus pitch depression; §2.5.5.3.1). Also, Löfqvist (1980: 487) reviews a study finding that while adult American English speakers had a larger VOT range for initial aspirated than unaspirated stops, two-year olds had the reverse, again suggesting that the wide range of VOT values available to voiceless aspirated stops is not necessarily auditorily helpful during acquisition even if it does imply an opportunity for less strict articulatory timing. Kingston et al. (2009: 34-38) provide further arguments that a voice contrast is not necessarily harder to aquire or maintain than an aspiration one. Possibly neither contrast is inherently more favored than the other; at any rate, they can certainly synergize, as in Swedish.

90

3.2.4.2 Laryngeal tension

Additional laryngeal enhancements include the variable use of vocal fold stiffening and/or laryngeal constriction in an unaspirated voiceless stop series contrasting with a series characterized by vocal fold slackening (§2.5.6). In these systems there is no voiceless unaspirated stop series that is consistently neither slack nor laryngeally constricted. Historically, such patterns could conceivably have developed from an earlier voiceless : voiced contrast via an intermediate stage where gestures regulating the aerodynamics of voice by manipulating laryngeal height and vocal fold tension (§2.6) with effects on pitch (§3.3.7.2) were reinforced by more substantial vocal fold tensing in the voiceless series and/or slackening in the voiced series. Tensing of the voiceless series was then open to reinforcement from additional laryngeal constriction mechanisms, which often synergize (§2.4), while the voiced series could have been further slackened by loss of medial compression which eventually resulted in devoicing in most positions (§2.5.5). Such natural phonetic development of pitch and voice quality distinctions with loss of closure voicing is reconstructed in some Mon-Khmer languages, where the first register has cues suggestive of creaky voice and/or the second register associates with slack voice (e.g. Brunelle 2005: 145-166, Wayland and Jongman 2003; further works on register cited in §2.5.4, 3.2.6.1). This suggests that the historical development of ejection in the non-slack unaspirated series in Xhosa (see Doke 1954, Hyman 2003) and Eastern Armenian (see Pisowicz 1976, Tragut 2009: 17-18) may not have been simply the addition of a random cue to the series to distinguish it from the slack stops, but in fact the gradual increase of phonetic elements already present.

3.2.4.3 Non-explosion and prenasalization

A range of enhancements is also reported with voiced stops, including nonexplosion and prenasalization (Table 3.1). Notably, the gestures responsible for these non-contrastive properties also occur in smaller degrees in plain voiced stops as articulatory strategies for sustaining voicing (§2.6); thus again the enhancements do not seem to be random additional cues for perceptual dispersion, but involve strategies already natural to the maintenance of voicing. Further, the more pronounced velic lowering and cavity resizing responsible for audible prenasalization and implosion not only sustain voicing but tend to amplify it, arguably increasing its perceptual salience (Flemming 2004: 258-261).

| Table 3 1 | Stop | inventories | with en | hanced | voice | series |
|-----------|---------|-------------|---------|--------|-------|---------|
| | 1. Diop | Inventories | | nunceu | VOICC | Series. |

| Туре | Examples |
|---------------------------------------|--|
| /T ^N D/ | San Colorado Mixtec (Flemming 2004: 259-260), Apinaye, Kewa, Luvale, Paez, Wantoat, Washkuk (Maddieson 1984), Lusi, Umbundu, Ura (Mielke 2007a) (total 10) |
| /T D ^{<} / | Maasai, Nyangi (Maddieson 1984; differently on Maasai see Mielke 2007a); other unspecified Bantu languages and Chinese dialects (§2.6.2) |
| /T T ^{h N} D/ | Hakka, Nambakaengo (Maddieson 1984), Pileni (Mielke 2007a). Sedang (Maddieson 1984) adds a creaky-voiced stop series |
| /T T ^h D ^{<} / | Eastern Cham (Brunelle 2005: 55), Sgaw Karen, Swahili (Maddieson 1984; differently on Swahili see Mielke 2007a) |

Here and in Table 3.2, the inventory types are defined using capital coronal symbols as cover symbols for all places of articulation. The cover symbol for voiced non-explosives $D^{<}$ includes stop series reported as implosive, preglottalized, and creaky, because these articulations do not appear to contrast in any language and the distinctions

between them are subject to extreme under-documentation (§2.6.2). It should be assumed that the primary sources are impressionistic auditory fieldwork unless otherwise stated.

To sum up, we have seen several cases where enhancement can be plausibly explained as a perceptually motivated increase or facilitation of articulatory or aerodynamic properties already present in the stop, rather than merely increasing the perceptual difference between two categories of sounds by adding properties that are phonetically unrelated to the properties those sounds already have. This fits well within Stevens and Keyser's (1989, 2010) view of enhancement.

3.2.5 Preponderantly voiced stop inventories

Where a language has a greater number of contrastive stop series with closure voicing than without closure voicing, the contrasts among series with closure voicing often involve the same gestures that we have already seen used non-contrastively as voiced stop enhancements. Several dozen examples are listed in Table 3.2.

Table 3.2. Preponderantly voiced stop inventories.

| Туре | Examples |
|--|---|
| /T D D ^{<} / | Angas, Dan, Doayo, Hamer, Ik, Kadugli, Katcha, Kpelle, Logbara, Margi, Mursi, Tama, Tarok (Maddieson 1984), Bari, Degema, Dhaasanac, Egene, Epie, Gwari, Lorma, Muktile, Mumuye, Mupun, Pero, Pulaar, So, Tirmaga, Zina Kotoko (Mielke 2007a), Mbatto (Stewart 1989, 1993) (total 29) |
| /T D ^N D/ | Siriono (Maddieson 1984), Abun, Chomo, Chori, Dholuo, Gade, Kinyamwezi, Korowai, Kutep, Mende, Tiv, Yukuben (Mielke 2007a); also Acenhnese (cf. Durie 1994) except that the third series is treated as /N ^D / by Ladefoged and Maddieson (1994: 104-106, cf. Botma 2004: 204) (total 11 or 12) |
| /T ^N D D ^{<} / | Fyem (Mielke 2007a), Lawa, Stieng (Huffman 1976: 579) (prenasalization occurs on only half the second series in Fyem and is optional in Stieng) |
| /T D ^N D D ^{<} / | Gbeya, Ngizim, Sara, Yulu (Maddieson 1984), Bata, Fulfulde (a Mali dialect), Giziga, Hurza, Lele, Mada, Mofu, Moloko, Muyang, Noon, Rao, Sango, Tangale, Zulgo (Mielke 2007a) (total 18) |
| $/T T^{<} D^{N} D D^{<}/$ | Ngiti (Mielke 2007a) |
| /T T' D ^N D D ^{<} / | Dahalo (Mielke 2007a) |

3.2.6 Why voiced aspirates are different

3.2.6.1 General

While cavity resizing and velic lowering can sustain stop voicing, serve noncontrastively as enhancements, and serve contrastively to create preponderantly voiced stop inventories, I have not found aspiration in any of these functions. It is worth considering whether this could be explained by the principle of economy explored above: avoidance of the accumulation of phonetically unrelated material within a phonemic category. The change of voiced non-explosive to voiced explosive stops and the development of breathy voice after voiced stops are both sound changes that have been reconstructed in a fair number of languages, so one might expect that the occurrence of both these changes could produce contrastively breathy voiced stops in a language without voiceless aspirates. But the rise of salient breathy voice after voiced stops is typically reported with concomitant closure devoicing, as in sundry Mon-Khmer languages (Brunelle 2005: 146-147, Diffloth 1982, Ferlus 1992, 1983, 1979, 1977, Gregerson 1976: 335, Headley 1998, Huffman 1976, Pinnow 1979: 123, Sidwell 2005: 197, 2006, Theraphan 1990, cf. Thurgood 2008: 20-21. Most of these sources rely on auditory impressions, but e.g. Brunelle, Gregerson, and Theraphan include acoustic data.) Gregerson (1976: 341-342) cautions that breathy voice as well as pitch and vowel quality differences often reconstructed as having developed in tandem with devoicing of original plain voiced stops may have been present in Proto-Mon-Khmer itself, which is conventionally reconstructed with the stop system */T D D^c/ (cf. Sidwell 2005: 197, 2006).

Only one language has come to my attention where contrastive breathy voice reportedly developed on stops without concomitant devoicing. Cao Bang (Tai; Pittayaporn 2007: 110-111) is thought to have undergone */b d 6 d/ > /b d b d/. In this case *Tr > T^h is also posited, however, changing the putative Proto-Tai stop system */T D D[<]/ to /T T^h D D/ and suggesting (Weiss 2009) that the development of a contrastive voiceless aspirate series may have facilitated the development of the breathy voiced series contrasting with the plain voiced series (cf. Kiparsky 2005a on "priming" effects).

3.2.6.2 Jakobson's universal and Proto-Indo-European

The limited distribution of voiced aspirates has sparked considerable interest for a long time, since the "standard" reconstruction of the longest and most extensively studied

prehistoric language, Proto-Indo-European, posits a stop system /T D D⁶/ which has no attested parallels (ch. 6). This problem was famously highlighted by "[a]rguably the most influential voice in the history of phonological feature theory" (Morén 2007: 10), Roman Jakobson:

To my knowledge, no language adds to the pair /t/ - /d/ a voiced aspirate $/d^h/$ without having its voiceless counterpart $/t^h/$, while /t/, /d/, and $/t^h/$ frequently occur without the comparatively rare $/d^h/$, and such a stratification is easily explainable (cf. Jakobson-Halle [1956: 43f.]); therefore theories operating with the three phonemes $/t/ - /d/ - /d^h/$ in Proto-Indo-European must reconsider the question of their phonemic essence. (Jakobson 1962[1958]: 528)

Jakobson's generalization about voiced aspirates has been called "Jakobson's universal" (Fallon 2002: 12-13) and has considerable support: clear examples of voiced aspirates – with consistent closure voicing and breathy aspiration involving substantially higher airflow than modal voice – do tend to come from stop systems containing at least four series of the form /T D T^h D^ĥ/. The majority of these hail from the Indian subcontinent, though a few other Asian and African examples are reported (Clements 2003: 209-211) including some Igbo dialects (Niger-Congo: Ladefoged and Maddieson 1996: 60, Maddieson 1984: 292). Similar also is /T D T^h T^ĥ/ ascribed to Ikalanga (Niger-Congo: Bantu; Mathangwane 1999) and the system reported in Cao Bang (§3.2.6.1).

Alleged counter-examples to Jakobson's universal include Madurese and Kelabit, two Austronesian languages reported with /T D D^h/ stop systems. Closer examination of the aspirates in these languages reveals quite different phonetic profiles from the Hinditype voiced aspirates (cf. Stuart-Smith 2004: 18), though the phonological picture is more challenging as we will see shortly (§3.2.6.3f).

Limited as our typological knowledge is, numerous linguists have argued that there is still benefit in investigating possible theoretical principles which would motivate the absence of certain logically conceivable segment inventory types in the data we do have (see Croft 1993, Garrett 1991: 795-96, Kortlandt 1993, Matthews 1982). We will now consider two explanations that have been proposed for the apparent restricted distribution of Hindi-type voiced aspirates.

Jakobson and Halle's (1956: 43) explanation relies on two ideas: (1) in obstruents the primary sound source is oral, while voice is only a secondary sound source, therefore not "optimal", and (2) consonants and vowels are under a universal pressure toward acoustic dispersion: consonants are acoustically "primarily characterized by reduction of energy" in contrast to vowels, so voiced aspirates are not "optimal" since they have unnecessary acoustic energy. What it means to call voice a secondary sound source and how this status is meant to oppose the non-contrastive occurrence of voice in obstruents is not clear to me; neither is the basis for the acoustic-intensity dispersion claim. Various works have claimed that languages often do increase the perceptibility of some contrasts through dispersion of acoustic cues $(\S3.2.4)$, though the generality and force of such trends seem limited (Hale, Kissock, and Reiss 2006). Ladefoged (2006) and Kirchner (1998) are examples of works recognizing a role of articulatory effort reduction among (at least diachronic) influences on phonology. But the plausibility of such arguments depends heavily on the details of particular cases, and the wider scope of such ideas in shaping phonologies as a whole is still unclear. For example, Jun (2004) persuasively argues that the relative infrequency of heterorganic nasal-stop clusters relates to their being harder to perceive, not more complex to produce, and Henke et al. (in press) argue that Modern Greek favors fricative-stop realizations of obstruent clusters because these are easier to perceive, even though this does result in more complexity of manner of articulation than fricative-fricative or stop-stop clusters. Further examples are treated in ch. 4, along with the question of whether and in what sense sound changes can optimize phonologies.

Another explanation offered for the apparent restriction that Jakobson identified on voiced aspirates has to do with *economy*, as summarized by Salmons (1993: 12, cf. Barrack 2003a: 3-4):

Martinet (1955:114-115, see now Martinet 1991 as well) makes a claim similar to Jakobson's somewhat earlier, namely that "a series of the type bh, dh, gh only appears to be attested in languages in which there also exist a series of voiceless aspirates, ph, th, kh." He finds this easily explicable in terms of economy. That is, a voiced aspirate series would be doubly marked – by voice and aspiration – within a three-series system.

Clements (2003: 309-311) also uses economy to explain his research finding voiced aspirates only in languages with both voiceless aspirates and voiced unaspirated stops. More general discussions adopting a favorable view of feature economy include Clements (2003, 2009), Ladefoged (1999: 587), and older works noted in Helgason and Ringen (2008: 624). In the previous sections we have seen evidence that economy is sensitive to enhancement, however, and the examples of enhancement that we reviewed involved the exploitation of phonetic material already present in the target sounds rather than the accumulation of phonetically unrelated properties. Defining feature economy as the avoidance of that type of accumulation apparently provides a more specific explanation for Jakobson's universal than acoustic dispersion, acoustic "optimality" of sound sources, or economy *tout court*. This explanation also covers more ground by simultaneously explaining the more general rarity or absence of aspiration functioning as an enhancement of voiced stops or as the crucial contrast responsible for preponderantly voiced stop inventories (§3.2.6.1).

Such typological generalizations need not be exceptionless to retain value within frameworks like Evolutionary Phonology (Blevins 2004) and Emergent Feature Theory (Mielke 2008a). Approaches of that sort tend to generate more detailed proposals about trends in language change and acquisition than about the set of possible languages. A basic insight is that inventories which sound changes are unlikely to produce should be rare, though they may occur (cf. Morén 2007: 21). The advantage of this perspective is that it seeks explanations for trends without insisting on unprovable all-or-nothing predictions about what is possible (which Maddieson 2007: 104 considers a permanently invalid exercise anyway). It is along such lines that Weiss (2009) examines the evolution

of voiced aspirates, highlighting *inter alia* two Austronesian languages that have attracted attention as possible counter-examples to Jakobson's universal. These are considered next.

3.2.6.3 Madurese

The Madurese stop series transcribed as voiced aspirates bh dh gh etc. contrasts with voiceless unaspirated and voiced unaspirated stop series (Barrack 2003, Hock 1991: 625-26, Woodhouse 1995), but the limited experimental evidence available on this language only gives evidence for a voiceless realization of the aspirate series (Cohn and Lockwood 1994, Cohn and Ham 1998). Both the aspirates and the plain voiced stops lower F1 and F0 in a following vowel (Cohn and Lockwood 1994: 79, 86), traits normally associated with voiced obstruents (see Blust 2004: 133, Mielke 2008a: 94-95, Vaux 1998b). Cohn and Lockwood found F0 lower by an average of 60 Hz for the female speaker and 20 Hz for the male speaker at voice onset after aspirated than unaspirated voiceless stops. For comparison, in a male Hindi speaker (Kagaya and Hirose 1975: 37) and in female Taiwanese speakers (Lai et al. 2009), F0 respectively averaged 10 Hz and 3 Hz higher at voice onset after aspirated than unaspirated voiceless stops, while studies of Thai (Perkins 2011: 2, 22-23, cf. Downing and Gick 2001: 74) have found F0 higher after aspirated than unaspirated voiceless stops in some speakers, with the reverse in other speakers and no significant effect in still other speakers. While pitch lowering and tongue raising could reflect a voiced historical origin of the Madurese aspirates, their observed voiceless closure and the conflicting cross-linguistic results on the pitch perturbation effects of aspirated vs. unaspirated voiceless stops weigh against analyzing them as synchronically voiced. No particular confidence can be placed in the use of voiced symbols in the standard orthography or in an ambiguous phonetic description of the aspirates as "soft" in a grammar from over a century ago (Kiliaan 1911 in Barrack 2003). On the basis of the pitch effect, Trigo (1991: 129) alludes to the Madurese aspirates as

similar to the Javanese slack-voiced stops, citing Kenneth Stevens, but the latter (p.c.) is decidedly agnostic on the laryngeal phonetics of the series.

On historical and comparative evidence, a voiced predecessor for the Madurese aspirate series is inferred by Stevens (1966), who considers the possibility that the Madurese aspirates and voiced stops split from a single parent voiced series with the conditioning environments being subsequently effaced by dialect admixture. There seem to be several typological parallels for the elements of this scenario. A complex array of historical factors has created a remarkably large number of doublets within Austronesian as a whole (Blust 2011). As for the diachronic addition of laryngeal contrasts to a stop inventory, Hajek (2010) finds aspiration developing by a combination of borrowing and lexically-diffusing innovations (often in initial voiceless stops with a liquid coming soon after) in several languages of the Malay peninsula that have had contact with Thai, which has phonemic aspirates. Hajek and Bowden (2002) describe how the Austronesian language Waimoa has developed contrastive voiceless aspirates, along with contrastive ejectives (currently known in only one other member of the family, Yapese). Other languages which appear to have enlarged their phoneme inventory by innovating most or all of their aspirated phonemes include Kusunda (Nepali isolate: Watters 2006) and Limbu (Sino-Tibetan: Mielke 2007a).

Alternatively, Nothofer (1975) derives the Madurese voiced stops from earlier glides and the aspirates from earlier voiced stops. Regardless of whether this or Blust's hypothesis is more accurate, positive evidence seems lacking for a stage at which the Madurese stop system possessed both a plain voiced series and a contrastive aspirate series with systematic voice during either closure or offset. By the time Madurese gained a contrast between aspirates and plain voiced stops, the aspirates may well have been completely voiceless. Arguments that voiceless stops which trigger pitch lowering on a following vowel are synchronically phonologically voiced in some abstract sense will be treated below (§3.3.7.2).

3.2.6.4 Kelabit and Lun Dayeh

In the Bario dialect of Kelabit and the closely related Long Semado dialect of Lun Dayeh, we find voiceless unaspirated stops and voiced unaspirated stops contrasting with "voiced aspirates" phonetically realized as [DT^h], where the aspiration is "apparently" weak or absent for "many" speakers at least in Bario (Blust 2006: 316, cf. 313). These sounds resemble clusters or geminates in their relatively long duration and in their restriction to intervocalic position in both dialects. Their partial devoicing and aspiration can be plausibly explained as having developed from oral air pressure buildup in earlier voiced geminates, a reconstruction strongly supported by comparative evidence (Blust 2006). Historically the aspirates derive from voiced geminates that developed in turn from earlier clusters in CVC reduplications and from single stops following initial stressed schwa. The "schwa is retained in most dialects of Lun Dayeh and Kelabit, but initial schwa from any source has been lost in other North Sarawak languages," where the reflexes of the voiced geminates thus occur both initially and medially and include voiced implosives, voiceless fricatives, and both voiced and voiceless stops (Blust 2010: 50). Exceptionally, Long Lellang Kelabit has aspirate reflexes of the geminates, optionally drops initial schwa before the aspirates (e.g. /(ə)bhar/ 'loincloth', /(ə)dhəh/ 'one'), and has lost earlier initial schwa everywhere else (Blust p.c.).

It is unclear whether the Kelabit aspirates should be treated as occupying one or two phonological timing slots. Blust (2006: 313-318) presents several arguments for a singleton analysis: (1) all other surface clusters result from schwa syncopation across a prefix-stem boundary, (2) phonetic aspiration does not otherwise occur in the language, nor does [\int], which often occurred in the release of the alveolar aspirate for some of his speakers, (3) high vowels lower before non-glottal codas but do not lower before aspirates, (4) the aspirates alternate under affixation with plain voiced stops, and (5) the cognates in various other Kelabit and Lun Dayeh dialects are single stops or fricatives. Individually, these arguments seem inconclusive. The first two explain some phonotactic behaviors at the expense of others: why are the aspirates phonetically fairly long, why are they banned from initial position in most Kelabit and Lun Dayeh dialects, and why in Long Lellang Kelabit does initial schwa delete optionally before the aspirates when it has fully deleted before other consonants? Against the duration objection, Blust (2006: 317) suggests that voiced aspirates may tend to be relatively long anyway because of their articulatory complexity and sequential nature (mostly voiced closure followed by glottal widening timed to occur mostly after release). In fact various studies have found the closure of voiced aspirates shorter than that of other stop series (§2.5.3.1, 4.4.2). Impressionistic estimates of phonetic length may not be a clear criterion for deciding how many timing slots the Kelabit aspirates occupy. The other, phonotactic factors just mentioned remain in conflict, some seeming to indicate one timing slot and others two.

Blust's third argument seems ambivalent: if glottal consonants are separate segments, why not the voiced components of the aspirates? Neither induce high vowel lowering. As for his fourth and fifth arguments, alternation predicts nothing about how many timing slots an alternant may fill, and different dialects/languages have different grammars. Blust integrates these last two arguments by asking why the synchronic Bario pattern has a voiced alternant of the aspirate while some of the cognates from other dialects are voiceless, but this question remains regardless of whether the aspirates are analyzed as singletons or not. Chance probably plays a major role in language evolution (e.g. McMahon 2000), but there is no reason to think that language learners are constrained or normally even able to consider linguistic prehistory or cross-dialect commonalities when inducing the grammar of a particular language.

Blust (2006: 327) observes that "the alternation of plain voiced stops and voiced aspirates in Kelabit appears under exactly the conditions that produce allophonic gemination in many other languages: if a preceding schwa is unstressed, the following

consonant is not geminated, but if it is stressed, gemination of the following consonant is automatic (Blust 1995)." This process geminates consonants following stressed schwa (which survives medially) in Kelabit, with the notable exception of plain voiced stops, which instead turn to aspirates in dialects like Bario (Blust 2006: 316-317, 324). If the aspirates are phonologically geminates, then the exception is removed. This is certainly not decisive, of course.

In fact none of the considerations so far seem decisive either way. A network of historical changes has created a synchronic asymmetry in Kelabit which may admit no tidy analysis, as is true of various patterns in other languages (§3.3.1). Unless it can be shown that users of a language are innately hardwired to assign all sound sequences in the language unambiguously to a particular number of segment timing slots, it may be more plausible to describe the Kelabit aspirates as sounds having some properties of singletons and some properties of geminates. Alternatively, one may simply conclude that the aspirates are singletons; a number of other contour segment types (§3.3.6.3) are commonly treated as singletons in the absence of evidence to the contrary.

Elsewhere in Austronesian, the cognates of the Kelabit aspirates have the form /DT/ in Ida'an Begak (Blust 2010: 56-57, 99-101, Goudeswaard 2005: 4), where loss of initial schwa has even brought these sounds into initial position, along with true voiced geminates (Blust 2006: 318). Voiceless aspirates developed from geminates and certain clusters in Southern Subanen (Lobel and Hall 2010), and voiceless aspirates arose from /DT/ clusters in Maranao, where they trigger "tensing and raising" of following vowels as in Madurese (though unlike in Madurese, the contrastive voiced stops only optionally have this effect; Lobel 2010).

For earlier literature on the Kelabit aspirates see Blust (1969: 90, 1973, 2002), Hayward (1989: 46), and Stuart-Smith (2004: 18). Similar aspirates are described as word-initial clusters in Zhu|'hõasi (Khoisan) along with prevoiced ejectives (see Ladefoged and Maddieson 1996: 63, 80, Henton, Ladefoged and Maddieson 1992: 87, and for some similar sounds in Zambian Tonga, Hayward and Lindsey 1988).

3.2.6.5 Conclusion

Jakobson's universal remains unrefuted insofar as it concerns Hindi-type voiced aspirates rather than aspirates that are systematically voiceless for all or part of their duration. A tendency for languages to avoid the accumulation of phonetically unrelated material within a phonemic category – i.e. a principle of economy sensitive to phonetic enhancement – seems to furnish plausible motive for Jakobson's universal, the lack of evidence that aspiration ever serves as the contrast responsible for preponderantly voiced stop inventories, and the lack of evidence for Hindi-type voiced aspirates occurring as an enhanced voice series (without contrastive unaspirated voiced stops).

The pitch-lowering effect of the Madurese voiceless aspirates and the phonetics and peculiar phonotactics of the Kelabit and Lun Dayeh half-voiced aspirates remove them from being counter-examples to any of these generalizations unless we formulate the generalizations in more abstract phonological terms (e.g. applying to [+spr, +voi] sounds) and argue convincingly that these Austronesian aspirate types belong in the same phonological category. I am unaware of arguments to that effect (see §3.3.6f).

Further support for motivating Jakobson's universal from the interaction of economy with enhancement comes from evidence that aspirates can be non-contrastively voiced at the same places of articulation which favor non-contrastive voicing in unaspirated stops. Didayi Gta (Austro-Asiatic: Munda; Pandey in press) reportedly has $/b^{fi} d^{fi}$ / without */p^h t^h/; filling out the unaspirated stop set /p b t d t d k g/, the only voiceless aspirates are marginally phonemic /t^h k^h/. The occurrence of /b^{fi} d^{fi}/ without */p^h t^h/ without */p^h t^h/ the fact that a fair number of languages have /b/ without

/p/ (Maddieson 1984) and /d/ without /t/ (Tryon 1994). The pattern of /b/ without /p/ has been attributed to the larger oral cavity of bilabial stops facilitating voicing (see Maddieson 1984, Mołczanow 2007: 61-62, and §2.6.1). The pattern of /d/ without /t/ may be compared with reports that voiced coronal implosives, which are very often retroflex even if other coronals in a language are not (Greenberg 1970: 129), are also the most common selection other than /6/ in inventories possessing only one non-explosive (Greenberg 1970: 126-128, cf. Maddieson 1984: 112; further on the example of Somali: Edmondson, Esling, and Harris 2004: 5). Similarly, in phoneme inventories with only one voiced stop (not described as any kind of non-explosive), the most commonly reported place of articulation is bilabial, followed by coronal (Maddieson 1984: 35). Aerodynamic arguments relating place of articulation with favorability to voicing in unaspirated stops seem applicable to the voiced aspirates of Didayi Gta as well, though analyses of highly gapped inventories are tentative since such inventories exhibit wide variation (cf. Salmons 1993: 68).

The next section explores the interaction of economy with enhancement in another area: the conditions under which ejectives undergo diachronic voicing.

3.2.7 Diachronic ejective voicing

Extensive diachronic connections and some synchronic variation have been reported between ejectives and voiced non-explosives (Bomhard 1981: 374-375, Carlson and Esling 2003: 188, Fallon 2002: 258-264, Schrock 2011: 20-21), but evidence for a direct, unconditional sound change from ejectives to voiced explosive stops is strikingly difficult to find. Miller (2007) surveys evidence on this question from a variety of sources, concluding that all the candidate instances are either unsuitably synchronically conditioned, reconstructed with non-explosive intermediaries, or (in a few cases) produced by historical sound changes where it is unclear whether non-explosive intermediaries were involved or not. For example, there has been widespread ejective voicing in the evolution of Chechen and Ingush (Northeast Caucasian) and there is widespread synchronic variation of ejectives with voiced stops in Slave (Athapaskan), but in both languages the voicing is limited to an environmentally conditioned process of lenition that embraces additional series. Fallon (2002: 275-77) also reports "quasivoiced" ejectives or ejective-like sounds in Kabardian and Munda, but Kabardian involves explicit reports of larynx lowering and slight implosion, while Munda is explicitly neutral to both larynx height and voicing. Thus none of the data presents a clear case of a direct, unconditional sound change from ejectives to voiced explosives.

A possible explanation is that relaxation of laryngeal constriction throughout stop closure is normally insufficient by itself to induce voicing throughout closure, because that typically requires additional, active strategies (cf. §2.6). Notably, the other type of closed-glottis stop besides ejective – voiceless implosive – is repeatedly described as having glottal closure released shortly before the release of oral closure, with the intervening period voiced (see Demolin 1995, Demolin and Vuillermet 2006, McLaughlin 2005, Ladefoged and Maddieson 1996: 87-89). This would fit with the negative or at least very low oral pressure typical of non-explosives, which is naturally maintained during glottal closure if the larynx does not rise significantly as it does in many ejectives (Fallon 2002). The majority of preglottalized stops are also reported as voiced (e.g. Maddieson 1984), which is compatible with the role of glottal closure in staving off oral air pressure buildup as already suggested – not to mention the pervasive documentary ambiguity between preglottalized stops and non-explosives already (§2.6.2).

Thus if an ejective series does not involve larynx raising or undergoes a sound change removing larynx raising, and if the same series also undergoes a sound change that causes glottal closure to be released sooner, then voicing seems likely to result through the rest of oral closure – to the degree that the period of glottal closure staves off

oral air pressure buildup. This is tantamount to saying that laryngeally constricted stops are prone to become voiced stops to the degree that they reduce oral air pressure buildup, which we have already seen proposed as the defining strategy of non-explosives. Put another way: when the period of glottal closure shortens or disappears in ejectives through sound change, the resulting sounds apparently do not develop voice as a phonetic enhancement unless this is economical – favored by aerodynamic conditions already present. (On articulatory mistiming as a source of sound change see Stuart-Smith 2004: 170; on vertical larynx movement variation observed in ejectives and resulting variation in pitch and phonation see Fallon 2002, Kingston 2005: 152.)

3.3 Features

3.3.1 Introduction

This section investigates the motives for phonetically natural and unnatural features (§3.3.2), the topic of how features and feature-correlate maps may be acquired (§3.3.3), and universal constraints on feature-correlate mapping (§3.3.4). Several major proposals about specific features are critiqued with a focus on the representation of aspiration (§3.3.6) and voice (§3.3.7). I conclude in favor of auditorily organized feature definitions for voice and aspiration, while noting that laryngeal correlates can also be organized in other ways which do not clearly align with either of these categories. Finally, I summarize evidence for several cases of feature ambivalence and their basis in phonetic singularities (§3.3.8, 3.3.9). The case for feature emergence develops and strengthens in each of these sections.

3.3.2 Phonetically natural and unnatural features

In a late and unfinished manuscript, Ladefoged (2006: 20-28) distinguishes two basic types of features, corresponding to two extremes in a continuum of possible phonological classes identified by Mielke (2008a). The first type of feature is phonetically natural, and may be most simply defined with either articulatory or auditory correlates depending on the case. Auditory correlates are particularly useful for sounds which can be produced in several different ways but which count as realizations of the same phoneme in a language, as for example English /r/ (Ladefoged 2006: 28), which has great articulatory variation among different speakers although it can be produced relatively consistently by individual speakers (see Mohanan et al. 2009: 6). The second basic type of feature is phonetically unnatural (Ladefoged 2006: 22):

There are other examples of patterns arising from historical processes that were, when they occurred, due to articulatory or auditory aspects of the sounds in question, but which now cannot be explained in terms of how they are currently heard or produced. These patterns have to be described in terms of *ad hoc* features that have neither an auditory/acoustic nor a physiological basis. ... It should be noted that any non-physical arbitrary feature that simply groups segments as a result of diverse historical events in a particular language or group of languages is not part of a universal phonetic feature set.

Ladefoged (2006: 27) continues: "In considering phonological universals we can use only phonological features that reflect an articulatory or auditory property of some kind". In other words, sound-system universals can involve features to the degree that the features are phonetically natural, but not all features are phonetically natural. In agreement with this insight, many other works (e.g. Albright and Hayes 2006: 14, Bach and Harms 1972, Blevins 2008, Buckley 2000, Hale 2007, Hale and Reiss 2000a, Hayes et al. 2008, Mielke 2008a, and Vaux 2008) document how historical extensions of
phonological patterns beyond their original phonetic motivations create phonetically unnatural phonological classes. These works argue persuasively against indiscriminately incorporating phonetic factors directly into cognitive grammatical representations.

The existence of phonetically unnatural features makes it extremely probable that the grammars of individual languages differ in their feature inventories. This ties in closely to another topic that has long challenged feature-based cross-linguistic comparison, again summarized by Ladefoged (2006: 30-31):

There are problems in the notion that each feature should be definable in terms of a specified property... Across languages there are often situations that are partially the same, and we would miss a significant generalization if we did not group them together. For example, there are similarities in the processes referred to as vowel harmony in African languages and register distinction in Mon Khmer languages... These distinctions often involve several articulatory parameters that are related to one another, such as tongue-root movements, tongue height variations, vertical larynx movements, and changes in laryngeal tension (and hence in phonation type). But these factors are not all describable in terms of a single feature. We undoubtedly need some way of expressing these similarities.

One solution is to view phonological features as language-specific regions in the inventory of possible gestures and cues. These regions would be defined by each learner during the process of grammar construction guided by learner analysis of ambient language data. Individual learners might induct slightly different mappings of features to phonetic correlates, causing systematic phonetic variation among speakers of the same language. We have already seen several examples of this (§2.5.5). Brunelle (2010: 20-21) considers what inter-speaker variation implies for the representation of the two contrastive stop series of Javanese in particular:

[U]nless we rigidly adhere to universalist assumptions, there is no reason to believe that all learners develop an identical phonological representation for the register contrast. As long as all speakers 1) have cognitive categories that correspond to the high register (tense stops) and the low register (lax stops); and 2) acoustically produce the contrast in a way that is similar to the way that the rest of their speech community (or of one well-established dialectal or social subgroup) produces it, the way in which the contrast is represented in individual grammars and articulated by different speakers can vary widely. Therefore, instead of proposing that the Javanese stop contrast is captured by a single articulatory feature, it seems more economical to propose that speakers have a contrast between two relatively abstract categories... and that they learn the phonetic properties that are associated to them through exposure and through trial and error.

Other recent phonetic, phonological, and psycholinguistic research marshals additional support for this framework. These works include Mielke (2008a) on the emergence of features, Samuels (2009a) on the emergence of feature geometry, Menn (2010) and Sosa and Bybee (2009) on the potential for general powers of human observation, analysis, and memory to handle the acquisition of phonology, and Archangeli (2011) and Mohanan et al. (2009) on what phonological computation might look like without an innate universal battery of constraints. The trend of minimizing the amount of innate structure posited is not limited to phonology, but owes some of its impetus to Chomsky's similar minimalist exhortations framed primarily in the context of syntax over the last several years (Samuels 2009a: 1-3). These broader themes of grammar design are pursued further in ch. 5; here we will continue focusing on the implications of emergent theory for features in particular.

Innate feature theories, according to Mielke (2008a), posit a finite, universal, cognitively innate set of features linked to phonetic correlates; the learner uses these features to construct representations of phonological classes and rules/constraints. Innate hierarchical organization of features is a popular development of innate feature theory since Clements (1985) and aims to group features with mutually dependent behaviors under shared nodes. Of course, the richness of the innate phonological endowment increases further and astronomically with Optimality Theory (Prince and Smolensky 1993) discarding rules for a universal constraint set, if the set is understood to be innate and fully present in all languages. While Optimality Theory remains officially agnostic on this point, it is difficult to motivate a learned universal constraint set (§5.5, 5.7).

In contrast, Emergent Feature Theory proposes that as definers of segment classes and as objects manipulated by phonological processes, features are abstract in the sense that they are not *necessarily* closely related to phonetic content (Mielke 2008: 9, 98-99). This matches Ladefoged's (2006) argument that both phonetically natural and unnatural, language-specific features exist. Further, since learners in the emergent framework are under no pressure to specify phonologically active classes using features drawn from an innate universal set, they are also under no pressure to create complex feature specifications for classes by combining features; instead, they may just as well assign one feature to each class unless there is positive, language-specific phonological evidence to the contrary (Mielke 2008a: 99). A possible case of the latter sort has to do with the independence of voice and aspiration in some Indic languages (§3.3.4).

One noteworthy way that phonetically unnatural features may develop is through sound change by generalization. While some sound changes result from across-the-board phonetic processes, others involve shifts in the boundaries of already existing phonological classes based on speaker generalizations that evidently involved cognitive evaluation of similarities in the phonetic content of sounds (Blevins 2004: 32-34). Mielke (2008: 85-95) discusses several phonological patterns that seem to have developed in this way. In each case, it appears that a phonetically natural phonological rule became unnatural when one of the segment classes involved in the pattern was expanded to include new sounds that shared with the members of the original class some phonetic trait which was not crucial to the original rule. Further illustration appears in an ongoing New Zealand English vowel merger (Warren, Hay, and Thomas, forthcoming; Warren and Hay, forthcoming; Hay, Warren, and Drager 2006) where increasingly younger speakers clearly tend to merge more than older speakers but retain considerable ability to distinguish non-merged tokens. This merger too seems to have begun in a narrow, phonetically natural environment and then analogically spread to broader, phonetically less natural environments. The speaker generalizations that seem necessary for such sound changes to occur involve categorical judgments about the phonetic traits of sounds.

3.3.3 Feature acquisition

Hale, Kissock, and Reiss (2006) distinguish three strictly separate modules in the human faculty that makes use of linguistic sound systems: a phonological computer which manipulates features, a perceptual transducer which translates acoustic input into features for the computer, and an articulatory transducer which translates features from the computer into a gestural score for the production system. These modules are unable to see each other's activities, and the transducers have very restricted learning ability. Features and their mapping through the transducers to percepts and gestures are presumed to be innate because features, acoustic signals, and gestures are such different entities. Hale and Reiss (2000b: 9-11) point out the difference between mental 3D comprehension and 2D images on a piece of paper which do not even need fully connected lines, and similarly the difference between the potentially equivalent complexity in mental plans for pushing a feather and pushing a bar of gold, on one hand, and the disparate effort required in actually pushing the two objects, on the other. When cognitive representations, physical stimuli, and physical production are so different, how could categorical mappings between them ever be learned?

Decades of psycholinguistic debate show that this question is far from merely rhetorical, however (Menn 2010: 93):

There has been intense controversy for many years between those who think that much knowledge of possible linguistic patterns is built into human minds genetically, so that what children need to do is figure out which of those patterns are used in the language around them, and those who think that what is built into our minds is not knowledge of language, but an extraordinary ability to respond to and extract the patterns that are present in the thousands of examples that we hear, examples of greetings and commands, questions and relative clauses, casual speech and prayers. One of the strongest arguments for the existence of this generic learning ability is when the general structural character of a familiar type of knowledge that many scientists had argued must be innate is unexpectedly replicated in a new context and in a new body of knowledge. Sign languages have garnered increasing attention for this very reason. Sign languages develop as a solution to the needs of particular minority populations and communicative contexts, so it is less plausible to treat the features of sign languages as universally innate to humanity. It would also be much more costly to do so, because the feature inventories and feature geometries of sign languages tend to be considerably larger than those of spoken languages – and as with speech, the feature geometries of sign languages reveal asymmetries that match human anatomy, increasing the redundancy of stipulating the same patterns in the innate cognitive endowment (Mielke 2008a: 15-19, Morén 2007: 19, 41, Samuels in press). Far more probably, "whatever the mechanism is that allows humans to learn emergent sign language features and combinations should also be available for use in learning emergent spoken language features and combinations" (Morén 2007: 41).

Other activities besides speech exhibit the same traits: functionally driven patterns emerge as general cognitive powers of categorical perception are applied to new types of data and new relationships are identified between causes and effects. Illiterate and improvisatory musicians learn to recognize and play notes, intervals, melodies, chords, and progressions that vary significantly among cultures – presumably without innately mapped connections between the specific patterns involved in the auditory, cognitive, and motor domains. Video games are mastered through learning to map hand movements to intricate audiovisual stimuli in order to get particular results within a framework that is often radically removed from any real-world context; here too there is no reason to presume a detailed innate endowment of gaming knowledge. In all these activities, various conceivable combinations of stimuli or gestures are avoided for transparently functional reasons (some chords are impossible to play, some color combinations are hard to read on-screen, etc.).

113

The complexity of learning to identify segments, assign phonetic cues to them, and remember the various bundles in which these cues typically associate does not seem fundamentally different from the complexity of these other situations. Not surprisingly, humans and other creatures are born with highly sensitive powers of general acoustic categorical perception enabling them to differentiate and remember sounds that they have never made and may never be able to make; for instance, chinchillas and quail can distinguish laryngeal, place, and manner distinctions of human phonology which they will never be able to produce (Blevins 2004: 54-55, cf. Samuels in press). As in the other activities mentioned above, learning also has a filtering aspect: over time the practitioner develops expectations of what is significant and pays less attention to what is not (see Menn 2010: 91, 187-190, 350-51). The linguistic tuning of this general ability to ignore some perceived differences in favor of others occupies much of "early phonological learning" (Blevins 2004: 55). Mielke (2008a: 43-44) discusses experiments showing that many of the phonetic differences which people learn to tune out during speech processing can still be perceived by the same people when they are not in speech-processing mode.

The most parsimonious explanation for how people learn these different kinds of knowledge (speech, sign language, music...) and for the functionally motivated patterns in the structure of these bodies of knowledge seems to depend on the connection between practice and categorical perception (cf. Morén 2007: 57-61). People and animals have an extensive ability to discover correlations between different domains by repeatedly applying very general analytical powers to available inputs, including passively received stimuli and the stimuli of their own actions. All that categorical perception requires is enough range in the data for dissimilar tokens to be isolated and contrasted (see further Bybee 2010). In the case of linguistic sounds, acoustics and articulatory physiology furnish natural quantal boundaries (Stevens and Keyser 1989, 2010; see e.g. the discussion of palatals in §4.3.1.1).

A further problem with innate feature theories is that the bulk of the complexity which must be attributed to the phonetic transducers is not static but stochastic and highly variable (Pierrehumbert 2001: 143). Learned knowledge is naturally variable if it is learned under relevantly varying conditions, but capturing such variation with a hardwired knowledge apparatus requires additional stipulations which to my knowledge have not yet been clearly formulated even hypothetically.

One last argument against Hale, Kissock, and Reiss's (2006) transducer model comes from research on the capacity of auditory systems to extract spatial/gestural information from auditory or visual input. Iskarous (2010) discusses recent strides in this area particularly regarding extraction of articulatory area functions for vowels from the acoustic signal, and Vaux (2009: 77) synthesizes much previous research on Motor Theory – in general, in connection with language, and particularly regarding mirror neurons, "which are activated by both execution and observation of manual and oral actions of both first and third person agents". To sum up, several streams of recent research seem to support the claim that instead of approaching language acquisition with a highly detailed innate knowledge of phonology, learners instead rely on "an extraordinary ability to respond to and extract the patterns that are present in the thousands of examples" that they encounter (Menn 2010: 93).

3.3.4 Universal constraints on feature-correlate mapping

If phonological features are language-specific regions mapped by learners in the universal space of possible gestures and cues, then the study of phonetically natural features must investigate the way that the inherent structure of that landscape can influence the way learners subdivide it. For example, the release noise of an ejective and the formant structure of a low front vowel are acoustically and articulatorily dissimilar enough that they are unlikely to be identified with a common natural phonetic feature, but the gradient phonetic relationships between whispery, strict breathy, and slack voice do invite natural phonetic association. Such considerations form a basis for exploring *phonetic geometry*. To place this idea in context, let us first review the concept of Feature Geometry.

Feature Geometry (e.g. Clements 1985, Clements and Hume 1995, Halle 1995, Halle, Vaux, and Wolfe 2000) manages to capture some cross-linguistic patterns in the phonological behavior of features by positing a universal tree structure - so that place assimilation, for example, can be formulated with a Place node dominating place features instead of requiring a simple listing of place features. In this way Feature Geometry improves on a theory which offers no formal explanation as to why some groups of features occur more often in phonological patterns than others do, just as features themselves improve upon theories which offer no formal explanation as to why some groups of segments occur more often in phonological patterns than others do. Conventional models of Feature Geometry are not without their difficulties, though. The extent to which their structure mirrors the structure of the human vocal tract raises a serious duplication concern if Feature Geometry is innate (Samuels 2009a: 54-56). Yet if Feature Geometry is not innate, then it can only claim universality if we enrich its formal structure far beyond what conventional "tree" diagrams can capture (§3.4) and simultaneously acknowledge that the grammars of individual languages only make use of parts of this structure. In response to such concerns, an alternative approach allows features to be grouped into larger categories (equivalent to various nodes in Feature Geometry) during the learning process in a phonetically motivated but language-specific fashion (Blaho 2008, Reiss 2003).

The rest of this section explores the implications of this move by illustrating how the structure of the speech apparatus constrains the possible and likely groupings of gestures and cues and how these constraints in turn affect both the number of features which can be mapped into a given phonetic space and the relative positions within that space which those features may occupy.

Several phonetic correlates are shown with some of their natural physiological, aerodynamic, and acoustic relationships in figure 3.1. Cues are italicized while non-

acoustic traits are not. Association lines (some of which are dashed to increase readability) connect phonetic elements in various relationships which either must or can occur, depending on the interaction of other associated factors. For example, the aerodynamic trait of a moderately strong transglottal pressure differential (sufficient for the voicing cue) is linked to several possible articulatory strategies, at least some of which must be present for the aerodynamic configuration to occur. When multiple factors must co-occur to entail a given effect, their association lines touch at that correlate; otherwise, converging association lines are generally separated by visible space, as is the case with the gestures associated to the transglottal pressure differential. It should be kept in mind that the dependents of vocal fold slackening ($\S2.5.5$) have particularly complex relationships, so the associations shown in that area are especially incomplete. I have also omitted distinctions and relationships among the muscles that manipulate the tongue (see Baer, Alfonso, and Honda 1988, Honda 1996, 2008: 16, 20, Trigo 1991: 115) and the muscles responsible for pharyngeal and epilaryngeal contractions (§2.4). Certain other finer interactions among phonetic elements are also not identified in figure 3.1. Increased airflow and resultant glottal turbulent noise are differentiated into lower and higher degrees, but otherwise quantity distinctions are generally omitted.

Figure 3.1. Phonetic geometry (partial).



 $^{^{2}}$ Here, positive (negative) spectral tilt = higher (lower) ratio of upper to lower harmonics than that found in modal voice. Spectral tilt is generally only seen in voiced contexts.

³ Not during stop closure.

The same phonetic information is reproduced in figure 3.2 and overlaid with an estimate of the maximum number of features among which these correlates could be distributed simultaneously. The estimate was made by first identifying the relatively auditorily distinct cues (excluding gradient differences such as between voiceless breath and hoarse whisper) and assigning them to feature categories as shown by the boxes. For convenience these categories are labeled with fairly conventional feature names (see e.g. Hall 2007, Halle, Vaux, and Wolfe 2000) except for [slack]. The choice not to represent slackening with the feature [voice] here is motivated by the following facts. Phonetically slack and voiced stops are systematically distinguished at the phonetic level in languages like Korean and Wu Chinese, and they are allophones assigned to different contrastive series in Qingtian Wu Chinese and Sivas Armenian (§2.5.5.4). Subsuming slack stops under a broad, low-frequency definition of voice also runs into additional problems (§3.3.7.2). Formally, the ideas of the feature category and the feature name [slack] have already been considered favorably by Chen and Downing (2011).

Second, it was observed that the non-modal types of voice at the bottom of the figure are phonologically composite and do not merit unique features in languages with phonological patterns that clearly differentiate voice from aspiration or glottal constriction. In Sanskrit, for example, the deaspiration but not devoicing of stops in reduplicants and word-finally suggests that voiceless and voiced aspirates are here losing a shared aspiration specification while remaining differentiated by voicing (see Janda and Joseph 1989: 248, Joseph and Janda 1988: 31). Supporting this, deaspirated voiced aspirates are indistinguishable from underlying unaspirated voiced stops (Selkirk 2004[1992]: 317). The related Indic language Marathi also has a word-final deaspiration process that targets both voiced and voiceless aspirates and produces stops that appear indistinguishable from other unaspirated voiced and voiceless stops (Selkirk 2004: 316, noting that the reports are impressionistic fieldwork). Such evidence matches the analysis

Figure 3.2. Suggested maximum number of features (bold) that can be simultaneously assigned in the phonetic space shown, with some exclusions at bottom (see discussion). Boxes show feature-cue assignment.









of various ancient Indian phonetic treatises, e.g. "*sōsmōsmaņām ghōsiņām śvāsanādau*" '[the specification] of aspirates (*sōsman*) and fricatives (*ūsman*) that are distinctively voiced (*ghōsin*) is breath (*śvāsa*) and voice (*nāda*)' (see Busetto 2003: 203, 213-214 citing *Rk-Prātiśākhya* 13.4-6).

In languages without such phonological evidence, the various non-modal types of voicing could in principle be specified by single features on a language-particular basis. It is moot to ask whether such features are phonetically more natural than composite feature specifications of non-modal phonations, though. That question boils down to whether the phonetic complexity of non-modal phonations should be treated as a natural unit or not, and there is no meaningful answer to the question without universal criteria. Therefore, in an emergent feature framework, there may be no clear basis for positing single phonetically natural features for the non-modal types of voicing. In an alternative framework with universal features, phonological evidence from languages like Sanskrit and Marathi might (if sufficiently widespread) support a universal composite representation of breathy voice. Such representations have been fairly popular in works focused on phonological features ([+spr] is combined with a specification for voice in Hall 2007, Halle and Stevens 2002[1971], Kehrein 2002, Kehrein and Golston 2004, and Vaux 1998a,b; Busetto 2003 discusses comparable analyses in the ancient Indian phonetics literature).

Since the cues in figure 3.2 (other than dependents of *voice*) are assigned to features, additional features cannot be simultaneously assigned to the non-acoustic phonetic correlates in the figure because such features would lack cues to make them recoverable in speech. Thus, in an emergent feature framework, the maximum number of phonetically natural features that can be simultaneously assigned to a set of correlates is limited by anatomy.

Figures 3.3 and 3.4 show possible assignments of some correlates to some features, coded by color. In figure 3.4 it should be noted that the correlation between



Figure 3.3. Association of some correlates to laryngeal features in Hindi stops.

Key: [±voi] blue, [+spr] red, [+voi, +spr] purple. Colors have not been exhaustively assigned; see discussion.



Figure 3.4. Example of phonetic correlate assignment to [slack], [spr], and [voi].

Key: [+spr] red, [+slack] dark blue, [+voi] light blue. Colors have not been exhaustively assigned; see discussion.

vocal fold slackening and low tone causes an overlap with tonology if a tonal system is present. Hence the restriction of slack stops to lower-toned syllables in some languages (Wu Chinese), though in others (Xhosa) slackening lowers both high and low tones below their normal range (§2.5.5). Two other comments on this figure are in order. First, slack voice is not highlighted as a correlate of [slack] here because slack voice might be better treated as an overlap of [slack] with phonetic voicing, at least in languages where slack stops are voiceless (on the question of voiced slack stops, see §3.3.7.2). Slack voice occurs automatically in vowels that follow voiceless slack stops because some distinctly [slack] cues are impossible during stops. This may be simply an acoustic fact that motivates particular gesture timings but does not itself have to be phonologically represented in temporal terms or with reference to [voice] (see further discussion of timing issues in §3.3.6.3). Second, slight amounts of glottal turbulent noise are associated with slack sounds in Korean, Wu Chinese, and for some speakers, Xhosa. A portion of the glottal noise correlate is colored dark blue in the figure to show this. This cue has been fiberoptically correlated with moderate PCA activity in Korean (§2.5.5.3.1).

Stops with substantial amounts of closure voicing occur alongside or even in contrast with slack stops in some languages (§2.5.5.4). In these situations, the voice correlate (colored light blue in fig. 3.4) can be implemented through various means except the combination and degree of larynx lowering and intrinsic laryngeal muscle suppression associated with the realization of [slack] in stops (dark blue, upper part of figure). The possible means include non-explosion (Xhosa) and ordinary modal voicing (Zulu, implying some combination of factors such as less larynx lowering and more medial compression than in the depressors) (§2.5.5.2).

126

3.3.5 Modeling phonetic variation

The requirement that features must be perceptible entails that the number of features which can co-exist in the phonetic space is considerably smaller than the number of phonetic correlates in the same space, as exemplified in figure 3.2. At the same time, there are different ways to subdivide a given range of correlates under a given number of features; for instance, a correlate bundle may be organized around particular cues or around an articulatory strategy that produces a wider range of cues in different contexts, respectively yielding an auditorily or articulatorily organized feature (§3.3.1). These considerations establish both restrictions and a range of possible variations in feature-correlate mapping which help fulfill Roman Jakobson's call for a limited "universal set of distinctive features tied to phonetics but flexible enough to allow for empirically demonstrated cross-linguistic variation" (Morén 2007: 10). In attaining this result we have not found it necessary to attribute detailed stipulations about features to the human cognitive endowment.

That fact in turn facilitates the explanation of sound changes in which particular cues are phonologically reanalyzed and assigned to different features (see Blevins 2004: 32-39). The challenge here for innate feature theory is that once the supposedly innate associations between features and their phonetic correlates can vary as required by the intermediate stages in these sound changes, there is no principled way to determine how much variation is allowed or why. If the potential variation range is gradient and not limited in a principled way by the innate feature machinery itself, then there is little meaning left in the notion of innate features and we are left locating the universal restrictions within phonetic geometry instead, as here.

A similar observation as this one about sound change can also be made about synchronic inter-speaker variation, of which we have already seen several examples. In an emergent framework, this variation is explained as a natural result of the same learning process that is responsible for the acquisition of features. Different learners achieve different results in their task of identifying and imitating the cue patterns of other speakers. The many-to-many relationship between gestures and cues opens up space for different learners to discover slightly different articulatory strategies for producing similar sounds, with some of these strategies further resulting in different acoustic side effects. In an innate-feature framework, however, inter-speaker phonetic variation occurs in spite of innate feature-correlate mappings and must be interpreted as evidence of what they do not cover. This means admitting that learners have considerable ability to analyze the acoustic signal and induce phonetic categories which show sometimes dramatic variations among speakers, but the ability to map these phonetic categories to phonological categories still supposedly requires an innate transducer. It is not clear how to justify that division of labor.

These considerations provide a coherent framework for evaluating several proposed definitions of the features responsible for two of the most important laryngeal distinctions – aspiration and voice – as well as related proposals about slack stops and larynx lowering.

3.3.6 Representing aspiration

3.3.6.1 [spread glottis]

The feature label [spread glottis] proposed by Halle and Stevens (1971) has been fairly popular in the phonological feature literature (e.g. Clements 1985, Clements and Hume 1995, Hall 2007, Halle 1995, Halle, Vaux, and Wolfe 2000, Iverson and Salmons 1995, 2011, Kenstowicz 1994, Vaux and Samuels 2005; Ridouane, Clements, and Khatiwada forthcoming, with further works listed p. 262). However, a simple articulatory correlation of the feature with glottal spreading has proven inadequate given the comparable degree and duration of vocal fold abduction found in voiceless unaspirated and voiced aspirated stops in Hindi (§2.3.2), the slight active abduction observed in the release of glottal stops (§2.4), and silent high abduction in geminates (Ridouane et al. forthcoming, cf. Ladefoged and Maddieson 1996: 70-72, Ridouane 2003). Abduction gestures may also span voiceless clusters in complex ways, creating atypical glottal width contours for some individual segments which nevertheless remain voiceless due to their intersegmental context (Ridouane 2008: 344, Yoshioka, Löfqvist, and Hirose 1979, 1980). This increases the appeal of non-articulatory explanations for vocal fold abduction patterns in stops.

3.3.6.2 [heightened subglottal pressure]

Opting for an aerodynamic focus, Chomsky and Halle (1968) claim that [heightened subglottal pressure] is a necessary condition for aspiration. This view gained little popularity (see Helgason and Ringen 2008: 608) and has since been rejected on phonetic grounds (Jessen 1998: 134-135, cf. Ladefoged 1967: 1-49, Ladefoged and Loeb 2009). One of the objections is that subglottal pressure (Ps) drops during stop aspiration and in intervocalic [fi] (Demolin 2007: 77-78, Ohala and Ohala 1972, cf. Ladefoged 1967: 43). More generally, pulmonic control of Ps does not appear sensitive to individual segments (Ohala 1990a: 42). Aerodynamic and acoustic experiments by Demolin (2007: 89) and Ohala (1990a) showed positively correlated Ps and intensity contours spanning multiple segments and even phrases (cf. Baer et al. 1976: 176). This is not surprising since acoustic intensity is a measure of loudness, which ranges widely across utterances depending on speaker judgments about ambient noise levels as well as emotion and focus (cf. Demolin 2007: 79). Loudness is also a common correlate of stress (Gordon and Applebaum 2010: 35), which has proven to be the environment where aspiration is strongest in some languages (Swedish: Jessen 2002: 175; American English: Hirose and Gay 1972: 147, Sawashima 1997: 71). However, stress associates with a subset of the

syllables in a word, not (like aspiration) directly with segments. Ladefoged (1967) and Ladefoged and Loeb (2009) conclude from aerodynamic, acoustic, and EMG measurements that stress correlates positively with increased Ps caused by activity of respiratory muscles including the intercostals, but that segmental Ps variations are determined as a rule by glottal and supraglottal articulations rather than actively controlled by the respiratory muscles. Demolin (2007: 90) and Honda (2008: 9) arrived at the same general conclusion, notwithstanding some possible exceptions (Demolin 2007: 78, 89-90). On the evidence, heightened subglottal pressure is not a necessary or even a usual correlate of aspiration (Stuart-Smith 2005: 173-174).

3.3.6.3 VOT and timing in phonological representations

Chomsky and Halle (1968: 327) are themselves unenthusiastic about another approach to laryngeal features, one relying on VOT (Lisker and Abramson 1964, 1971, Kagaya and Hirose 1975: 43). Most phonologists have followed suit in the sense that phonological specifications of subsegmental timing information are generally avoided (see the string of works cited at the beginning of §3.3.6.1) with a few particular exceptions. We will touch on these before considering problems with VOT based representations of laryngeal distinctions.

Tonology is probably the clearest case where multiple, mutually incompatible features can associate with a single segment in a temporal sequence that needs to be phonologically specified. This is a consequence of a larger generalization: tonal specifications associate with segments in many-to-many fashion that can generate contour tones on single segments as well as level tones spanning multiple segments. Leben (2011) reviews how this fact motivated the development of autosegmental phonology, where properties not typically restricted to individual segments (and thus sometimes called "suprasegmental") were assigned their own representational tiers separate from the segment tier and capable of being associated with it in languagespecific ways.

There has been major debate about whether affricates and prenasalized stops should be represented with conjunctions of both values of the features [cont] and [nas] and whether the values' temporal sequence needs to be included in the phonological representation (e.g. Clements and Hume 1995: 254-256). With affricates this seems largely not the case (Hall 2007: 331, Kehrein 2002) though there may be some exceptions (Rood 1975). The representation of prenasalized stops is less clear for other reasons (Beddor and Onsuwan 2003, Herbert 1986, cf. Bradshaw 1978, Cohn and Riehl n.d., Feinstein 1979, Hayward and Muljono 1991, Reid 2000).

In a much more unusual proposal, Steriade (1993, 1994) argued for phonological specification of laryngeal timing information in stops, based on gestural sequences in Huautla Mazatec that are far more complex than any we have considered so far in relation to the laryngeal distinctions in stops. Golston and Kehrein (1998) highlight concerns with her analysis, discuss Pike and Pike's (1947) original proposal that the sequences are multi-segmental, and suggest a simpler analysis in which some of the laryngeal distinctions are associated with nuclei rather than onsets (cf. Kehrein 2002: 128-130). Here it suffices to note that this particular debate has attracted attention precisely because it is unusual. In a sense it is an exception supporting the more general fact that the encoding of laryngeal timing information in contrastive phonological representations has not found widespread favor with phonologists.

VOT-based representations of laryngeal contrast raise several questions. Voiced aspirates, for instance, have modal voicing both during stop closure before the aspiration and in the vowel after it (Dixit 1979: 429-430, Stuart-Smith 2004: 163-164). In defending VOT-based models, Lisker and Abramson (1971: 142, cf. 1964: 418-419) simply note that voiced aspirates have non-modal phonation and that the phonological unity of their aspiration with that of voiceless aspirates is not self-evident. This view is also reflected in Ladefoged's feature proposals, which do not relate voiceless aspiration to voiced

aspiration (see Ladefoged 1999; Schiefer 2004[1992]: 297 on Ladefoged 1971; Dixit 1979: 431 on Ladefoged 1973). However, phonological data does support the unity of both aspiration and voice in some Indic languages (§3.3.4).

VOT models of aspiration also have trouble with (post-)aspirated stops wordfinally and preconsonantally (Bhatia 1974; further examples of word-final post-aspirated stops: Iverson and Salmons 2011, Vaux and Samuels 2005; aspirates before consonants including stops: Laver 1994: 239-240; both: Kehrein and Golston 2004: 335, 340, 348). Finally, Ridouane et al. (forthcoming 271-275) emphasize how timing-based definitions of aspiration cannot pin down a particular timing: vocal fold abduction overshooting release does not work in devoiced sonorants or fricatives, which in some languages pattern phonologically with aspirate stops (Vaux 1998a), nor in fricative-stop onset clusters, which can have a single abduction gesture whether the stop is aspirated or not. (For further discussion of "unwarranted phonologization of VOT" see Vaux and Samuels 2005: 406, also Halle 1973: 928, Löfqvist 1995: 105-106.)

Timing is still clearly important in the implementation of laryngeal distinctions; for instance, Kagaya and Hirose (1975: 43) emphasize that the glottal width difference in their voiceless unaspirated and voiced aspirated stops is one of timing rather than degree. Ridouane et al. (forthcoming 279) note that this creates a quandary for features defined primarily in terms of articulatory correlates, while an auditory definition of the feature responsible for aspiration easily motivates the necessary timing constraints without requiring temporal elements in phonological representations: "Timing relations follow from the requirement that the acoustic goal associated with the feature be manifested in the signal." The relevant cue will require a particular articulatory sequence in time, but those articulatory details are not necessary to the acoustic definition of the cue itself.

132

3.3.6.4 Auditory definitions: intensity, duration, and turbulent noise

Acoustic correlates are prioritized in the feature definitions of Jakobson, Fant, and Halle (1954: 11-13; cf. Jakobson and Halle 1956: 33-34). They identify aspiration as a phonetic correlate of their [tense] feature, which is not properly a laryngeal feature but rather correlates with duration and intensity and is measured over release noise in stops (pp. 36-39). Segmental intensity differences are notoriously variable and gradient (Parker 2002, 2008), and this use of [tense] is mainly rejected in later phonological feature literature (including the works cited at the start of §3.3.6.1). Jessen (1998: 254-268, 2001) argues for a revival of the feature. While acknowledging various serious representational issues that remain unresolved, such as the lack of a formal laryngeal category subsuming both aspiration and voice, he emphasizes the phonetic transparency of his proposal – which is in fact fairly abstract, taking some of the phonetic correlates of the abstract [voice] proposed by Kingston and Diehl (§3.3.7.2) and making them simultaneously dependent on [tense] too, as "substitute correlates" that may occur instead of the latter's primary correlate of aspiration. Aspiration itself does not require articulatory tensing (cf. strict breathy voice: §2.5.3), nor are acoustic duration and intensity the only appealing bases for auditory definitions of the feature responsible for aspiration.

Both voiced and voiceless aspirates have glottal turbulent noise lasting from a stop edge into the adjacent vowel. Stuart-Smith (2004: 164-166) notes proposals by Davis (1994) and Dixit (1982, 1987) that aspiration should be defined in terms of this period of turbulent noise. Ridouane et al. (forthcoming 277) similarly identify "aspiration noise, i.e. aperiodic energy in the second and higher formants" as a necessary component of the definition of [+spread glottis]. The acoustic definition can be further refined; for instance, aspiration noise has high bandwidth or energy dispersal rather than concentrating amplitude at narrower frequencies as stridents do, and the extent to which it obscures vowel formants varies (see Harrington 2010a,b, Stevens 1998, 1999).

Since voiceless fricatives tend to have moderate to high vocal fold abduction, they tend to have glottal turbulent noise as well (see Jaiswal 2011: 4-5, Rice 1994: 134-135, Ridouane et al. forthcoming 5, Vaux 1998a, with further glottal width data in Saito 1992: 35, Sawashima 1997: 71). The same type of noise may also occur to some degree in voiced fricatives, but the turbulent noise of oral frication (Stevens 1998, 1999) makes detecting the presence of glottal noise difficult in fricatives generally (see Stuart-Smith 2004: 206, Vaux 1998a: 509). Phonological classes defined by aspiration but not including the class of voiceless fricatives would thus need to correlate explicitly with glottal turbulent noise *not* masked by another turbulent sound source. On the other hand, phonological classes including both voiceless fricatives and aspirated stops would be specified as [+spr] (or the equivalent), which in these cases would have to correlate with both turbulent noise and vocal fold abduction (cf. Ridouane et al. forthcoming) because the masking effect of oral frication on glottal turbulent noise in voiceless fricatives entails that there is no single acoustic noise pattern shared by voiceless (but not voiced) fricatives and prevocalic stop aspiration.

The existence in several languages of phonological classes including voiceless fricatives and aspirated stops has prompted the proposal that voiceless fricatives tend to receive the specification [+spr] at some level of grammar even in languages without clear phonological evidence for this (Clements 1985: 248, Jessen 1998: 186, 254, Rice 1988, 1994, Vaux 1998a, and others cited in Kingston et al. 2009: 32, Ridouane et al. forthcoming 263). The same idea has been extended to voiceless sonorants (Lombardi 1994: 151 in Clements 2003: 294, Vaux 1998a). In an emergent framework, though, phonological feature specifications have no reason to be induced in a given language without phonological evidence (Mielke 2008a: 99).

3.3.6.5 Broader issues

A final and fairly obvious fact to recognize in defining the phonetic correlates of the feature responsible for aspiration contrasts is the gradience of the correlates themselves. In Navajo, for instance, /k/ has a VOT of over 40 ms, /k^h/ over 150 ms; in Hindi, French, and Thai, by contrast, voiceless unaspirated stop VOT is around 20 ms or less (Ladefoged 2001: 127-131; for further cross-linguistic VOT comparisons see Lisker and Abramson 1964). If the feature responsible for aspiration contrasts is correlated with glottal turbulent noise, the correlation cannot be simply with presence vs. absence of this cue but rather with higher vs. lesser degrees. Degree here is measured relatively easily in terms of duration, though this does not exclude the possibility that listeners also attend to other cues like acoustic intensity. At least some studies have correlated greater peak glottal width with stops that are perceived as more strongly or heavily aspirated (Hong, Kim, and Niimi 2002, cf. Fischer-Jørgensen and Hirose 1974: 250); greater abduction entails higher transglottal airflow, which could in principle generate higher-intensity turbulent noise within a certain range of conditions. Vaux and Samuels (2005: 401) discuss research arguing that intensity of aspiration is perceptually even more important than duration for some speakers.

The concerns addressed in this section bear on broader questions about the relationship between articulatory and acoustic components in feature definitions. In a basic sense, all features have both articulatory and acoustic definitions because they must be both produced and heard (Ladefoged 2006: 22, Mohanan et al. 2009: 4, Robbins 1977: 60). Presumably most languages can be understood without the listener actually seeing the speaker's face and throat, and in this sense acoustic input is sufficient for the hearer (though users also learn to infer speech information from watching a speaker whether or not the speaker is audible, cf. Keating 2012, Scarborough et al. 2009). At the same time, since mature speakers have passed the babbling stage (§5.2.1), they normally do not need to *test* their productions on their own auditory systems through online repetition in

ordinary speech. In this sense, articulatory input is commonly sufficient for mature production, though the speaker's production-perception feedback loop plays an important role in learning, in production monitoring (Wedel 2011, cf. Dorman, Freeman, and Borden 1976), and evidently in sound change as speaker expectations shift during adolescence and even adulthood (Kerswill 1996, Sankoff and Blondeau 2007). Thus, articulatory and acoustic information are not corequisites for feature recovery; articulatory information is generally sufficient for feature production, and acoustic information is generally sufficient for feature production, and acoustic information for producing a set of cues, speakers may under-articulate orally, laryngeally, or pulmonically, which can lead to production tokens that are imperceptible. Oral under-articulation has been analyzed as a major source of lenition patterns (Kirchner 1998). One dramatic case that seems to involve pulmonic under-articulation happens with utterance-final vowels in Blackfoot (Algonquian) and Oneida (Iroquoian), which are normally acoustically imperceptible although still orally articulated (Gick et al. 2012).

Glottal turbulent noise presumably has acoustic qualities that tend to distinguish it from other kinds of aperiodic noise, so that listeners who have learned to produce it can also infer the involvement of the glottal sound source when they hear it. Consequently, the presence of that sound source and the presence of the acoustic cue are not necessarily corequisites for language users to identify the presence of the feature [+spread] (see Ridouane et al. forthcoming 261, 277-278). Instead, the question of whether the definition of this or any other feature is primarily articulatory or auditory depends on how the feature's correlates are organized. As Ladefoged (2006: 30) alludes, thanks to the "complex, many-to-many relationship" of gestures and cues, some features may marshal a range of articulations to generate more or less the same bundle of cues under different conditions, while other features may prioritize relatively simple gestural attributes that produce a wider range of acoustic effects under different conditions. The former sort of feature is clearly auditorily motivated, the latter articulatorily motivated. A third possibility is for some features to have simpler, more one-to-one mappings between gesture and cue bundles; in those cases it is moot to ask whether the feature is more articulatory or more auditory.

3.3.6.6 Conclusion

Our review of aspiration indicates that an auditory redefinition of the feature responsible for it has several advantages. The primary correlate which determines the organization of the other correlates that map to this feature appears to be glottal turbulent noise, though phonological classes including voiceless fricatives and aspirated stops but excluding voiced fricatives seem to rely on a more complex definition such as turbulent noise *and* vocal fold abduction, due to oral frication masking any acoustically distinct glottal turbulent noise pattern.

In languages where the definition of this feature is organized on a primarily auditory basis, the feature may need a new name. I leave this to future debate. For ease of reading here, I retain [spread], abbreviated as [spr] following Hall (2007).

3.3.7 Representing voice

3.3.7.1 Articulatory or acoustic [voice]

Chomsky and Halle (1968: 326-329) focus on the articulation of sounds commonly classified as voiced and identify this especially with two features: non-tensed supraglottal musculature and adducted vocal folds, the latter correlating with [+voice] while actual vocal fold vibration is treated as merely a consequence of the proper articulatory and aerodynamic conditions. However, subsequent work (Ladefoged and Maddieson 1996, Clements and Osu 2002) has identified a greater range of articulatory factors that combine in various ways to make voicing possible. The requirements differ depending on stricture (cf. §3.4). Jakobson, Fant, and Halle (1954: 18, 26) and Ladefoged (1999: 611) emphasize the auditory correlates of [voice]. The primary cue of [voice] in this view is periodic fundamental frequency within the range perceived as pitch, manifested as the voicing bar at the bottom of spectrograms. The vocal folds are the only speech sound source which produce such a sound, since other periodic sound sources in the vocal tract oscillate at much lower frequencies (trilling: Stevens 1999, Shadle 2010). Ladefoged (1999: 611, 2006) concludes that the definition of [voice] is auditorily organized. In other words, audible periodic vocal fold movement is the correlate which motivates the various articulatory and aerodynamic correlates often associated with phonological [voice]; their normal function is to sustain audible voicing (§2.6).

Some of those correlates also naturally associate with other cues (cf. Clements and Hallé 2010: 4), such as audible duration differences in the ratio of consonant closure to a preceding vowel, which can compensate for loss of audible periodic vocal fold movement under special circumstances like whispered speech. (I appreciate Bert Vaux bringing this to my attention; see also the discussion of context-sensitive cue availability in §3.5.) It should be noted that if features emerge on a language-specific basis, then this auditorily organized view of [voice] would only be expected to hold for languages where the majority of sounds specified for that feature do involve audible periodic vocal fold movement which cannot be attributed merely to voicing leak from adjacent sonorants (cf. §2.5.5.2.3, 3.2.2). The organization of correlates expressing a laryngeal contrast in other languages may be quite different (§3.3.9).

3.3.7.2 Abstract low-frequency [voice]; slack and stiff stops

Another proposal, developed in Kingston and Diehl (1994, 1995), Jessen and Roux (2002), and Kingston et al. (2009) among others, defines [voice] as a relatively abstract feature associated with low frequency but not necessarily with closure voicing. The low-frequency cues include F0 lowering on the following vowel. Kingston et al. (2009) claim that this relationship is actively controlled by speakers who deliberately covary different cues because they integrate perceptually. However, Hoole and Honda (2011) critique this claim in detail using EMG and other laboratory evidence. They conclude that the origin of consonant pitch-perturbation effects is indeed mechanical, though it can be actively enhanced in language communities that come to view the pattern as phonologically significant. If Hoole and Honda's account does more justice to laryngeal physiology than Kingston and Diehl's, this does not rule out perceptual integration of similar cues in an emergent-feature account, but in that framework there is no particular reason to call any abstract emergent feature [voice] rather than something else unless phonetic voicing happens to be a salient cue of the feature.

An advantage of not equating [voice] with low-frequency cues is that the phonetic and phonological behavior of slack stops and slack voice is different from voice per se: not only can slack stops have voiceless closure and not only can they apparently occur in unpredictable distribution relative to plain voiced stops (§2.5.5.4), but it appears that entities other than stops can be slack. There are sonorant depressors in the Nguni languages (Jessen and Roux 2002, Traill et al. 1987), nasal onsets trigger the same effects on following vowels as slack stops do in Korean (Kim and Duanmu 2004: 62), and syllables with some sonorant onsets or with no underlying onset consonant can contrastively bear the same cues as syllables with slack stop onsets in some dialects of Chinese (§2.5.5.3.2), prompting analyses that attribute the correlates of slackness to syllables or morphemes rather than to segments (see Chen 2008, Ga et al. 2011, Steed 2011). Notice that in these non-segmental Chinese analyses, slackness associates with only a subset of the tone inventory, but slackness still cannot be identified with a particular tone.

All of this suggests a phonetically natural feature defined by correlates similar to those found with various strategies for sustaining voice in obstruents, except that phonetic voicing itself is not consistently included and obstruence is not necessarily involved. While the goal of [voice] would be audible voicing, the goal of this distinct feature would be vocal fold slackening, typically with effects on both pitch and phonation (§2.5.5). One reasonable name for this sort of feature would be [slack], distinct from both [voice] and tone features. (Further problems with trying to conflate these three categories under the same feature representation scheme will be discussed shortly.)

Whether any segment can then be specified as both [+slack] and [+voice] depends on additional factors. In the case of slack sonorants, the answer depends on whether sonorant voicing is ever phonologically specified (cf. §3.3.7.4). In the case of stops, impressionistic descriptions in some languages of phonetically voiced stops consistently triggering audible breathiness and salient pitch lowering on the following vowel might conceivably be identified as cases of voiced slack stops (e.g. Brao: Mon-Khmer; Huffman 1976: 579), but this could be hard to motivate phonologically in the absence of contrast between voiced slack stops and voiced but not slack stops.

One kind of voiceless counterpart to slack stops typically raises pitch and increases spectral tilt, giving impressions of creaky or similar voice qualities; this stop type has been described contrasting with slack stops in Javanese and Korean, and conditioning a register that contrasts with a modally voiced register in Chanthaburi Khmer, among other languages (§2.5.4-2.5.6). One possible name for this sort of feature, again distinct from [voice] and tone features, would be [stiff]. The next section ends with a glance at [slack] and [stiff] as compared to two other proposed feature terms, [lowered larynx] and [raised larynx]. First, though, we will examine a model which puts the terms [stiff] and [slack] to a different, much broader use.

3.3.7.3 [stiff], [slack], and larynx height

In an earlier proposal about laryngeal features, Halle and Stevens (2002[1971]) *replaced* [±voice] with different combinations of [±stiff vocal folds] and [±slack vocal folds]. This move quickly sparked empirical objections (see Keating 1988: 17-22) and never gained much popularity (Kenstowicz 1994: 39), but it has still played an important

role in several works on laryngeal features (e.g. Ahn and Iverson 2003, Avery and Idsardi 2001, Iverson and Ahn 2007, Iverson and Salmons 2003a,b, Vaux 1998a,b).

In this model, the stop types /T T' T^h/ are [+stiff], which correlates with CT and to some extent thyroarytenoid (TA) activity (Halle and Stevens 2002: 46). However, we have seen that CT activity does not correlate with voiceless stop articulations in some Hindi, Japanese, and American English speakers (§2.3.3), and a study of Korean found CT activity roughly comparable in all three stop series including the slack (lenis) stops (§2.5.5.3.1). Hirose and Gay (1972: 162) concluded that "there is no EMG evidence, in the form of increased CT, VOC or LCA activity, to support the concept of [+stiff] vocal cords" as being a consistent component "for the production of voiceless obstruents" in the languages they reviewed (Hombert et al. 1973: 43 note further studies to the same effect). In addition to positing these apparently inaccurate phonetic details, Halle and Stevens' model also ignores the role of vocal fold spreading in voiceless unaspirated stops (Keating 1988: 18-19).

Similarly, though the model assigns [+slack] to voiced stops (Halle and Stevens 2002: 51, Halle 2009: 71-72), we have seen that contrastive slack stops are typically voiceless in the environments where they can be phonetically distinguished from voiced stops (§2.5.5.1). The model represents Korean voiceless lenis stops as [+spread, -stiff, -slack] (Halle and Stevens 2002: 51), but the phonetic evidence indicates that they are better classified as slack stops; however, [-spread, -stiff, +slack] is already Halle and Stevens' representation of ordinary voiced stops, and [+spread, -stiff, +slack] is their representation of Hindi-type voiced aspirates. The Korean lenes thus pose a problem for the model's claim to provide distinct and phonetically accurate representations for all three types of sounds. On top of these issues, Kingston has observed (Vaux, p.c.) that replacing [+voice] with other features as in the Halle-Stevens model formally ignores the purpose of articulatory strategies that facilitate voice in obstruents.

Even the model's aptitude for representing the consonant-tone interactions that inspired it is questionable. Not only voiced but also phonetically voiceless consonants

function as tone depressors in some Bantu languages and Wu Chinese dialects (Jessen and Roux 2002, Schachter 1976; Chen and Downing 2011). The pitch lowering is dramatic in Zulu and interacts with tonology in Wu, while phonetic stop voicing in both languages does not have these effects. Voiceless aspirates also lower the pitch of following vowels in Madurese (§3.2.6.3). These effects cannot be represented as spreading [+slack] from the consonant to the vowel unless modally voiced obstruent stops, voiceless slack stops, and Madurese voiceless aspirates are all [+slack] and not [+stiff], but the feature [+spread] is incapable of distinguishing all these sounds from one another and from the Hindi-type voiced aspirates. Another, more general problem is that Halle and Stevens' model predicts similar pitch effects of consonants on both preceding and following vowels, in spite of a striking lack of evidence in the case of preceding vowels (Hombert et al. 1972: 42, Ohala 1973: 7).

Besides these phonetic challenges, the model faces multiple phonological ones. It represents both phonation and tone with the same features, but these parameters are fairly independent in some languages (DiCanio 2008: 177, Keating 1988: 21, Yip 2002: 57-59). While some tone-phonation combinations seem auditorily more difficult to distinguish than others, a number of languages resolve this problem by lengthening non-modally phonated vowels (Blankenship 2002, Gordon 1998) or by realizing contrastive tone and contrastive non-modal phonation on different portions of a vowel (Silverman 1997: 236). It is unclear how the full range of phonation-tone combinations would be represented in the Halle-Stevens model. The model also handles no more than three level tones, though systems with four or five are attested (DiCanio 2008: 177, Fromkin 1972: 51, Yip 2002: 57-59), and it treats mid tone as formally unmarked, though mid tone has a more restricted distribution than low tone in some languages (Fromkin 1972: 56-50). The representation of voicelessness and high tone with [+slack] also makes it impossible to describe intervocalic consonant voicing as a simple assimilatory phenomenon while adjacent vowel tones vary; vowel devoicing between

voiceless unaspirated consonants cannot be represented as assimilatory either (Fromkin 1972: 51).

Finally, ejectives and voiceless implosives lack distinct representations in the Halle-Stevens model, unlike a number of other stop types not known to contrast in any language. After reviewing Ladefoged's (1973) criticisms of this inconsistency, Iverson (1983: 333) suggests a "strictly non-contrastive vertical laryngeal movement feature" which could be applied to this problem. The feature [lowered larynx] ([LL]) has also been posited as a non-contrastive specification of Madurese voiced and aspirated stops (Cohn 1993a, Trigo 1991) and as a specification differentiating first and second register stops in Eastern Cham (Brunelle 2005: 248). Larynx lowering can create cues distinct from vocal fold slackening in the case of true implosives, where sufficiently rapid and substantial larynx lowering can generate a distinctive crescendo during closure as well as an audibly ingressive release (§2.5.5.2.3, 2.6.2); in such cases, [LL] seems a phonetically reasonable way to distinguish voiceless implosives from voiceless ejectives. As an articulatory feature, [LL] also seems a plausible way of associating several of the cues of larynx lowering which co-occur in a number of register languages, as proposed by Brunelle (2005: 211-212, 236; §2.5.4), though some of these cues also result from tongue-root advancement (see Lindau 1979, and Tiede 1993 using MRI data). Other problems noted above with the Halle-Stevens model remain, however.

To conclude our discussion of Halle and Stevens (2002), while they may be right that voiceless and voiced stops tend to condition higher and lower pitch respectively on following vowels for mechanical reasons (Hoole and Honda 2011, cf. Halle 2009: 72), incorporating this fact into phonological specification by representing binary voice and pitch distinctions with the same pair of features creates a number of phonetic and phonological difficulties. An auditory feature [voice] is transparently phonetically motivated and does not face these problems, and in an emergent framework, it naturally correlates with various articulatory strategies that help sustain voicing, among their other effects (fig. 3.1). However, the [LL] feature to which Iverson (1983) alludes may be preferable over [voice] for expressing some register distinctions. In this role, [LL] might also be descriptively more accurate than [slack] in languages where the lower register has pitch or vowel quality cues but does not associate with slack (weakly breathy) voice quality (on this variable see Svantesson and House 2006, Wayland and Jongman 2003). Comparing the appropriateness of [LL] and [slack] for representing the lower register, and similarly of [raised larynx] ([RL]; Trigo 1991, cf. Moisik and Esling 2011, Moisik, Czaykowska-Higgins, and Esling 2011) and [stiff] for representing the higher register, requires a more complete phonetic and phonological picture than space has allowed us to develop here.

3.3.7.4 Split treatments: Sonorant/Laryngeal Voice and underspecification

Another important proposal about laryngeal features handles reported cases of voiced obstruents patterning with sonorants by replacing [voice] with two Feature Geometry nodes: Laryngeal Voice (LV), found only in obstruents, and Sonorant Voice (SV), found in all sonorants and itself replacing the feature [son] (Rice 1993, 2005, Rice and Avery 1991). Besides questions about how this model handles voiceless sonorants and lateral obstruents (Yip 2011), there are more general concerns that we will consider here.

The Sonorant Voice model has two major effects. First, it enables phonological classes containing sonorants and voiced phonetic obstruents to be defined as SV by stipulating that the voiced obstruents are phonologically sonorant, or in the new terminology, SV, even though they surface as obstruents. Second, in languages where the voice of sonorants is phonologically active, the relevant rules are supposed to be formulated with SV, which is present in all sonorants by definition. At the same time, in languages where the voice of sonorants is not phonologically active while the voice of obstruents is, the relevant rules are formulated with LV instead.
The first achievement appears unnecessary. It seems needless to replace [son] with SV and then use SV to capture classes including sonorants and voiced obstruents, because these classes can be captured simply with [+voi], while classes containing only voiced obstruents can be distinguished as [+voi, -son]. The second major provision of the SV proposal, a simple representational distinction between cases where phonologically active voice involves the voice of sonorants and cases where it does not, offers a solution to a problem which is arguably better solved in a different way. We will review the original problem, then compare solutions.

In a combination of autosegmental tiers with underspecification (see Goldsmith and Noske 2005: 9-10, Steriade 1995: 465-466), the spread of voice from one obstruent to another had been modeled as an operation extending the association of [+voi] from one [-son] segment to the next. When the process operated across intervening sonorants without affecting or being affected by them, the question was how to capture the fact that their voice values were invisible or "transparent" to the process. Given the fact that sonorants are predictably voiced in many languages and that lexicons are more efficiently modeled if predictable information is omitted in them and filled in later by the grammar, it seemed a simple step to delay filling in sonorant voice until after obstruent voice assimilation, thus explaining why sonorant voice was invisible to such assimilation. In addition, if [-voi] was underspecified in voiceless obstruents on the grounds that it too is predictable given the [+voi] specification of voiced obstruents, then assimilation of voiceless to voiced obstruents would be a feature-filling operation as well. This approach ran into complications, however, in languages where sonorant voicing is phonologically active under conditions sensitive to lexical information, as in heteromorphemic postnasal stop voicing in Puyo-Pongo Quechua (Botma 2011 §2.2, Rice 1993) and in Japanese, where postnasal stop voicing in Yamato-stratum words feeds Rendaku, an obstruent voice dissimilation process in morphosyntactically conditioned types of compounds (Rice 2005). Such lexical and morphosyntactic information was supposed to have been lost by the level of grammar at which predictable features would be filled in, for a variety of

reasons that collectively played a major role in the paradigm of Lexical Phonology which was highly influential at the time (see Goldsmith and Noske 2006: 7-9, Kaisse and McMahon 2011).

Each of the key components of this approach to modeling phonologically inactive sonorant voice has subsequently been called into question. The psycholinguistic appropriateness of modeling segmental phonology with autosegmental tiers is debatable (Ussishkin 2011), and the properties of the relationship between autosegmentalism and underspecification theories remain unclear (Goldsmith and Noske 2006: 13). Steriade (1995) raised challenging questions about using underspecification to model transparency with tiers, and eventually the idea that predictable features are inactive (and hence underspecifiable) until late in the phonology was found plagued with significant problems (Samuels 2009a: 77-98). Even the general architecture of Lexical Phonology has proven at best a description of trends with exceptions (Kaisse and McMahon 2011).

Given formal problems with how tiers handle spreading, Samuels (2009a: 132-146) recommends replacing the Spread operation with a Copy operation. A similar approach has already proven fruitful in modeling long-distance consonant assimilations (Rose and Walker 2004). For example, obstruent voicing can be modeled as a process that searches for the nearest [-son] segment and copies [+voi] to that target. This leaves [voi] intact as a feature shared at some level by both sonorants and obstruents, facilitating the representation of processes where voice is phonologically active between sonorants and obstruents. Models which split the representation of voice into different features or nodes lack this advantage (Itô, Mester, and Padgett 1995: 577).

The recognition of flawed uses of underspecification is the more welcome because they have led to questionable increases in explanatory apparatus more times than just with the SV proposal. In at least one case where a predictable feature seems phonologically active, the paradox that it could not be unspecified at the same time was initially resolved not by creating new features but by invoking neutralization. The supposedly inactive feature [-voi] was left unspecified, and its apparent activity (spread) was reanalyzed as deletion of [+voi] from the target. The deletion then needed to be motivated on independent grounds. This led to the privative claim (Lombardi 1991) that phonological [-voi] does not exist and that apparent assimilation of one obstruent to the voicelessness of another is always an unrelated positionally triggered loss of laryngeal specifications. Among various problems with this claim (see Kim 2002, Steriade 1995) was the discovery of languages having voicelessness assimilation without positional devoicing. Wetzels and Mascaró (2001) highlighted Ya:thê: for this purpose, but San'ani Arabic (Watson 2002: 250-251) serves as well. Effectively to neutralize this discovery, another feature was then brought into play (Iverson and Salmons 2003a): [-voi] would be replaced in any given analysis by zero or [stiff] depending on whether non-specification or specification was preferable. Thus both [+voi] and [-voi] have been subjected to splitting, with the result being the pair SV/LV in one case and zero/[stiff] in the other.

Given innate universal features, allowing voiced obstruents to be specified with LV or SV (or allowing voiceless obstruents to be specified with [stiff] or zero) based simply on which is more convenient for labeling phonological classes and formulating rules or constraints is arguably too arbitrary if features have universal phonetic correlates – and not arbitrary enough if they do not. Given emergent features, phonological classes containing voiced sonorants and voiced obstruents may be defined with single features in particular languages, but there is no reason to label the voiced obstruents with "sonorant voice" because this is phonetically inaccurate.

3.3.7.5 Conclusion

Given the multiplicity and complexity of strategies for initiating and sustaining voice, it is not surprising to find such a diverse range of proposals replacing a simple voice distinction with articulatory or more abstract models of laryngeal features that jointly handle both voice and other phonologically important categories. The various shortcomings of the proposals we examined are resolved, however, with an auditorily organized definition of [voice] as a phonetically natural feature that leaves room for other ways of organizing correlates into more or less similar features within the universal phonetic geometry (§3.3.4). One of these alternative feature trends can be referred to as the [slack] type (§3.3.7.2). The next section presents a broader look at how singularities within the universal phonetic space can pose challenges for attempts at universal definitions of phonetically natural features.

3.3.8 Feature ambivalence and sonorance

Surveys of the 6000+ phonological classes compiled in P-base (Mielke 2007a) reveal that nasals and laterals phonologically pattern sometimes with continuants and sometimes with non-continuants, even in the same language (Mielke 2008a: 56-77, Samuels 2009a: 51), and similarly that glottal continuants pattern sometimes with sonorants and sometimes with obstruents (Miller in press, Parker 2011 §4, Ratelle 2009). Mielke (2008a) uses the term *feature ambivalence* for this type of situation: thus it appears that nasals and laterals are cross-linguistically ambivalent toward the feature [cont] and glottals toward [son]. Glides could also be considered ambivalent toward [cons] since they pattern sometimes with consonants, sometimes with vowels, in ways that are not explicable by syllable structure alone and that require a variety of underlying

representations (see Hayes 1989: 300, Levi 2011, Nevins and Chitoran 2008, Nevins and Vaux 2007, Padgett 2008, Svantesson et al. 2005).

Each of these situations can be plausibly explained phonetically. Nasals are orally stopped but still have unimpeded egressive airflow through the nose, while laterals similarly are oral-centrally stopped but still have lateral egress; thus both sound types have continuous airflow along some parameters but not others, raising questions about how to define the feature [cont] and whether to split it into multiple features that are superfluous for most other segments (Mielke 2008a: 74-77). Similarly, glides have less constriction than most consonantal segments but often have slightly more constriction and lower acoustic intensity than the most similar vowels, causing debate about the definition of [cons] and whether to split it into two features [cons] and [vocalic] to distinguish finer degrees of stricture (see Levi 2011, Nevins and Chitoran 2008, Padgett 2008, Parker 2002). Again, this distinction is superfluous for other segments.⁴

The [son] ambivalence of glottal continuants has plausible phonetic causes as well. Glottal continuants typically have higher oropharyngeal air pressure (e.g. Parker 2002) and weaker formants (Clements 2006a: 3) than modally voiced vowels with identical oral articulations; yet this is due to the turbulence caused by vocal fold spreading, and no supraglottal constriction need be present (see Ladefoged and Maddieson 1996: 325-326, McCarthy 1994: 193). Partly due to this fact, glottal continuants can show considerably less formant damping than stops and fricatives (see Clements 2006a: 3, Clements and Osu 2002). In explosive stops and fricatives with salient frication, there is a more consistent association between airflow obstruction, increased oral air pressure, and formant damping (including reduced amplitude or even complete failure of voice), while in most other sounds there is a complementary association between freer airflow, lower oral air pressure, and clearer formant activity. The most commonly proposed phonetic correlates of [son] are drawn from this set of

⁴ This particular use of [vocalic] should not be confused with other, earlier proposals that used the same feature name, such as Chomsky and Halle (1968: 303-303) and Jakobson, Fant, and Halle (1952: 18-20). Neither of those earlier uses has found much subsequent favor, as can be seen from the major sources on feature theory cited at the beginning of §3.3.6.1, but the reasons are beyond our scope here.

properties (see Parker 2002: 48, 135, Clements and Osu 2002), but glottal continuants diverge from most other sound types in behaving like sonorants with respect to only some of these properties and like obstruents with respect to others.

Non-explosives behave inconsistently with respect to the same correlates, but in roughly the opposite way that glottal continuants do, by having reduced oral air pressure on one hand with complete oral airflow stoppage and little or no formant activity on the other. Given these properties of non-explosives and glottals, Clements and Osu (2002) propose splitting [son] into two features, [son] correlating with acoustic properties and [obstruent] correlating with oral air pressure buildup. Again, this distinction is superfluous for most other sounds, which is one reason why obstruence and sonorance are generally considered antonyms (e.g. Kenstowicz 1994).

Having the concept of feature ambivalence in mind is helpful in considering a few broader issues related to sonorance. In a framework with a closed, universal feature set, the main advantage of correlating sonorance with oral air pressure that does not rise much above ambient air pressure (cf. Parker 2002: 135-145) is that this readily captures the sonorant-like behaviors of both canonical sonorants and ambivalent categories including glottal continuants and non-explosives (see Botma 2011, Clements and Osu 2002, Miller in press), while obstruent-like behaviors shown by the same ambivalent sound types can be attributed respectively to glottal widening and stophood. Glottal widening and stophood both attenuate the sound signal as compared to modal voice with an unobstructed supraglottal tract, thus arguably reducing sonority (Miller in press) and making a sound more likely to pattern with obstruents regardless of its categorical [son] value. This frees [son] to be defined in terms of correlates which other well-motivated features do not cover, like relative oral air pressure, which we already infer is actively controlled by speakers ($\S2.6$). The feature [son] can then perform phonological work which other well-motivated features cannot do as easily. For example, nasal harmony systems are known where sonorants and non-explosives are targeted by nasalization spread while other stops are not (see Botma 2011, Clements and Osu 2002); this could be

attributed to the targets sharing lower oral air pressure, which associates naturally with nasality since nasal egressive airflow is unobstructed. In this approach, non-explosives would be sonorant stops, higher in sonority than other stops due to their sonorance but lower in sonority than other sonorants due to their stophood.

Ladefoged (1999: 615) argues that if the definition of [son] is separated from the cues of voicing and periodic formant activity, then the tendency of sonorants to be voiced implies that speakers identify the phonetic combination [+son, +voi] as a "salient psychological percept... a rather far-fetched notion for which there is no evidence". Defining sonorance in relation to voice seems to be the most popular alternative to aerodynamic definitions (see Botma 2011 §2.2, 4, Parker 2002: 48, 135). Some problems with geometric redefinitions of voice and sonorance were discussed above (§3.3.7.4); here let us reconsider Ladefoged's argument for a feature [son] defined by acoustic correlates that include voice. Ladefoged argues that without audible voicing playing a key role in motivating the organization of the phonetic correlates of [son], the definition of the feature becomes psychologically implausible. However, just as correlates can be combined on an auditory basis under a common feature ($\S3.3.1$), there seems no reason to exclude the possibility of auditorily motivated combinations of *features* (as well as feature combinations phonetically motivated in other ways: ch. 4). The example here would be defining [son] aerodynamically and explaining its relationship with [voice] as motivated by more or less the same auditory strategy that Ladefoged wanted to attribute to the feature [son] itself. Both ideas are coherent, but the aerodynamic definition of [son] results in less overlap with the definition of [voice] than does Ladefoged's auditory correlation of [son] with periodic formant activity. As a further note on psycholinguistic realism, the potential for psychological categories with multi-modal sources is also wellestablished both in general (§5.6.3) and for phonological categories in particular: Botma (2011 §5) discusses an experiment that found English stops likelier to be heard as aspirated when synchronized with a puff of air on the listener's hand, "suggesting that

information from the auditory and the tactile domain may combine to form a salient psychological percept."

This argument for an aerodynamic definition of [son] depends, though, on a more general principle of prioritizing the parsimonious distribution of correlates among as few features as possible. This makes sense in an innate-feature perspective, but in an emergent framework it is unmotivated because there is no minimal universal feature set, only correlates which can be mapped to abstract features on a language-particular basis. In that perspective, attention turns instead to the maximum number of features possible within a given phonetic space and to natural phonetic relationships among the correlates (§3.3.4), including cases of feature ambivalence.

The emergent analysis seems supported by the fact that all the cases of feature ambivalence which we have just examined are plausibly motivated by *phonetic* singularities: regions within phonetic space where one finds divergence of properties that otherwise tend to correlate. Phonetically motivated feature ambivalence reinforces a challenge that innate feature theories already face for a variety of other reasons (cf. Mielke 2008a: 15-28, 56-77, 98, 117): how to avoid both redundant and incomplete coverage of the facts. In an innate feature theory, cases of feature ambivalence entail either features parametrization (e.g. defining nasals as continuant on a language- or pattern-specific basis) or the expansion of the set of well-motivated features with additional features that are superfluous for the vast majority of sound types (e.g. defining laterals as [laterally continuant] but not [centrally continuant]). Neither option is convincing. Why should innate features fail to retain both universal correlates and a decent functional load in specifying contrasts, and why should they consistently fail precisely where phonetic singularities are involved? This amounts to saying that innate features not only largely reflect the phonetic apparatus, which has already raised redundancy concerns (e.g. Samuels 2009a: 55-56), but more specifically that wellmotivated innate features with universal correlates just happen to copy the most easily generalizable aspects of the production apparatus while failing to cover the rest. In lieu of such incomplete yet redundant coverage, it is attractive to investigate how both phonetically simple and less straightforward patterns might be acquired by the same general mechanisms. Mielke (2008a) undertakes this at length.

Below (§3.4) we consider another pattern which is similar to feature ambivalence in depending on phonetic singularities, but which involves three phonetically natural factors instead of two (like nasality and continuance). These three factors are voice, the feature responsible for aspiration, and stricture. The asymmetry in their interaction has especially interesting implications for the discreteness of features and the adequacy of conventional Feature Geometry models.

3.3.9 Laryngeal ambivalence in Germanic

The identity of the laryngeal contrast in English and German stops has been much debated. One series is consistently voiceless but has variable aspiration largely correlated with stress; the other is generally unaspirated but varies in voicing depending largely on the sonorance of adjacent sounds. Iverson and Salmons (1995, 2003b) represent the contrast with [spr], treating the variable voicing of the second series as passive; Kingston et al. (2009) argue for [voi], partly because of evidence that the second series is actively voiced word-medially between vowels in English (§2.6.1). The [voi] model must specify information about aspiration phonetically, and the [spr] model must specify information about voicing phonetically. These phonetic directions must differ from those in, say, Swedish, where one series is consistently voiceless aspirated and the other is consistently voiced in the majority of environments for the majority of speakers tested by Helgason and Ringen (2008).

If features emerge, the laryngeal contrast in English or German stops could theoretically involve a somewhat abstract phonological feature mapping to language- and even environment-specific bundles of phonetic correlates. Though we would expect the contents of these bundles to be diachronically related, it might not be the case that all the correlates associated in any environment with a given series center around a single correlate that synchronically motivates the others in a phonetically transparent way. For example, some degree of intervocalic stop voicing (devoicing) in voiceless (voiced) stops, or of aspiration (deaspiration) in unaspirated (aspirated) stops, could arise historically for phonetic reasons unrelated to lexical contrast. Several possible sources of change in voicing and aspiration have already been discussed (active and passive voicing strategies §2.6.1, presonorant voicing §3.2.3, unconditional devoicing §3.2.6, degree of aspiration correlating with stress §3.3.6.2, aspiration arising from imprecise articulator coordination §3.2.4). Subsequently, a learner might perceive and intentionally imitate or even enhance such phonetic patterns, again regardless of whether he or she perceives them as lexically contrastive or as making some other correlate more perceptually salient.

Such a process could result in a certain degree of complexity (cf. Docherty 1992: 128-129) in which it would be hard to determine whether a laryngeal contrast was organized primarily around a glottal width difference (e.g. Jessen and Ringen 2002: 192), a difference in oral air pressure during closure (cf. Butcher 2004), presence vs. absence of aspiration noise (with resistance to closure voicing switching in as a secondary correlate in post-stress intervocalic stops, for instance), or susceptibility to audible periodic vocal fold movement. In this last scenario, the key fact would be that the unaspirated series associates with greater amounts of voicing – not only through greater susceptibility to closure voicing, but also through the lengthening of preceding voiced vowels and through shorter VOT for following vowels.

Conceivably, speakers of a language might perceive a contrast as a combination of such scenarios. Individuals might also differ in their mental characterization of a contrast, thanks to different perceptions of its phonetic implementation which in turn correlate with inter-speaker phonetic variation (cf. Brunelle 2010: 20-21). Valuing descriptive rigor and insight will certainly entail identifying natural relationships among the phonetic correlates of a feature in order to detect the degree to which the feature's correlate bundle is synchronically organized in natural phonetic terms (§3.3).

3.4. Beyond Feature Geometry trees: larynx-stricture interactions

Stops are the most common segments to contrast in voice or aspiration (Kehrein 2002, cf. Maddieson 1984, Mielke 2007a). Contrastive voice is rarer in fricatives than in stops but still relatively common (Maddieson 1984: 28, 53). Distinctive aspiration occurs even more rarely in voiceless fricatives, while fricatives associated with breathy voice seem exceptionally rare (Jacques 2010, Kehrein 2002, Ladefoged and Maddieson 1996: 178). The role of abduction in fricative contrasts is not limited to control of (post-)aspiration, however. Some languages have two contrasting fricative series distinguished by duration, degree of airflow, and susceptibility to voicing; the latter two properties suggest a difference in glottal width involving control of abduction, though it does not necessarily audibly overshoot stop edge to produce aspiration (Welsh: Kibre 1997: 17, Ball and Müller 1992: 82-85; Dutch: van Rooy and Wissing 2001: 302, 311, 328). As for sonorants, they are voiced by default and appear to contrast only in presence or absence of vocal fold abduction; in other words, contrast among [n n^h n n^f] is not reported (Kehrein 2002).

To analyze the possible role of perception in motivating some of these patterns, let us temporarily limit the term *aspiration* to a substantial period of glottal turbulent noise (§3.3.6.4) between the release of a supraglottal constriction (stricture) and the onset of modal voicing in a following sonorant. Let us also define *aspirate types* as types of aspirated sounds distinguished by stricture, and *salience* as perceptual salience. As we have just seen, the number of languages in which a given aspirate type occurs apparently decreases in tandem with stricture, suggesting that stricture degree is important to the viability of aspiration as a contrastive cue (cf. Kehrein 2002: 117). The relative rarity of contrastively aspirated fricatives, and the predominance of sibilants among them, has also been related to salience (Jacques 2011: 1519). A review of the acoustic evidence and typological data suggests the following generalization: aspiration is more salient the more it differs acoustically from the immediately preceding stricture, and this measure of salience correlates broadly with both the number of languages in which a given aspirate type occurs and the number of segments that belong to each aspirate type in individual languages.

Thus, the most salient aspirate type is an aspirated stop, where the relatively quiet stop closure has high perceptual contrast with the turbulent glottal noise of the aspiration. Aspirated stops are also the aspirate type found in the most languages, and if a language has one aspirated stop it usually has several (Maddieson 1984, Mielke 2007a).

The next most salient aspirate type is an aspirated sibilant fricative, where the relatively high-amplitude high frequency of the oral constriction has moderate perceptual contrast with the lower-amplitude and wider-bandwidth (more distributed) frequency of the aspiration. Aspirated sibilant fricatives are contrastive in several languages (Jacques 2011, Kehrein 2002: 81, Kehrein and Golston 2004: 332, Ladefoged and Maddieson 1996: 179) and in complementary distribution with unaspirated allophones in Halh Mongolian (Svantesson et al. 2005). In most of these cases there are also fricatives at other places of articulation, which are not reported as aspirated in the sources just listed. Palatal, alveo-palatal, velar, and labiodental aspirated fricatives are reported but markedly less often than aspirated sibilants, perhaps because they involve less strident noise (Jacques 2011: 1519-1520, who proposes the implicational hierarchy s^h > $c^h \int^h s^h > c^h x^h t^h t^h$).

The remaining aspirate types seem to have low salience because both the constriction period and the aspiration are characterized by fairly low-amplitude, high-bandwidth aperiodic noise. For example, $[\theta^h n^h l^h r^h]$ and $[\theta n l r]$ respectively sound

very similar. I am aware of no languages in which these pairs of sounds contrast. As already mentioned, Kehrein (2002) finds no contrast between voiceless and breathy sonorants either. Botma (2011) observes that voiceless nasals tend to occur in languages with contrastive stop aspiration, supporting the suggestion already made by various other writers (see Kehrein 2002, Lombardi 1991, Vaux 1998a) that the voicelessness of sonorants and the aspiration of stops have the same feature specification. The auditory redefinition of [spr] as glottal turbulent noise would serve this purpose.

These patterns create an important phonological assymetry: as one moves from stops through strident to non-strident fricatives to sonorants, the phonetic space that languages are found using to implement independent [voi] and [spr] features goes from being clear and abundantly attested to being vanishingly small. Stricture thus apparently correlates with the number of viable laryngeal configurations and contrasts involving voice, abduction, and timing, creating a set of interesting asymmetries in the possible phonological interactions of laryngeal features. Feature Geometry was designed to represent universals in phonological feature behavior (e.g. Clements 1985, 2009), but it does not seem possible to capture the patterns that we have just examined using the standard tree structure of Feature Geometry. It is not clear what alternative representational structures would be more successful.

Ohala (2005: 8-9) has similarly argued that the natural relationships among phonetic correlates which recur cross-linguistically in phonological patterns are too complex to be captured by tree structures with strict dominance (where each element is immediately dependent on only one node) and that the solution is to model the phonetic naturalness of phonological patterns through sound change rather than in the formal representations of grammar. This point has been taken up by the emergent (Mielke 2008a), evolutionary (Blevins 2004), and minimalist streams of phonological research (Blaho 2008, Hale and Reiss 2000a,b, Morén 2007, Samuels 2009a), which continue to develop with considerable overlap (Archangeli 2011, Mohanan et al. 2009). Some of the implications are explored further in the next two chapters.

Before we conclude this chapter, one further insight about features can be gleaned from the arguments for emergence that we have developed so far. We have just seen how it becomes harder to tease correlates apart into two separate features [spr] and [voice] as stricture decreases, and in the previous section we saw how some correlates tend to pattern together except for particular types of sounds defined by singularities in the universal phonetic space. These patterns exemplify variation in the number of correlates available and in how freely they are available for phonological distinctions. As shown next, two opposite extremities in this range of variation appear to be occupied by a pair of phenomena that we touched on earlier in the chapter: enhancement and (some kinds of) positional neutralization.

3.5 Comparison of enhancement with partial positional neutralization

Phonetic correlates that are naturally related in terms of physiology or physics often combine in bundles for the realization of language-specific phonological categories (for example, Lisker 1986 identifies sixteen distinct phonetic properties that vary according to the stop "voicing" contrast in English). In the emergent framework promoted in this chapter, these language-specific phonological categories are features which may be more or less phonetically natural depending on how naturally related their phonetic correlates are. This has implications for our understanding of enhancement and partial positional neutralization.

In this framework, enhancement is readily viewed as the grouping of a larger bundle of *correlates* under a particular feature than is found under features with similar correlates in various other languages. For example, the laryngeal stop distinction in Swedish correlates with both aspiration and voice, while in various other languages it relies on one or the other (§3.2.4). This contrasts with a view of enhancement as the addition of redundant *features* to phonological specifications. For example, the tendency of voiceless fricatives to involve some vocal fold abduction has been related to a hypothetical default specification of voiceless fricatives with the feature [+spr] at some level of phonological representation distinct from mere phonetic implementation, with this default then being deactivated in languages with contrastively aspirated fricatives (Avery and Idsardi 2001: 48, Vaux 1998a: 509). For a somewhat similar treatment of different stages in the evolution of Southeast Asian register contrasts see Brunelle (2005: 268-272). In the emergent framework, this accumulation of features is needlessly complex except in particular languages where additional phonological evidence motivates multi-feature specifications of the enhanced categories. In all other cases, enhancement is more simply modeled as the association of a larger number of correlates with a feature in one language than with similar features in other languages.

At the other extreme of variation in feature-correlate mapping, partial positional neutralization can be viewed as a reduction of the number of correlates associated with a feature in a particular environment. This is precisely how Yu (2011) analyzes partial positional neutralization, contrasting it with previous works that treat the same phenomenon as a product of orthographic interference, phonetic analogy, or turbid representations. Yu relates much of this previous work to a more modular view of the phonetics-phonology interface, where the subtle cues which betray contrast in partially neutralized surface forms have no presence in the phonological representation. The phonological representation thus categorically neutralizes the segments in question, and the subtle evidence of contrast in the phonetic forms must be treated as the reappearance of a distinction that had been either erased or somehow hidden by the phonology. Yu reviews studies concluding that orthographic interference cannot explain all reported cases of partial neutralization, and he argues that experiments which found speakers much more consistent in producing than in perceiving partially neutralized distinctions can be explained without invoking categorical neutralization at the phonological level (cf.

Ernestus 2011 §2.5-2.6). Explanation of partial neutralization by phonetic analogy obviously complicates the architecture of the grammar by reintroducing previously effaced distinctions, and turbidity vastly enriches phonological representations to distinguish elements which end up being pronounced from elements which are only "projected" (structurally present but abstract). Yu's model of partial neutralization appears simpler and more natural: speakers simply map some features to smaller sets of correlates in some word positions than in others.

Yu refers to this conception of partial positional neutralization as "covert contrast," but in the emergent framework pursued here it is worth considering whether and in what sense users *hide* the contrast at any level of representation or implementation. Conceivably, a partially neutralized contrast has a higher profile in the speaker's memory as tied to production than in the auditory processing system, due to the lower perceptibility of the contrast in the neutralization position; recall the systematic production of reportedly inaudible contrasts in utterance-final position in Blackfoot and Oneida (§3.3.6.5). Independent evidence for some divergence in speakers' knowledge of production and perception has already been found in the different influences which these two kinds of knowledge exert on loanword adaptation (see Kang 2011 §3, Martin and Peperkamp 2011, Ussishkin 2011 §2.3) and in studies finding poorer auditory awareness of allophonic than contrastive distinctions, even when the allophonic distinctions are clearly systematically controlled in production (Martin and Peperkamp 2011 §2.1).

3.6 Conclusion

This chapter has argued for a principle of feature economy that biases language users against inducing grammars with unnecessarily large combinations of phonetically *unrelated* correlates in the realization of phonological specifications. The force of economy is seen in how often the laryngeal properties of stops in languages that lack laryngeal contrasts in stops are either conditioned by adjacent sounds or reduced to a uniform minimum (voiceless unaspirated). Similar effects are seen when all stop or obstruent series undergo positional laryngeal neutralization. When only a subset of the stop or obstruent inventory neutralizes, though, the neutralized outcome has a wider range of possible realizations, just as a stop series has a wider range of possible laryngeal specifications when they express laryngeal contrasts.

Phonetically *related* correlates commonly combine in bundles for the realization of language-specific phonological categories (features). In the emergent view explored here, enhancement is the grouping of a larger bundle of correlates under a particular feature than is found under a feature with similar correlates in various other languages, while partial positional neutralization is the reduction of the number of correlates associated with a feature in a particular environment. This perspective contrasts with previous analyses where enhancement is the addition of redundant features to phonological specifications and where partial neutralization is considered a symptom of orthographic interference, phonetic analogy, or turbidity invoked to allow complete neutralization of the contrast at some level of phonological representation. These previous analyses accommodate a more strictly modular view of grammar in which the phonetics and phonology components cannot "see" inside one another, while the emergent analysis presented here seems to align with abundant and increasing evidence that human cognition is much less strict than previously supposed in separating different kinds of knowledge either within memory or during processing (§5.6.3).

This chapter also noted that correlates which play an enhancing role in voiced stops in some languages frequently occur contrastively in preponderantly voiced stop inventories in other languages, while aspiration, which appears to have no natural enhancing role in voiced stops, does not seem to occur as the contrastive factor responsible for any preponderantly voiced stop inventory either. This is plausibly explained as another effect of feature economy, allowing "Jakobson's universal" to be modeled coherently within an emergent feature framework. Alleged counter-examples to Jakobson's universal were reanalyzed. A couple of other phenomena deriving from the natural relationships among phonetic correlates were discussed in this chapter. Several cases of feature ambivalence were attributed to phonetic singularities. We also reviewed evidence against Feature Geometry being innate and against strict domination in Feature-Geometry hierarchical structure. Non-cognitive motives for cross-linguistic feature co-occurrence trends are explored further in ch. 4, with the argument that innate cognitive stipulations about such patterns are redundant.

Finally, the middle of the chapter reviewed several major proposals about the representation of aspiration and voice. Since laryngeal muscles generally are not in a oneto-one relationship with auditorily distinguishable types of stops, and since the timing of laryngeal articulations required for various types of consonants is more complex than a single VOT parameter can capture, auditorily organized definitions seem appropriate for both aspiration and voice – at least in languages where the dominant allophones of those categories are respectively realized with glottal turbulent noise and phonetic voicing. In Emergent Feature Theory, though, there are no grounds for forcing a particular feature label like [voice] onto other ways of associating phonetic correlates with phonological categories (such as slack stops) merely because the categories are somewhat similar to prototypical voice. Instead, phonological features are language-specific categories induced by learners in ways influenced by the physics of speech; features can be compared on the basis of similarity in their phonetic correlates and distributional properties, and feature types vary widely in both frequency and phonetic naturalness. The implications are developed in ch. 5 with acquisition-related arguments that grammars differ in their contents rather than simply in how the contents are organized.

4 Feature Co-Occurrence Trends

4.1. Introduction

This chapter analyzes over seventy feature co-occurrence trends and concludes that they can be reduced, with some clarification in feature definitions, to around a dozen positive conditionals whose content can be plausibly derived from an even smaller number of non-cognitive factors. This finding for segmental feature patterns is mirrored in other domains of phonology by discoveries supporting the phonetically motivated emergence of sonority (see Henke et al. in press, Miller in press) and of syllable constituent weight (Browman and Goldstein 1988, Goldstein 2009, Goldstein et al. 2007, 2008, Shaw et al. 2009 on the different timing patterns of complex onsets vs. codas; cf. Goedemans 1998, Topintzi 2006a,b).

4.2. Feature co-occurrence trends

The most comprehensive list of cross-linguistic feature co-occurrence trends that I have found is that of Calabrese (1992, cf. 2005). Calabrese formalizes these patterns as hierarchically organized negative inventory constraints, evidently innate in the initial knowledge state rather than language-particular, and variously deactivated in individual languages to capture phonological patterns.

The constraints are of two kinds: prohibitions, stating feature combinations that never occur, and marking constraints which ban cross-linguistically avoided feature combinations and which are deactivated in languages that use those combinations. Each constraint bans a pair of features, optionally in combination with further features which are represented as a periphery using rule environment notation. Prohibitions are asterisked, while in the marking constraints, the first feature in the pair is understood as implying avoidance of the second, which is underlined. Thus prohibitions have the form *[A, B] (/ [_, C, D, etc.]) and marking statements have the form [A, <u>B</u>] (/_, C, D, etc.]).

The purpose of the constraint hierarchy is that cross-linguistically more freely distributed sounds are mentioned by fewer constraints, while cross-linguistically less freely distributed sounds are mentioned by more constraints including constraints that tend to be dependent on other constraints. Dominant constraints cannot deactivate without the deactivation of their dependents, so inventories can only contain the more marked sounds banned in the dependent constraints by also permitting the less marked sounds banned in the dominant constraints. Stops are treated as the unmarked obstruents, obstruents the unmarked consonants, nasals the unmarked sonorants; hence stops are undominated, while nasals dominate liquids.

Calabrese's (1992) seventy-odd inventory constraints are shown in Table 4. Dependency is shown by indenting the dependent constraint under the dominant constraint. Where a constraint immediately depends on more than one other constraint, the indentation system shows the nearer dominant constraint if it is adjacent on the page while the remaining dominant constraints are linked to the dependent with association lines. I have saved space by collapsing Calabrese's environmental notation, turning e.g. [A, B] (/ [_, C, D...]) into [A, C, D, ... B], and by specifying rhotics as [+rhot] rather than [+son, +cons, -nas] (which in Calabrese's hierarchy is read as non-lateral by default because laterals are more marked than rhotics). Constraints that I suggest should be retired are marked †. Possible additional constraints are shown in parentheses. Arguments follow in §4.3.

A striking aspect of the overall system is the large amount of content overlap. Most of the first three columns are distinguished only by [-cont], [+cont], and [+son/ +nas], for example, while for the most part each of the first several rows is distinguished only by one of Lab, Dor, [ant], [dist], [voi], [+cst], [+spr], [strid], and [+lat]. (Calabrese treats the place categories Lab etc. as Feature Geometry nodes rather than as features; cf. Halle, Vaux, and Wolfe 2000.)

| Stops | Fricatives | Sonorants | Rhotics |
|-----------------------------|------------------------------|-----------------------|-----------------------------|
| | [-son, <u>+cont]</u> | [+cons, <u>+son]</u> | [+son, +cons, <u>-nas]</u> |
| [-cont, Lab] | [+cont, <u>Lab]</u> | [+nas, <u>Lab</u>] | *[+rhot, Lab] |
| [-cont, <u>+rd</u>] | | | [+rhot, <u>Gut]</u> |
| [-cont, <u>Dor</u>] | [+cont, <u>Dor]</u> | [+nas, <u>Dor]</u> | *[+rhot, Dor] |
| [-cont, <u>-ant]</u> | [+cont, <u>-ant]</u> | [+nas, - <u>ant]</u> | [+rhot, <u>-ant]</u> |
| [-cont, <u>+dist]</u> | [+cont, - <u>dist]</u> | [+nas, <u>+dist]</u> | [-ant, +rhot, <u>-dist]</u> |
| [-cont, <u>+voi]</u> | [+cont, +voi] | [+son, <u>-voi]</u> | ([-son,+cont, +rhot]) |
| [-cont, +cst] | [+cont, <u>+cst</u>] | [+son, +cst] | |
| [+voi, -cont, <u>+cst]</u> | [+voi, +cont, <u>+cst]</u> | | V <u>Laterals</u> |
| [-cont, +spr] | [+cont, +spr] | [+son, <u>+spr]</u> | [+son, <u>+lat]</u> |
| [+voi, -cont, <u>+spr]</u> | $[+voi, +cont, \pm spr]$ | | *[+lat, Lab] |
| *[-cont, +strid]† | [+cont, <u>-strid]</u> | *[+son, +strid] | [+lat, <u>Dor]</u> |
| [-ant, -cont, <u>-dist]</u> | [+voi, +cont, <u>-strid]</u> | | [+lat, <u>-ant]</u> |
| *[-cont, +lat] | [+cont, -son, <u>+lat]</u> | *[+lat, +nas] | [-ant, +lat, <u>-dist]</u> |
| *[-cont, +RTR] | [+cont, <u>+RTR</u>]† | *[+nas, Gut] | *[+lat, Gut] |
| [Dor, -cont, <u>-ATR]</u> | [Dor, +cont, <u>-ATR]</u> | [+nas, <u>+cont</u>] | |
| | (*[-son, +cont, +nas]) | | ([-son, +cont, +lat]) |
| | | | |

Table 4. Constraint set adapted from Calabrese (1992).

| <u>Opposites</u> | | <u>Vowels</u> | |
|---------------------|----------------|--------------------------|--------------------------------|
| *[-cons, -son]† | *[+hi, +lo] | [+lo, -cons, <u>-bk]</u> | [+lo, -cons, <u>+ATR</u>] |
| *[-cons, -cont]† | *[+spr, +cst]† | [+lo, -cons, <u>+rd]</u> | [-hi, -cons, <u>+ATR]</u> |
| *[-cons, +suction]† | | [-lo, -cons, <u>-hi]</u> | [+hi, -cons, <u>-ATR]</u> |
| | | [-cons, <u>+nas</u>] | [+bk, -lo, -cons, <u>-rd</u>] |
| | | | [-bk, -lo, -cons, <u>+rd]</u> |

The complexity of the constraint set reduces further on closer inspection. Some constraints, like the prohibitions *[+spr, +cst] and *[+hi, +lo], trivially follow from definitions of those features as incompatible opposites – though we have seen both physiological and phonological evidence that in fact this is probably not the case with [+spr] and [+cst] (§2.4). In connection with incompatible opposites it bears noting that a number of other gesture combinations have been identified as possible but phonetically antagonistic – that is, relying on components that clash or are in tension with one another. These combinations include voice and supraglottal stop closure (§2.6), nasal egress and oral frication (Gerfen 2001), high vowels and tongue root retraction in vowel harmony in African languages (Downing 2009a,b), and high vowels and pharyngealization in emphasis harmony in Arabic (Thompson 2006). I find none of these antagonistic combinations reported as occurring systematically at the phonetic level except in phonological contrast with less antagonistic configurations containing each of the antagonistic gestures apart from the other.

Most of the remaining constraints in Table 4 can be attributed to a fairly small set of phonetic explanations. I sketch these next, then defend the methodology and explore the results further in §4.4.

4.3. Explanations

4.3.1 Place

By far the most common type of place distinction in stops is a three-way distinction between labial, coronal, and velar (Maddieson 1984: 31-32). In addition to any articulatory motives, this pattern seems to have a plausible acoustic motive in the way it disperses the patterns of formant perturbations that stop releases at different places of articulation cause on following vowels (Morén 2007: 46-58, Schwartz et al. 2012; further on acoustic cues of stop place see Delattre, Liberman, and Cooper 1955,

Harrington 2010a, 2010b: 171-219, Lindblom and Sussman 2012, McLeod et al. 2001, Sussman and Shore 1996). In this section we examine more detailed asymmetric trends in place of articulation.

4.3.1.1 Coronals

The role of Lab and Dor in Calabrese's marking statements is almost purely to formalize Cor as the unmarked place without ever mentioning it. The arguments that have been advanced for a special status of all or some coronals in various aspects of phonology are many and varied, but the notion that the various special behaviors of coronals were generally a result of coronals having a simpler cognitive representation (e.g. Paradis and Prunet 1991) has lost some popularity. Peculiar behaviors of coronals have more recently been derived from non-cognitive factors like articulation, acoustics, and frequency (see Frisch 1997: 167-168, Gafos 1996: 3-10, 131-174, Jun 2004, Walker, Byrd, and Mpiranya 2009), and arguments have accumulated that several phenomena are inconsistent with across-the-board claims of coronal "unmarkedness" (e.g. Clements 2009: 60 citing Yoneyama, Beckman, and Edwards 2003 on dorsals being more frequent and acquired earlier than coronals in Japanese; also Howe 2004, Hume 2003, Hume and Tserdanelis 2002, Rice 2009).

This has gone hand in hand with a general shift away from Sagey's (1986) call for behavioral asymmetries in sounds to be formalized through a universal markedness hierarchy where increased markedness correlates with increased representational complexity (Mielke 2008a: 24; more generally Haspelmath 2006, 2008a). Consequently, the two-term non-liquid constraints containing Lab or Dor in Table 4 do not seem to be an illuminating way of formally acknowledging the diverse factors for unique behaviors of coronals.

As for the sub-classification of coronals, Calabrese's [dist] constraints indicate that stops and nasals generally prefer to be apical while posterior coronal liquids prefer to be laminal, but variation in the tongue shape of coronals across languages and across speakers of the same language turns out to be considerable and under-documented (see Dart 1991, Gafos 1996: 132, Ladefoged 2005: 6). This is not surprising given that the front end of the tongue is flexible and curved, inherently presenting a fluid set of multiple surfaces, and is also a uniquely agile articulator, moving quickly with minimal displacement of the formants of neighboring sounds and capable of producing similar acoustic effects from articulatory shapes that differ considerably in the exact details (see Gafos 1996: 9-10, 132-171, Steriade 1995: 145-147). While Ladefoged (2005) and Gafos (1996) conclude that traditional generalizations about the laminality of different types of coronals in various languages are fairly simplistic and misleading, Gafos makes considerable progress in relating phonological behaviors of coronal fricatives to articulatory details. On the whole, the evidence weighs against phonetically arbitrary features being needed to capture cross-linguistic phonological trends distinguishing different types of coronals. If we were to project from the incomplete state of current research, the prediction would be that whatever coronal-dependent features end up being validated will be phonetically natural. It seems too early to discuss [dist] feature trends any further here.

The notion that coronals tend to be anterior (encoded within Calabrese's markedness hierarchy by [-cont, <u>-ant]</u>, [+cont, <u>-ant]</u>, [+nas, <u>-ant]</u>, [+rhot, <u>-ant]</u>, [+lat, <u>-ant]</u>) can be attributed partly to an artifact of analysis, partly to quantal factors, and partly to a hypothesis developed shortly below (§4.3.1.4), that omissible gestures increasing the complexity of a sound are omitted in many languages and rarely exclusively relied on in any language. The artifact of analysis is that the relative backness of coronals is often not specifically documented unless it is contrastive, leading to an impression that coronals not contrasting for [ant] are [+ant] rather than [-ant]. This impression might actually not be phonetically demonstrable in some languages. If no language has only retroflex coronals, however, this may indicate that the default position of the tongue tip is relatively anterior and that curling it back to produce posterior apicals

and retroflexes takes more effort. Finally, the fact that most languages avoid noncontrastive, unconditional realization of their consonantal coronals as palatals (which are [-ant, +dist]) may be related to the relative indistinctness of that place of articulation (see Stevens and Keyser 1989: 98): the lips and front and back of the tongue are distinct enough to suggest that the boundaries between them are effectively quantal, but palatals do not enjoy such a boundary with either more anterior coronals or dorsals and their acoustics are accordingly less easily distinguishable than those of anterior coronals and other places are from each other. The impression that coronals tend to be anterior may thus derive from a combination of different factors.

4.3.1.2 Rounding

Rounding correlates with lip protrusion, whereas [p b] etc. show lip compression (see Ladefoged 1999: 594-595). Labial stops obviously need to press the lips together, while in low vowels for example it is relatively difficult to round and protrude the lips because they are already spread to accommodate the fact that the mouth is more open in low vowels than in any other segment. The constraints [-cont, $\pm rd$] and [$\pm lo$, -cons, $\pm rd$] can thus be derived from articulatory facts. The widespread correlation of backness with rounding among vowels expressed in the last two constraints in Table 4 has been explained acoustically (see Maddieson 2008a) as a way of reinforcing lower formants.

4.3.1.3 Liquids and nasals

Notwithstanding the articulatory complexity of laterals and rhotics (Ladefoged and Maddieson 1996), their highly limited place distribution has a clear articulatory basis. Laterals must be lingual (coronal or dorsal) because laterals are defined by air escaping over the sides of the tongue. Most sounds classified as rhotic are taps, flaps, or trills, which require rapid movement that may be easier with the coronal than with the labial and uvular articulators and that appears to be impossible with the velar and pharyngeal articulators. (On labial taps and flaps see Olson and Hajek 2001, 2003; on labial trills, Ladefoged and Maddieson 1996.) The remaining sounds which have been classified as rhotics are glides with articulatory place trajectories and formants similar to those of the rhotics already mentioned. These glides are typically apical, retroflex, or bunched coronal, or uvular (e.g. [½] in German: Kohler 1999). The corresponding syllabic sounds are not widely used as vowels in the world's languages, while the syllabic equivalents of glides that have not been analyzed as rhotic ([j w β q u]) are more often used as vowels ([i u \pm y u]). Thus a combination of factors may explain the constrained place distribution of laterals and rhotics. The relative rarity of non-coronal laterals may also relate to the greater agility of the coronal articulator already noted, though I do not recall if I have seen this connection explicitly made in the literature.

The only nasals judged articulatorily impossible by the International Phonetic Association (1999) are those produced behind the velum and the attached uvula, because air cannot be cut off in this region and still flow through the nasal passage. I assume Calabrese's prohibition *[Gut, +nas] reflects this, but I am not sure whether it includes uvular nasals. Uvulars are discussed in the next section.

4.3.1.4 Uvulars, laryngeals, and clicks

Uvular articulation (which I presume is referred to in [Dor, -cont, <u>-ATR</u>], [Dor, +cont, <u>-ATR</u>]) involves movement of the back of the tongue not only up but backward toward the uvula, which differs from the targets of more anterior places of articulation in being neither firm (like the teeth and palate) nor subject to well-controlled movements itself (like the lips, which in bilabials are both articulators and articulatory targets for one

another). This complexity and instability, especially as compared to velars, may contribute to the relative rarity of uvulars and to the fact that their presence usually implies the presence of velars too (though there are exceptions: Maddieson 2008c).

The prohibition *[-cont, +RTR]) presumably bans stops with the tongue root as the primary articulator. In light of Esling et al. (2005), this appears to be correct, but it also appears that pharyngeals in general have aryepiglottic constriction rather than tongue root retraction as their primary articulator. If no obstruents are primarily articulated with tongue root retraction, then the constraint [+cont, <u>+RTR</u>] (which refers to fricatives in Calabrese's hierarchy) should be a prohibition too. The best feature representation of pharyngeals will not concern us here, but from recent literature on laryngeals (§2.4) it appears that stops primarily articulated with a strong constriction in the upper larynx, including the aryepiglottic folds, play the phonological role of pharyngeal stops and are not only relatively rare but articulatorily quite complex. Pharyngeal continuants also appear relatively difficult to produce, since their considerable tendency to perturb surrounding formants indicates that it is not easy to navigate abruptly to and from a pharyngeal articulation (unlike with coronal liquids, §4.3.1.3).

Another instance where phonological rarity may be connected with the greater effort or complexity in implementation is the case of clicks, which require two place gestures (one velar, one ante-velar) coordinated to produce suction. (They do not require a separate suction feature; see Howe 2003. Here I refer only to clicks that are phonological, i.e. contrastive and lexically represented; phonetic clicks and other nonpulmonic consonants including clicks are also attested non-systematically in various languages as epiphenomenal realizations of other segments or segment sequences: see Ohala 1995, Simpson 2010.)

While our understanding of the correlation between effort/complexity and frequency is not advanced enough to permit many predictions, it is clear that languages do adhere to some limits of this sort; otherwise nothing would prevent the sound changes which produce less common and more difficult or complex sounds from leading to more

languages with such sounds, or to languages where such sounds occurred much more often.

Perhaps a key factor is the independence of the gestures involved. All languages (see Ladefoged and Maddieson 1996, Maddieson 1984) employ place, the simple distinction between a closed and open oral passage (stops vs. vowels), modal voice as an aerodynamic default in vowels, and voicelessness as an aerodynamic default in at least some of the stops. Place is related to stricture, stricture to aerodynamics, and aerodynamics to voice. Without these elements there would not be enough segments to create a lexicon (cf. Clements 2009: 35). Laryngeal constriction and double place on the other hand are complex and not indispensible for a reasonably sized lexicon. It makes sense for independently activated, logically omissible activities to have lower cross-linguistic frequency. This in turn recommends that we consider the extent to which cross-linguistically avoided phonological configurations can plausibly, if still only hypothetically, be connected to the theme of omissible complexity or difficulty. We will consider the theoretical implications of this in greater depth shortly (§4.4) after surveying the remaining feature co-occurrence constraints mentioned above.

4.3.2 The larynx and aerodynamics

4.3.2.1 General

The constraints [+son, <u>-voi</u>], [-cont, <u>+voi</u>], and [+cont, <u>+voi</u>] in Calabrese's hierarchy express the well-established generalization that voice is default in sonorants but antagonistic in obstruent stops and fricatives. The aerodynamic basis for this is one of the most well established phonetic explanations in phonology (\S 2.6).

There are several cross-linguistically important relationships among [voi], [son], [spr], [cst], and [cont] which Calabrese's hierarchy does not capture but which were explored in chs. 2 and 3, including the absence of a [spr] contrast in voiceless sonorants

and the reason for default voice in non-explosives. Generalizations that were made earlier (§2.6, 3.4) about how ranges in possible laryngeal feature contrast are controlled by cooccurring supralaryngeal features are difficult even to represent with an abstract feature tree but easily explained from phonetic facts.

What the two-term constraints in Table 4 involving [+spr] and [+cst] express is that these gestures are contrastive on orally placed segments in only a minority of languages. This fact about frequency may be related to the fact that contrasts in vocal fold abduction and laryngeal constriction require greater effort to produce and process, because there are more configurations to keep track of. Laryngeal constriction is also a logically omissible way to increase articulatory complexity, as noted above. Whether aspirated stops are a more complex articulation than unaspirated stops or the reverse is difficult to determine (§3.2.4)

4.3.2.2 Nasals and liquids

The active lowering of the velum necessary for categorically nasal sounds appears to be another instance of omissible effort. While the default position of the velum is arguably somewhat lower than a fully raised position, producing moderately nasal sounds (Francis Nolan p.c.), it must often be lowered even further to produce distinctively nasal sounds, suggesting that distinctively nasal sounds are not an articulatory default but typically require active velic lowering. Similarly, raising the tongue center but keeping the tongue sides low creates a more complex tongue configuration than keeping the sides and center relatively uniform. Such observations may relate to why not all languages have lateral or nasal sounds (Maddieson 1984: 61, 74) and why nasal vowels imply both oral vowels and at least phonetically nasal consonants. (Usually nasal vowels imply phonologically nasal consonants too, but in some languages nasality is contrastive only on vowels and predictable on consonants: Clements and Osu 2003.) The rarity of nasalized fricatives has been persuasively related to the aerodynamic difficulty of maintaining enough oral airflow for frication while also having nasal airflow (see Shosted, Ohala, and Sprouse 2005, Gerfen 2001). Finally, a dispreference for mixed egress paths plus the other complexities already mentioned may relate to why phonemic nasal-laterals are avoided; it is also possible that "the presence of antiresonances from *two* 'sidebranch' resonators, the nasal cavity and the short cavity behind the lateral's coronal closure... would result in an ambiguous percept" (Nolan p.c.). These patterns are expressed in Calabrese's constraint hierarchy with [+son, +<u>lat</u>], [-cons, <u>+nas</u>], [+nas, <u>+cont</u>], and *[+lat, +nas], as well as the implied constraint [+cons, <u>+nas</u>] which he formulates as [+cons, <u>+son</u>] because nasals are the unmarked sonorants in his hierarchy.

The prohibition on nasalized laterals cannot extend to surface distinctions, however. In Sanskrit (Renou 1961) a word-final nasal becomes a nasalized /l/ through sandhi when the following word begins with /l/, and in Catalan (Bonet and Mascaró 1997: 124), "in loans or acronyms, like *Chandler*, *Bentley*, or *INLE* (Instituto Nacional del Libro Español), the nasal becomes a (nasalized) lateral: [tʃaĨlər], [bɛĨli], [iĨle]." Since the same sources present each language as allowing surface [nn] and [ll], the geminate laterals with nasalization appear to be perceptually distinct even though nasalized laterals are not phonemic.

That rhotics are absent in some languages (the import of [+son, +cons, <u>-nas</u>] within Calabrese's hierarchy) can be attributed to factors as various as the articulation of rhotics. The iterative articulation of trills is uniquely complex. The distribution of taps and flaps is often uniquely restricted: they are favored in environments where the tongue is already moving from one position to another, which may relate to why a number of languages exclude them from word-initial position (see Ladefoged and Maddieson 1996: 216-17). However, the claim that rhotics are unmarked with respect to laterals (the import of [+son, +cons, <u>-nas</u>] dominating [+son, <u>+lat</u>] in Calabrese's hierarchy) does not seem

justified; rhotics and laterals each occur in about 80% of the 548 languages and 627 language varieties in Mielke (2007a).

4.3.2.3 Fricatives

The intermediate stricture and turbulent pressure of fricatives seem to require more effort and precision than either stops or vowels do, and this conceivably relates to the fact that while all languages have stops and vowels, fricatives are absent from nearly ten percent of the languages in the UPSID database (see Maddieson 2008b). The constraint corresponding to these observations in Calabrese's hierarchy is [-son, \pm cont].

Phonetically, stridence is an acoustically noisy property available only to fricative sounds (see Ladefoged and Maddieson 1996, Maddieson 1984: 51-58). This is reflected in Calabrese's constraints *[-cont, +strid] and *[+son, +strid]. The acoustic salience and hence perceptual clarity of strident fricatives may relate to the fact that non-strident fricatives imply at least one strident fricative in most languages (see Maddieson 1984, 2008b); this is reflected in Calabrese's hierarchy as [+cont, -strid]. Whether stridence is phonologically associated only with fricatives is another question (see §3.3.6.3 on the representation of affricates).

The generalization that strident fricatives may be all voiceless but are never all voiced ([+voi, +cont, <u>-strid</u>] dominated by [+cont, (-son,) <u>+voi</u>] in Calabrese's hierarchy) appears correct. It does not extend to non-strident fricatives. Three Australian languages are reported as having only one fricative each, where the fricative is voiced non-strident (Maddieson 2008b). Some languages have multiple non-strident voiced fricatives without corresponding voiceless ones and without having strident voiced fricative phonemes at all (e.g. Spanish: Martínez-Celdrán 2008; Central Buang and Western Fijian: Tryon 1994). Unlike strident sibilant voiced fricatives, other voiced fricatives frequently have glide variants or are not clearly distinguished from phonetic glides in the literature (Cser

2003: 122). Since stridence is heightened acoustically salient turbulence (Stevens and Keyser 1989), audibly strident voiced fricatives presumably require a stronger airstream and greater oral air pressure than other voiced fricatives. This requirement must be met at the same time that voicing is sustained. The unique aerodynamic needs of strident and especially sibilant voiced fricatives may relate to why they have a more restricted distribution than other voiced fricatives, and why they are not reported with glide variants as often as other voiced fricatives are.

One phenomenon that does not appear explicitly in Calabrese's hierarchy is that of fricatives with a nasal, lateral, or rhotic component. True nasalized fricatives occur phonetically, but both language data and aerodynamic reasoning make it highly doubtful that they occur contrastively (§4.3.2.2). Lateral and rhotic fricatives occur fairly infrequently and involve fairly complex articulations (Ladefoged and Maddieson 1996). In coronal trills, for example, maintaining high enough stricture to impede airflow and create frication impressionistically seems prone to hinder the vibrating motion of the trill itself, the same way that dribbling a basketball is harder the closer you keep it to the ground. Manipulating one or both sides of the tongue to achieve frication while the tongue center maintains stop contact may be only slightly easier. The rarity of lateral and rhotic fricatives may thus relate to the theme of omissible complexity and effort.

Phonological constraints reflecting these facts have been added to Table 4 in parentheses: *[-son, +cont, \pm nas], [-son, +cont, \pm lat], and [-son, +cont, \pm rhot]. The last of these might be better framed as a prohibition if rhotic fricatives are never contrastive, but I have not ascertained this.

4.3.3 Vowels

Among Calabrese's vowel constraints, we have already discussed *[+hi, +lo], [+lo, -cons, +rd], [-cons, +nas], [+bk, -lo, -cons, -rd], and [-bk, -lo, -cons, +rd]. The

remainder have plausible phonetic explanations. Low vowels have less physical room for front/back distinctions because both the tongue and the jaw move down and back when opening to the low position; hence a phonological contrast in [bk] is rare with low vowels. Calabrese's [+lo, -cons, <u>-bk</u>] predicts that low vowels behave phonologically as back vowels, but Mielke (2008a: 156) found that the phonological patterning of low vowels with back vs. front vowels is fairly balanced.

The idea that non-high vowels tend to be [-ATR] while high vowels tend to be [+ATR] (i.e. [+lo, -cons, +ATR], [-hi, -cons, +ATR], [+hi, -cons, -ATR]) might be explained by a preference for the tongue root and the tongue body to move together: moving the tongue body up may be easier if the tongue root advances somewhat, and having the tongue body in a lower position may be easier if the tongue root does not advance as much.

The constraint [-lo, -cons, <u>-hi</u>] expresses the trend for mid vowels to imply high and low vowels but not vice versa, which seems to be partly an analytical artifact (vowels cannot phonologically occupy a middle level in a vowel system unless there are both higher and lower vowels) and partly the result of dispersive tendencies (Flemming 2003). The latter, however, are not well enough understood for us to predict the degree of vowel distribution in acoustic space given only the height and backness relations of the vowels to one another (Hale, Kissock, and Reiss 2006).

Finally, Calabrese's prohibitions *[-cons, -son] and *[-cons, -cont] are trivially true or false depending on feature definitions: true if [-cons] is correlated to a lack of impedance or pressure incompatible with [-son] or [-cont], and false otherwise (cf. §3.3.8).

4.3.4 Interim reduction

Some of the constraints in Table 4 proved empirically unsatisfactory. Some, such as *[+hi, +lo], proved redundant given their feature definitions. A principle that omissible actions are not defaults can account for all constraints treating [+F] as marked where [+F] correlates with an omissible gesture. Nearly twenty constraints in Table 4 can be set aside for reasons such as these.

The remaining more than fifty negative constraints reduce to roughly a dozen conditional statements as exemplified in (1), where \rightarrow marks a one-directional implication and \leftrightarrow marks a bidirectional implication. Notably, the feature [son] is key to eight of the twelve statements in (1), and both [son] and [cons] are key to six. This is not surprising if [son] is defined in terms of pressure, because pressure implies some kind of control on airflow and the controls in this case are provided by other feature specifications, primarily involving stricture.

While the contents of Table 4 thus reduce to a dramatically smaller set of conditionals in (1), the minimum number of non-cognitive explanations sufficient to cover these patterns in light of the foregoing discussion seems even lower. Around half a dozen non-cognitive motivating factors are shown in (2) and keyed to (1) with superscript letters. This is not to say that only these factors are involved, but merely that the valid and non-trivial information content of Table 4 seems plausibly derivable from these factors. The ramifications of this claim are discussed next.

(1) Conditional restatement of feature co-occurrence trends

```
[+cons] \rightarrow [-rd],^{a} \{Cor (weak)^{b} \rightarrow [+ant]^{a}\}^{5}
[\alpha son] \leftrightarrow [\alpha voi]^{c}
[\alpha son] \leftrightarrow [\alpha cont]^{c}
[-son] \rightarrow [+cons]^{c}
[+strid] \leftrightarrow [+cont, -son, +cons]^{d}
[+nas] \leftrightarrow [+son,^{d} + cons^{e}]
[+nas] \rightarrow [-cont]^{e}
[+lat] \rightarrow Cor,^{b} [+son],^{f} [+cons],^{d} [-nas]^{e}
[+rhot] \rightarrow [+cont],^{d} [+son]^{f}
Dor, [-son] \rightarrow [+ATR]^{a}
[+lo, -cons] \rightarrow [-ATR, -rd]^{a}
[\alpha bk] \leftrightarrow [\alpha rd] / [-lo, -cons]^{g}
```

(2) Possible motives

^a easier articulation, given the antecedent

^b functional arguments for coronal uniqueness, such as articulatory ease and minimization of formant perturbation

^c aerodynamic dispersion (least vs. most obstruction/pressure) minimizing

intermediate configurations with their more complex articulations

^d comprehensive feature definition

- ^e favoring of oral and central egress and avoidance of egress path splitting
- ^f [α son] \leftrightarrow [α cont] (itself possibly motivated by *c*)
- ^g mutually reinforcing percepts

⁵ Read, "Consonantal segments tend to be non-round (possibly due to motive a), they weakly tend to be coronal (possibly due to b), and if they are coronal they tend to be anterior (possibly due to a)."

4.4. Implications

4.4.1 Introduction

The previous section has repeatedly invoked articulatory or perceptual difficulty and complexity in offering explanations for why some sound patterns are rare or nonexistent. The role of effort in phonetically based explanations in phonology is controversial. Here I hope to clarify some of the controversies.

First, effort-related explanations do not necessarily imply that phonology is biased to make languages become globally easier to speak over time. Understanding why requires us to distinguish sound changes from the static aspects of the speech production and processing systems which often influence them (§4.4.2). Second, it has been claimed that a counter-intuitive hypothetical bias favoring complexity or difficulty explains just as much data, with just as much validity, as a more intuitively plausible bias favoring ease or simplicity; the two biases are consequently supposed to cancel one another out, leaving effort-related explanations essentially useless in phonology. I argue that a wider perspective refutes this claim (§4.4.3). Implications for grammar design are treated in the next chapter.

4.4.2 Why sound change is not teleological

Ohala (1993: 262-264) and Blevins (2004: 71-78, 205, 278-280) argue that sound change is not driven by speaker goals such as making a language easier to speak or hear. It does not follow that unintentional effort reduction plays no role in sound change, however. The following analogy from Mielke (2008a: 82-83) provides a good starting point for establishing the sense in which sound changes often involve effort reduction
without the nature of sound change itself being driven by a goal of making languages easier to speak.

Due to the layout of the *qwerty* keyboard, some typographical errors are more likely than others... Acknowledging the role of the layout of the keyboard (or vocal tract) in what types of deviations from a target are most likely does not amount to saying that the result of these errors is more natural or optimal than the intended target...

Although it is hard to predict when a typing error will be committed, considering the layout of the keyboard and the content of the spell checker makes it possible to predict which deviations from the target are likely to occur, and which of those are likely to persist. Taking into account the reality in which typing (or language use) occurs does not require any sense of optimization...

In the keyboard analogy, certain combinations or sequences of keystrokes involve more difficult or complex finger movements than others, such that failure to strike some key is likelier in these cases than in many others. An example is failure to capitalize, or continuing to capitalize one or two letters longer than intended. These reductions or mistimings are not evidence that the activity of typing has a natural goal of striking fewer keys or of capitalizing more letters in general. However, these kinds of mistakes do reflect the fact that people are likelier to fail to perform all parts of a task or to perform the parts at the right times when the task is more complex, difficult, or fast-paced. This last consideration is why more typing mistakes are made at higher speeds.

In the phonetic realm, similar considerations apply to articulatory reduction and mistiming. Some sounds or sound sequences are more complex or difficult to produce than others, and thus likelier to be performed incompletely or with gestural mistiming. Reduction and mistiming are also likelier in faster or less careful speech styles. Another indicator of the relationship between difficulty and speed is that correctly and completely produced sounds may simply have longer average production times when they are relatively articulatorily complex. Take for example the following situation in Nootka dialects of Nuuchahnulth (Wakashan: Carlson and Esling 2003: 189):

The simple glottal stop of /wi:?u:/ 'nephew'... lasts about 100 ms, while the glottalized resonant series in Nootka, consisting of /²m, ²n, ²j, ²w/, has an articulatory structure consisting of simple glottal stop plus the resonant, lasting nearly 200 ms. Pharyngeals, such as the /ʕ/ in /ʕiħu:/ 'to cry after', require 250 to 300 ms to complete all of the articulatory maneuvers associated with the segment including preparatory constriction, radical occlusion and gliding into a subsequent segment.

Without such connections between articulatory complexity and speech rate, there would be no reason for errors like gesture reduction to be more common in faster styles of speech. A similar but perhaps more obvious connection holds between speech accuracy and attention. Thus, casual speech is often characterized by both higher speed and less complete or accurately timed articulatory performance.

One type of gestural mistiming is coarticulation, which can lead to assimilations (Blevins 2004: 144). Assimilations can simplify an articulatory sequence. Gestural reduction also has a simplifying effect by omitting all or part of a gesture. Both coarticulation and gestural reduction can thus reduce the effort required to articulate a sequence of sounds. Since coarticulation and gestural reduction are common articulatory sources of sound change (Blevins 2004: 142, 291-192), their effects can accumulate over time. However, sound systems are complex enough that simplifying the production of a word in one respect often makes it more complex in another (cf. Clements 2009: 34). The frequency of simplifying sound changes does not entail a global drive for languages to become easier to speak over time.

Crucially, a bias for languages to avoid certain kinds of effort or complexity – as well as absolute bounds which no languages are found to transgress – stem in large part from static properties of the speech production and processing systems. These properties are not diachronic anymore than the properties of the keyboard we considered earlier. The structure of the keyboard is not an aspect or goal of typographical error, and the definition of what is simple or easy to produce or perceive depends on properties of the speech system which are not aspects or goals of language change. Instead, these static and universal properties explain why some sound changes are more common than others, and also why sound changes never seem to produce certain outcomes, even cumulatively where every step in the chain is attested and unproblematic (§3.3.3, 3.2.6.1). As Kiparsky (2004, 2006) emphasizes, it is not clear how this latter fact would be handled by a theory which derives the explanation of sound patterns primarily from sound change.

4.4.3 Why phonology is not dysfunctional

Another important factor to consider in defining the role of effort in shaping sound patterns is the multiplicity of dimensions in which effort can be measured, as noted by Hale and Reiss (2000a,b). For example, neutralization can simplify articulation but increase input-output mismatch and ambiguity, forcing hearers to increase their reliance on context to determine underlying forms and meaning. This is a case where articulatory simplification arguably decreases perceptual ease. Loss of articulatory gestures can also induce greater articulatory complexity, as when vowel syncope creates consonant clusters. On the other hand, maintenance of existing contrasts can be attributed to a drive for communicative clarity and ease of processing, even though more contrasts mean more articulatory complexity (cf. Clements 2009). All of these situations can co-exist in a language, leading to phonological intricacies where complexity appears to increase in some ways while decreasing in others.

The frequency of situations in which effort or complexity increases in some ways while decreasing in others at first glance appears equally compatible with three different hypotheses: that languages favor upper bounds on effort and complexity, that they favor lower bounds, or neither. Hale and Reiss (2000a,b) label the first two hypotheses functional and dysfunctional and contrast them using the following examples: while the functional explanation would claim that a drive for articulatory ease merged contrast and led to increased perceptual difficulty in one case while a drive for perceptual ease preserved contrast in spite of consequent articulatory complexity in another case, the dysfunctional explanation of the same data would claim exactly the opposite – that a drive for perceptual difficulty (OBFUSCATE) was what led to the merger of contrast in the first place while a drive for articulatory difficulty (No PAIN, No GAIN) motivated preservation of contrast in the second case. Hale and Reiss (2000a,b) conclude that because some data sets are locally consistent with both kinds of explanation, neither kind should be used. This seems questionable for the following reason.

Human behavior in general shows much more sensitivity to upper than lower bounds on effort. Humans value devices that decrease cost and increase profit. Profit is whatever is desired; cost is what must be exchanged for it. The mere increase of articulatory and perceptual energy expenditure has no discernible profit, so we must presume its status is cost. The phrase No PAIN, No GAIN itself means that cost must be exchanged for profit, not that cost is profit. For example, it is reasonable to believe that water seeks the lowest available space in response to gravity, because that is where water is generally found and there is a wealth of other evidence for gravity regardless of how imperfect our understanding of it may be. The existence of sand bars in rivers is not evidence that water instead seeks to stay out of the sea. Similarly, it is reasonable to believe that the goal of speech is to communicate, because communication generally occurs when people speak, speech is intentional, and there is a wealth of evidence that humans and other creatures intentionally communicate in other ways. The existence of phonological complexity is not evidence that people speak with the intention of creating complicated sound patterns instead.

The dysfunctional approach to phonological explanation is thus invalidated by a lack of evidence for any profit motive, which means that it is not available to neutralize the value of the opposite approach – correlating the absence or rarity of certain sound patterns with the relatively high though still physically feasible effort and complexity they would involve. A model that acknowledges this correlation is more informative than

a model that ignores it by formulating the universal restrictions as innate cognitive stipulations. And while the measurement of these properties is itself complex, as in several of the allegedly effort-increasing examples from Blevins above, the overall picture – including the principles of cost and profit, dozens of feature co-occurrence trends (§4.3), and sound changes involving coarticulation and reduction (§4.4.2) – indicates that languages extensively avoid sound patterns which are relatively difficult to produce or process, while *there is no persuasive evidence that languages avoid anything because it is simply too easy* or that they ignore effort or complexity altogether.

As noted earlier, the number of interacting properties in any given sound pattern makes effort and complexity hard enough to measure that they have very limited predictive or quantifiable value as explanatory principles. My purpose in arguing that they still play a substantial role in the shape of sound systems is not to promote a more predictive formal model of sound patterns, but simply to highlight the tremendous extent to which sound patterns can be plausibly derived from properties of the speech production apparatus. Once an extensive set of such derivations is granted – regardless of all the details in each particular case – it becomes massively redundant to derive many of the same patterns from innate cognitive stipulations.

This insight readily finds a place in theories that deny effort reduction or simplification a major role in sound change, because there are plenty of other phonetic forces to discuss as sources of many sound patterns, especially when viewed as products of sound change. The reason I have emphasized considerations of effort and complexity in this chapter is that they appear to play an extensive role in the synchronic feature co-occurrence trends examined in §4.3.

4.5. Conclusion

The wide extent to which feature co-occurrence trends can be motivated by phonetic factors without recourse to detailed innate cognitive stipulations supports the general claim of emergent and evolutionary approaches to the design of the phonological component of the grammar and its relationship to sound change. In this perspective, the structure of the phonetic apparatus manifests itself as a causal factor in grammar design when it overrides faithful inheritance during acquisition (just as the structure of the keyboard only manifests itself when a typing mistake is made). The override is a historical event reflecting structural aspects of the phonetic apparatus and resulting in a change of grammar without those aspects themselves being components of the grammar – just as typographical errors change a text and reflect structural aspects of the keyboard and of the human hand without those aspects themselves being part of the text.

The phonetic forces that shape the linguistic evolution of sound systems can involve considerations of effort and complexity without becoming teleological or arbitrary, but the least redundant grammar design theory generally seems to exclude such phonetic forces from playing a direct role in the grammar: instead, it is their products – sound patterns instantiated in word forms – that are extracted into the grammar during acquisition. This claim is developed further in the next chapter, where I argue that innate content-specific constraints unnecessarily duplicate learnable input-output mappings, that profuse generation unnecessarily duplicates typological exercises which do not belong in grammars, and that online serial execution of rules sometimes unnecessarily duplicates steps in acquisition.

5 Grammar

5.1 Introduction

The last two chapters have argued that the details of crosslinguistically common features and feature co-occurrence patterns emerge on a language-specific basis from non-cognitive, phonetic factors that operate through sound change. This conflicts with the influential framework of Optimality Theory, which posits that grammars systematically differ only in the ranking of universal constraints. This chapter briefly reassesses some of OT's goals and components and compares them with alternative approaches to grammar design.

5.2 Grammar

5.2.1 Definition

I will use the term *grammar* for the knowledge system required for mentally processing auditory input and mentally planning speech production in a given language, where the purpose of the production function is to encode meaning in sound for a listener and the purpose of the processing function is to decode meaning from sound produced by another speaker (cf. Ussishkin 2011). This code involves a mostly arbitrary mapping between sound and meaning, where sounds and meanings each have a systematic organization that allows an infinite number of possible constructions from a finite, known set of elements (see Chandler 2011, Martinet 1949, 1957, 1983, Menn 2010: 101-127, 185-199). For obvious reasons, what speakers know about their language has primarily been explored by analyzing what they say, though brain imaging technologies now also

allow us limited observations of how the brain reacts during listening (Menn 2010: 196-197, 213, 390, Singer 2009, Woods et al. 2011).

In light of the errors and other idiosyncracies of individual speakers, as well as the fact that speakers never make all the statements which their language properly allows them to make or understand, it is not surprising that the distinction between knowledge and actions has played a major role in linguists' efforts to define the object of their study. Saussure explored the distinction in terms of *langue* versus *parole* (Saussure 1989[1916], W. Gordon 2004), while Chomsky has written of *competence* versus *performance* (Chomsky 1965: 3, cf. Aaronson and Rieber 1975, Morén 2007: 4). Both pairs of terms have been used in widely and somewhat counter-productively divergent ways (see G. Miller 1975, Trabant 2005). In particular, Chomsky emphasized the study of linguistic competence through intuitions about what speakers know rather than experiments measuring what they do, with a focus on finding the most restrictive model for defining the competence of a non-existent ideal speaker rather than emphasizing what actual speakers reveal that they have in common (Grace 1989, G. Miller 1975).

In this chapter we will be concerned with parsimonious modeling of what speakers systematically do – most often and in common – in order to use particular languages. This includes both actual speech and the psychological behaviors involved in producing and processing it. Not all of a person's subconscious brain activity in association with a language is necessarily part of its grammar, because cognition is multimodal, readily storing and activating large quantities of highly diverse mental associations (§5.6.3). The definition of grammar used here accounts for this by equating grammar with the competence needed for a particular kind of performance: using a language, not merely playing with it or knowing things about it. To clarify this, we now look briefly at a couple of activities – early childhood babbling and lexical innovation, especially loanword adaptation – that have sometimes been treated as windows into grammatical knowledge. I argue that these activities can change grammars but that

grammars do not execute them; therefore, there is no reason to think that the patterns in these activities are codified within grammars.

5.2.2 Development

Early childhood babbling (de Boysson-Bardies and Vihman 1991, Hayes 2004, Menn 2004, Samuels in press) has been treated as evidence for highly detailed, innate, universal components of grammar (McCarthy 2002: 208-213) because it reveals crosslinguistic trends that cannot be attributed to ambient adult languages – not simply because the babbler has not yet learned the ambient languages, but more importantly because the babble trends can violate them. Such trends do not have to be derived from innate structures in cognitive competence, however. Instead, the cognitive side of babble patterns can be viewed as simply the child's analytic generalizations about the production system he is exploring. Sounds and sound patterns favored by articulatory physiology and acoustic physics as simpler or easier to produce or perceive than others can naturally become more frequent in the course of generic, undirected babbling as the child exercises his or her production system, and it is readily conceivable for some of these patterns to become habits. In other words, innovative babble patterns can be responses to pressure from the production system in its early, more exploratory stages. As children master new articulations, their production-driven patterns will change and the constraints will change. Likewise, as a child becomes more skilled at imitating ambient adult speech, production patterns that violated the grammar of the ambient language will recede. At the same time, it has also been found that production practice begins influencing perception from an early age (DePaolis et al. 2011, Vihman et al. 2006, Vihman and Keren-Portnoy in press), notwithstanding the fact that perception develops much more rapidly than production ability (Blevins 2004: 219-236, Hayes 2004: 191-192, Menn 2010, Mohanan et al. 2009: 4).

189

In terms of Mielke's keyboard analogy (§4.4.2), erroneous patterns during early acquisition (of language or typing skills) that do not reflect the "correct" pattern (ambient adult speech or a text to be typed) simply reveal information about the structure of the production apparatus (the vocal tract or the keyboard). It is redundant to derive the patterns of the mistakes from innate content-specific constraints (such as innate phonological constraints motivating obstruent devoicing, or innate typological constraints motivating outputs like *teh* for *the, from* for *form*, and so on). While much remains to be done in early-acquisition research, at this juncture there seems no mandate to derive content-specific patterns in early babbling from a universal, innate set of content-specific constraints; much recent experimentation points rather toward emergence (e.g. Menn and Vihman in press, Velleman and Vihman 2006, Vihman et al. 2006: 364, Vihman et al. 2009).

If babble patterns can diverge from the grammar of the target language under the influence of the production apparatus, there is no reason to rule out similar influence in cases of borrowing or lexical innovation that reveal phonologically innovative patterns. In both cases, the speaker is exploring sound combinations that are not already conventionalized in the target language, and in the context of this creative freedom it is unsurprising if factors other than the grammar of the target language exert some influence. These could include performance-related factors including those earlier suggested as shaping common cross-linguistic feature co-occurrence trends (ch. 4). Another factor even more clearly in evidence though is the simple recognition of the difference between what is already true of the target language and what is not. Insights of this sort are not necessarily part of the grammar, if a grammar is a system for processing and producing grammatical utterances in a given language, because these processing and production activities do not themselves require the speaker to activate much knowledge about what the language is not.

For example, there is no evidence for [q] being present anywhere in English grammar for most speakers of English, but rather than viewing this fact as a component

of the English grammars of these speakers, we might want to consider it simply a generalization about what their English is *not*.⁶ Such generalizations are relevant in language contact situations, when English speakers have to decide how to borrow words containing [q] in the source language. But when the words are naturalized with [q]-less forms in English, then constraints mentioning [q] are no more necessary for the perception and production of grammatical English utterances than they were before. Underlying and surface forms in the source language (see Nevins 2008b, Nevins and Braun 2008). This holds true regardless of whether we attribute the avoidance of [q] in English borrowing to the fact that English already lacks [q] before borrowing, or whether we attribute it to phonetic rationales for the relative cross-linguistic rarity of uvulars. Both kinds of factors could even be involved simultaneously.

These considerations are similar to the rationale for distinguishing between synchronic grammar and diachronic trends: phonetic reasoning might explain the relative rarity of sound changes producing uvulars, and the uvular-free evolution of English might be taken as an example of this – yet none of these considerations properly belong in the formulation of synchronic English grammar, because they are unnecessary for modeling the processing and production of grammatical English utterances.

Loans and neologisms are often adapted to the native patterns of the borrowing language, and this can provide evidence for the phonology of the latter (Pierrehumbert 2001: 139). However, when borrowing and lexical innovation do result in innovative phonological patterns, they can be seen as *changing* the grammar of a language. The new grammar abides by the same definition as the old: it is the knowledge system for using the target language (cf. Nevins and Vaux 2008: 37). This is reason to keep the study of borrowing, lexical innovation, and babbling distinct from the concept of grammar. Further on the various influences that shape loanword phonology, see Haspelmath (2008b), Kang (2011), Martin and Peperkamp (2011), and Ushiskin (2011).

⁶ [q] does occur as an allophone of /k/ for some speakers of Multicultural London English (Nolan p.c.).

5.3 Optimality Theory and its classic schema

OT was introduced by Prince and Smolensky (2002[1993]) and elaborated in Kager (1999), McCarthy (2008a, 2002), and Smolensky et al. (2006) among others. As summarized by Smolensky et al. (2006:1: 495-502, 523-525), OT *defines* the set of possible languages as the permutation of the ranking of a universal set of constraints. These constraints do not restrict inputs; rather, the ranked constraints of a given language map any linguistically possible input to the most grammatically correct output, by either recommending or penalizing specified output properties per se (markedness constraints) or identifying respects in which outputs should match inputs (faithfulness constraints). OT also provides ways to *visualize* this system which may be psycholinguistically inaccurate with respect to some details in the interest of making others more comprehensible. After summarizing the simplified schema that has received the most attention in the works just cited, we will consider how more detailed or psycholinguistically realistic models of OT grammar acquisition and computation depart from it.

In this classic schema, the execution of the production function of the grammar has three main phases. First an underlying form is fed into a generative component (*Gen*) which emits every linguistically possible output in total disregard for language-particular criteria of grammaticality (freedom of analysis). Then these output candidates are fed into an evaluative component (*Eval*) which contains a ranked set of violable constraints. Each candidate is evaluated against each constraint, and constraint violations are recorded. Finally, the candidate with the lowest violation score is selected as the actual output (winner) while the other candidates (losers) are discarded. The principle that candidates are scored independently of one another against a single constraint ranking is called parallelism: the candidates can be evaluated simultaneously (in parallel), and the constraints bear simultaneously on each candidate. Evaluation only occurs after blind, profuse generation.

Leading OT computational work on acquisition (Alderete et al. 2005, Albright and Hayes 2011: 7, Hayes 2004: 194, Merchant and Tesar 2005, Tesar 2007a, 2009, Tesar and Prince 2003, Tesar et al. 2003) and mature production (Karttunen 2006, Riggle 2004) emphasizes the need for narrowing down the set of structures that are evaluated. This includes the set of underlying forms that are attributed to the lexicon in the course of the learning process and the set of output candidates evaluated during production. Such "smart" models (Kager 1999: 26) explicitly restrict generation based on prior evaluation. Logically, the smartest conceivable mature grammar would normally generate only winning outputs.

This makes it especially important to understand that the proliferation of losers in the OT schema is not actually a property that OT attributes to real grammars. The schema is only an approximate way of thinking about what happens when a given input form triggers nerve activity in the brain. Activation triggered by the input must spread among the relevant neural networks and stabilize in whatever form best collectively satisfies the preferences (weights) of the nerve connections. According to Smolensky et al (2006:1:40), grammar constraints are physically expressed as distributed patterns of these weights. The constraints, or weight patterns, evaluate the nerve activity triggered by the input and direct this activity until it settles into the correct output form. The activation patterns prior to that optimal equilibrium do not systematically encode any grammatical information: they do not realize features, morphemes, or words. Thus the correct output form is normally the only form that the grammar actually produces; the losing output candidate forms evaluated in the OT schema are figments of illustration.

A connectionist network computes an optimal representation by navigating through a space of subsymbolic representations [neural patterns] to ultimately construct the activation pattern that realizes an optimal symbolic structure [output form]. In no sense do such algorithms "generate alternative structures, evaluate them, and pick the best." In general, the only symbolic structure ever realized by such an algorithm

is the optimal one; at prior stages of computation, the algorithm constructs activation patterns that have no symbolic interpretation. (Smolensky et al. 2006:1:525)

If the losers never actually exist, they cannot actually be evaluated in parallel. It follows that where parallelism is allegedly crucial for the satisfactory analysis of some phonological patterns, the arguments at most validate only the simultaneous presence of certain kinds of structure in grammatical directions, not a need for simultaneous evaluation of multiple output candidate forms. This conclusion appears true in the cases I have investigated (§5.4). However, if proliferation of losing output candidates is unnecessary, then the large number of constraints which might be required to filter out the profusion of losers is also unnecessary. As a result, not only does a universal constraint set not need to be innate, but it does not need to be present in individual grammars at all. This motivates an approach to grammar design and acquisition in which the details of grammatical directions are efficiently extracted from observed surface forms during acquisition and are subsequently used for the efficient construction of surface forms during speech production (§5.5).

5.4 Necessary parallelism?

5.4.1 Imdlawn Tashlhiyt Berber

Core OT texts use several languages to illustrate their claim that strict parallelism is sometimes necessary. Prince and Smolensky (2002: 22) begin with syllabification in Imdlawn Tashlhiyt Berber (ITB) (Berber: Afroasiatic). They describe the mainstream pre-OT approach to this problem as:

one in which the structure is built (more-or-less) bottomup, step-by-single-step, with each step falling under the constraints appropriate to it. Taking this seriously... this would mean... that you first locate the nuclei – the heads of syllables; then project

higher order structure that includes the onsets; then project the structure that includes postnuclear consonantism. In ITB, however, as in many other languages, the availability of nuclei depends on the choice of onsets: an early step in the derivational constructive procedure, working on a low level in the structural hierarchy, depends on later steps that deal with the higher levels. Indeed, the higher level constraint is very much the more forceful...

In the theory advocated here... we expect exactly this kind of interaction. The whole output is open to inspection; how we choose to inspect it, or how we are forced by UG to inspect it, is not determined by the course that would be taken in bottom-up construction. The potential force of a constraint is not indexed to the level of structure that it pertains to, and under certain circumstances (UG permitting or demanding), constraint domination will be invoked to give higher-level constraints absolute priority over those relevant to the design of lower structural levels.

In fact, ITB does not call for such unrestrained interaction potential between different structural domains, but only for a "constructional algorithm" (ibid.) with a few slightly complex steps. As is generally true, the process can be modeled with a preference for each step to be as simple as possible, which may increase the number of steps, or with a preference for as few steps as possible, which will mean some steps involve more forces acting simultaneously. But either way, no step can be modeled in simpler terms than the data will allow. Pre-OT derivational phonology allowed a given rule to execute as complex an operation as necessary, and even in gradualist post-OT theories requiring "one" change at a time, the definition of "one" is determined by what the data allows (e.g. McCarthy 2010a).

Perhaps some parts of the construction process are necessarily more complex in ITB than in some other languages. ITB onsets and nuclei, for example, may need to be identified simultaneously, perhaps as a rising span of a certain size in a form's sonority curve (the graph one would make by plotting each segment's value in the sonority hierarchy: see Miller in press, Samuels 2009a: 119-121). This would indeed entail multiple forces working together, namely sonority and span size (limited by avoidance of complex margins). But it does not follow that the desired structure cannot be built

according to a theoretically satisfying plan. Even the fact that simultaneous onset-plusnucleus assignment requires scanning multiple segments at once during syllabification is not a problem; scanning multiple segments to fulfill a requirement is sometimes necessary during evaluation in OT as well.

The real item of interest in the ITB case may be that earlier proponents of the more-and-simpler steps approach to output construction had become used to building only one syllabic constituent per step, while specific properties of ITB may make this inelegant. But again this does not require abandoning directed construction for blind, profuse generation followed by evaluation of entire output candidate structures. That kind of parallelism does not essentially improve the analysis of ITB.⁷

5.4.2 Tongan

The next language Prince and Smolensky (2002) employ is Tongan, where they affirm that "it is important to have the entire structural analysis present at the time of evaluation" (p. 31). Their data can be summarized as follows: where stress falls on the first of two identical vocalic morae, the output joins these morae in one long vowel, while where stress falls on the second of two such morae, the output maps each mora to a separate short vowel with hiatus between: *húu* [hu:] vs. *huúfi* [hu.'u.fi]. The pattern can be analyzed in terms of three requirements: that the last (bimoraic) foot be right-aligned with the word edge, that foot boundaries not fall within syllables, and (least forceful of all) that identical consecutive vocalic morae not be separately syllabified.

⁷The syllable theory that Prince and Smolensky (2002) develop largely from ITB data also formally predicts a significantly greater permutational array of viable grammars than is empirically justified, and arranges them according to principles that do not appear to be operative, such as sets of grammars whose classes of segments banned from syllable margins are defined as increasingly large chunks of the sonority hierarchy starting from the top end. See Pater (2008: 15) on subsequent rejection of margin constraints, and Dell and Elmedlaoui (2002) for a better presentation of the ITB data.

Prince and Smolensky discuss "the view that structures are literally *built by re-write rules* in a strictly bottom-up fashion, constructing layer upon layer" (p. 30) and reject an analysis of this sort which generates intermediate forms such as [hu.u], concluding that while such a "bottom-up" approach would work if "the package of constraints relevant to syllable structure completely and universally dominates the constraint-package pertaining to foot structure" (p. 31), it does not work in Tongan. Yet all this only suggests that a perspicuous constructional algorithm for Tongan cannot assign stress after building all syllable structure. This may be interesting as a refutation of certain pre-OT assumptions about the order in which prosodic constituents must be constructed, but it does not show that the Tongan structures must be generated without access to any set of directions or requirements and then evaluated entire. One possibility is to assign morae first, then stress, then syllable boundaries. Having mora specification precede other syllabic structure is not a new idea (e.g. Hayes 1989: 260). Another possibility is that at least some moraic and syllable structure can be induced simultaneously from intersegmental gesture timing information (§4.1).

5.3.3 Lardil

According to Prince and Smolensky (2002: 137), Lardil has a pattern where "the crux of the matter is that the grammar must determine which *total* analysis is wellformed – a task impeded by the use of serial algorithms to build structure step-by-step." Basically, a /CVC/ input is deemed too short, so it is augmented. If the second input C can be a coda, then augmentation makes it a coda by adding -CV; otherwise, augmentation adds only -V. Augmentation is thus looking not merely at syllabified forms, but (according to Prince and Smolensky) at the effect its own application will have on syllabification, maximizing stem right-alignment with a syllable boundary. While the set of situations where this alignment fails to occur (see Prince and Smolensky 2002: 109, 113 n. 59) suggests a possible alternative, simpler motive for the pattern – maximization of stop onsets – Prince and Smolensky's analysis itself can be reformulated in constructional terms. Abstracting with those writers away from a complication involving the tapped /r/,⁸ we can simply say that if the stem-final C is a licit sonorant coda, then the augment is CV with C a regressively homorganic stop, and that the augment is V otherwise (i.e. when the stem-final C is a stop, and maybe when it is *y* or *w*, though examples are not given). Just inspecting this statement in light of the Lardil inventory reveals that the process favors stem/syllable right-alignment when licit codas are involved, and also that it avoids geminates and favors stop onsets.

Exactly how to formalize the process description, including the choice of which possible underlying forces to foreground as positive motivations for the process, is a separate issue. The point here is simply that a theoretically reasonable set of directions can be found for constructing the Lardil pattern without blind and profuse generation of losing output candidate forms.

Imdlawn Tashlhiyt Berber, Tongan, Lardil, and the handful of other prosodic problems that Prince and Smolensky (2002: 31-32) pass over very briefly in a single paragraph may indeed show that previous serial derivational theories drew the lines in the wrong places between different constructional steps, and Prince and Smolensky undertake a worthy task in critiquing this state of affairs. But proof that directed construction should be abandoned is lacking. The basic principle of a constructional approach is that in any given process, the input should supply enough information for the directions to work with. In the cases Prince and Smolensky develop, this information is indeed present in the specification of the string of input segments and their relevant features (in some cases perhaps including intersegmental timing information). Given this information, language-specific sets of directions can successfully guide the identification

⁸Their suggestion that some of the details they choose not to treat "may be relevant to a later level of phonology than we discuss" (p. 108) is curious for an approach that emphasizes direct mapping from input to output without internal hidden layers, as McMahon (2000: 25) notes.

or construction of everything else – relative sonority, moraicity, stress, syllable boundaries and foot structure – *for* generation rather than *after* it.

5.4.4 Southern Paiute, Yidin, and Axininca Campa

McCarthy (2002: 138-91) mentions a few other cases, but as far as I can tell, they fare the same as the previous ones. Southern Paiute has a reduplication process which according to McCarthy's data can be directed by the constructional algorithm, "Prefix a copy of root-initial Stop – Vowel – Nasal to roots beginning with that string; and let the reduplicant segment with the Nasal specification receive a place feature from the following root-initial stop. Otherwise, prefix CV." This reflects the fact that reduplicants have codas where this creates the one type of cluster which is licit in Southern Paiute, homorganic nasal plus stop, but not otherwise. The relationship between static phonotactic constraints and a constructional algorithm is an important issue, but here our only concern is with McCarthy's claim that the Southern Paiute pattern presents an ordering paradox for rule-based approaches and needs to be treated with a model that involves the generation of multiple incorrect candidates evaluated in parallel with the right one. This claim seems incorrect in light of the constructional algorithm just described.

Another case is Yidin, which in light of McCarthy's data can be governed by the algorithm, "Dissect a word into two-syllable feet starting from the left edge. Next, if any foot has a long vowel in the second syllable, make all feet in that word iambic; otherwise, make all feet in a word trochaic." Or in McCarthy's own terms, "If an even-numbered syllable contains a long vowel... then that syllable and all other even-numbered syllables are stressed. Otherwise, odd-numbered syllables are stressed" (2002: 150). Again, McCarthy claims that this pattern merits an OT treatment, with generation of multiple incorrect candidates evaluated in parallel with the right one, but it is not apparent why. It

is coherent and feasible first to define syllables, then assign feet, then scan for underlying length – which secures all the structure mentioned in the *if* clauses in McCarthy's generalization about the pattern – and finally to assign stress based on length in relation to feet, which secures the structure mentioned in his *then* clauses.

Axininca Campa (McCarthy and Prince 1993, cf. Prince and Smolensky 2002: 19) has a reduplication requirement that the base fill the bimoraic minimum length requirement for the prosodic word; bases which do not do this are augmented with (C)V as necessary. The base is copied in the reduplicant, along with this (C)V augment. McCarthy and Prince (1993: 79, cf. 88-89) claim an ordering paradox for serial constructive approaches, reasoning that the triggering environment for augmentation includes the reduplicant, which must therefore predate augmentation, but the reduplicant copies the base including the augment, so the augment must predate reduplication. What triggers augmentation is not necessarily the presence of the reduplicant, however, but can instead be simply the presence of the reduplication morpheme (see Samuels 2009a: 182). The appropriate constructive algorithm will direct that if the reduplication morpheme is present, then the base should be checked for bimoraicity; if the base is too small, it should be augmented to the bimoraic length. The reduplicant is then built (or phonologically realized) by copying the whole base, augment included. Whether or not this algorithm is more elegant than what traditional generative rule-writers would have devised for Axininca Campa is irrelevant for our purposes; what matters is that it is a successful constructional algorithm.

McCarthy and Prince (1998: 119) review some of the above examples as well as a few others, remarking, "Crucial evidence distinguishing serialist from parallelist conceptions is not easy to come by; it is therefore of great interest that reduplication phonology interactions supply a rich body of evidence in favor of parallelism." In surveying their analyses of these cases, however, I reach essentially the same conclusion as in the discussion above. The reduplicative processes have somewhat complex constructional algorithms which, when triggered by the presence of the at least partially specified base concatenated with the appropriate reduplicative morpheme, call for copying of the base in the reduplicant *plus* certain additional developments which could be realized with either more serially ordered rules or rules with more complex directions. While some early rule-based theories may not have explicitly accounted for such situations, the notion that constructional algorithms cannot handle these patterns in a clear, efficient, and restrictive fashion seems mistaken.

5.4.5 Southeastern Tepehuan

Finally, Southeastern Tepehuan (Kager 1999: 177-188) syncopates the vowel in even-numbered syllables that are open, a process that could be attributed to the syncopated vowels being unstressed while the remaining vowels have secondary stress – apart from the main-stress vowel, of course. Main stress falls on the first syllable if the second underlying syllable is light, otherwise on the second syllable; from an output perspective, this means main stress falls on the second mora after syncope.

Kager objects to the idea of attributing syncope to stress patterns by asserting that there is no phonetic evidence for secondary stress, and offers instead a proposal that consonant-faithfulness constraints crucially conflict with prosodic constraints prescribing that every word has only one foot (sc. only one, primary, stress, and no evidence for secondary stress) and that as few syllables as possible should be footless. Hence, syllables are minimized by deleting vowels, but because deleting all unstressed vowels would result in unacceptable consonant clusters, consonant-faithfulness constraints effect the preservation of roughly every other vowel to prevent this. This interaction between segmental and prosodic constraints is portrayed by Kager as thoroughly intersecting multiple levels of structure and providing strong support for the classic OT analysis where (to repeat Prince and Smolensky 2002: 22) "the entire structure" is "present at the time of evaluation". But as McCarthy (2008b) points out, the proposed forces motivating this kind of pattern are equally well satisfied by entirely different choices of vowel deletion, so what dictates which vowels are deleted? I assume one must add some stipulation. In a constructional approach, this could be a requirement that after the stressed syllable, odd-numbered vowels are favored for syncope while even-numbered ones are favored for preservation (to ensure, for example, that /tiitirovin/ yields [tiitropin] and not *[tiitirpin], a candidate Kager does not consider). Without connection to a cross-linguistic principle of alternating secondary stress, such a stipulation does not seem particularly explanatory. It is also serial since it refers to stress, which is not underlying (see McCarthy 2007: 66).

The requirement could be rewritten without reference to stress, but the result would be even more ad hoc. In OT, markedness constraints cannot refer to the input, and since stress is not underlying in Southeastern Tepehuan, faithfulness constraints cannot refer to stress. So we would have to choose among faithfulness constraints recommending the preservation of certain vowels, and markedness constraints recommending medial two-consonant clusters after certain vowels. Neither solution bears much resemblance to Kager's intentions, and neither seems very explanatory. It is also somewhat difficult to define "certain vowels". A faithfulness-dominant solution could recommend a faithful first-syllable vowel, faithful second-syllable vowel if the second syllable is underlyingly heavy and the first syllable is underlyingly short, and faithful even-numbered vowels thereafter (along with a markedness constraint forbidding consonant clusters above a certain size, apparently two). A markedness solution recommending as many as possible medial two-consonant clusters starting from the left edge of the word (without consonant epenthesis, [cons] changes, or input-output linearity violations) would work with words where the underlying second syllable had a short vowel, but not with words where that vowel was long – and markedness constraints cannot refer to underlying structure.

Southeastern Tepehuan syncope may indeed be "deeply *parallel*" (Kager 1999: 179) in a number of senses, but it is not parallel in the sense that evaluation must work

with profusely generated complete output candidates, with no reference to the input other than to check faithfulness. A constructional approach actually seems not only viable but superior. As a parting note on this language, Kager's claim that there is no evidence for secondary stress may be circular. Suppose that the underlying vowels synchronically selected for syncope originally received the weakest degree of stress (i.e. none) and that all other vowels (at least medially) had secondary stress: syncope would then remove the bulk of the evidence for the difference between unstressed and secondarily stressed vowels, and the originally secondary stress level would easily be reanalyzed as a new unstressed level.

5.4.6 Review

The examples in all the languages we have reviewed involve the interaction between different domains of structure (syllables, morae, stress, epenthetic augments) not present in the input. Prosody is the realm *par excellence* where multiple domains interact, and it is no accident that OT's main claims to improved analysis of particular data mostly involve prosodic problems, while the best-known phenomena that seem to require serial derivation involve non-surface-true generalizations about interactions *within* the same domain, typically segmental phonology (cf. McMahon 2003, 2000: 24). However, even the multi-domain problems can be solved constructionally and do not require a blind, profuse Gen discrete from Eval.

Parallelism in a broader sense – the consideration of multiple factors or options before reaching a decision – remains an attractive concept, correlated with the notion of avoiding the inelegant breakdown of process descriptions into any unnecessary ordered steps. But of course that notion predates OT. While the imposition of extrinsic ordering on all generative events, even those that do not require it, is a hallmark of both *SPE*-style generative phonology (see McCarthy 2007) and gradualist post-OT theories (e.g. McCarthy 2007, 2009b, 2010a), at least in the former tradition there was considerable freedom to replace certain groups of ordered rules with single, more complex rules that could look for multiple properties at once and react accordingly. The Elsewhere Condition is an example of such a rule structure (see McCarthy 2002: 139, 149).

5.5 The architecture of phonology

5.5.1 Grammars differ in content, not just its permutation

The question – How much grammar can be learned rather than innate? – is championed among others by Chomsky under the rubric of minimalism (Samuels 2009a: 1-3). Much minimalist work has been done in syntax, but interest in the same question has been gaining ground in phonology (Becker and Potts 2011 §5). We saw this in the last two chapters in discussions of several recent works on the emergence of phonological features and feature co-occurrence patterns from non-cognitive factors. Several arguments were reviewed (especially in §3.3.3) for the learnability of features and, by extension, feature co-occurrence patterns. We now pursue formal implications of these ideas for grammar design theory. Except as noted, the discussion will remain neutral on whether features are learned or innate, focusing instead on how feature cooccurrence patterns could be learned in a language-specific fashion as the building blocks of phonologies.

It would be superfluous for learners to acquire grammatical directions that have no effect or activity in their language, so the most natural expectation is that grammars differ in their contents, not just in how the contents are organized (Hale and Reiss 2000a: 164). The details of cross-linguistic patterns that emerge during early childhood babbling which are not part of the ambient adult language, or in loanword adaptation without having been previously present in the borrowing language, can be attributed to phonetic factors without being treated as structures that all learners encode into their grammars (cf. §5.2.2). Thus, while OT is agnostic about whether its universal constraint set is learned as "a consequence of universal functional pressures like the architecture of the human vocal tract and perceptual system, or the task demands of communication" rather than being genetic (Smolensky et al. 2006:1:42-43), the vastly more parsimonious hypothesis is that these functional pressures only create grammatical structure on a language-specific basis – when a learner adopts a particular pattern not present in the observed usage of mature speakers, or when borrowers adopt a particular pattern not previously present in the borrowing language. In other words, cross-linguistic trends in babbling and loanword adaptation are evidence of linguistic knowledge which is only incorporated into actual grammars to the extent that it changes them. It is redundant to load grammars with a universal constraint set either innate or derived from these forces, because such a resource is unnecessary for either speaking or understanding any language. Additional arguments against the idea that the same constraint set is present in every language are presented by Ellison (2000).

The notion that grammars can differ in content, not just ranking, actually accords well with the claim of Smolensky et al. (2006:1:40) that grammatical constraints are part of the abstract (symbolic) information level, implemented at the neural level in a distributed fashion. Given that Smolensky et al. (2006) envision both grammatical forms and grammatical constraints as distributed information patterns within neural networks, given that grammatical forms are clearly learned, and given that neural connection weights can also be learned (ibid. 1:8), a reasonable hypothesis is that content-specific grammatical constraints are learned too, rather than genetically hardwired into the nervous system – and that they are incorporated into the grammar as needed, not in superfluous profusion.

5.5.2 Users optimize grammars, not just words (and not typologies)

Riggle (2004) discusses a grammar implementation model which during the learning phase establishes possible input-output paths, gives each segment of each path a constraint-violation score, and then identifies the optimal path so that subsequent mature grammar execution can simply follow that. The goal is to dispense with a certain amount of "online" optimization: once certain successful generalizations have been found, they do not need to be rediscovered during every applicable speech act. In this respect, Riggle's model aims at being smart (§5.3). It is also admittedly highly simplified since it is restricted to building outputs one segment at a time from left to right. Of interest though is that parts of Riggle's model aim at restricting online generation and optimization by having the learner lock in successful generative sub-paths and not have to recreate them for every applicable speech act.

Finally, with the entire grammar represented as a single preoptimized evaluation function it is possible to detect and generalize recurring patterns across the optimization of various input strings and thereby construct... a transducer that directly generates optimal output forms from underlying forms. In effect, this move obviates the need for word-by-word optimization since all of the optimization is done in the process of the creation of the transducer... to create a machine that generates all and only the optimal candidates... Specifically, I'll present here an algorithm for constructing transducers that map input forms directly to the output forms that would be selected as optimal under a given ranking of a set of constraints. With such transducers it will be possible to generate optimal output forms without the need to do optimization on each input/output pair. By encoding optimal parses directly into a finite state transducer it will thus be possible to circumvent the need for on-line optimization. (Riggle 2004: 16, 20, 43, 138-39; emphasis added)

In the final pages, Riggle addresses the danger that obviating optimization during grammar execution might be viewed as anti-OT:

Another aspect of the model developed here that should receive further scrutiny is the representation of OT grammars... as transducers. There is some tension between ranked sets of universal constraints on one hand and finite state machines on the other. The former is far more elegant in terms of describing the range of possible languages and the relatedness of languages to one another while the latter is far more transparent in terms of describing the input/output mapping defined by a particular grammar. One might ask if one of these representations of phonological grammars was more "right" or more illuminating than the other.

Considering my algorithm for turning ranked sets of constraints into finite state transducers that map inputs directly to optimal outputs, it might seem as if I've taken the OT out of Optimality Theory. That is, one might think that the low-level description of the computation of optimality is somehow closer to the "true" phonology and should therefore obviate the need for constraints and rankings. Such a conclusion would be mistaken, first and foremost, because ranked constraints provide elegant descriptions of the relations that are defined by phonological grammars and of the relationships among various grammars (languages), while the algorithms that I present here represent merely one method for computing these relations. (2004: 173-74)

Distinct topics seem conflated here, amounting to a covert switch of subject from the nature of grammars to ways of comparing grammars. If a grammar is the cognitive system speakers use for recognizing and producing speech in a particular language, why would any grammar need to be "elegant in terms of describing the range of possible languages and the relatedness of languages to one another", let alone sacrificing clarity in the description of that particular language for the sake of these cross-linguistic revelations? We are given no reason to suppose that foregrounding cross-linguistic insights is a task of individual grammars. Granting that "ranked constraints provide elegant descriptions... of the relationships among various grammars," such descriptions cannot be equated *with* grammars, and no argument is given for requiring them to be components *of* a particular grammar.

Another claim in the quotation is that "ranked constraints provide elegant descriptions of the relations that are defined by phonological grammars". This claim,

assuming it is distinct from the one just analyzed about relationships among grammars, must refer to relations within an individual grammar. If so, it seems to conflate the difference between actual relations and possible ones and to make essentially the same mistake as the first claim. The actual relations that a grammar has to capture to get from inputs to outputs – which are the only relations we can prove that it does capture – are precisely the relations that Riggle's algorithms attempt to isolate by reducing online optimization "to create a machine that generates all and only the optimal candidates" (2004: 43). What other relations exist within a grammar, except for latent possible rearrangements that are entirely counterfactual? It is not a grammar's job to indicate what other grammars of the same language would have looked like, or what grammars of other languages do look like.

Riggle's comments emphasize that reducing online optimization as an activity of grammars is not necessarily better than keeping it, but the justifications given for the latter seem to switch topic from grammar to cross-linguistic typology. What if one's focus stays fixed on what grammars are – or better yet, what if one focuses on the larger picture of not only what grammars do, but also how they are acquired? The issue of locking in successfully learned patterns is then seen as a simple one of retention. In what sense is it optimal to learn the same item over again every time the item is needed? In most other contexts, successfully learning something normally implies retaining it for a significant time so that it does not have to be relearned every time it is needed. If this notion is carried over into grammar design, the optimal grammar is one where as much optimization as possible has already been carried out by the time the speaker reaches maturity.

A theory which incorporates this concern into its core will plainly differ from one which treats the search for smart grammars as a merely implementational issue of no interest for defining the essence of the theory. Reinventing the wheel is either a theoretically fundamental part of every act of driving, or it is not. But the two models of what travel entails are radically different. Formalisms which facilitate permutational typological exercises are interesting, but there is no reason to consider such activities part of the grammars of individual languages.

5.5.3 Grammars don't handle the rich base

The preceding arguments that the detailed contents of grammars are efficiently learned and that grammars consequently differ in content, not just ranking, clash with a core OT doctrine: Richness of the Base (RotB). According to this hypothesis, grammars are so designed that if all linguistically possible inputs (sequences of all possible feature combinations, Smolensky 1996: 4-5) were fed through a grammar, each input would map to a grammatical output for that particular grammar (Tesar and Smolensky 2000: 75, cf. McCarthy 2008a, Kager 1999). Smolensky et al. (2006:2:233) present the argument for RotB this way (emphasis original):

OT is a highly output-oriented grammatical theory. The strongest hypothesis is that *all* systematic, language-particular patterns are the result of output constraints – that there is no other locus from which such patterns can derive. In particular, the *input* is not such a locus. Thus, for example, the fact that English words never begin with the velar nasal η cannot derive from a restriction on the English lexicon barring η -initial morphemes. Rather, it must be the case that the English grammar *forces* all its outputs to obey the prohibition on initial η . This requirement amounts to a counterfactual: *even if there were* an η -initial lexical entry in English, providing an η -initial *input*, the corresponding *output* of the English grammar would *not* be η -initial. Thus, the absence of η -initial words in English must be explained within OT by a grammar – a ranking of constraints – with the property that *no matter what the input*, the *output* of the grammar will not be η -initial.

The fact that language learners are most commonly exposed to correct outputs and not to incorrect outputs which are clearly identifiable as incorrect makes it inherently difficult for learners to construct grammars capable of identifying all possible losing output candidates in conformity with the Richness of the Base requirement, a problem which continues to challenge both the mechanics and the psycholinguistic realism of OT learning theories in spite of attempts to resolve the difficulty by biasing learning algorithms *against* simple (faithful) input-output mappings (see Albright and Hayes 2011: 7-8). That bias becomes even more problematic given various studies showing that learners initially tend to prefer faithful mappings and that they learn morphophonological alternants later *even* when the marked alternants reflect phonetically well-grounded constraints (see Albright and Hayes 2011: 13).

The quotation above confirms that requiring grammars to handle the Rich Base of all cross-linguistically possible inputs is a consequence of the premise that grammars differ only in constraint ranking plus the premise that only output properties can be simply recommended or penalized (markedness constraints: §5.3). I have already argued against the first premise (§5.5.1). The next section develops an argument against the second premise, leading ultimately to the conclusion that individual languages do not handle the Rich Base (§5.7).

5.6 Generalized partial representations

5.6.1 Introduction

The idea that content-specific grammatical directions (such as rules or constraints) can be learned by generalizing about recurring content patterns in observed data has been explored by Albright and Hayes (2006: 4-5), with a somewhat similar line of research in Boersma and Pater (2007). Albright and Hayes develop a procedure for deriving rules from paradigm forms. The forms are organized and subjected to iterative comparison in a way "that lines up the segments that are *most similar* to each other"; material shared by multiple forms "is collapsed using the feature system", while material that does not match across multiple forms "is designated as optional" and represented minimally. The

point of interest is that the eventual rule contains feature specifications and segment sequencing information that have been taken directly from encountered words by generalization. Whether the generalization mechanism is formulated to produce rules or constraints, the key point is that it operates largely by finding and extracting elements common to the representations of multiple words that are observed to participate in the same phonological process.

I propose modeling most grammatical directions not as a special type of operation or object (rules or constraints) distinct from the phonological representations of words, but rather as generalized partial representations extracted from patterns observed in surface forms during acquisition. In speech production, grammatical representations are combined with idiosyncratic information about a particular string of morphemes (word parts stored in an abstract underlying form in the speaker's mental lexicon) to create a more complex representation of the desired word. This complex representation contains both the underlying forms of the input morphemes and the grammatical information which other theories encode in phonological rules or constraints. After constructing this representation, the speaker applies general principles to read from it the correct phonological output – the set of categorical specifications to be implemented as gradient real-time gestures in the actual utterance of the word.

By default, lexical inputs to the phonology activate the representations of relevant grammatical patterns on the basis of content match – either identity between two elements or, possibly, a category-to-instance relation between them identified during acquisition on the basis of shared phonetic content (issues with formalizing such information and how it would relate to underspecification are broached in §5.6.3). Restricted sets of lexical items that exceptionally do not activate grammatical patterns on the basis of content match would be inhibited with respect to those patterns. Grammatical patterns can also activate or inhibit one another, but because grammatical patterns form a closed set, these interactions can be modeled as part of the grammar rather than

automatically triggered by content relations. Several possible types of activating and inhibitory relationships between representations are illustrated below.

Once a lexical input and all relevant grammatical representations are in conjunction, reading the output from this total phonological representation involves evaluating all the information within each timing slot and must include a protocol for adjudicating content conflicts. Such conflicts arise when some combination of the lexical input and the grammatical patterns within the total representation make clashing recommendations about a particular kind of element (e.g. about the value of a feature) in a given timing slot. While rule-based theories resolve this sort of clash largely through rule ordering and OT handles it through constraint ranking, the representational approach to grammar suggested here uses static licensing relationships. A single grammatical representation (definition of a grammatical pattern) will typically consist of two domains of phonological elements, both sensitive to intersegmental timing information. One domain licenses the other; from the viewpoint of speech production we can call them parent and daughter. Daughters consist of information not already available in underlying forms (the equivalent of information encoded in markedness constraints in OT and generated by rules in rule-based theories). Outputs are normally read by combining the lexical input with the daughters; when a daughter element clashes in a given timing slot with a lexical input element or with an element of another grammatical representation, the winning element is the one activated through the highest number of licensing associations, counted both directly and transitively. The daughters of licensing associations that are specially marked as inhibitory are excluded from this calculation, along with any material that cannot be activated without being licensed by them.

These ideas have important antecedents. The concept of licensing figures in Itô, Mester, and Padgett (1995) and Steriade (1995), but the details are generally very different; in particular, grammatical patterns do not license each other. Psychological node theories (e.g. MacKay 1987) and connectionist network theories (Garson 2010) also employ licensing relationships of sorts, as connections between nodes or nerves, with thresholds for activation across those connections. Some of these concepts will be discussed below as they relate to the way the present model would be neurally implemented.

The following sections apply this model to several well-known types of pattern interaction. Conventional terms for these patterns are incorporated into the discussion. These terms tend to imply ordered processes, but it should be kept in mind that the pattern representations being proposed here are not ordered processes.

5.6.2 Illustrations: epenthesis and spirantization

To begin, suppose a language learner progresses to the point of sorting out stems from certain suffixes on the basis of their combinatorial behaviors. The suffixes begin with vowels, and the stems end in two-consonant clusters. The learner notices that when the same stems occur with no visible suffix, ε normally intervenes between the two consonants. This is a generalization about parts of the representations of multiple words and suffixes. The learner extracts this insight from the relevant observed forms and stores it as a generalized partial representation defining this pattern. Specifically, the generalization is that the presence of ε between two consonants at the end of a word and the absence of ε elsewhere is systematic (grammatical) while the identities of the two consonants are unpredictable but consistent properties of particular morphemes which must be stored in those morphemes' lexical entries. The presence of two consonants without an intervening unpredictable vowel at the end of a word licenses an intervening ε , as schematized in (3). Association lines connect parent (licensing) to daughter (licensed) elements. (3) Generalized partial representation of ε -epenthesis



Tiberian Hebrew (Semitic) has a productive pattern of this form, illustrated in (4) with an example from Green (2004: 43). For more data see Bruening (2000), Coetzee (1999), Green (2004), and Idsardi (1997a,b, 1998, 2000). I will abstract away from a few more narrowly conditioned cases where the epenthetic vowel is other than ε ; for details, see Coetzee (1999).⁹

(4) Epenthesis in TH /qod $\int \rightarrow$ [qoð ε f] 'sacredness'



The unsuffixed input /qodʃ/ has a word-final consonant cluster (df #) which matches the parent material in (3) (CC#) as shown by the vertical association lines. This match activates the grammatical representation of epenthesis. From the conjunction of the input with this relevant grammatical pattern, the phonological output is read as the totality of active material, which here includes everything since no element is inhibited. The output

 $^{^{9}}$ A very few nouns, a number of jussive/imperfective verb forms, and verbs with a 3sg fem. -*t* do not epenthesize. Here I follow what seems to be the more widely accepted analysis that this non-epenthesis is exceptional. Green (2004, cf. Bruening 2000) alternatively treats it as the default and requires all epenthesizing forms to be lexically marked as trochaic, which introduces a new (metrical) category of lexical diacritic. Under that analysis, the rest of my discussion of Tiberian Hebrew would simply require the additional metrical diacritic in the UR of epenthesizing forms. This would not affect my arguments below.

is shown at the bottom of (4) in parentheses to emphasize that it is already contained within the conjunction of the input with the grammar. The output actually licensed by (4) is an incorrect *[qode \int] rather than [qoðe \int] because we have not yet accounted for the continuance of the second consonant.

The latter property is due to how epenthesis interacts with another Tiberian Hebrew grammatical pattern wherein non-emphatic postvocalic singleton obstruents are spirants (see Rendsburg 1997: 74-75, as well as the sources on epenthesis above). The phonetic nature of the excluded emphatic consonants is uncertain, but the most common suggestions are glottalization as in Ethiopic Semitic and dorsopharyngealization as in Arabic (Thelwall and Sa'adeddin 1999, Rendsburg 1975: 73-76). The exclusion of emphatics from this pattern must be specified in the grammar; we will use [-RTR] provisionally here (see Idsardi 1998: 50). The fact that vowels are predictably continuant is represented here as V([+cont]) (§5.6.3).

(5) TH postvocalic spirantization



When the learner has noticed that a more narrowly conditioned grammatical pattern can be captured by combining the representations of independently motivated and less narrowly conditioned grammatical patterns (cf. Kiparsky 1982: 233-234, Vaux 2008: 9), nothing prevents the learner from establishing this connection in the grammar. Thus in /qodʃ/ \rightarrow [qoðɛʃ], instead of positing the parochial grammatical direction represented in (6), the learner may combine the independently needed representations of epenthesis (3) and spirantization (5) as shown in (7). (6-7) Representations of TH epenthesis and postvocalic spirantization in /qod $\mathfrak{f} \rightarrow$ [qoð $\mathfrak{e}\mathfrak{f}$] 'holiness'

(6) Parochial



(7) Generalized



The parochial grammatical direction in (6) and the generalized grammar in (7) differ purely in their internal associations. In (6) there is no analysis, simply a stipulation that the string V([+cont]) – C[-RTR] – C # (using dashes to separate timing slots) licenses [+cont] in the third timing slot in the string and an epenthetic ε inserted after this slot. In contrast, the same information is analyzed in (7) as a conjunction of the
representations of (3) epenthesis and (5) spirantization, shown by labeling the appropriate association-line clusters as ⁽³⁾ and ⁽⁵⁾.

5.6.3 Underspecification and multi-modal cognition

Having the lexical input /qodʃ/ activate the epenthesis and spirantization patterns as shown above requires a type of knowledge that we have not yet discussed: the user must be able to identify consonants and vowels.

Brain imaging studies indicate that the vowel-consonant distinction is psychologically significant in auditory processing and speech planning (Singer 2009, Woods et al. 2011), and the distinction is phonetically evident in broad activity patterns of several laryngeal muscles (§2.2, second to last paragraph). It is also phonologically evident in patterns like epenthesis, which seems to have a higher-than-chance frequency of inserting consonants between vowels or vice versa. Consonant or vowel epenthesis often relates to existing syllable structure (Samuels 2009a: 122) or to the perceptibility of segments already present; e.g. some consonants lose their most salient cues unless a vowel follows (Henke et al. in press, Wright 2004).¹⁰ Thus the consonant-vowel distinction seems psycholinguistically real, phonologically active in a wide array of languages, and phonetically natural. Yet how is knowledge about this distinction configured in relation to lexical underlying forms? Is every consonant token somehow

¹⁰ This summary of the natural phonetic motives for consonant and vowel epenthesis is supported by the fact that its explanatory coverage seems to have principled boundaries: common types of epenthesis not covered by this explanation appear plausibly derivable instead from natural phonetic variations in how intersegmental transitions are coordinated or perceived. These variations include failure to synchronize wholly or partly independent gestures (Ohala 2005: 10-11), decelerated transition between a stop and a more open segment (see Chomsky and Halle 1968: 319) or vice versa (Green 1997: 56), and historical development of stop release noise (Hyman 2003: 19-24) and concomitant higher-pressure airflow (Ohala 1983: 204-205) into a fricative in the constricted environment of a following high vowel (cf. Hall and Hamann 2003).

marked as such in the lexicon, or are lexical consonant tokens unspecified for the property of being a consonant, with this information being supplied by the grammar?

Notice how this question is about a different kind of structure from (3). There and in all the other representations of grammatical patterns explored below, the parents and daughters are strings distributed across multiple segmental timing slots, and everything in the parent string must be present in order to license the daughter string. The question of how users know that /d/ is a consonant, however, is a question about a list and does not involve multiple timing slots: there is simply a set of elements, every one of which counts as a consonant in and of itself. The question we just asked about this set is, in effect, whether the members are individually identified as consonants in the lexicon, or whether instead the list is stored in the grammar with its members as the parent material and its name ("consonant") as the daughter – so that a consonant in a lexical input is only formally recognized as a consonant when it has activated the grammatical list for the consonant category.

Either option is feasible. The second one conserves memory by preventing the lexicon from duplicating in multiple places a piece of information (which segments are consonants) that can easily be stored in one place. However, the brain does not seem to value that kind of conservation as much as we might expect. Instead of storing different pieces of information in separate places, the brain has an incredible propensity for translating them into different activation patterns distributed across many nerves and regulated by learned weights on the nerve connections (§5.3). Instead of vetting incoming stimuli and refusing to save information deemed already presented in memory, the brain stores information about repeated or similar experiences in a fashion that allows the mental representation of tokens to influence more general ideas and expectations about types (a focus of exemplar theories: Dogil forthcoming, Pierrehumbert 2001, Walsh et al. forthcoming). Tokens of vastly different categories also readily activate one another.

Each memory is not a completely separate item, like a photograph in a drawer or a file in your computer would be... [P]eople do better on an exam when they are tested in the same room that they had lessons in than they do when they are given the same exam in a different room, so they must have learned information about the room along with the lessons... Clearly, our brains are not particularly selective. All the things that are going on at a given instant are linked together (and as we get older, we seem to hold onto more and more of this incidental information...)

When two sets of stimuli occur together many times... the connections among them get strengthened... Knowing about the larger structures of the brain, including the language sensitive areas, is critical for understanding the effects of brain damage, but we have to work at the level of the neurons to understand how we learn and process language... Organizing [linguistic] processes into levels and sublevels does not mean that each level is isolated from the others [discretely modular and serially active]... the assembly line metaphor is not a good approximation to what goes on in the brain. (Menn 2010: 81-84, 95, 110; cf. 61, 113-115, 162, 258, 331, 390)

These insights are based on a vast body of experimental findings, some of which involve speech errors. For instance, word blends and substitutions have shown that phonological content can prime lexical retrieval:

D. G. MacKay (1973b) found that words involved in blends were phonologically as well as semantically similar with greater than chance probability, and Dell (1980) showed that phonological and semantic similarity independently and reliably influence these errors. Dell (1980) also reported a parallel phenomenon for word substitutions... These findings indicate that lower level processes (the phonological representation) can influence higher level processes (the selection and misselection of which word gets produced), and such phenomena are problematic for theories postulating separate perception and production components, with strictly topdown processes for production... (MacKay 1987: 34)

Syllable frequency also influences phonetics, even though this contradicts a strictly modular view of phonology as a computer that manipulates morphosyntactic input specified with only lexical phonological information and that emits outputs for phonetic implementation without these different domains being able to influence one another (see Pierrehumbert 2001: 137-138, Walsh et al. forthcoming, 13; cf. Bybee 2002:

59-60, Sosa and Bybee 2009: 483 on phonetic sound changes sensitive to word frequency).

All this raises questions about the appropriateness of models that emphasize underpecification as key to understanding grammatical processes. Storing predictable phonological specifications in a separate place from lexical information and laying great importance on when that information enters into the phonological construction of word forms are ideas that strongly evoke the limited power of the computers which we are currently able to build; these ideas relate much less clearly to the highly distributed nature and liberality of association and activation which characterize the brain. This is certainly not to say that there are no characteristic orders in which different grammatical aspects of an utterance are constructed (semantic, syntactic, morphological, etc.), but it does impel us to scrutinize the phonological evidence for underspecification carefully.

In Steriade's (1995) broad review of rule-based underspecification theories, the cases where predictable information allegedly needs to be specified after the application of some rules center around the problem of modeling phonological processes as the local manipulation of association lines between autosegmental tiers without letting the association lines of elements that participate in those processes cross over the association lines of non-participants. Delaying specification of some non-participating elements until after a process is one solution to the problem. We have already noted independent arguments against this use of underspecification, though (§3.3.8). The only other arguments I have encountered for not specifying predictable material until after a certain stage of derivation have to do with lexical economy, and we have just seen that the psycholinguistic realism of such reasoning is doubtful.

Consequently, I adopt the working hypothesis that if we refrain from investigating when (or at what points in the associations among neural activity patterns) different parts of the grammar access intrasegmentally predictable information, the essential structure of grammatical patterns and their interrelationships can still be analyzed insightfully. This chapter accordingly remains agnostic about whether languages underspecify any phonological category simply because it is predictable (see Samuels 2009a: 77-98, also Goldsmith and Noske 2005, Mohanan 1991, Odden 1992, Steriade 1995, Vaux 2010).

Theories of underspecification do impact another important topic: theories of phonological operations. For example, post-vocalic spirantization can be represented in process-oriented formats as continuance assimilation from vowels to stops, unless [+cont] is deliberately excluded from the vowel specification on the grounds that continuance is predictable in vowels (so Idsardi 1998). If any instance of contextually predictable information is phonologically unspecified, then phonetically natural phonological processes dependent on that information cannot be represented in a phonetically natural way unless their application is deferred until after the unspecified material has been filled in. In personal communication, Idsardi suggested that the specifiability of contextually predictable information relates inversely to how crosslinguistically common its predictability is, so that vowels can never be specified as [+cont] because by definition they are always continuant in every language. However, information about cross-linguistic feature co-occurrence trends is normally not available to learners constructing grammars as posited by Emergent Feature Theory. This leaves learners with two ways to handle at least some phonological patterns that depend on predictable information: induce contextually predictable feature specifications based on patterns that can be deduced from the final phonological shapes of words, or construct phonetically unnatural phonological representations of the same patterns in order to have them apply to underspecified forms.

Both options are indicated noncommittally in the representation of postvocalic spirantization above (5): the first element in the parent string is spelled V([+cont]), indicating a specification of the vowel category which may or may not include [+cont]. If [+cont] is included, then the phonological representation contains contextually predictable information but also reveals a phonetically natural analysis of postvocalic spirantization as the result of an assimilation operation between adjacent segments. If [+cont] is not included, then the representation is more economical but also more

arbitrary; the daughter element seems to be inserted *ex nihilo*. In that case one might ask why vowels license [+cont] on a following stop rather than licensing, say, simultaneous palatalization and rounding. However, in an emergent framework, the answer to this question is provided by the theories of sound change and acquisition, not by the theory of representations of phonological patterns, so the phonetic unnaturalness of representations involving formal *ex nihilo* insertion is not a reason to reject such representations.

In other words, it is already clear from a variety of other cases that phonetically unnatural grammatical patterns occur in languages thanks to sound changes cumulatively obscuring their predecessors' conditions (§3.3.1), so grammars have to be capable of representing phonetically unnatural phonological patterns. This suggests that phonetically unnatural representations could be constructed for some phonetically natural patterns that depend on predictable information, if learners prefer not to specify such predictable information in phonological representations. If so, there would be no clear objection to a theory of phonological representations in which general postvocalic stop spirantization and labio-palatalization were represented with the same structure; the fact that the former occurs while the latter seems unattested would be explained by the nature of acquisition and sound change, not by the nature of phonological representations. Regardless of how hard it may be to learn unnatural patterns, the formal properties of mental representations of learned information are not necessarily *responsible* for whether the information is hard to learn, any more than for whether the information has cross-linguistically common or rare properties (see Hale, Kissock, and Reiss 2006).

In summary, the arguments we have seen for features needing to be specified after a certain point in phonological derivation relied on a psycholinguistically unconvincing way of modeling transparency in autosegmentalism, and the argument that a feature needs to be specified before a certain point in phonological derivation in order to avoid a phonetically unnatural representation involving *ex nihilo* feature insertion depends on asking the theory of representations to handle what may actually be the responsibility of acquisition and sound change, at least in an emergent framework. In the absence of convincing evidence that predictable phonological information is not available throughout the construction of phonological representations, we will adopt the working hypothesis that it may or may not be, and that further knowledge of the matter is unnecessary for insightful analysis of the essential structure of phonological patterns.

5.6.4 Feeding

A pattern which activates another is traditionally said to feed it. Feeding and three other kinds of relationships discussed below (counterfeeding, bleeding, and counterbleeding) are traditionally captured in serial rule-based approaches by different rule orders (Mascaró 2011). There appears no satisfactory way to capture counterfeeding and counterbleeding in OT without resorting to serial derivation (McCarthy 2007).

Epenthesis feeds spirantization in Tiberian Hebrew (see the sources under (3)). This combination of (3) and (5) is represented in (8) with an example in (9).

(8) TH epenthesis feeding spirantization



(9) TH epenthesis feeding spirantization in $/malk / \rightarrow [melex]$ 'king'



Something missing in (9) is a representation of the insight that the words which have *a* before a stem-final consonant cluster generally change it to ε in unsuffixed forms containing epenthetic ε (see e.g. Coetzee 1999, Green 2004). This pattern is shown in (10), aligned with epenthesis which activates it in (11), and exemplified with /malk/ in (12). (The notation V[+lo] for a low vowel may be taken as informal; regarding the vowel category and the notation ε ([-lo]) for the predictable fact that ε is not low, see §5.6.3.)

(10) TH *a* assimilating to ε



(11) TH epenthesis feeding a-assimilation



(12) TH epenthesis feeding *a*-assimilation and spirantization in $/\text{malk}/ \rightarrow [\text{melex}]$ 'king'



5.6.5 Counterfeeding

When pattern x logically could feed pattern y but does not, it is conventional to say that x counterfeeds y. The following counterfeeding example is taken from a dialect of Bedouin Arabic dialect where in traditional terms there are two productive processes: a process "raising short /a/ to a high vowel in a nonfinal open syllable" and "a process of epenthesis that breaks up final consonant clusters" (McCarthy 2007: 10, 26; 34). The epenthetic vowel is represented here only as it appears in this word; since its quality is not relevant to the analysis, we abstract away from any other complications in epenthetic vowel quality in this dialect. Epenthesis is shown in (13), *a*-raising in (14), their interaction in (15), and a word exemplifying the interaction in (16).

(13) Epenthesis in a Bedouin Arabic dialect



(14) a-raising in a Bedouin Arabic dialect



(15) Epenthesis counterfeeding vowel raising in a Bedouin Arabic dialect



(16) Bedouin Arabic counterfeeding in /gabr/ \rightarrow [gabur]



Since no plausible cross-linguistic synchronic principle has been found for predicting when a process that can feed another will do so and when it will not, stipulation is required. This is achieved by extrinsic rule ordering in traditional generative phonology (in this case, *a*-raising before epenthesis). No satisfactory way has been found to express counterfeeding in the classic parallel mode of OT (McCarthy 2007, esp. 34-35 on the failure of Local Constraint Conjunction).

The representational alternative I suggest here is an inhibitory connection between one pattern (epenthesis) and the other (*a*-raising): activation of the former licenses suppression of the latter. The inhibition is indicated by placing a double bar through the association line that joins the two patterns where they match and by greying out the suppressed pattern. As a result of this active suppression, matches between the input and the inhibited pattern are no longer strong enough to activate the latter; this is shown by dotting the appropriate association lines.

5.6.6 Bleeding and counterbleeding

Bedouin Arabic also provides the example $/\int arib-at / \rightarrow [\int arbat]$. This is described as showing a process "deleting short high vowels in nonfinal open syllables" (McCarthy 2007: 10) along with the process of raising that we saw in (14). The deletion pattern is shown in (17) and its interaction with *a*-raising is shown in (18), with a word illustration in (19).

(17) Short high vowel deletion in a Bedouin Arabic dialect

(18) Vowel deletion bleeding *a*-raising in a Bedouin Arabic dialect



(19) Bedouin Arabic bleeding in $/ \text{arib-at} \rightarrow [\text{arbat}]$



The inhibitory association is constructed in the timing slot where the inhibiting pattern has a daughter incompatible with a parent element in the inhibited pattern. Otherwise, the lack of content match between these two elements would leave them unassociated and both patterns would be activated by their content matches with the lexical input.

In that type of situation, where pattern x could naturally bleed y but does not, the conventional expression is that x counterbleeds y. Counterbleeding is a type of non-interaction between patterns. An example of counterbleeding occurs in Catalan, where coda cluster simplification counterbleeds (fails to bleed) regressive place assimilation. The two patterns are shown in (20-21), with a word illustrating their interaction in (22) (data from McCarthy 2007: 40-41). Only coronals undergo the place assimilation, and only non-coronals trigger it (Steriade 1995: 126, de Lacy 2002).

(20) Catalan regressive place assimilation



(21) Catalan coda cluster simplification



(22) Regressive place assimilation counterbleeds coda cluster simplification in Catalan /ben-k/ → [beŋ]



If (21) bled (20), the output would be [bɛn].

Since no plausible cross-linguistic synchronic principle has been identified for predicting when a pattern which could bleed another will do so and when it will counterbleed instead, stipulation is required; extrinsic rule ordering is used in rule-based phonology, but no satisfactory way has been found to model counterbleeding with parallelism in OT (McCarthy 2007: 33-34, 47-51, highlighting the failure of his earlier Sympathy Theory and various other patches proposed for OT in this area).

There has been considerable debate over which of the two, bleeding or counterbleeding, should be treated as a default with respect to the other (see McCarthy 2007, Kiparsky 1982: 72-74). In the present model, counterbleeding is formally simpler because there is no inhibitory association, but this does not necessarily mean that either pattern is formalized as being more natural than the other.

5.6.7 Do something only/except when...

The grammatical representations discussed above have a binary basic structural division between licensing material and licensed material. However, the representational format explored here easily allows for non-binary grammatical representations that straightforwardly account for additional phenomena. These phenomena include patterns described under the headings "do something only when..." and "do something except when..." in Baković (forthcoming), to which the reader is referred for examples.

In the present framework, do-something-only-when interactions can involve a grammatical representation which is unitary, containing no internal associations. It may even be a single feature. It achieves grammatical status by serving as a parent element crucial to the activation of multiple other grammatical representations. Do-something-except-when triggering can involve a grammatical representation which is unitary in the same sense, achieves grammatical status by being a daughter to multiple other grammatical representations, and suppresses them along with itself as soon as it is activated. Conflicting activation patterns and self-inhibition after activation are staple concepts in neurology (e.g. Lieberstein 1971, Smith and Jahr 2002, Yu and Choe 2006) and have also been explored in psychological node theories (see MacKay 1987: 34, 141-164).

In one famous do-something-except-when example, three-consonant clusters are avoided in Yawelmani Yokuts surface forms even though various rule interactions would otherwise produce them (Baković forthcoming 13, Kisseberth 1970). Other rules thus apply except when they would generate CCC. In the present model, any time a lexical input plus relevant grammatical patterns jointly license a CCC string in Yawelmani Yokuts, that string would activate an inhibitor which would suppress the string itself and its grammatical parents. The CCC inhibitory structure here is equivalent to a constraint *CCC, but its impact on output construction differs from both the classic OT schema (where there is no guided output construction: §5.3) and frameworks involving constraints that trigger rules which repair the outputs of previous rules (e.g. Calabrese 2005). Here, the CCC inhibitor does not trigger repair of any output form, because the output form is only read once the grammatical representations relevant to the input have reached their proper activity levels in conjunction with one another.

Another possible kind of do-something-except-when relationship is seen in Duke of York gambits. For example (McCarthy 2007: 23), Nuuchahnulth a) unrounds wordfinal uvular stops but b) rounds uvulars after round vowels. When a round vowel is followed by a word-final uvular stop, the latter is unrounded in the output, showing in OT terms that the constraints effecting (a) outrank those effecting (b), and in ordered-rule terms that (a) follows and undoes (b). The present model would have two grammatical representations in which a) a word-final uvular stop licenses [-rd] and b) a uvular stop after a round vowel licenses [+rd]. An input with a word-final sequence of round vowel plus uvular stop could be treated as activating both representations on the basis of content match, with (a) stipulated as inhibiting (b) to resolve the clash between them.

When there is no independent need for the unitary elements that condition some do-something-only/except-when relationships, those elements can in principle be duplicated in each of the other grammatical representations that they help license or inhibit. In rule-based terms, this is equivalent to incorporating the avoidance of CCC into each of the rules that would otherwise produce CCC. This entails some redundancy and

loss of insight (Baković forthcoming 13-15). Such redundancy does not impair the grammar's practical ability to handle the data, however. In contrast, McCarthy (2007) argues that OT's practical ability to handle counterfeeding and counterbleeding with universal constraints is seriously impaired. This bears on the central concern of Baković (forthcoming). Baković identifies do-x-only/except-when-y patterns as opaque when surface conditions are not enough to predict the occurrence of x, and he emphasizes that the failure of rule mechanisms to capture these patterns complements the shortcomings of OT in handling other opaque patterns like counterfeeding and counterbleeding. Thus neither rules nor OT offers an ideal treatment of all of these phenomena. The difference in practical capacity that we have just noted between how rules handle do-somethingonly/except-when patterns and how OT handles counterfeeding and counterbleeding suggests that the balance in this case weighs in favor of rules, however, particularly since the way information is stored and organized in human memory does seem to exhibit considerable redundancy ($\S5.6.3$). At the same time, the representational approach to grammatical directions presented here appears to avoid the shortcomings we have just seen with rules and constraints alike. It does involve sequences of activation flow and suppression throughout networks that realize grammatical patterns, but it is already known that such activities are neurally common.

5.6.8 Summary

Encoding grammatical patterns as representations that contain and share licensing relationships enables transparent yet not intrinsically serial analyses of what have traditionally been viewed as major kinds of process interaction, including opaque interactions. The computational efficiency of this model for both acquisition and mature execution is considerable. Compared with the classic OT schema, this model reduces the output candidate search space to a grand minimum. Additionally, if grammatical directions are learned, then the model renders superfluous the presence in individual grammars of most of the constraints that OT assigns through its universal constraint pool; instead, the information equivalent of crucially related subsets of OT constraints is learned in a vastly more compressed form via the extraction of partial representations from observed words.

This embodies the emergent (Mielke 2008), evolutionary (Blevins 2004), and substance-free (Hale and Reiss 2000a,b, Hale 2007) understanding of the relationship between innate cognitive ability, the learning process, and cross-linguistic trends in detailed phonological patterns. What the present model makes especially explicit is that observed word forms mediate between the phonological contents of synchronic grammar, on one hand, and phonetic and other non-cognitive forces that shape sound patterns diachronically, on the other. This mediating role of word forms is highlighted through the idea that a grammar's phonological feature information inheres in representations extracted from observed words rather than being located in constraints that are innate or learned through babbling.

Conceivably, some conjunctions of grammatical patterns (e.g. in 12 above) and even frequently recurring elements of the output in such complex representations are remembered by language users rather than discovered afresh for each use – just as most people remember the lower values in the multiplication table rather than recomputing them every time they are needed. This possibility about grammars is transparently captured in the present model by letting complex structures like the grammatical material in (12) (i.e. 12 minus the input and output rows) remain stable in a user's memory. With serial rules or the classic OT schema, in contrast, it seems less obvious how frequently accessed patterns in the grammatical machinery could be stored as "prefabricated" elements in memory so that they would not need to be recompiled at every use. This observation seems theoretically interesting because it identifies a category of operations in both rule-based theories and the classic OT schema which appear to be unnecessary for modeling or computing grammars. While it is common for information to be equally configurable in either declarative or serial-operational terms (see Garson 2010), the transformational equivalence of those two styles of format does not make them equally transparent in representing actual grammars with psychological realism. This section has presented a fresh search for such representations.

5.7 Output-drivenness

Languages use a wide range of possible input-output mappings. There are languages with underlying distinctions absolutely neutralized in all positions on the surface; for example, Blackfoot, Wichita, and Uyghur both have two underlyingly distinct segments that neutralize to surface i (Frantz 1991, Rood 1975, Comrie 1997), Menomini similarly has two underlyingly distinct kinds of n (Bever 1961: 33-34, 75-80), and Yucatec Maya has two underlyingly distinct kinds of h (Orie and Bricker 2000). There are languages with surface distinctions that reflect no underlying distinction, as is abundantly illustrated by extensive, systematic, and categorical consonant allophonies in the Numic branch of Uto-Aztecan (Charney 1993, Elzinga 2000, Sapir 1992[1930]). And of course in all languages there are faithfully surfacing underlying distinctions.

One way to capture these different kinds of input-output mapping is with grammatical directions that directly define the *maps*, which necessarily means including reference to both inputs and outputs. In OT, however, the only constraints that directly describe maps are faithfulness constraints which spell out non-occurrences of input-output mismatch. In other words, instead of treating it as the default for things to remain as they are, OT makes this type of case the only context in which constraints can refer directly to maps. All other OT constraints directly refer only to outputs, a much celebrated property that has been termed *output-orientedness* (Tesar 2008, Smolensky et al. 2006:2:233, cf. Prince and Smolensky 2002: 1).

As a result, OT must capture some types of mapping circuitously and even then problems remain. For example, the case of underlying distinctions absolutely neutralized at the surface could be partly described in OT by distinguishing faithful from unfaithful surface phones, though the choice of which was which might be arbitrary. However, forcing Wichita *t* to surface as [ts] before only one of the two kinds of *i* (Rood 1975) would require inter alia a markedness constraint banning output [ti] only when the *i* was the right sort. There are no properties which distinguish these two kinds of *i* at the surface level, but OT's output-orientedness by definition requires markedness constraints to refer only to outputs.

In contrast to OT, the representational model presented in this chapter defines grammatical directions *as* map components. In this model, restricting evaluation to output forms makes no sense. Evaluation (identification of the correct phonological output) is performed on total maps. Total maps are composed of grammatical representations associated with a lexical underlying form, and the output is identified as a portion of the total map's content. This captures a more efficient and intuitive picture of outputs than is possible under the requirement that content-specific constraints apply solely to output forms. That requirement forces grammars to use larger constraint sets so that they can handle inputs they never actually receive by assigning those inputs to outputs via mappings that are otherwise unnecessary, in order to handle violations of a set of generalizations that we cannot show are ever actually violated in the language (cf. §5.5.3). Mohanan et al. (2009: 8) have already argued that such a restriction may be inappropriate in an emergent framework. This chapter's constructive, representational approach renders the restriction pointless.

In place of grammars that are output-oriented in the sense of describing inputoutput mismatches only in terms of output forms, the present model accesses a different concept for which I will appropriate the term *output-drivenness*. This term has been used in various ways before. Some writers seem to use it synonymously with OT outputorientedness (Liu 2009, Hyman 2006: 4, Blevins and Garrett 2004: 120, Vaux 2001). Tesar (2008) redefines it as a logically possible formal property of grammars whereby no possible input maps to a grammatical output unless all possible inputs that are more similar to that output map to it also. Output-drivenness by this definition clearly relates to RotB, since both deal with mapping all possible inputs to grammatical outputs.

This chapter presents a view of grammar that can be described as output-driven in a very different sense: the derivation of grammar from observed outputs during acquisition is maximized to reduce innate/learnable duplication problems. In this framework, the analysis of actual outputs during acquisition controls the creation of more grammatical structure than in grammars with innate content-specific constraints. Outputdriven grammars by this definition contain only the information that they need to operate and that learners can reasonably be expected to learn, rather than all the information that a typologist would want in order to elegantly describe grammars as permutations of each other. Grammars of this sort are optimized during acquisition.

As described at the outset (§1.3), this emergent view of phonology entails certain ideas about sound change too: a large percentage of sound changes are predicted to occur during acquisition as people construct grammars that differ slightly from the grammars of an older age group. These differences are more substantive than just the reranking of universal constraints, and because they occur during the process of creating grammatical directions more or less from scratch (without knowing anything about the grammars inside the older age group's heads), there is no reason to model the differences as the addition, deletion, or reordering of rules in grammars initially matching those of the older age group. These claims inform the methodology for reconstructing parts of the sound systems of prehistoric languages by retracing the sound changes that have affected daughter languages. In order to compare the daughter languages and make insightful guesses about past commonalities from which they diverged, it proves important to understand natural phonetic variation in sounds and different ways of mapping phonological categories (features) onto bundles of phonetic correlates (cf. §3.3.4). This is illustrated in the next chapter.

6 Reconstruction

6.1. Introduction

Chapters 2 and 3 developed a perspective on the laryngeal phonology of stops, and chapters 4 and 5 developed insights on the emergence of feature patterns within phonology in general. This chapter applies those findings to one of the most notorious unsolved problems in historical linguistics: the laryngeal contrasts of the Proto-Indo-European (PIE) stops, and the status of an apparent gap in one series where a labial is expected.

The comparative evidence for the PIE stop series is presented first, followed by a novel reconstruction and a brief summary of why this reconstructed system is synchronically realistic (§6.2). The evolution of this reconstructed system into the daughter languages is then systematically traced in both phonological and phonetic terms, with typological parallels and again with reference to the theoretical principles developed in previous chapters (§6.3). The proposal is contrasted with previous solutions to the same problems; shortcomings are identified in these but found absent in the new proposal (§6.4). A similar routine is performed on the problem of the apparent labial gap, with a critical typology of possible reconstructions (§6.5.1), a new recommendation, and a detailed diachronic elaboration of the consequences in the daughter branches (§6.5.2). A separate controversy about the number of PIE stop series is reviewed (§6.6). Finally, the results are summed up in the context of an evaluation of the historical-comparative method and the role of subsegmental information in reconstruction (§6.7).

Etymological research on Proto-Indo-European is more extensive than for any other proto language in the world. I propose no new etymologies here but focus only on the phonological analysis of the system which has already been reconstructed on the bases of hundreds of cognate sets. For this reason and also for reasons of space, I avoid discussing individual word forms and focus on *segments* instead. Extensive lists of cognates and discussions of the reconstructions of particular words will be found in the numerous works cited from the Indo-Europeanist literature.

6.2. The Proto-Indo-European stop system

6.2.1 Comparative evidence

The simplest reading of the comparative evidence is that PIE had three series of stops, one of them voiceless unaspirated and the other two voiced. This reflects the majority of the daughter languages. The "standard" or mainstream traditional reconstruction of PIE assumes this shape and differentiates the voiced series with aspiration, following the Indic reflexes. (Standard grammatical sources include Clackson 2007, Fortson 2004, Meier-Brügger 2003, Mayrhofer 1986, Sihler 1995. The best lexical resource is Rix et al. 2001.)

The comparative evidence is given in Table 6.1 in a simplified form highlighting the main reflexes for each of the three series according to standard sources like those just listed, except for Italic where the analysis of the second series follows Stuart-Smith (2004).

Table 6.1. Comparative evidence for the PIE stops.

| | In. | G | It. | Gm. | Arm. | С | BS | Hitt.1,2 | Alb. | Ir. | Toch. |
|------|---------------------------------|---------|------------|-----|----------------------------|---|----|--------------------|------|-----|-------|
| I. | Т | Т | Т | Θ | T^{h} | Т | Т | T ^h , T | Т | Т | Т |
| II. | $\mathbf{D}^{\mathbf{\hat{n}}}$ | T^{h} | Θ/E | D/Đ | $\mathbf{D}^{\mathrm{?h}}$ | D | D | T, D | D | D | Т |
| III. | D | D | D | Т | Т | D | D | T, D | D | D | Т |

A fourth series T^h is found in Indic, but cognates elsewhere generally have the same reflexes as the first series, so Indic T^h is generally not traced back to a separate

phonemic PIE series (see the above sources, Barrack 2002: 78, Lehmann 1996: 57, Salmons 1993: 8, and §6.6). The interpretation of Hittite is uncertain (see Melchert 1994); two possibilities are given here separated by commas. Series II is aspirated in numerous Armenian dialects, but it is unclear whether the series was associated with breathy voice in their common source (Garrett 1991: 798-99, Pisowicz 1976).

The standard reconstruction of the three series as T D⁶ D (also written T D^h D) creates an inventory with no attested parallel, notwithstanding various counter-claims (see §3.2.6.3f) which have sometimes been counter-balanced in their turn by inaccurate criticisms.¹¹ On the assumption that PIE was a natural human language, any systemic properties of PIE for which we find no attested parallels deserve closer study. Are they simply empirical rarities, or do broad theoretical considerations oppose them and suggest that the standard reconstruction is incorrect?

I argue that the latter is true and propose a revised reconstruction, taking into account the Indo-European comparative evidence, global typological evidence, and a coherent feature theory of the sounds involved. This reconstruction is presented in (23). After defining the synchronic phonology of this system and motivating it from the findings in previous chapters, I analyze how it evolved in the daughter languages (§6.3), showing that the paths of sound change have good typological parallels. (The nature and importance of this kind of diachronic typology has already been described and modeled by Hayward 1989, Job 1989, 1995, Ohala 1993, and Stuart-Smith 2004, among others; cf. Ohala 1990b: 268.) I then compare my proposal with previous reconstructions (§6.4).

¹¹ E.g. D seems absent in inflectional and derivational morphemes, while D⁶ is not, and this has been treated as a markedness problem (see Clackson 2007: 46, Hopper 1973: 141, 155, 157). But the premise that languages have a strong tendency to select affix segments based on a markedness hierarchy is not corroborated by the evidence. Bybee (2005) concluded from her typological study of verbal affixes in twenty-three "maximally unrelated" languages that while languages obey some obvious exclusions, such as one against using only phonetically complex consonants (like coarticulated or non-pulmonic ones), there is otherwise only a weak, fractional trend toward patterned exclusions, and that these are due to a variety of reasons including phonetic reduction (which privileges the non-complex) and the reuse of old affix material.

For readability, asterisks are generally omitted from hypothetical forms; attested forms are labeled as such or with the appropriate language name.

6.2.2 A new proposal

I propose that the PIE stop series were voiceless unaspirated, voiced unaspirated explosive, and voiced non-explosive (§2.6.2). This inventory type is cross-linguistically well-attested (§3.2.5).

(23) The PIE stop series

| I. /T/ | [T] |
|--------------------------|-----------|
| II. /D/ | [D] |
| III./D ^{<} / | $[D^{<}]$ |

The suggestion to redefine the second and third series as unaspirated voiced explosive and voiced implosive or non-explosive was made by Haider (1985) and developed by Weiss (2009) and Kümmel (2010); all three restrict this redefinition to Pre-PIE, leaving the standard PIE model essentially intact with whatever problems it already had. Miller (2006) suggested attributing /T D D[<]/ to PIE itself. Miller (2008) and in more detail Kümmel (2010) highlight the dearth of traditional PIE ND clusters as typologically supporting a reconstruction of traditional D as implosive or non-explosive D[<].

6.2.3 Place of articulation issues

In addition to positing the contrasts in (23), I follow the standard reconstruction in positing that the PIE stops contrasted for place of articulation, including labial, coronal,

velar, and labiovelar - with one probable exception. In the third series, the expected labiovelar non-explosive was probably not realized as such because that seems to be an unattested type of sound (see Ladefoged and Maddieson 1996, Maddieson 1984). Labialvelar stops such as $[\overline{qb}]$ and $[\overline{qb}^w]$ with oral rarefaction effects like those of nonexplosives are attested, however (Greenberg 1970, though contrastive non-explosion is also reported in voiced labial-velar stops: Salffner 2004: 101). Extensive diachronic interplay has been observed among labial-velars, bilabial non-explosives, and labiovelars (Cahill 2006, Williamson 2004, Grégoire 2003, Ikekeonwu 1999, Lewis 1997, Ladefoged and Maddieson 1996, Demolin 1995, Ohala and Lorentz 1977, Rountree 1972, Greenberg 1970). The attested sound typologically most comparable to a hypothetical $/d^{w}/$ thus appears to be $[\widehat{qb}^{w}]$ or $[\widehat{qb}]$. Labial-velar stops are attested in contrast with labiovelar stops, languages with $\overline{/qb}$ but no $\overline{/kp}$ are more common than the reverse type, and languages with $\overline{(qb)}$ but no $\overline{(kp)}$ tend to have fewer gaps elsewhere in the system than languages of the reverse type (Cahill 2008). All of these facts harmonize well with a PIE containing a voiced labial-velar stop but no voiceless one. Instead of a voiced labiovelar non-explosive in the third series, therefore, I posit a voiced labial-velar.

Having surveyed the synchronic character and typological viability of the nonexplosive reconstruction in (23), we now turn to the diachronic implications. In the next section, I show that the non-explosive reconstruction fares well in this regard.

6.3. Diachronic outcomes

Although systematically non-explosive or sonorant reflexes of the third series do not surface in any of the traditional twelve branches, this can be seen as a historical accident. The boxes in Table 6.2 with the commentary underneath show that the three stop series evolved according to three separate trends. Each of the twelve branches follows one of these basic trends.

Table 6.2. Diachronic trends.

| PIE | Ind | G | It | Gm | Arm | С | BS | Alb | Ir^{12} |
|--|-----------|---------|----------------|-----|----------------|---|----|-----|-----------|
| I. T | Т | Т | T ¹ | Θ | T ^h | Т | Т | Т | Т |
| II. $(\mathbf{D}_{\mathbf{D}})^{(\mathbf{f})}$ | $a D^{h}$ | T^{h} | Θ/Đ | D/Đ | $D^{?h}$ | D | D | D | D |
| III. $D^{<}$ | D | D | D | Т | Т | D | D | D | D |

(*a*) Series III became explosive like II, while II acquired a turbulent noise correlate (§3.3.6.6) and, in Greek, devoiced. Devoicing and aspiration of a phonologically plain voiced stop series while a voiceless unaspirated series remains unchanged is also reconstructed in Western Kammu (Mon-Khmer; Svantesson and House 2006: 310), Lawa Umpai (Mon-Khmer; Sidwell 2006: 39) and Madurese (§3.2.6.3); further parallels to the total devoicing of the aspirates in Greek are found in Armenian dialects (Pisowicz 1976) and Indic languages (Stuart-Smith 2004). The shift of nonexplosives to explosives is well-attested (Greenberg 1970).

(*b*) Given voice as a default implied by non-explosion in series III (§2.6.2, 4.3.2), the loss of voice there could be taken as a default effect attendant on loss of non-explosion, or it could have postdated this. In series II, voice is not a default implied by any other property, and it survives. Series I undergoes glottal widening, manifested as aspiration in Armenian and ultimately with frication in Germanic (cf. the discussion of vocal fold abduction in voiceless fricatives, §3.3.6.4).

By treating aspiration as original to the dominant phone of series I in PIE, Gamkrelidze and Ivanov (1995) remove the main Indo-European examples of general aspiration of a voiceless series, but Khasi (Mon-Khmer) provides a non-Indo-European parallel (see Sidwell 2006: 36, cf. 2005: 197). For parallels to the loss of non-explosion in series III see (*a*); for parallels to the obstruent devoicing in series II see Khmer and

¹²Hittite interpretations 1 and 2 fall in groups b and c respectively. Tocharian is omitted because it merges all three series to voiceless obstruents, except for a zero reflex of the series III coronal (Clackson 2007).

Khmu Yuan (Mon-Khmer: Ferlus 1992, Sidwell 2006: 39), Scottish Gaelic (Green 1997: 44, Gillies 1993: 148; contrast Modern Irish: Green 1997: 41-43), Welsh (Ball and Müller 1992), and Alsatian German (Keller 1961: 121).

(*c*) This group shows loss of non-explosion in III and merger of III with II. For loss of non-explosion see (*a*).

Looking at the evolution of the stop system as a whole, we see an earlier conclusion (§3.2.6.5) borne out in the fact that aspiration and voice did not both stabilize on II except in Proto-Indo-Iranian, which developed contrastive T^h in addition to T and D. This T^h is preserved most straightforwardly in Sanskrit. Its development there may have been contingent on several other factors working together as follows.

Several PIE stems are traditionally reconstructed with clusters of a voiceless stop followed by $*h_2$, a consonant of unknown realization conventionally called a laryngeal (see the standard sources in §6.2.1; further discussions and sources are found in Reynolds et al. 2000 and Winter 1965). Sooner or later the cluster came to be realized as a voiceless aspirate series [T^h], at least in Indo-Iranian. In other branches of Indo-European, the reflex of PIE /Th₂/ is generally the same as that of PIE /T/, with original /h₂/ vanishing in this environment as in nearly all others. The development of voiceless aspirates from /Th₂/ in Indo-Iranian may be compared to the development of phonetic voiceless aspirates from /Th/ clusters in Mon and Khmer, though the latter aspirates may remain synchronically bisegmental since processes such as infixation regularly break up the clusters in morphophonemic alternations (see Ferlus 1992: 82, 1983: 77-78).

It may be significant that the inherited stock of T^h cases in Indo-Iranian are mostly coronals, since the greater quickness of the coronal articulator (Fourakis, Botinis, and Niryanakis 2002, Frisch 1997: 167-68, Gafos 1996: 3-10, 131-74) might make monosegmentalization of /Th/ easier at that place. Further, since PIE h_2 conditioned

adjacent non-high vowels to *a* and is in fact the only known systematic environment in which PIE *a* occurred, accounting for the vast majority of reconstructions with that vowel, the laryngeal in the /Th₂a/ sequences had relatively low functional load to begin with. Conceivably this made it somehow easier to lose (cf. Bermúdez-Otero 2007: 512). The single branch in which this aspiration was clearly not lost, Indo-Iranian, is also the single branch in which the non-high vowels merged and the functional load of the distinction between *a* and *e*, *o* was lost – creating a unique opportunity for the functional load of the reflex of PIE h_2 to increase naturally and to stabilize with the realization of voiceless aspiration. This in turn helped license the unique rise of a voiced aspirate series with robust breathy offset and closure voicing. The interplay of these contrastive relationships is probably not entirely accidental.

6.4. Comparison with other reconstructions

6.4.1 Other kinds of voiced stops

The non-explosive reconstruction in (23) proves superior to several models that redefine the second series in other ways. While space prevents me from considering all possibilities in detail, I have tried to incorporate all the interesting ones through judicious discussion of typologically attested phonological categories and phonetic patterns.

Voiced glottalized explosive stops are probably the most similar alternative to non-explosives; indeed they are poorly distinguished in the literature from non-explosives and do not appear to contrast with them (§2.6.2). Since non-explosives do not require glottalization, however, there seems no point in reconstructing it for PIE. The objection of Barrack (2002: 91) and Gamkrelidze (1989: 99), that a bilabial "gap" becomes even less tolerable if the bilabial is non-explosive, is treated in §6.5.

Prenasalized stops, a logical possibility casually noted by Gregory Iverson (p.c.) as a conceivable value for one of the PIE voiced stop series, would have to occur in

various clusters, given what we know about PIE phonotactics. Yet some of these clusters seem typologically bizarre, such as [$\{r, l, z\}^nD$]. In the languages with CNC sequences described by Herbert (1986: 183-184), he finds phonological evidence that the nasal is syllabic, and he concludes that prenasalized consonants tend to occur in languages that avoid syllable codas (p. 186). Syllabic nasals are already reconstructed for PIE, but with developments unrelated to the stop series (see the standard sources in §6.2.1). I conclude that prenasalized stops are no better an interpretation of the third PIE stop series than voiced glottalized stops.

Finally, reconstructing PIE series II and III as prevoiced aspirated and plain voiced explosive, with the aspirates being like those of Kelabit and Lun Dayeh (§3.2.6.4), would leave unexplained the much wider distribution of PIE series II, particularly in post-consonantal position; it is also unclear how such a reconstruction would handle the rarity of series III in postnasal position (Kümmel 2010) and the problem of the bilabial gap (§6.5).

6.4.2 Ejectives

Ejective models have probably been the most popular approach to revising PIE in the last several decades, under the heading of Glottalic Theory (Gamkrelidze and Ivanov 1995, Vennemann 1989, Kortlandt 1978, Hopper 1977a, 1973, cf. Fallon 2002). But ejective models require most of the daughter branches to undergo a subsequent T' > D sound change that clashes with what trends we can see in the typological evidence and with the most plausible reading of phonological universals underlying them – unless one posits a non-explosive intermediary (T' > D[<] > D), in which case there seems little reason to have the ejective stage at all. (On the typological evidence see Barrack 2004, 2003a,b, 2002, Job 1995, 1989, and for evidence from alleged loan etyma, Vennemann 2006; on relevant phonological theory see §3.2.7.) Barrack (2002: 91) and Gamkrelidze (1989: 99) object that a reconstruction back to $D^{<}$ runs afoul of a bilabial problem often referred to as the "*b* gap". As we will see shortly (§6.5.1.2), there is a solution to this problem that does require the non-explosives to have developed after the breakup of PIE, and hence demands a different realization such as ejectives in PIE itself; this solution is not the only possible one, however, and it is relatively implausible in that it requires every daughter branch to have innovated [6] after the breakup.

There is also a subtler problem with ejective models. Given the restricted distribution of aspiration (§3.2.6), the stop inventory /T T' D/ is phonologically well-formed, but /T T' D^f/ with the third series phonologically specified for both voice and aspiration is ill-formed. At the same time, any three-series PIE stop inventory with the first series specified as aspirated (/T^h/) creates a diachronic typological problem since it would force unconditional phonological deaspiration of [T^h] throughout most branches of the family – a situation without adequate parallel as far as I am aware. The nearest possible parallel that has been brought to my attention, Austrian German (Anna-Maria Adaktylos p.c.), is too dissimilar in both the number of series involved and their phonetic realizations (see Moosmüller and Ringen 2004). The dearth of evidence for general deaspiration without merger of laryngeal distinctions has influenced arguments about the reconstruction of the Proto-Tai stop system as well (see Pittayaporn 2009: 88). Thus, ejective models run into trouble if either of the non-ejective series is phonologically specified for aspiration.

Yet it appears that they run into another problem if aspiration is not specified. Within an ejective approach to PIE it seems that CVC roots have phonological agreement in voice but not in laryngeal constriction or aspiration (Gamkrelidze and Ivanov 1995: 29, cf. D. G. Miller 1977a). However, Hanson (2001) generalizes from typological evidence that such root voicing agreement implies similar agreement in another laryngeal feature, such as constriction or aspiration. Since neither of these features is available because they are not specified in a well-formed ejective model of PIE, root voice agreement is not expected there either. (For an alternative treatment of the relevant root patterns that does not rely on a radical redefinition of the second series, see Iverson and Salmons 1992.)

6.4.3 Fricatives

Prokosch (1939) suggested reconstructing series II as fricatives, but the suggestion works about equally well with either II or III. Reconstructing voiceless fricatives would require a typologically unsupported general voicing of that series in most of the daughter branches, but voiced fricatives would escape that problem. A voiced fricative series II would have to despirantize directly to voiced aspirates in Indic and perhaps phonetically in Armenian, a change for which I have no typological parallels, though it does not seem phonetically implausible (see Stuart-Smith 2004: 206, Vaux 1998a: 509). Otherwise, the only diachronically questionable aspect of reconstructing either II or III as voiced fricatives is the general despirantization that would be required in all the daughter branches.

I do not have parallels for such a change in the normal evolution of any language, only examples restricted in various ways – mostly to allophones at some place of articulation, to borrowing, and to child speech. Modern Hebrew has despirantized earlier $[\theta \ \delta \ \gamma]$ allophones of [t d g] (Idsardi 1997a). Comanche (Numic: Uto-Aztecan) has reportedly despirantized earlier $[\delta \ \gamma \ z]$ allophones back to their dominant allophones [t k ts] (McLaughlin 2000, Charney 1993). We do find a despirantization of phonemic dental fricatives in German (*du, der, dorf* cf. Eng. *thou, the, thorpe*), but conceivably this relates to dental fricatives being relatively rare cross-linguistically. Second-language varieties of English are well known for realizing / $\theta \ \delta$ / as [t^(h) d]. Second languages and borrowing patterns present examples of despirantization at additional places of articulation, including postvelar (Gaelic [lox] 'loch', German [box] 'Bach' yielding American English [lɑk], [bɑk]) and even labiodental (e.g. an Indian reporting Marathi as his first language and speaking English sub-fluently mentioned India's [p^heməs p^hilm stɑız] 'famous film stars', personal experience; cf. Iverson and Salmons 2008: 10, 13). For despirantizations in child speech see Vaux and Miller (2011). While better examples would be nice, the notion of general despirantization does not seem theoretically problematic.

As far as PIE itself is concerned, a reconstruction with T D D assigned in some order to the three PIE series leads to sound changes for which I lack clear typological parallels, though the changes themselves are not necessarily phonetically unreasonable.

6.4.4 Conclusion

The survey in the previous section indicates that the most coherent and typologically well supported reconstruction of the three PIE series may be the /T D D[<]/ reconstruction as shown in (23). This reconstruction not only departs less from the comparative evidence than most of the alternatives, but it also has an attractive instability in the inherently temporary nature of non-explosion and other strategies for staving off oral air pressure buildup (§2.6) which may relate to why non-explosives have been estimated to occur systematically in only ten percent of the world's languages (Maddieson 1984). The non-explosive reconstruction also has more specific advantages which we will see in dealing with the next problem.

6.5. The labial "gap"

6.5.1 Critical typology of reconstructions

There is little or no comparative evidence for a bilabial in the third series (Clackson 2007, Fortson 2004, Meier-Brügger 2003, Sihler 1995, Mayrhofer 1986, in

spite of Szemerényi 1996: 145, 1985: 11-12, Djahukian 1990: 5-7, Wescott 1988, Johansson 1900: 390, and Havers 1946: 96-7 and Meid 1989 in Salmons 1993). If the third series is voiced, this is typologically surprising given the other attributes of the stop system, such as its lack of gaps (which enervates comparisons with highly gapped languages, e.g. Woodhouse 1995: 180, 1993: 3; cf. Iverson and Salmons 1996, Mac and Longacre 1960: 27, Longacre 1957. Gaps have proven typologically relevant to stop voicing patterns elsewhere; Cahill 2008.) There are several possible responses to the problem posed by this apparent dearth of the PIE series III bilabial, a problem we will loosely call the "*b* gap" for convenience. Below I have tried to schematize the possible solutions exhaustively.

6.5.1.1 PIE had a /D^(<)/ series with the bilabial unusually rare or absent

The first possibility is that PIE series III was an unaspirated voiced stop series and the bilabial member of the series was rare or non-existent even though this is typologically unusual. This solution is naturally unpopular because historical linguists prefer reconstructions with broad structural traits that resemble those found in attested languages. Additionally, it is unclear what would motivate the development of this reconstructed system from a less unusual earlier system. That motivated sound changes can produce structurally atypical traits which later disappear is well known,¹³ but in this case no plausibly motivated sound change is identifiable.

The only attested case I have found where the bilabial member of a phonologically voiced stop series may have undergone an unconditioned sound change

¹³Compare the rise and subsequent disappearance in English and Welsh of non-back rounded vowels (which are cross-linguistically even rarer than back unrounded vowels: Maddieson 2008a), or the weakening of final *-m* producing paradigms with zero accusative singular morphology but non-zero nominative singular morphology in Old Icelandic and Old French, a cross-linguistic anomaly subsequently disappearing from both languages (see Dixon 1994). I admit these are not particularly good examples: the first is not rare enough to be exceptional, and the other is not phonological.

not shared by the rest of the series is when the series is [6 d g...] (Greenberg 1970). But assigning this value to pre-PIE series III, with [6] then becoming zero or merging with some other segment in PIE itself while the other stops remain voiced, raises the question of what value to assign to series I and II without causing typological problems (§6.2-6.4).

Another logical possibility is having a pre-PIE voiced stop series III with the bilabial later undergoing unconditional articulatory undershoot, or lenition, to a fricative or glide which then by a plausible separate sound change disappears or merges with another segment in PIE. The nearest example of such a labial-only lenition that I have found, in Nkore-Kiga (Gurevich 2004: 177-78, Poletto 1998, Kirchner 1998: 108, 123, Muzale 1998, Taylor 1985), appears in spite of somewhat heterogeneous documentation to be a post- or inter-vocalic lenition which targets just the bilabial member of the voiced stop series $(*b > \beta)$ thanks to the fact that its historical effects on the coronal (*d > r) are masked by numerous loans with unlenited d. The lenition environment still seems conditioned, though. Broad studies of lenition or articulatory undershoot (Gurevich 2004, Kirchner 1998) do reveal that multiple previously proposed universal place constraints are invalidated by the diverse empirical data, but these studies also suggest that lenition tends to target either the environment of adjacency to sounds with lower stricture, or particular sounds that are cross-linguistically less widely distributed, like voiced velar obstruents. This weakens the case that voiced bilabial stops would lenite unconditionally while their counterparts at other places of articulation did not.

6.5.1.2 PIE had a bilabial gap, but it was natural

The next possibility is that PIE series III had some realization in which a bilabial gap would not be unusual. To avoid creating unusual gaps in the daughter languages (§6.5.1.1), this unobjectionable gap in PIE must have been filled in each of the branches

before or while the rest of series III was changing. However, the possible original realizations for III in this scenario run into various wider problems that were mostly identified in §6.4. Ejectives, for example, would mean a well-attested [p'] gap (Maddieson 1984) but would also entail either an unattested direct T' > D sound change or voicing through an intermediate non-explosive stage; in the latter case, [6] must have been innovated in each of the branches before the rest of the series had finished becoming non-explosive, to prevent an unusual gap from recurring in each branch. Similarly, reconstructing series III as voiceless aspirates with a [p^h] gap would force several daughter branches to undergo unconditional $T^h > D$, which is typologically and phonologically dubious since it would involve unconditional deaspiration (§6.4.2) with a simultaneous and questionable addition of voice. As far as I know, a solely bilabial gap in a large system of stops which otherwise has few or no gaps is not attested when the gap is a voiced non-explosive, creaky-voiced, or voiced preglottalized stop. There do not seem to be any other, better possibilities.

6.5.1.3 Traditional PIE /b/ merged with some other segment(s) before PIE

Another conceivable option is that PIE had no bilabial gap. The semblance of one is due to a subsequent merger of the series III bilabial with another segment, which occurred separately in the daughter branches. This opens up multiple options. One is that different mergers occurred in different branches and the cognate sets illustrating this divergence have not been recognized because the historical evidence was not examined well enough. This suggestion is not convincing until someone produces new and satisfying cognate sets. The remaining options involve the same merger occurring in each of the branches.
Merger with /w/ has been proposed (see Sihler 1995), but in its simplest form this would require the branches to undergo unconditional articulatory undershoot in only the series III bilabial. Such lenition seems unattested as we have already seen (§6.5.1.1). A revised version of this merger could have an originally non-explosive third series lose non-explosion except for the bilabial (an attested pattern $\S6.5.1.1$), followed by secondary changes introducing contrastive [b]; the lone bilabial non-explosive would then constitute its own series and presumably be free to become a glide without incurring the "labial-only lenition" objection, followed by merger with /w/. This same mechanism would work with merger to /m/ (an option noted by Mayrhofer 1986: 100). Both /6/ > /w/and $\frac{6}{>}$ m/ are attested (Pittavaporn 2007: 106-107, Haider 1985: 11, Haudricourt 1975, Greenberg 1970: 137), although in several of these languages, one finds implosives at other places of articulation becoming resonants as well, and in none of the languages have I found it stated that this type of change affected the bilabial implosive only. Another possible merger outcome is the series III labiovelar or labial-velar, but while diachronic interplay among $[6, \overline{qb}, \overline{qb^w}, q^w]$ is attested (§6.2), I do not have a parallel for this particular directionality. All of these scenarios require the same merger to occur in each of the daughter branches, however, which seems relatively unlikely.

Merger of the series III bilabial (traditional /b/) with the series II bilabial (traditional /b^h/) is another story. Barrack (2004) suggests that such a merger might have occurred in Pre-PIE and points out that it would have contributed to the high frequency of traditional PIE /b^h/. Locating the merger in Pre-PIE, however, would have caused the problematic gap in PIE itself that we are interested in avoiding. Let us therefore examine what the merger might have looked like if it had played out *after* the breakup of PIE.

6.5.2 Merger of traditional PIE /b/ with /b^f/ after PIE

6.5.2.1 Introduction

This section develops a model of the evolution of PIE in which traditional PIE /b/ and /b^{fh}/ merge in all of the daughter branches. In seven of the branches, this merger is already posited in the traditional reconstruction as part of a more general merger between series II and III. These branches therefore require no special arrangements. They will be referred to as "the simpler cases" and discussed fairly briefly below. In the five remaining branches, however, series II and III did not merge. The hypothesis that /b/ merged with /b^{fh}/ in these branches therefore entails more complex explanations, which will be presented in detail for each branch.

A key trait of the following proposal is that nearly all the forces motivating the merger in each of the daughter branches are already posited as being at work in the stop systems of those branches given the model of PIE in (23). The remaining sound changes that are additionally required for this merger hypothesis are fairly well supported both empirically and in terms of phonetic plausibility.

The only element not already entailed by the non-explosive model which this following proposal posits as recurring in more than one daughter branch (specifically five) is so phonetically natural that it may be likelier to occur than not. This is an element of timing and has the following basis. Higher pressure is correlated with the posterior end of the place spectrum, where the oral cavity is smallest and air pressure builds up fastest. One well attested correlate of this fact is that aspiration tends to be stronger in more posteriorly placed stops (Billerey-Mosier 2003: 6, Kobayashi 1999: 2, Gordon and Applebaum 2006: 163-64). Another is the already mentioned fact that voiced series are attested with unevenly distributed oral air pressure, such that the most anterior member, the bilabial, is non-explosive while the other members are explosive. From such facts I

draw the simple inference that pressure-increasing sound changes, including loss of nonexplosion and development of aspiration, may first become noticeable at the posterior end of a series and then spread forward. I will refer to the effect of this phenomenon on bilabials as "labial delay".

I now present the hypothetical individual branch developments, supported typologically and with phonetic arguments. For data on the branches, see the standard sources in §6.2.

6.5.2.2 The simpler cases

A general merger between series II and III has already long been reconstructed in seven branches of the Indo-European family: Baltic, Slavic, Albanian, Iranian, Hittite, Tocharian, and Celtic. (Separate complications affect the series III labiovelar in Celtic and some cases of the series III coronal in Tocharian, but the details are irrelevant here.) The general merger in each of these branches has traditionally been reconstructed as PIE /D⁶/, /D/ merging to /D/. Under the non-explosive hypothesis, we have instead PIE /D/, /D[<]/ merging to /D/. The typological support for the changes involved was given earlier (§6.3).

6.5.2.3 Armenian and Germanic

In Armenian, series I and III increased in pressure, with I aspirating and III then losing non-explosion and devoicing. The bilabial member of III experienced labial delay, only becoming explosive later. It then merged with the most similar other segment, the bilabial of series II which at this stage was phonologically either voiced or slack (cf. §6.2). If the members of series III became explosive and voiceless simultaneously, then no more need be said, but I do not have a typological parallel for such a development. Alternatively, they lost non-explosion before voicing, but in this process they did not become confused with series II because the latter had become slack by then.

The Germanic situation is similar to the Armenian one, but series I eventually spirantized. There is debate over the contexts in which II may also have spirantized before the breakup of Proto-Germanic. No evidence for breathy offsets is reconstructed for II based on the Germanic daughter languages, but the following analysis does reconstruct this trait in order to handle a special circumstance.

In certain loanwords, stops that were arguably voiced in the source languages (Vennemann 2006) ended up as voiceless unaspirated in Germanic, including some labial examples. The main point of interest is the origin of Gmc /p/. None of the examples are securely traced back to PIE – if they were, there would be less of a PIE "*b* gap" problem. Hence it is commonly suggested that Gmc /p/ derives from borrowings, possibly supplemented by certain other innovative processes within Germanic which we cannot identify. The question is then how these borrowings fit into the evolution of the Germanic stop system from the model of PIE presented in (23). A solution is given in successive diachronic stages presented from left to right in (24). An asterisk marks the suggested earliest form of the innovated segment that became Proto-Germanic /p/. Here and in the discussion of the other branches later, parentheses have their standard value of rendering what they enclose optional or uncertain; in (24) for example they place the voicing of various segments in question, to be discussed further below. Finally, lowercase Roman numerals in this and subsequent tables and discussion refer to diachronic stages, while the uppercase Roman numerals continue to distinguish the contrastive stop series.

| | PIE | (i) | (ii) | (iii) | (iv) |
|-----|----------------------|---------|--------------|----------------|------|
| Ι | Т | T^{h} | | | |
| II | D | Ţ | | | D/Ţ |
| | 6 ¹⁴ | | | b | |
| III | | | (ồ) * | р | |
| | ɗ | | (Å) | t | |
| | g | | (ĝ) | k | |
| | $\widehat{gb}^{(w)}$ | | (ĝ)w | k ^w | |

(24) Evolution of the Pre-Germanic stop system

At the first stage of development (i), series I was phonetically aspirated and II had become slack (on the notation, see §2.5.6). These changes made it possible for the nonbilabial members of III to lose non-explosion without merging with any of the other series, yielding stage (ii). The parenthesized voiceless diacritic indicates that III may have devoiced in certain contexts but the details are not reconstructed.

It is at stage (ii) that the loanwords in question were borrowed. Plain voiced stops in the source forms were considered phonetically most similar to series III, so they were borrowed into that series, devoicing in whatever contexts that series was voiceless. Plain voiced bilabials in the source forms were adapted with the non-place characteristics of the non-bilabial members of the series – that is, explosive and perhaps devoiced under some conditions. This type of bilabial is marked with an asterisk in (24). The inherited series III bilabial, which was uniquely still non-explosive, now contrasted with an explosive plain voiced bilabial and was thus reanalyzed as occupying its own series.

¹⁴ Originally part of series III; becomes a separate series at stage (ii).

After this, series III completely devoiced while the bilabial non-explosive became explosive, producing stage (iii). Alternatively, the bilabial non-explosive merged with series II more quickly. At any rate it had undergone this merger by stage (iv), which approximately corresponds to Proto-Germanic. By this time, series I may also have spirantized. There are two possibilities for how series II evolved to reach this stage. One is that the series lost breathy offset, voicing intervocalically like the allophones of slack stops in some languages (§2.5.5.3) and voicing at least partially in word-initial presonorant position, a pattern that again has typological parallels (3.2.3). This option essentially results in the conventionally reconstructed Proto-Germanic system. As an alternative, II may have remained slack from stage (i) all the way through Proto-Germanic itself. The other sound changes mentioned in these scenarios, including labial delay, deaspiration, devoicing, loss of non-explosion, and spirantization were discussed earlier (§6.3, 6.5.2.1).

At stage (ii), just after the borrowings that we have discussed had taken place, the Pre-Germanic stop system would have been as follows:

(25) Pre-Germanic stop system at stage (ii)

| Ι | p^{h} | t^{h} | \mathbf{k}^{h} | \mathbf{k}^{wh} |
|-----|--------------|------------------------|---------------------------|----------------------------|
| II | ÿ | ţ | ķ | <u>k</u> ^w |
| III | (b) | $(\overset{\circ}{Q})$ | $(\mathbf{\mathring{g}})$ | $({\bf \mathring{g}})$ |
| IV | 6 | | | |

If the third series in this inventory was underlyingly voiceless /T/ and just phonetically voiced in some environments (such as postnasal), then the system is like that of Xhosa (§2.5.5.2.2). On the other hand, if the third series had a stronger tendency to show variable amounts of closure voicing in general, then the first three series might

form a system like Sivas Armenian or Qingtian Wu (§2.5.5.4). Either way, this analysis of the evolution of the Germanic stop system has typological parallels at each of its stages, including the key stage when the borrowings that helped create eventual Proto-Germanic /p/ occurred.

6.5.2.4 Italic

In Italic, series II spirantized and initially devoiced, while the non-bilabial members of III became explosive. Secondary innovations then introduced contrastive tokens of [b] (seen mostly in the various [b] lexemes without regular PIE etyma), isolating the non-explosive structurally into a series of its own. Subsequently, the non-explosive underwent articulatory undershoot, ending up with the stricture of a bilabial fricative [β]. A similar but more drastic change in a bilabial non-explosive's stricture, resulting in a glide, has been reported elsewhere (Greenberg 1970), and glide/fricative variation is well attested for bilabials (e.g. Iverson and Salmons 1996, Maddieson 1984) and at other places (Cser 2003: 122, Howe 2003: 163, Martínez-Celdrán 2008, Vaux and Miller 2011). The Italic reflex never decreased stricture to the point of being a glide, however; instead, it merged with the by now identical reflex of series II, stabilizing as a fricative and devoicing initially.

This brings us to an early stage of Italic contrasting voiceless stops, voiced stops, and fricatives that are voiceless initially and voiced in inter-voiced environments, reflecting the revised understanding of Italic developed by Stuart-Smith (2004). This is represented in stage (iii) in (26) below. A general stopping of medial voiced fricatives and the reduction of the voiceless non-sibilant fricatives to /f h/ bring us to the Latin system, at stage (iv). As in (24), the innovative /b/ at stage (ii) is asterisked.

(26) Evolution of the PIE stop system in Italic

| | PIE | (i) | (ii) | (iii) | (iv) |
|-----|----------------------|---------|--------------------|---------|-------------|
| Ι | Т | | | | |
| II | D | Θ-, -Đ- | | | f-, h-, -D- |
| III | 6 | | (β) | φ-, -β- | f-, -b- |
| | | | b* | | |
| | ď | | d | | |
| | g | | 9 | | |
| | $\widehat{gb}^{(w)}$ | | g ^w , w | | |

It should be observed that apart from articulatory undershoot of the lone nonexplosive, everything in this hypothesis is already posited in the non-explosive model (§6.3).

6.5.2.5 Indic

Unlike Latin and Greek (below), Indic eschewed devoicing in series II thanks to having previously phonemicized a separate series of voiceless aspirates. With these in place, II developed a voiced realization with breathy offset, while III lost non-explosion. For the aerodynamic reasons already discussed, however (§6.5.2.1), the series II bilabial had weaker breathiness than the other members of that series, as it still does in Hindi (Kobayashi 1999: 2). This made it more easily confusable with the bilabial of series III, and the two segments merged. The outcome of the merger was unique, however. Due to the robust stabilization in Indic of the Grassmann's Law pattern dictating that a stop be unaspirated when the next stop in the word was an aspirate (see standard sources in §6.2.1), I posit that the merged bilabial was analyzed as /b/ when the next stop in the word was an aspirate and as /b^ĥ/ otherwise. Subsequently, /b^ĥ/ may have developed a more saliently aspirated realization, and it is clear that a limited number of other /b/ tokens were innovated, partly through loanwords.

This analysis not only exploits the Grassmann pattern so characteristic of Old Indic, it also reflects the attested lexical distribution and frequency of /b/ and /b⁶/ in Sanskrit (see Whitney 1896 §50). While one could expect /b/ to be more common than /b⁶/ in any given language on the assumption that /b/ has a simpler specification and that it occurs in more languages than /b⁶/, this expectation is decisively disappointed by Sanskrit, where /b/ is in the minority and my impression of the lexicon (e.g. Benfey 1866) is that most tokens of /b/ with Indo-European etymologies are in the Grassmann environment (i.e. with an aspirate following). The expectation would also be theoretically unfounded. The number of languages in which a sound occurs and its textual or lexical frequency within a particular language do not necessarily correlate (Croft 1993), and cases of two similar phones neutralizing to the phonetically more complex one "appear to be quite common" according to Hale (2007: 161-162), who discusses /ē ō/, /ei ou/ neutralizing to [er ou] and initial /r, /wr/ neutralizing to [wi] in the history of English.

6.5.2.6 Greek

In Greek, series II became slack, subsequently aspirating and devoicing, while III became explosive. Due to labial delay, the series II bilabial lagged behind the rest of the series in these changes. This effectively left a $/p^h/$ gap, which is well attested (e.g. Maddieson 1984). Labial delay may also have kept the series III bilabial non-explosive at this stage. The labial-velar retained the tendency for oral rarefaction induced by its double place articulation (§6.2), since this was independent of the non-explosion strategy being lost in the rest of the series. This yielded stage (i) in (27) below.

(27) Evolution of the Greek stops¹⁵

| | PIE | (i) | (ii) | (iii) | (iv) |
|-----|----------------------|-----------------|------|----------|---------------------------------|
| Ι | p t k | | | | |
| | k ^w | | | | p, t |
| II | b | | | | p ^h |
| | d | t ^h | | | |
| | g | k ^h | | | |
| | g ^w | k ^{wh} | | | p ^h , t ^h |
| III | 6 | | b | | p ^h |
| | ď | d | | | |
| | g | g | | | |
| | $\widehat{gb}^{(w)}$ | | | /6/ [gb] | b, d |

The lone non-explosive /6/ then lost non-explosion and merged with /b/, yielding stage (ii). This phoneme subsequently became slack on its way to devoicing and aspiration in stage (iv), since it was still a target of the the trend that had already produced those changes in the rest of original series II.

One of the environments where this trend was likely to reach /b/ soonest was before /u/, and this may have already occurred at stage (ii). The high stricture of /u/ would narrow the air passage and naturally increase air pressure slightly (cf. aspiration before the highest vowels in Bantu: Hyman 2003, Mathangwane 1999). The complexity of transitioning from the lip compression of the stop to the protruded lip rounding of the

¹⁵ This abstracts away from certain other developments, including palatalizations induced by PIE y and the merger of labiovelars with velars next to $*\tilde{u}$ or *w fairly early in Greek.

vowel would additionally support an extra narrow labial aperture until the transition to a protruded articulation was completed, further facilitating pressure increase. This plus any inherited breathiness would naturally favor aspiration and attendant devoicing in this environment, more than perhaps any other. It is possible that some or all of these factors caused a phonetic difference between /b/ in this environment and in other environments, which could have contributed to the Mycenaean Greek Linear B syllabary (Thompson 2010) adopting an optional sign for the bilabial aspirate before u while aspirates did not have distinct signs in any other circumstances.

While the Mycenaean syllabary did distinguish the voiced coronal stop from the voiceless unaspirated and aspirated coronal stops, it did not otherwise distinguish voicing, so it leaves us in the dark concerning the laryngeal features of the bilabial series. Mycenaean may correspond to stage (ii), but the data is equally consistent with alternative interpretations including even the inherited series II and III bilabials remaining distinct in Mycenaean. Mycenaean preserved a distinct place value for the labiovelars and may have developed the series III labial-velar to $[g^w]$, the value traditionally assumed.

In the other dialects, the velar component of the labial-velar was phonologically reanalyzed as simply a device for avoiding oral air pressure buildup. Hence the labial-velar adopted the phonological status of /6/. This development appears to be typologically supported by the extensive diachronic interplay between labial-velar stops and bilabial non-explosives mentioned earlier (§6.2); as just one example, a single Igbo phoneme is realized as [\hat{gb}] in some dialects, [6] in others (Ikekeonwu 1999: 109). While the non-Mycenaean dialects may have reanalyzed the labial-velar simultaneously with the other developments that produced stage (ii), the result is presented as stage (iii) for clarity. The remaining developments concern non-Mycenaean dialects only.

The new $\frac{6}{[gb]}$ subsequently shifted its realization in most environments to [6] and then [b] (both attested changes, cf. Greenberg 1970: 136-37). At the same time or

slightly earlier, /b/ devoiced and aspirated, finishing the process that had already encompassed the rest of series II and filling the labial gap in the aspirate series. This brings us to stage (iv). One complication is that before \acute{e} in some dialects, /6/ [\widehat{gb}] shifted to /d/ instead. The other labiovelars developed similarly: $k^w > t / \{\acute{e}, \acute{t}\}$, else > p; $k^{wh} > t^h$ / $_{\acute{e}}$, else $> p^h$, though Allen (1957a) notes that the evidence regarding the aspirate before \acute{t} is tenuous. Allen (1957a) proposes that [w] > [q] / $_{\acute{e}}$, triggering palatalization of the preceding stop, followed by loss of the labial component and simplification of the remainder to a plain coronal. Within the present model, I suggest that analogy with the voiceless labiovelars induced the voiced labial-velar to palatalize similarly to them; that is, given the rise of a palatal gesture coarticulated with the voiceless labiovelars, one developed in the voiced labial-velar as well, regardless of the fact that the labial gesture in the former was rounded and vocalic while in the latter it was compressed and stopped. On possible reasons for the inconsistent role of \acute{t} in triggering palatalization, see Allen (1957a).

6.5.2.7 Conclusion

The proposal that the original voiced bilabial stops merged catches a free ride on wider mergers in seven of the daughter branches and hinges on the independently supported phenomenon of labial delay in the remaining five. In Italic, the merger additionally involves a stricture reduction that is already reconstructed for series II; in Indic, Grassmann's Law was involved, and in non-Mycenaean Greek, the loss of the labiovelars. Thus the proposed solution to the "*b* gap" problem is almost entirely driven by mechanisms already needed to derive the daughter languages from the non-explosive model of PIE, while all of the alternative solutions considered in §6.5.1 either proved considerably less parsimonious or were problematic in other ways. This adds to the other, broader advantages of the non-explosive reconstruction identified in §6.4.

6.6. Alleged voiceless aspirate etyma

Earlier (§6.2), I asserted that contrastive PIE voiceless aspirates are not justified by the comparative evidence. Against this claim, Elbourne (2000, 1998) lists twenty-five annotated etymologies. Other scholars have generally not embraced his argument, though it has been cursorily alluded to as a possible solution to the standard three-series reconstruction (e.g. Stuart-Smith 2004: 5). Here I briefly summarize Elbourne's etyma, citing them only according to his numbering (some are further discussed in de Decker 2010). I conclude that even if the majority of his list were accepted apart from the question of aspiration, they still present no convincing case for PIE /T^h/.

Roots where voiceless aspiration is solely evidenced in Sanskrit and *a priori* might thus be a Sanskrit innovation (nos. 7, 8, 13, 14, 18) make up a fifth of the list. There is no reason to suppose that innovated aspiration cannot spread fairly irregularly by lexical diffusion; other languages which appear to have innovated most or all of their aspirate phonemes include Limbu (Sino-Tibetan: Mielke 2007a) and Kusunda (Nepali isolate: Watters 2006).

Roots where *s*- precedes the stop make up another third of the list (nos. 3, 5, 11, 12, 22-25). A PIE in which T^h is disproportionately attested with a preceding *s* would be an odd one; by their very frequency in the list (which increases percentage-wise if one ends up suspending any of Elbourne's other etymologies), the *s*-cases weigh against reconstructing a distributionally normal, fully phonemic T^h series in PIE.

One root (1 'mussel') has only Sanskrit and Greek cognates. One root (2 'laugh') favors an onomatopoeic interpretation. Three roots (nos. 19, 21, 23) have independent evidence for "laryngeals" in the right position to trigger aspiration (see the discussion of Indo-Iranian in §6.3). Some of the remaining seven roots are clearly ambiguous for reasons that space prevents me from giving here; laryngeals could be involved with members of this set too.

6.7. Conclusion

The giant of early historical linguistics, Ferdinand de Saussure, is famous for the dictum: "dans la langue il n'y a que des différences" (1972: 166) – "in language there are only distinctions." Mayrhofer (1986: 98) in his monumentally meticulous work on PIE phonology echoes similar sentiments when he emphasizes that one need not choose between different reconstructions if they differ only in subsegmental details, because "die Gleichungen bleiben dieselben... Zahl und Oppositionsverhältnis der Phoneme ändern sich nicht" – "the correspondences remain the same... the number of phonemes and their contrastive relationships do not alter" (similarly Penney 1998: 154). Speculation on proto segmental content has never been absolutely excluded by Indo-Europeanists, but as seen in quotations like these, there has been an understandible reluctance to expect much fruit from the exercise (cf. Kaye 1997: 529). The tracking of segmental oppositions over time is relatively well understood, but how could proto content ever be realistically reconstructed?

Understood purely in terms of abstract segmental oppositions, the synchronic phonology of a proto language is no more than a tally of algebraic symbols representing proto segments, a set of symbol groupings representing proto phonological classes, and occasionally a few rules where classes interact. There is nothing typologically unusual about the abstract tally, map, or rules that one would uncover for PIE by studying only abstract segmental oppositions; many other languages have about the same number of segments and at least the same number of classes and rules. But viewed at the subsegmental level, PIE presents problems that have perplexed historical linguists for nearly as long as the discipline has existed. It is no surprise that intuitive concern for the phonetic content of phonemic categories has long been treated as an important factor in the reconstruction process (Fox 1995: 81). Only the development of phonetically informed theories of subsegmental phonological categories (feature theories) in the last several decades has made possible a formal and systematic, rather than intuitive and fundamentally inconclusive, approach to these problems.

By feature theory I refer here not simply to the study of abstract contrast specifiers in particular languages, but to the wider study of phonetic cues and the various phonological roles they play. Feature theory serves as a formal way to investigate how phonetic properties associate with phonological categories; these associations are not random and unconstrained, any more than indiscriminate groups of segments can make phonological inventories. Feature co-occurrence trends have proven highly restricted by articulatory and other performance-related factors. Knowledge of these restrictions provides guidance as we explore not only the patterns of synchronic sound systems but also the evolution of sound systems over time – the paths of sound change (Stuart-Smith 2004: 1-7, cf. Anttila 1971: 649).

From the investigation in this chapter and the previous ones, it appears that the long-standing difficulty in reconstructing the PIE stop system has been due to the comparatively high number of relevant restrictions on feature patterns which were previously little understood. Antagonisms between voice and oral air pressure buildup and between voice and vocal fold abduction, particularly in stops with their unique aerodynamics, make possible asymmetric consonant systems that are highly sensitive, meaning that relatively few associations between phonetic correlates and phonemic categories can be changed without unacceptably disturbing the rest of the system. Until these repulsions and attractions are identified at the subsegmental level, attempts to characterize such a system are unhelpfully coarse and are likely to make very limited sense. But a systematic typology of possible reconstructions using a cross-linguistically informed understanding of how phonological categories are mapped to phonetic correlates radically changes the situation, boosts the process of elimination, and in the

267

case of PIE reveals that the space of viable solutions is remarkably small. Specifically, a plausible reconstruction of PIE seems to be found in a system with three stop series contrasting in voice and (non-)explosion, with the subsequent merger of /b/ and /6/ in five of the daughter branches and more general mergers of /D/ and /D[<]/ in the other branches effacing the traces of original /6/.

7 Conclusion

We have explored in detail how phonetic correlates map to feature categories, how such correlate bundles interact in commonly reported feature co-occurrence trends, and the implications of these patterns for grammar design and sound change. Rather than deriving these highly specific patterns from abstract structure hardwired in cognition, this work has identified plausible non-cognitive sources for them. Consequently it seems that innate cognitive stipulations for these patterns are redundant and that the highly specific content which OT's universal constraint set aims to capture can be derived more efficiently and with arguably greater neural plausibility in a model where learners construct grammatical representations of phonological patterns on a language-specific basis. A corollary is that sound change generally involves reanalysis of phonetic data rather than altering the organization of any cognitively inherited phonological categories – a fact important for the methodology of diachronic reconstruction, as we saw in a fresh assessment of one of the oldest unsolved problems in historical linguistics, the evolution of laryngeal distinctions in the Indo-European stops.

This work contributes to a growing body of research focused on removing innate/learnable duplication problems. While most language-specific phonological information seems best removed from the innate endowment, I have not concluded that phonetic content and phonological cognition never interact directly. In each of the three main dimensions of this work – developing phonological profiles of specific sound types, exploring implications for grammar design, and historical problem solving – I have argued that cognition maintains extensive awareness of the phonetic values of the variables it manipulates.

Bibliography

This bibliography includes some sources that are not cited in the text but that were consulted while drafting it. Online articles were taken from academic sites as a rule. Due to access limitations, it was convenient in eight percent of the entries to cite book passages from Google Books. Online articles with dates exclusively in square brackets were dated using the Internet Archive (http://www.archive.org/web/web.php) because I did not find explicit dates within the files. The reader is invited to report link rot to brett.a.miller@colorado.edu.

- Aaronson, Doris and Robert W. Rieber (eds.) (1975) *Developmental Psycholinguistics and Communication Disorders*, Part IV: The Relationship between Linguistic Competence and Linguistic Performance, *Annals of the New York Academy of Sciences* vol. 263, 5-287.
- Adjarian, Hratchia (1899) 'Les explosives de l'ancien arménien étudiées dans les dialectes modernes', *La Parole: Revue internationale de rhinologie, otologie et laryngologie* 1, 119-127, <u>http://gallica.bnf.fr/ark:/12148/bpt6k57626142/f121.image</u> (retrieved 5 January 2012).
- Ahn, Sang-Cheol and Gregory K. Iverson (2004) 'Dimensions in Korean Laryngeal Phonology', *Journal of East Asian Linguistics* 13, 345-379, <u>http://www.uwm.edu/~iverson/dimenskor.pdf</u> (MS dated 2003) (retrieved 29 July 2008).
- Albright, Adam and Hayes, Bruce P. (2006) 'Modeling productivity with the Gradual Learning Algorithm: the problem of accidentally exceptionless generalizations', in Gisbert Fanselow, Caroline Fery, Matthias Schlesewsky and Ralf Vogel (eds.) *Gradience in Grammar: Generative Perspectives*, Oxford: Oxford University Press, preprint version

http://www.linguistics.ucla.edu/people/hayes/papers/ModelingProductivityWithGLA AlbrightHayes.pdf (retrieved 5 February 2009).

Albright, Adam and Bruce P. Hayes (2011) 'Learning and learnability in phonology', in John Goldsmith, Jason Riggle, and Alan C. L. Yu (eds.) *The Handbook of Phonological Theory*, second edition, Oxford: Wiley-Blackwell, <u>http://www.linguistics.ucla.edu/people/hayes/papers/AlbrightHayesLearningAndLearn</u> abilityInPhonology.pdf (retrieved 18 February 2012).

- Alderete, John (2001) 'Dominance effects as transderivational anti-faithfulness', *Phonology* 18, 201-253, <u>http://www.ai.mit.edu/projects/dm/featgeom/alderete-antifaith.pdf</u> (retrieved 19 June 2010).
- Alderete, John, Adrian Brasoveanu, Nazarré Merchant, Alan Prince, and Bruce Tesar (2005) 'Contrast analysis aids in the learning of phonological underlying forms', in

The Proceedings of WCCFL 24, 34-42,

http://www.lingref.com/cpp/wccfl/24/paper1204.pdf (retrieved 29 August 2009).

- Allen, W. Sidney (1950) 'Notes on the phonetics of an Eastern Armenian speaker', *Transactions of the Philological Society* 49, 180-206.
- Allen, W. Sidney (1955) Review in *Bulletin of the School of Oriental and African Studies* 17/1, 191 of Isidore Dyen (1953) *The Proto-Malayo-Polynesian Laryngeals*.
- Allen, W. Sidney (1957a) 'Some problems of palatalization in Greek', Lingua 7, 113-33.
- Allen, W. Sidney (1957b) 'Aspiration and the Harauti Nominal', *Studies in Linguistic Analysis*, special volume of the Philological Society, 68-86.
- Allen, W. Sidney (1973) '*Khthön, ruki* and related matters', *Transactions of the Philological Society* 72, 98-126.
- Allen, W. Sidney (1978[1965]) Vox Latina. A Guide to the Pronunciation of Classical Latin, Cambridge: Cambridge University Press.
- Allen, W. Sidney (1995) 'On "tenseness" in Modern Icelandic', *Transactions of the Philological Society* 93/1, 1-16.
- Alwan, Abeer, Shrikanth Narayanan, and Katherine Haker (1997) 'Toward articulatoryacoustics models for liquid approximants based on MRI and EPG data. Part II. The rhotics', *Journal of the Acoustical Society of America* 101/2, 1078-1089.
- Andrews, J. Richard (2003) *Introduction to Classical Nahuatl*, revised edition, Norman, Oklahoma: University of Oklahoma Press.
- Anttila, Arto (2007) 'Variation and optionality', in de Lacy (2007), 519-536.
- Anttila, Raimo (1971) 'Phonetics in reconstruction and phonetic reconstruction', in André Rigault and René Charbonneau (eds.) *Proceedings of the Seventh International Congress of Phonetic Sciences*, University of Montreal and McGill University, 22-28 Aug 1971, The Hague: Mouton, 649-651.
- Arbeitman, Yoël L. and Allan R. Bomhard (eds.) (1981) *Bono homini donum: Essays in Historical Linguistics in Memory of J. Alexander Kerns* (vols. 1 and 2), Amsterdam: John Benjamins.
- Archangeli, Diana (1984) Underspecification in Yawelmani Phonology and Morphology, doctoral dissertation, MIT.
- Archangeli, Diana (1997) Optimality Theory: An Overview.
- Archangeli, Diana and Douglas Pulleyblank (1994) *Grounded Phonology*, Cambridge, Massachusetts: MIT Press.
- Armagost, James L. and Wick Miller (2000) 'Are voiceless vowels phonemic in Comanche?' in Casad and Willett (2000), 57-82.
- Aron, Adam (2007) 'The neural basis of inhibition in cognitive control', MS, <u>http://www.aronlab.org/Pubs/aron2007ns.pdf</u> (retrieved 18 June 2010).
- Aronson, Howard I. (1997) 'Georgian phonology', in Kaye (1997), 929-940.
- Avery, Peter and Idsardi, William J. (2001) 'Laryngeal dimensions, completion and enhancement', in Hall (2001), 41-70.

- Bach, E. and R. T. Harms (1972). 'How do languages get crazy rules?' in R. P. Stockwell and R. K. S. Macaulay (eds.) *Linguistic change and generative theory*, Bloomington: Indiana University Press, 1-21.
- Baer, Thomas, P. Alfonso, and K. Honda (1988) 'Electromyography of the tongue muscle during vowels in /əpVp/ environment', *Annual Bulletin of the Research Institute of Logopedics and Phoniatrics* 22, 7-20, <u>http://www.umin.ac.jp/memorial/rilp-tokyo/R22/R22_007.pdf</u> (retrieved 24 February 2012).
- Baer, Thomas, Thomas Gay, and Seiji Niimi (1976) 'Control of fundamental frequency, intensity, and register of phonation', Haskins Laboratories Status Report on Speech Research SR-45/46, 175-185, <u>http://www.haskins.yale.edu/sr/SR045/SR045_11.pdf</u> (retrieved 13 January 2012).
- Baer, Thomas, Anders Löfqvist, and Nancy S. McGarr (1983) 'Laryngeal vibrations: a comparison between high-speed filming and glottographic techniques', Haskins Laboratories Status Report on Speech Research SR-73, 283-291, http://www.haskins.yale.edu/sr/sr073/SR073 15.pdf (retrieved 20 February 2012).
- Baković, Eric (2006) 'A revised typology of opaque generalizations', appeared in *Phonology* 24/2, 217-259, <u>http://roa.rutgers.edu/view.php3?roa=850</u> (retrieved 27 August 2009).
- Baković, Eric (forthcoming) 'Opacity and ordering', to appear in J. Goldsmith, A. Yu, and J. Riggle (eds.) *The Handbook of Phonological Theory*, second edition, prepublication version, <u>http://ling.auf.net/lingBuzz/000926?referer=http%3A%2F</u> %2Fworks.bepress.com%2Febakovic%2F46%2F (retrieved 23 March 2012).
- Baldi, Philip and Ruth Johnston-Staver (1989) 'Historical Italic phonology in typological perspective', in Vennemann (1989), 85-101.
- Ball, Martin J. and James Fife (1993) The Celtic Languages.
- Ball, Martin J. and Nichole Müller (1992) Mutation in Welsh, New York : Routledge.
- Ballard, Elaine and Donna Starks (2004) 'Liquids: laterals and rhotics or much more?' *Proceedings of the 2004 Conference of the Australian Linguistic Society*, <u>http://ses.library.usyd.edu.au/bitstream/2123/102/1/ALS-20050630-EB.pdf</u> (retrieved 30 July 2008).
- Barrack, Charles M. (2002) 'The Glottalic Theory revisited: a negative appraisal,' *Indogermanische Forschungen* 107, 76-98.
- Barrack, Charles M. (2003a) 'The Glottalic Theory revisited: a negative appraisal. Part II; The typological fallacy underlying the Glottalic Theory,' *Indogermanische Forschungen* 108, 1-16.
- Barrack, Charles M. (2003b) 'Kortlandt's glottalic account of Proto-Indo-European: a response,' *Interdisciplinary Journal for Germanic Linguistics and Semiotic Analysis* 8/2, 145-166.
- Barrack, Charles M. (2004) 'Labial gap in Proto-Indo-European', MS, University of Washington.

Barrack, Charles M. (2005) 'The labial gap in Proto-Indo-European', *Interdisciplinary Journal for Germanic Linguistics and Semiotic Analysis* 10/2, 231-241, <u>http://chomikuj.pl/grzegorj/Dokumenty/Lingwistyka/Barrack5,1129405850.pdf</u> (retrieved 3 January 2012).

Battistella, L. (1990) Markedness: The Evaluative Super-structure of Language, Albany.

- Beatty, John Joseph (1974) *Mohawk Morphology*, Greeley: University of Northern Colorado, Museum of Anthropology.
- Becker, Michael, Nihan Ketrez, and Andrew Nevins (in preparation) 'The surfeit of the stimulus: analytic biases filter lexical statistics in Turkish voicing deneutralization', <u>http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-publications/beckerEtA1_08_The-Surfeit.2</u> (retrieved 25 September 2010).

Becker, Michael and Katherine Flack Potts (2011) 'The emergence of the unmarked', ch. 58 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.)

The Blackwell Companion to Phonology, Oxford: Wiley-Blackwell.

- Beckwith, Charles (2007) 'On the Proto-Indo-European obstruent system', *Historische Sprachforschung* 120, 1-19.
- Beddor, Patrice Speeter and Chutamanee Onsuwan (2003) 'Perception of prenasalized stops' [in Ikalanga], presentation, 15th International Congress of Phonetic Sciences, Barcelona 2003, 1-4, <u>http://www-</u>

personal.umich.edu/~beddor/pubfiles/Beddor_Onsuwan_2003.pdf (retrieved 13 February 2012).

- Beekes, Robert S. P. (1995) *Comparative Indo-European Linguistics*, Amsterdam: John Benjamins.
- Bell-Berti, Fredericka (1975) 'Control of pharyngeal cavity size for English voiced and voiceless stops', *Journal of the Acoustical Society of America* 57, 456-461.
- Bell-Berti, Fredericka and Hajime Hirose (1975) 'Velar activity in voicing distinctions: a simultaneous fiberoptic and electromyographic study', *Journal of Phonetics* 3, 69-74.
- Bellem, Alex (2005) "'Pharyngeal" and "pharyngealization" in the Caucasian languages: phonological correlates vs. phonetic realizations', paper presented at the 13th Manchester Phonology Meeting; abstract

http://salford.academia.edu/AlexBellem/Papers/562046/Pharyngealand_Pharyngealisa tionin_the_Caucasian_Languages_Phonological_Correlates_vs_Phonetic_Realisations (retrieved 24 February 2012).

- Bellem, Alex (2007) *Towards a Comparative Typology of Emphatics*, PhD dissertation, University of London.
- Bellem, Alex (2009) 'The "problem" of pharyngealization and its role in the sound systems of North-East Caucasian languages', poster presented at the International Workshop on Pharyngeals and Pharyngealization, Newcastle; abstract <u>http://www.ncl.ac.uk/linguistics/assets/documents/AlexBellem.pdf</u> (retrieved 24 February 2012).
- Benfey, Theodore (1866) *A Sanskrit-English Dictionary*, London: Longmans, Green, and Co, Google Books download.

- Berko, Jean (1958) 'The child's learning of English morphology', *Word* 14, 150-177, <u>http://childes.psy.cmu.edu/topics/wugs/wugs.pdf</u> (retrieved 27 August 2011).
- Bermúdez-Otero, Ricardo (2003) 'The acquisition of phonological opacity', MS <u>http://roa.rutgers.edu/files/593-0403/593-0403-BERMUDEZ-OTERO-0-0.PDF</u> (retrieved 10 July 2010).
- Bermúdez-Otero, Ricardo (2005) 'Phonological change in Optimality Theory', in Keith Brown (ed.) *Encyclopedia of Language and Linguistics*, second edition, Oxford: Elsevier, pre-publication draft http://www.kul.pl/files/165/monograf/reading/changeinoptimality.pdf (retrieved 21

http://www.kul.pl/files/165/monograf/reading/changeinoptimality.pdf (retrieved 21 February 2012).

- Bermúdez-Otero, Ricardo (2007) 'Diachronic phonology', in de Lacy (2007), 497-518.
- Bermúdez-Otero, Ricardo (2011) 'Cyclicity', ch. 85 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *Companion to Phonology*, Oxford: Wiley-Blackwell.
- Bernsten, Jan (1996) *Runyakitara: Uganda's 'New Language'*, MS, Michigan State University.
- Berry, George Ricker (1901) *The Classic Greek Dictionary*, New York: Hinds and Noble.
- Berry, David A. (2001) 'Mechanisms of modal and nonmodal phonation', *Journal of Phonetics* 29, 431-450, <u>http://www.phonetik.uni-muenchen.de/~hoole/kurse/hslarynx/berry_jphon_29_4_2001.pdf</u> (retrieved 31 January 2012).
- Bever, Thomas Gordon (1961) *Leonard Bloomfield and the Phonology of the Menomini Language*, PhD dissertation, MIT,
- http://www.ai.mit.edu/projects/dm/theses/bever67.pdf (retrieved 16 March 2012).
- Bhatia, Tej K. (1974) 'A study of aspirated consonants as spoken and recognized by Hindi speakers', *Studies in the Linguistic Sciences* 4/2, 25-32, <u>http://www.ideals.illinois.edu/bitstream/handle/2142/26477/SLS1974v4-2_bhatia.pdf?</u> <u>sequence=2</u> (retrieved 8 February 2012).
- Bickel, Balthasar (2003) 'Belhare', ch. 34 in Graham Thurgood and Randy J. LaPolla (eds.) *The Sino-Tibetan Languages*, London: Routledge.
- Bickley, C. A. and Kenneth Stevens (1986) 'Effect of a vocal-tract constriction on the glottal source; experimental and modeling studies', *Journal of Phonetics* 14, 373-382.
- Billerey-Mosier, Roger (2003) 'An acoustic analysis of Kiowa stops', MS, UCLA, <u>http://www.linguistics.ucla.edu/people/grads/billerey/papers/RBM_Kiowa_stops.pdf</u> (retrieved 29 July 2008).
- Bird, Sonya (2004) 'Timing differences in St'át'imcets glottalized resonants: linguistic or bio-mechanical?' *Proceedings of the 10th Australian International Conference on Speech Sciences and Technology*, 328-333,

http://www.assta.org/sst/2004/proceedings/papers/sst2004-322.pdf (retrieved 19 June 2010).

- Bird, Steven (1995) *Computational Phonology: A Constraint-Based Approach*, Cambridge: Cambridge University Press.
- Blaho, Sylvia (2008) *The Syntax of Phonology*, PhD dissertation, University of Tromsø, <u>http://nytud.academia.edu/SylviaBlaho/Papers/911595/The_syntax_of_phonology_A_radically_substance-free_approach</u> (retrieved 16 February 2012).
- Blankenship, Barbara (2002) 'The timing of nonmodal phonation in vowels', *Journal of Phonetics* 30, 163-191.
- Blauert, Jens (1997) *Spatial Hearing: The Psychophysics of Human Sound Localization*, Cambridge, Massachusetts: MIT Press.
- Blevins, Juliette (1993) 'Klamath laryngeal phonology', International Journal of American Linguistics 59, 237-279.
- Blevins, Juliette (1997) 'Rules in Optimality Theory: two case studies', in Igy Roca (ed.) *Derivations and Constraints in Phonology*, Oxford: Oxford University Press, 227-260.
- Blevins, Juliette (2001) 'Klamath /s'/ clusters', *International Journal of American Linguistics* 67, 467-474.
- Blevins, Juliette (2003) 'One case of contrast evolution in the Yurok vowel system', *International Journal of American Linguistics* 69/2, 135-150, http://www.journals.uchicago.edu/doi/abs/10.1086/379681 (retrieved 29 July 2008).
- Blevins, Juliette (2004) Evolutionary Phonology: The Emergence of Sound Patterns.
- Blevins, Juliette (2006a) 'Phonetically-based sound patterns: typological tendencies or phonological universals?' MS,

http://web.archive.org/web/20110114061555/http://email.eva.mpg.de/~blevins/pdf/lab phon10.pdf (retrieved 25 February 2012).

Blevins, Juliette (2006b) 'A theoretical synopsis of evolutionary phonology', *Theoretical Linguistics* 32/2, 117-166,

http://washo.uchicago.edu/seminar/Blevins2006_TL.pdf (retrieved 4 February 2012).

- Blevins, Juliette (2007) 'A theoretical synopsis of evolutionary phonology', MS, Max Planck Institute for Evolutionary Anthropology, <u>http://ebookbrowse.com/blevins-2007-evolutionary-phonology-target-pdf-d49810032</u> (retrieved 6 February 2012).
- Blevins, Juliette (2008) 'Natural and unnatural sound patterns: a pocket field guide', in Klaas Willems and Ludovic De Cuypere (eds.) *Naturalness and Iconicity in Language*, Amsterdam: John Benjamins, 121-148; citations from draft version http://email.eva.mpg.de/~blevins/pdf/webpub2008c.pdf (retrieved 16 Sept. 2010).
- Blevins, Juliette (2009) 'Phonetic explanations for recurrent sound patterns: diachronic or synchronic?' in Raimy and Cairns (2009), 325-336.
- Blevins, Juliette and Andrew Garrett (1998) 'The Origins of Consonant-Vowel Metathesis', *Language* 74/3, 508-556.
- Blevins, Juliette and Andrew Garrett (2004) 'The evolution of metathesis', ch. 5 in Hayes et al. (2004), 117-156.
- Blust, Robert A. (1969) 'Some new Proto-Austronesian trisyllables', *Oceanic Linguistics* 8, 85-104.

- Blust, Robert A. (1973) 'A double counter-universal in Kelabit', *Papers in Linguistics* 7, 309-324.
- Blust, Robert A. (1978) *The Proto-Oceanic Palatals*, Memoir (Polynesian Society) no. 43, Wellington.
- Blust, Robert A. (1980) 'More on the origins of glottalic consonants', *Lingua* 52, 125-156.
- Blust, Robert A. (1988) *Austronesian Root Theory: An Essay on the Limits of Morphology*, Studies in Language companion series 19, Amsterdam.
- Blust, Robert A. (1995) 'Notes on Berawan consonant gemination', *Oceanic Linguistics* 34, 123-138.
- Blust, Robert A. (2000) 'Low-vowel fronting in Northern Sarawak', *Oceanic Linguistics* 39/2, 285-319.
- Blust, Robert A. (2001) 'Some remarks on stress, syncope, and gemination in Mussau', *Oceanic Linguistics* 40/1, 143-150.
- Blust, Robert A. (2002) 'Kiput Historical Phonology', *Oceanic Linguistics* 41/2, 384-438.
- Blust, Robert A. (2004) 'Austronesian nasal substitution: a survey', *Oceanic Linguistics* 43/1, 73-148.
- Blust, Robert A. (2006) 'The origin of the Kelabit voiced aspirates: a historical hypothesis revisited', *Oceanic Linguistics* 45/2, 311-338.
- Blust, Robert A. (2010) 'The Greater North Borneo Hypothesis', *Oceanic Linguistics* 49/1, 44-118.
- Blust, Robert A. (2011) 'The problem of doubleting in Austronesian languages', *Oceanic Linguistics* 50/2, 399-457.
- Boeckx, C. (2006) Linguistic Minimalism, Oxford: Oxford University Press.
- Boersma, Paul (1997) *The Elements of Functional Phonology*, MS, University of Amsterdam,

Boersma, Paul and Claartje Levelt (2003) 'Optimality Theory and phonological acquisition', MS,

http://www.fon.hum.uva.nl/paul/papers/OT_phon_acq_ARLA2003.pdf (retrieved 19 June 2010).

Boersma, Paul and Joe Pater (2007) 'Constructing constraints from language data: the case of Canadian English diphthongs', presentation, NELS 38, University of Ottawa, 27 Oct. 2007,

http://www.fon.hum.uva.nl/paul/presentations/CanadianRaisingNELS2007.pdf (retrieved 19 June 2010).

- Boersma, Paul and David Weenink (2010) *PRAAT: Doing Phonetics by Computer*, free software, version 5.1.43, <u>http://www.fon.hum.uva.nl/praat/</u> (retrieved 2010).
- Boisson, Claude (1989a) 'Lettre aux participants de l'Atelier sur l'indo-européen du Congrès de Pavie, 22 juillet 1985', in Vennemann (1989), 259-263.

- Boisson, Claude (1989b) 'Letter to Theo Vennemann, 7 Sept. 1985', in Vennemann (1989).
- Bomhard, Allan R. (1981) 'Implications of "Rhotacism in Hieroglyphic Luwian" in Arbeitman and Bomhard (1981), Part I, 29 (response to Yoël Arbeitman and Gilbert-James Ayala, 'Rhotacism in Hieroglyphic Luwian', same volume, 25-28).
- Bomhard, Allan R. (1981b) 'Indo-European and Afroasiatic' in Arbeitman and Bomhard (1981), Part I, 351-474.
- Bomhard, Allan R. (1988) 'Recent trends in the reconstruction of the Proto-Indo-European consonant system,' *Historische Sprachforschung* 101/1, 2-25.
- Bomhard, Allan R. (2008) 'The Glottalic Theory of Proto-Indo-European consonantism and its implications for Nostratic sound correspondences', MS, <u>http://www.nostratic.ru/books/(217)Nostratic%20Sound%20Correspondences.pdf</u>, <u>http://www.nostratic.ru/index.php?page=books&lang=en</u> (retrieved 16 February 2012).
- Bomhard, Allan R. and John C. Kerns (1994) *The Nostratic Macrofamily: A Study in Distant Linguistic Relationship.*
- Bonet, Eulalia and Joan Mascaró (1997) 'On the representation of contrasting rhotics', in Fernando Martínez-Gil and Alfonso Morales-Front (eds.) *Issues in the Phonology and Morphology of the Major Iberian Languages*, Washington, D.C.: Georgetown University Press, 103-126.
- Bonneau, A., Djezzar, L. and Y. Laprie (1996) 'Perception of the place of articulation of French stop bursts', *Journal of the Acoustic Society of America* 100/1, 555-64, abstract <u>http://www.ncbi.nlm.nih.gov/pubmed/8675848</u> (retrieved 20 July 2010).
- Bonvillain, Nancy (1978) 'Linguistic change in Akwesasne Mohawk: French and English influences', *International Journal of American Linguistics* 44/1, 31-39.
- Bonvillain, Nancy (1985) 'A note on *ne*?', *International Journal of American Linguistics* 51/4, 349-351.
- Botma, Bert (2004) *Phonological Aspects of Nasality: An Element-Based Dependency Approach*, Utrecht: LOT,

http://www.lotpublications.nl/publish/articles/000706/bookpart.pdf (retrieved 13 February 2012).

- Botma, Bert (2011) 'Sonorants', ch. 9 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Bradshaw, Joel (1978) 'The development of an extra series of obstruents in Numbami', *Oceanic Linguistics* 17/1, 39-76.
- Bradshaw, Mary M. (1997) 'A phonology-phonetics mismatch: voice in consonant-tone interaction', *Studies in the Linguistic Sciences* 27/1, 17-31, http://www.ideals.illinois.edu/bitstream/handle/2142/11571/SLS1997v27.1-05Bradshaw.pdf (retrieved 1 December 2011).

- Bradshaw, Mary (1999) *A Cross-Linguistic Study of Consonant-Tone Interaction*, PhD dissertation, The Ohio State University, http://linguistics.osu.edu/files/linguistics/dissertations/Bradshaw1999.pdf (retrieved 18 February 2012).
- Brosnahan, Leonard Francis and Bertil Malmberg (1976) Introduction to Phonetics, Cambridge: Cambridge University Press.
- Browman, Catherine and Louis Goldstein (1988) 'Some notes on syllable structure in articulatory phonology', Haskins Laboratories Status Report on Speech Research SR-93/94, 85-102, <u>http://www.haskins.yale.edu/sr/SR093/SR093_06.pdf</u> (retrieved 23 March 2012).
- Brown, Jason Camy (2008) *Theoretical Aspects of Gitksan Phonology*, PhD thesis, University of British Columbia,

https://circle.ubc.ca/bitstream/handle/2429/7578/ubc_2009_spring_brown_jason_cam y.pdf?sequence=1 (retrieved 4 February 2012).

- Bruening, Benjamin (1999) 'Morphological templates and phonological theory', presentation, Second Mediterranean Meeting on Phonology, Msida, Malta (cited in Green 2004).
- Bruening, Benjamin (2000) 'Morphological templates and phonological opacity', University of Delaware class assignment, http://www.ling.udel.edu/bruening/home/opacity1.pdf (retrieved 5 September 2009),

http://web.archive.org/web/20060308194825/http://www.ling.udel.edu/bruening/home /opacity1.pdf (retrieved 11 March 2012).

- Brunelle, Marc (2005) Register in Eastern Cham: Phonological, Phonetic and Sociolinguistic Approaches, PhD dissertation, Cornell University, <u>http://aix1.uottawa.ca/~mbrunell/Eastern%20Cham%20register.pdf</u> (retrieved 30 July 2008).
- Brunelle, Marc (2008) 'Diglossia, bilingualism, and the revitalization of written Eastern Cham', *Language Documentation and Conservation* 2, 28-46, <u>http://scholarspace.manoa.hawaii.edu/bitstream/handle/10125/1848/brunellesmall.pdf;</u> jsessionid=5828590D00976D0152B82353907B0B0E?sequence=10 (retrieved 16 January 2012).
- Brunelle, Marc (2009) 'Contact-induced change? Register in three Cham dialects', Journal of the Southeast Asian Linguistics Society 2, 1-22, <u>http://jseals.org/JSEALS-2.pdf#page=7</u> (retrieved 24 January 2012).
- Brunelle, Marc (2010) 'The role of larynx height in the Javanese tense ~ lax stop contrast', in Raphael Mercado, Eric Potsdam, and Lisa Demena Travis (eds.) *Austronesian and Theoretical Linguistics*, Amsterdam: John Benjamins, 7-24.
- Brunner, Jana (2005) 'Supralaryngeal mechanisms of the voicing contrast in velars', ZAS Papers in Linguistics 39:1:144 (master's thesis),

http://janabrunner.org/pdfs/brunner_ZASPIL_mag.pdf (retrieved 13 February 2012).

Brunner, Jana, Susanne Fuchs, and Pascal Perrier (2011) 'Supralaryngeal control in Korean velar stops', *Journal of Phonetics* 39/2, 178-195, prepublication draft

http://hal.archives-ouvertes.fr/docs/00/59/07/38/PDF/Brunneretal_final.pdf (retrieved 13 February 2012).

Brunner, Jana, Susanne Fuchs, Pascal Perrier, and Hyeon-Zoo Kim (2003) 'Mechanisms of contrasting Korean velar stops: a catalogue of acoustic and articulatory parameters', *ZAS Papers in Linguistics* 32, 15-30,

http://janabrunner.org/pdfs/brunneretal_zaspil32_2003.pdf (retrieved 13 February 2012).

- Bubeník, Vít (1983) *The Phonological Interpretation of Ancient Greek: A Pandialectal Analysis.* Toronto: University of Toronto Press.
- Buckley, Eugene (1992) 'On the sonority of glottalized sonorants', Linguistic Society of America paper, 9-12 January 1992, Philadelphia, http://babel.ling.upenn.edu/~gene/papers/glott_son.pdf (retrieved 28 July 2008), http://web.archive.org/web/20060830201415/http://babel.ling.upenn.edu/~gene/papers

/glott son.pdf (retrieved 2 December 2011).

- Buckley, Eugene (1994) *Theoretical aspects of Kashaya phonology and morphology*, Stanford: Center for the Study of Language and Information.
- Buckley, Eugene (2000) 'On the naturalness of unnatural rules', *Proceedings from the Second Workshop on American Indigenous Languages*, UCSB Working Papers in Linguistics 9; page numbers not given, except pdf numbering 1-14, http://babel.ling.upenn.edu/~gene/papers/nat_unnat_rules.pdf (retrieved 29 July 2008).
- Busetto, Luca (2003) 'Fonetica nell' India antica', in *Studi Linguistici e Filologici Online* (Rivista Telematica del Dipartimento di Linguistica dell' Università di Pisa), Pisa, <u>http://www.humnet.unipi.it/slifo/articolo%20Busetto.pdf</u> (retrieved 29 July 2008).
- Busetto, Luca (2007) 'Per un inquadramento fonatorio e indoeuropeistico delle rotazioni', in Luca Busetto (ed.) *Symposium amicitiae: scritti in ricordo di Pier Giorgo Negro*, Milan: Qu.A.S.A.R. s.r.l., 39-70.
- Butcher, Andrew (1992) 'Intraoral pressure as an independent parameter in oral stop contrasts', *Proceedings of the 4th Australian International Conference on Speech Science and Technology*, Canberra: Australian Speech Science and Technology Association, 298–291, <u>http://www.assta.org/sst/SST-92/cache/SST-92-AcousticPhonetics2-p2.pdf</u> (retrieved 16 January 2012).
- Butcher, Andrew (2004) "Fortis/Lenis" revisited one more time: the aerodynamics of some oral stop contrasts in three continents', *Clinical Linguistics and Phonetics* 18, 554-57; prepublication draft <u>http://www.flinders.edu.au/speechpath/FortLenRev.pdf</u> (retrieved 29 July 2008).
- Bye, Patrik (2003) 'Opacity, transparency, and unification in the phonology of Tiberian Hebrew', in *Proceedings of NELS* 33, 475-495, <u>http://www.hum.uit.no/a/bye/Papers/hebrew.pdf</u> (retrieved 10 July 2010).
- Bybee, Joan (2002) 'Lexical diffusion in regular sound change', in David Restle and Dietmar Zaefferer (eds.) *Sounds and Systems*, Trends in Linguistics Studies and Monographs 141, Berlin: Mouton de Gruyter, 59-74.

- Bybee, Joan (2005) 'Restrictions on phonemes in affixes: a crosslinguistic test of a popular hypothesis', *Linguistic Typology* 9, 165-222.
- Bybee, Joan (2010) *Language, Usage, and Cognition*, Cambridge: Cambridge University Press.
- Byrd, Andrew M. (2010) *Reconstructing Indo-European Syllabification*, PhD dissertation, University of Kentucky Linguistics Faculty Publications 55, <u>http://uknowledge.uky.edu/cgi/viewcontent.cgi?article=1054&context=lin_facpub</u> (retrieved 17 February 2012).
- Cahill, Michael (2006) 'Perception of Yoruba word-initial [gb] and [b]', 37-41 in Olaoba F. Arasanyin and Michael A. Pemberton (eds.) Selected Proceedings of the 36th Annual Conference on African Linguistics: Shifting the Center of Africanism in Language Politics and Economic Globalization, Somerville: Cascadilla, http://www.lingref.com/cpp/acal/36/paper1404.pdf (retrieved 29 July 2008).

Cahill, Michael (2008) 'Why labial-velar stops merge to /gb/', Phonology 25, 379-398.

- Calabrese, Andrea (1992) 'A constraint-based theory of phonological inventories', *Phonologica 92: Proceedings of the 7th International Phonology Meeting*, Torino: Rosenberg and Sellier, 35-50.
- Calabrese, Andrea (1995) 'A constraint-based theory of phonological markedness and simplification procedures', *Linguistic Inquiry* 26/3, 373-463.
- Calabrese, Andrea (2005) Markedness and Economy in a Derivational Model of Phonology.
- Campbell, Lyle (1974) 'Phonological features: problems and proposals', *Language* 50/1, 52-65.
- Campbell, Lyle and Ronald W. Langacker (1972) 'Proto-Aztecan Vowels: Part I', *International Journal of American Linguistics* 44/2, 85-102.
- Campbell, Lyle W. (1997) *The Historical Linguistics of Native America*, Oxford: Oxford University Press.
- Canger, Una (1988) 'Nahuatl dialectology: a survey and some suggestions', *International Journal of American Linguistics* 54/1, 28-72.
- Carlson, Robert (1994) A grammar of Supyire, Berlin: Mouton de Gruyter.
- Carlson, Barry F. and John H. Esling (2003) 'Phonetics and physiology of the historical shift of uvulars to pharyngeals in Nuuchahnulth (Nootka)', *Journal of the International Phonetic Association* 33/2, 183-193.
- Casad, Eugene H. and Thomas L. Willett (eds.) (2000) Uto-Aztecan: Structural, Temporal, and Geographic Perspectives, Hermosillo: Universidad de Sonora.
- Catford, J. C. (1977) *Fundamental Problems in Phonetics*, Bloomington: Indiana University Press.
- Cerrato, L. and M. Falcone (1998) 'Acoustic and perceptual characteristics of Italian stop consonants', in *Proceedings of the International Conference of Speech and Language Processing*, Sydney, Australia, <u>www.speech.kth.se/~loce/papers/ICSLP98-Paper463.doc</u> (retrieved 27 March 2008)

- Chafe, Wallace (1968) 'The ordering of phonological rules', *International Journal of American Linguistics* 34/2, 5-36.
- Chandler, Daniel (2011) 'Semiotics for beginners', <u>http://www.aber.ac.uk/media/Documents/S4B/sem08a.html</u> (retrieved 30 December 2011).
- Charney, Jean Ormsbee (1993) A Grammar of Comanche, Lincoln, Nebraska: University of Nebraska Press.
- Chen, Yiya (2008) 'The acoustic realization of vowels in Shanghai Chinese', *Journal of Phonetics* 36/4, 629-648.
- Chen, Yiya and Laura J. Downing (2011) 'All depressors are not alike: a comparison of Shanghai Chinese and Zulu', in Sónia Frota, Gorka Elordieta, and Pilar Prieto (eds.) *Prosodic Categories: Production, Perception and Comprehension*, Studies in Natural Language and Linguistic Theory 82, Dordrecht: Springer, 243-266.
- Childers, Donald G. and Chun-Fan Wong (1994) 'Measuring and modeling vocal source-tract interaction', *IEEE Transactions on Biomedical Engineering* 41/7, 663-71
- Chitoran, Ioana (1998) 'Georgian harmonic clusters: phonetic cues to phonological representations', *Phonology* 15, 121-41.
- Cho, T. and Patricia Keating (2001) 'Articulatory and acoustic studies of domain-initial strengthening in Korean', *Journal of Phonetics* 29/2, 155-190.
- Cho, Taehong and Peter Ladefoged (1999) 'Variation and universals in VOT: evidence from 18 languages', *Journal of Phonetics* 27, 207-229.
- Cho, T., S.-A. Jun, and P. Ladefoged (2002) 'Acoustic and aerodynamic correlates of Korean stops and fricatives', *Journal of Phonetics* 30, 193-228.
- Cho, Young-mee Yu (1991) 'On the universality of the coronal articulator', in Paradis and Prunet (1991), 159-180.
- Chomsky, Noam and Morris Halle (1968) *The Sound Pattern of English*, New York: Harper and Row.
- Chomsky, Noam (2005) 'Three factors in language design', Linguistic Inquiry 35, 1-22.
- Chomsky, Noam (2007) 'Approaching UG from below', in U. Sauerland and H. Gärtner (eds.) *Interfaces* + *Recursion* = *Language*?, Berlin: Mouton de Gruyter 1-29, <u>http://www.punksinscience.org/kleanthes/courses/UCY10S/IBL/material/Chomsky_U</u> <u>G.pdf</u> (retrieved 1 September 2011).
- Clackson, James (1994) *The Linguistic Relationship between Armenian and Greek*, Publications of the Philological Society 30, Oxford: Blackwell.
- Clackson, James (2007) Indo-European Linguistics: An Introduction, Cambridge University Press.
- Clements, G. N. (1990) 'The role of the sonority cycle in core syllabification', in J. Kingston and M. Beckman (eds.) *Papers in Laboratory Phonology 1: Between the Grammar and the Physics of Speech*, New York: Cambridge University Press. 283-333, <u>http://www.ai.mit.edu/projects/dm/featgeom/clements90.pdf</u> (retrieved 29 July 2008).

Clements, G. N. (2000) 'Phonology', in Bernd Heine and Derek Nurse (eds.) *African Languages*, 123-60.

Clements, G. N. (2003) 'Feature economy in sound systems', *Phonology* 20, 287–333, <u>http://nickclements.free.fr/publications/2003a.pdf</u> (retrieved 3 Febrary 2012).

- Clements, G. N. (2005a) 'The role of features in phonological inventories', Journée "Les géometries de traits/feature geometries", presentation, Université de Paris 8 et Fédération TUL, 3 December 2005, <u>http://lpp.univ-paris3.fr/equipse/nick_clements/Clements_powerpoints.pdf</u> (retrieved 21 January 2009).
- Clements, G. N. (2005b) 'The role of features in phonological inventories', MS, http://nickclements.free.fr/publications/2005b.pdf (retrieved 5 February 2012).
- Clements, G. N. (2006a) 'Does sonority have a phonetic basis? Comments on the chapter by Bert Vaux', MS (29 January 2006), <u>http://ed268.univparis3.fr/lpp/publications/2006_Clements_Does_Sonority.pdf</u> (retrieved 21 January 2009).
- Clements, G. N. (2006b) 'La phonétique quantique et les primitives phonologiques: notions de base', Journée ATALA "Recherches actuelles en phonétique et phonologie: un état de l'art", ENST, Paris, 21 January 2006, <u>http://ed268.univ-</u> <u>paris3.fr/lpp/publications/2006_Clements_Does_Sonority.pdf</u> (retrieved 21 January 2009).
- Clements, G. N. (2009) 'The role of features in phonological inventories', in Raimy and Cairns (2009), 19-68.
- Clements, G. N. and Elizabeth Hume (1995) 'The internal organization of speech sounds', in John Goldsmith (ed.) *The Handbook of Phonological Theory*, 245-306.
- Clements, G. N. and Annie Rialland (2005) 'Africa as a phonological area', MS, <u>http://ed268.univ-paris3.fr/lpp/publications/2005_Clements_Africa_as.pdf</u> (retrieved 30 July 2008).
- Clements, G. N. and Sylvester Osu (2002) 'Explosives, implosives, and nonexplosives: the linguistic function of air pressure differences in stops', in Carlos Gussenhoven and Natasha Warner (eds.) *Laboratory Phonology* 7, Berlin: Mouton de Gruyter, 299-350.
- Clements, G. N. and Sylvester Osu (2003) 'Ikwere nasal harmony in typological perspective', in Patrick Sauzet and Anne Zribi-Hertz (eds.) *Typologie des langues d'Afrique et universaux de la grammaire*, Paris: L'Harmattan, vol. 2, 70-95, <u>http://nickclements.free.fr/publications/2003c.pdf</u> (retrieved 22 May 2010).

Coetzee, Andries W. (1999) Tiberian Hebrew Phonology.

- Coetzee, Andries W., René Kager, and Joe Pater (2009) 'Introduction: phonological models and experimental data', *Phonology* 26, 1-8.
- Cohn, Abigail C. (1993a) 'Consonant-vowel interactions in Madurese: the feature Lowered Larynx', *CLS* 29/1 = *Papers from the 29th Regional Meeting of the Chicago Linguistic Society*, vol. 1, 105-119.

- Cohn, Abigail C. (1993b) 'Voicing and vowel height in Madurese: a preliminary report', *Oceanic Linguistics Special Publications*, No. 24, Tonality in Austronesian Languages, 107-121.
- Cohn, Abigail C. and William H. Ham (1998) 'Temporal properties of Madurese consonants: a preliminary report', *Working Papers of the Cornell Phonetics Laboratory* 12, 27-51.
- Cohn, Abigail C. and Katherine Lockwood (1994) 'A phonetic description of Madurese and its phonological implications', *Working Papers of the Cornell Phonetics Laboratory* 9, 67-92.
- Cohn, Abigail C. and Anastasia K. Riehl (n.d.) 'The internal structure of nasal-stop sequences: evidence from Austronesian', MS, <u>http://velar.phonetics.cornell.edu/pdfs/Cohn_Riehl_WP.pdf</u> (retrieved 16 February 2012).
- Colarusso, J. J., Jr. (1981) 'Typological parallels between Proto-Indo-European and the Northwest Caucasian Languages' in Arbeitman and Bomhard (1981), Part I, 475-557.
- Colarusso, J. J., Jr. (1988[1975]) *The Northwest Caucasian Languages: A Phonological Survey*, Ph.D. dissertation, Harvard University; Garland Outstanding Dissertations in Linguistics, New York: Garland.
- Coleman, John (1998) *Phonological Representations: Their Names, Forms, and Powers*, Cambridge: Cambridge University Press.
- Coleman, John (n.d.) 'Variation in the qualities of epenthetic vowels in Tashlhiyt Berber', MS, <u>http://www.phon.ox.ac.uk/jcoleman/TPS.html</u> (retrieved 1 December 2011).
- Coleman, John (2001) 'Phonetics and phonology of Tashlhiyt Berber syllabic consonants', *Transactions of the Philological Society* 99/1, 29-64.
- Comrie, Bernard (1997) 'Uyghur phonology', chaper 45 in Alan S. Kaye (ed.), *Phonologies of Asia and Africa*, vol. 2, 913-25.
- Connell, Bruce (1997) 'Mambila fricative vowels', to appear in Vicki Carstens (ed.) *Proceedings of the Annual Conference on African Linguistics* 28, draft, http://lucy.ukc.ac.uk/dz/ACAL28/ACAL28paper.html (retrieved 8 December 2011).
- Connell, Bruce and Amalia Arvaniti (eds.) (1995) *Phonology and Phonetic Evidence: Papers in Laboratory Phonology IV*, Cambridge: Cambridge University Press, 386-401.
- Cowgill, Warren (1985[1984]) '[Review of memorial volume Kerns]', Kratylos 29: 6.
- Crist, Sean (2001) *Conspiracy in Historical Phonology*, PhD dissertation, University of Pennsylvania, <u>www.panix.com/~kurisuto/publications/crist_thesis.ps</u> (retrieved as html 2 September 2009).
- Croft, William (1993[1990]) *Typology and Universals*, Cambridge: Cambridge University Press.
- Cser, András (2003) *The typology and modelling of obstruent lenition and fortition processes*, Budapest: Akademiai Kiado.

- Cser, András (in press) 'The role of sonority in the phonology of Latin', in Stephen Parker (ed.) *The Sonority Controversy*, Berlin: de Gruyter.
- Cumming, Ruth E. (2010) 'Speech rhythm: the language-specific integration of pitch and duration', PhD dissertation, University of Cambridge, http://www.dspace.cam.ac.uk/handle/1810/228685 (retrieved 27 March 2012).
- Cun, Xi (2005) 'The phonetic characteristics of implosives in Tow Chinese dialects', paper for Phonetics and Phonology in Iberia 2005, <u>http://webs2002.uab.es/filologiacatalana/papi/files/cun-xi.pdf</u> (retrieved 13 December 2011).
- Dankovičová, Jana (1999) 'Czech', in International Phonetic Association (1999), 70-73.
- Dart, Sarah N. (1991) 'Articulatory and acoustic properties of apical and laminal articulations', *UCLA Working Papers in Phonetics 79*, http://repositories.cdlib.org/cgi/viewcontent.cgi?article=1081&context=uclaling

http://repositories.cdlib.org/cgi/viewcontent.cgi?article=1081&context=uclaling (retrieved 28 July 2008).

- Darwin, Christopher J. (2008) 'Spatial hearing and perceiving sources', in William A. Yost (ed.) *Auditory Perception of Sound Sources*, New York: Springer, 215-232.
- Davies, William D. (1999) Madurese, München: LINCOM Europa.
- Davies, William D. (2010) A Grammar of Madurese.
- Davis, K. (1994) 'Stop voicing in Hindi', Journal of Phonetics 22, 177-193.
- de Boysson-Bardies, Bénédicte and Marily M. Vihman (1991) 'Adaptation to language: evidence from babbling and first words in four languages', *Language* 67/2, 297-319.
- de Decker, Filip (2010) 'What is the Greek counterpart of Sanskrit *th*?', presentation handout, Greek and Latin from an Indo-European Perspective 3, Bratislava, 7-11 July 2010, <u>https://uhdspace.uhasselt.be/dspace/handle/1942/11372?language=en</u> (retrieved 16 February 2012).
- de Lacy, Paul (2002) The Formal Expression of Markedness, PhD dissertation, University of Massachusetts, Amherst, <u>http://roa.rutgers.edu/files/542-0902/542-0902-LACY-0-0.pdf</u> (retrieved 22 May 2008).
- de Lacy, Paul (2006) *Markedness: Reduction and Preservation in Phonology*, Cambridge: Cambridge University Press.
- de Lacy, Paul (2007) 'The interaction of tone, sonority, and prosodic structure', in de Lacy (ed.) (2007), 281-307.
- de Lacy, Paul (ed.) (2007) *The Cambridge Handbook of Phonology*, Cambridge: Cambridge University Press.
- de Saussure, Ferdinand (1989[1916]) *Cours de linguistique général*, édition critique par Rudolf Engler, Wiesbaden: Otto Harrassowitz, vol. 1.
- de Saussure, Ferdinand (1972) *Cours de linguistique générale*, Paris: Charles Bally et Albert Séchehaye, as quoted in 'Saussure: bibliographie et citations extraites du cours', Centre International d'Etude de la Philosophie Française Contemporaine <u>http://ciepfc.rhapsodyk.net/article.php3?id_article=50</u> (retrieved 7 May 2008).
- Delattre, Pierre C., Alvin M. Liberman, and Franklin S. Cooper (1955) 'Acoustic loci and transitional cues for consonants', *The Journal of the Acoustical Society of*

America 27/4, 769-73, <u>http://www.haskins.yale.edu/Reprints/HL0012.pdf</u> (retrieved 17 March 2011).

- Dell, François (2008) 'Singing in Tashlhiyt Berber, a language that allows vowel-less syllables', Conference on the Syllable, CUNY Phonology Forum, 17-19 January 2008, <u>http://www.cunyphonologyforum.net/SYLLABSTRACTS/Dell.pdf</u> (retrieved 1 December 2011).
- Dell, François (2009) 'Singing in Tashlhiyt Berber, a language that allows vowel-less syllables', MS, <u>http://crlao.ehess.fr/depot/dell/Tashlhiyt_singing.pdf</u> (retrieved 5 December 2011).
- Dell, François and Mohamed Elmedlaoui (1985) 'Syllabic consonants and syllabification in Imdlawn Tashlhiyt Berber', *Journal of African Languages and Linguistics* 7, 105-130, <u>http://crlao.ehess.fr/depot/dell/Dell_Elmedlaoui1985.pdf</u> (retrieved 22 September 2011).
- Dell, François and Mohamed Elmedlaoui (2002) *Syllables in Tashlhiyt Berber and in Moroccan Arabic*, Dordrecht: Kluwer.
- Dell, François and Mohamed Elmedlaoui (2010) 'Syllables and gemination in imperfective stems in Tashlhiyt Berber', MS, <u>http://crlao.ehess.fr/docannexe.php?</u> id=1194 (retrieved 5 December 2011).
- Demolin, Didier (1995) 'The phonetics and phonology of glottalized consonants', in Connell and Arvaniti (1995), 368-85.
- Demolin, Didier (2002) 'The search for primitives in phonology and the explanation of sound patterns: the contribution of fieldwork studies', in Carlos Gussenhoven and Natasha Warner (eds.) *Laboratory Phonology*, vol. 1, Berlin: Walter de Gruyter, 455-514.
- Demolin, Didier (2007) 'Phonological universals and the control and regulation of speech production', in Maria-Josep Solé, Patrice Speeter Beddor, and Manjari Ohala (eds.) *Experimental Approaches to Phonology*, Oxford: Oxford University Press, 75-92.
- Demolin, Didier and Véronique Delvaux (2001) 'Whispery voiced nasal stops in Rwanda', presentation, Eurospeech 2001, Scandinavia, <u>http://perso.telecom-</u> <u>paristech.fr/~chollet/Biblio/Congres/Audio/Eurospeech01/CDROM/papers/page651.p</u> <u>df</u> (retrieved 28 January 2012).
- Demolin, Didier, Hubert Ngonga-Ke-Bembe, and Alain Soquet (1999) 'Phonetic characteristics of an unexploded stop in Hendo', Proceedings *ICPhS*, San Fransisco, 1047-1050, <u>http://www.ulb.ac.be/philo/phonolab/publi/ICPhS99/1047.pdf</u> (retrieved 29 July 2008).
- Demolin, Didier, Hubert Ngonga-Ke-Mbembe, and Alain Soquet (2002) 'Phonetic characteristics of an unexploded palatal implosive in Hendo', *Journal of the International Phonetic Association* 32, 1-15.
- Demolin, Didier and Marine Vuillermet (2006) 'Voiceless implosives: a comparison between American and African languages', presentation, Rara Conference, Leipzig, 30 March 2006, <u>http://elisa.ddl.ish-</u>

<u>lyon.cnrs.fr/fulltext/Vuillermet/Vuillermet_2006_Demolin_voiceless_impl.pdf</u> (retrieved 18 February 2012).

DePaolis, Rory A., Marilyn M. Vihman, and Tamar Keren-Portnoy (2011) 'Do production patterns influence the processing of speech in prelinguistic infants?', *Infant Behavior and Development* 34, 590-601, <u>http://www.york.ac.uk/media/languageandlinguistics/documents/staff/publications/De</u>

Paolis,%20Vihman%20&%20Keren-Portnoy%202011.pdf (retrieved 25 February 2012).

- Devine, Andrew M. (1974a) 'Aspiration, universals, and the study of dead languages', *Working Papers on Language Universals* 15, 1-24.
- Devine, Andrew M. (1974b) 'Etruscan language studies and modern phonology: the problem of the aspirates', *Studi Etruschi* 42/3, 123-151.
- DiCanio, Christian Thomas (2008) *The Phonetics and Phonology of San Martín Itunyoso Trique*, PhD dissertation, Berkeley, <u>http://linguistics.berkeley.edu/dissertations/DiCanio_dissertation_2008.pdf</u> (retrieved 11 January 2012).
- Diffloth, Gérard (1982) 'Registres, dévoisement, timbres vocaliques: leur histoire en katouique', *Mon-Khmer Studies* 11, 47-82.
- Dinneen, Francis P. (1990) 'A 17th century account of Mohawk' in Francis P. Dinneen and E.F.K. Koerner (eds.) North American Contributions to the History of Linguistics, Studies in the History of the Language Sciences 58, Amsterdam: John Benjamins, 67-86.
- Dixit, R. Prakash (1975) *Neuromuscular Aspects of Laryngeal Control: with Special Reference to Hindi*, PhD dissertation, University of Texas at Austin.
- Dixit, R. Prakash (1979) 'Inadequacies in phonetic specifications of some laryngeal features: evidence from Hindi', in Harry Francis Hollien and Patricia Hollien (eds.) *Current Issues in the Phonetic Sciences*, Current Issues in Linguistc Theory 9, Amsterdam: John Benjamins, 423-434.
- Dixit, R. Prakash (1982) 'On defining aspiration' in S. Hattori and K. Inoue (eds.) *Proceedings of the Thirteenth International Conference of Linguists*, 606-610.
- Dixit, R. Prakash (1987) 'In defence of the phonetic adequacy of the traditional term "voiced aspirated", *Proceedings of the Eleventh International Congress of Phonetic Sciences* 2, 145-148.
- Dixit, R. Prakash (1989) 'Glottal gestures in Hindi plosives', *Journal of Phonetics* 17, 213-237.
- Dixit, R. Prakash and P. F. MacNeilage (1981) 'Cricothyroid activity and control of voicing in Hindi stops and affricates', *Phonetica* 37(5-6), 397-406, abstract <u>http://www.researchgate.net/publication/15919138_Cricothyroid_activity_and_control_of_voicing_in_Hindi_stops_and_affricates</u> (retrieved 25 February 2012).
- Dixon, R. M. W. (1972) *The Dyirbal Language of North Queensland*, Cambridge Studies in Linguistics 9, Cambridge: Cambridge University Press.
- Dixon, R. M. W. (1994) Ergativity, Cambridge: Cambridge University Press.

- Dixon, R. M. W. (2010[1977]) *A Grammar of Yidin*, Cambridge: Cambridge University Press.
- Djahukian, G. B. (1990) 'A Variational Model of the Indoeuropean Consonant System', *Historische Sprachforschung* 103, 1-16.
- Docherty, G. J. (1992) 'The timing of voicing in British obstruents', Netherlands Phonetics Archives 9.
- Docherty, Gerry and Paul Foulkes (1999) 'Sociophonetic variation in 'glottals' in Newcastle English', *Proceedings of the International Congress of Phonetic Sciences* 99, 1037-40, <u>http://www-users.york.ac.uk/~pf11/ICPhS99-glottals.pdf</u> (retrieved 9 December 2011).
- Dogil, Grzegorz (forthcoming) 'Hard-wired phonology: limits and latitude of phonological variation in pathological speech', to appear in *Papers in Laboratory Phonology* 10, draft circulated in University of Colorado Boulder Phonology course taught by Rebecca Scarborough, fall 2008.
- Dogil, Grzegorz and Luschützky, Hans Christian (1990) 'Notes on sonority and segmental strength', *Rivista di Linguistica* 2/2, 3-54.
- Doke, Clement M. (1926) *The Phonetics of the Zulu Language*, Johannesburg: University of Wittwatersrand Press
- Doke, Clement M. (1931) *A Comprehensive Study of Shona Phonetics*, Johannesburg: University of Witwatersrand Press.
- Doke, Clement M. (1954) The Southern Bantu Languages, London.
- Donohue, Mark (1999) A Grammar of Tukang Besi, Berlin: Walter de Gruyter.
- Donohue, Mark (2002) 'Which sounds change: descent and borrowing in the Skou family', *Oceanic Linguistics* 41/1, 171-221.
- Dorman, M. F., F. J. Freeman, and G. J. Borden (1976) 'The effect of delayed auditory feedback on phonation: an electromyographic study', Haskins Laboratories Status Report on Speech Research SR-45/46, 187-195, http://www.haskins.yale.edu/sr/SR045/SR045_12.pdf (retrieved 18 February 2012).
- Downing, Laura (2009a) 'On pitch lowering not linked to voicing: Nguni and Shona group depressors', *Languages Sciences* 31, 179-198.
- Downing, Laura (2009b) 'Optimality Theory and African language phonology', in Masango Matondo et al. (eds.) *Selected Proceedings of the 38th Conference on African Linguistics*, 1-16, <u>http://www.lingref.com/cpp/acal/38/paper2131.pdf</u> (retrieved 20 April 2010).
- Downing, Laura J. and Bryan Gick (2001) 'Voiceless tone depressors in Nambya and Botswana Kalang'a', *Proceedings of the Annual Meeting, Berkeley Linguistics Society* 27, 65–80, <u>http://www.elanguage.net/journals/index.php/bls/article/viewFile/688/546</u> (retrieved 6 December 2011).
- Dresher, Elan (2009) *The Contrastive Hierarchy*, Cambridge: Cambridge University Press.

- Dresher, Elan, Glye Piggott, and Keren Rice (1994) 'Contrast in phonology: overview', MS, University of Toronto, <u>http://www.chass.utoronto.ca/~contrast/dpr.pdf</u>, published version cited as *Toronto Working Papers in Linguistics* 13/1, iii-xvii.
- Dressler, Wolfgang U., Prinzhorn, Martin, and Rennison, John R. (eds.) (1992) *Phonologica 92: Proceedings of the 7th International Phonology Meeting*, Torino: Rosenberg and Sellier.
- Dunkel, George (1981) 'Typology versus reconstruction' in Arbeitman and Bomhard (1981), Part II, 559-569.

Dunn, John Asher (1995) Sm'algyax: A Reference Dictionary and Grammar of the Coast Tsimshian Language, Seattle: University of Washington Press.

- Durie, Mark (1994) 'Aceh', in Tryon (1994), 407-420.
- Dutta, Indranil (forthcoming) 'Durational, f0 and spectral characteristics of Hindi stops', to appear in *Phonology and Phonetics of Turbulent Sounds: Papers from the Conference on Turbulences, Zentrum für Allgemeine Sprachwissenschaft, Typologie und Universalienforschung (ZAS)* [Berlin, Germany, 13th-14th October, 2006], Berlin: Elsevier, author's abstract <u>http://www.zas.gwz-berlin.de/turbulences/abstract/Dutta_ab.pdf</u> (retrieved 29 July 2008).
- Edmondson, Jerold A. (2000) 'Voice quality in the Ye and Bai languages: the role of the aryepiglottic folds', *The Fifth International Symposium on Languages and Linguistics*, Hochiminh City: Ho Chi Minh City University of Social Sciences and Humanities, 59-84, <u>http://sealang.net/sala/archives/pdf8/edmondson2000voice.pdf</u> (retrieved 2 March 2012).
- Edmondson, Jerold A. (2004) 'The aryepiglottic sphincter and the source-filter theory: evidence from some endangered languages', MS, <u>http://ling.uta.edu/~jerry/lsa4.pdf</u> (retrieved 14 January 2012).
- Edmondson, Jerold A. and John H. Esling (2006) 'The valves of the throat and their functioning in tone, vocal register and stress: laryngoscopic case studies', *Phonology* 23, 157-191.
- Edmondson, Jerold A. and K. Gregerson (1993) 'Western Cham as a register language', in Edmondson and Gregerson (eds.) *Tonality in Austronesian Languages*, *Oceanic Linguistics* no. 24, 61-74.
- Edmondson, Jerold A., John H. Esling, and Jimmy G. Harris (2004) 'Supraglottal cavity shape, linguistic register, and other phonetic features of Somali', MS, <u>http://ling.uta.edu/~jerry/somali.pdf</u> (retrieved 12 May 2010).
- Edmondson, Jerold A., John H. Esling, Jimmy G. Harris, and Tung-chiou Huang (2005) 'A laryngoscopic study of glottal and epiglottal/pharyngeal stop and continuant articulations in Amis—an Austronesian language of Taiwan', *Language and Linguistics* 6/3, 381-396,

http://www.ling.sinica.edu.tw/eip/FILES/journal/2007.7.30.5239504.52333432.pdf (retrieved 14 January 2012).

Edmondson, Jerold A., John H. Esling, Jimmy G. Harris, Li Shaoni, and Lama Ziwo (2001) 'The aryepiglottic folds and voice quality in Yi and Bai languages:
laryngoscopic case studies', *Mon-Khmer Studies* 31, 83-93, draft <u>http://ling.uta.edu/~jerry/yi-bai.pdf</u> (retrieved 2 March 2012).

- Edmondson, Jerold A., John H. Esling, Jimmy G. Harris, and J. Wei (2004) 'A phonetic study of the Sui consonants and vowels', *Mon-Khmer Studies* 34, 47-66, at http://www.sealang.net/archives/mks/pdf/34:47-66.pdf (retrieved 11 February 2012).
- Edmondson, Jerold A., John H. Esling, Li Shaoni, J.G. Harris, and Lama Ziwo (2000) 'The aryepiglottic folds and voice quality in Yi and Bai languages: laryngoscopic case studies', *Minzu Yuwen* 6, 47-53.
- Edmondson, Jerold A., Cécile M. Padayodi, Zeki Majeed Hassan, and John H. Esling (2007) 'The laryngeal articulator: source and resonator', *Proceedings of the XVIth International Congress of the Phonetic Sciences*, Saarbrücken, 6-10 August 2007, 2065-2068, <u>http://www.icphs2007.de/conference/Papers/1674/1674.pdf</u> (retrieved 18 February 2012).
- Egerod, Søren (1970) 'Distinctive features and phonological reconstructions', *Journal of the American Oriental Society* 90/1, 67-73.
- Elbourne, Paul (1997) *The Evidence for Voiceless Aspirates in Proto-Indo-European*, MPhil thesis, Oxford University.
- Elbourne, Paul (1998) 'Proto-Indo-European voiceless aspirates', *Historische Sprachforschung* 111, 1-30.
- Elbourne, Paul (2000) 'Plain Voiceless Stop plus Laryngeal in Indo-European', *Historische Sprachforschung* 113, 2-28.
- Ellison, T. Mark (2000) 'The universal constraint set: convention, not fact', in Joost Dekkers, Frank Reinoud Hugo van der Leeuw, and Jeroen Maarten van de Weijer (eds.) *Optimality Theory: Phonology, Syntax, and Acquisition*, 524-53.
- Elzinga, Dirk (2000) 'AnOTher look at Shoshone taps and spirants' in Casad and Willett (2000), 83-89.
- Emmeneau, M. B. (1933) 'A note on the development of the Indo-European dental groups', *Language* 9/3, 232-236.
- Ernestus, Mirjam (2011) 'Gradience and categoricality in phonological theory', ch. 89 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.

Esling, John H. (2000) 'Pharynx research', web page,

http://web.uvic.ca/ling/research/phonetics/jipa96.htm (retrieved 26 March 2012).

- Esling, John H. (2002a) 'The laryngeal sphincter as an articulator: tenseness, tongue root and phonation in Yi and Bai', presentation, LSA Annual Meeting, 5 January 2002, <u>http://web.uvic.ca/ling/research/phonetics/media/LSA2002.pdf</u> (retrieved 26 March 2012).
- Esling, John H. (2002b) 'The laryngeal sphincter as an articulator: tenseness, tongue root and phonation in Yi and Bái', slide presentation with video files, accompanying the phonetic descriptions in John H. Esling and Jerold A. Edmondson (2002). 'The laryngeal sphincter as an articulator: tenseness, tongue root and phonation in Yi and Bai', in A. Braun and H.R. Masthoff (eds.) *Phonetics and its Applications: Festschrift*

for Jens-Peter Köster on the Occasion of his 60th Birthday, Stuttgart: Franz Steiner Verlag; <u>http://web.uvic.ca/ling/research/phonetics/festschrift/index.htm</u> (retrieved 26 March 2012).

Esling, John H. (2002c) 'Laryngoscopic analysis of Tibetan chanting modes and their relationship to register in Sino-Tibetan', presentation, International Conference on Spoken Language Processing (ICSLP),

http://web.uvic.ca/ling/research/phonetics/icslp02.htm (retrieved 27 March 2012).

- Esling, John H. (2006) 'States of the glottis', *Encyclopedia of Language and Linguistics* 129-32, <u>http://www.sciencedirect.com/science/article/pii/B0080448542050987#b0018</u> with embedded film clips (retrieved 18 February 2012).
- Esling, John H. and Jimmy G. Harris (2003) 'An expanded taxonomy of states of the glottis', *Proceedings of the International Conference of the Phonetic Sciences XV*, Barcelona, vol. 1, 1049-1052,

http://web.uvic.ca/ling/research/phonetics/ICPhS/ICPhS2003_EslingHarris.pdf (retrieved 17 March 2011).

Esling, John H. and Jimmy G. Harris [2004] 'States of the glottis: an articulatory model based on laryngoscopic observations', presentation, International Conference on Voice Physiology and Biomechanics,

http://ncvs2.org/icvpb/presentations/EslingHarrisICVPB.pdf (retrieved 11 January 2012).

- Esling, John H. and Jimmy G. Harris (2005) 'States of the glottis: an articulatory model based on laryngoscopic observations', in W. J. Hardcastle and J. Beck (eds.) *A Figure of Speech: A Festschrift for John Laver*, Mahwah, New Jersey: Lawrence Erlbaum Associates, 347-383.
- Esling, John H., Katherine A. Fraser, and Jimmy G. Harris (2005) 'Glottal stop, glottalized resonants, and pharyngeals: a reinterpretation with evidence from a laryngoscopic study of Nuuchahnulth (Nootka)', *Journal of Phonetics* 33/4, 383-410.
- Everett, Daniel (2003) 'Theoretical implications of post-lexical Structure Preservation in Suyá', paper, Eleventh Manchester Phonology Meeting, 22 May, http://www.llc.ilstu.edu/dlevere/docs/suyahnd.pdf (retrieved 29 July 2008).
- Evers, Vincent, Henning Reetz and Aditi Lahiri (1998) 'Crosslinguistic acoustic categorization of sibilants independent of phonological status', *Journal of Phonetics* 26, 345-370, <u>http://www.ling-</u>phil.ox.ac.uk/files/people/crosslinguistic acoustic categ.pdf (retrieved 5 December

<u>phil.ox.ac.uk/files/people/crosslinguistic_acoustic_categ.pdf</u> (retrieved 5 December 2011).

Fais, Laurel, Sachiyo Kajikawa, Shigeaki Amano, and Janet F. Werker (2010) 'Now you hear it, now you don't: vowel devoicing in Japanese infant-directed speech', *Journal of Child Language* 37, 319-340,

http://infantstudies.psych.ubc.ca/uploads/forms/1269446878FaisEtAl_2010.pdf (retrieved 14 February 2012).

Fallon, Paul D. (2002[1998]) The synchronic and diachronic phonology of ejectives.

- Farnetani, Edda and Daniel Recasens (2010) 'Coarticulation and connected speech processes' in William J. Hardcastle, John Laver, and Fiona E. Gibbon (eds.) *The Handbook of Phonetic Sciences*, 316-52.
- Feinstein, Mark H. (1979) 'Prenasalization and syllable structure', *Linguistic Inquiry* 10/2, 245-278.
- Feng, Ling (2007) 'The articulatory and acoustic study of fricative vowels in Suzhou Chinese', *Proceedings of the International Congress of Phonetic Sciences XVI*, 573-576, <u>http://www.icphs2007.de/conference/Papers/1321/1321.pdf</u> (retrieved 8 December 2011).
- Ferlus, Michel (1977) 'Etude d'une strate de changements phonétiques dans l'ancien Cambodge', *Mon-Khmer Studies* 6, 59-67,

http://sealang.net/sala/archives/pdf8/ferlus1977etude.pdf (retrieved 26 July 2008).

Ferlus, Michel (1979) 'Formation des registres et mutations consonantiques dans les langues Mon-Khmer', *Mon-Khmer Studies* 8, 1-76,

http://sealang.net/sala/archives/pdf8/ferlus1979formation.pdf (retrieved 26 July 2008).

- Ferlus, Michel (1983) 'Essai de phonétique historique du môn', *Mon-Khmer Studies* 12, 1-90, <u>http://sealang.net/sala/archives/pdf8/ferlus1983essai.pdf</u> (retrieved 26 July 2008).
- Ferlus, Michel (1992) 'Essai de phonétique historique du khmer', *Mon-Khmer Studies* 21, 57-89, <u>http://sealang.net/sala/archives/pdf8/ferlus1992essai.pdf</u> (retrieved 26 July 2008).
- Finnian, Kelly, and Naomi Harte (2010) 'A comparison of auditory features for robust speech recognition', presentation, 18th European Signal Processing Conference (EUSIPCO-2010), Aalborg, Denmark, 23-27 August 2010, <u>http://www.eurasip.org/Proceedings/Eusipco/Eusipco2010/Contents/papers/15692908</u> 59.pdf (retrieved 18 February 2012).
- Fischer-Jørgensen, Eli and Hajime Hirose (1974a) 'A preliminary electromyographic study of labial and laryngeal muscles in Danish stop consonant production', Haskins Laboratories Status Report on Speech Research SR-39/40, 231-254, http://www.haskins.yale.edu/sr/SR039/SR039_15.pdf (retrieved 11 January 2012).
- Fischer-Jørgensen, Eli and Hajime Hirose (1974b) 'A note on laryngeal activity in the Danish "stød", Haskins Laboratories Status Report on Speech Research SR-39/40, 255-259, <u>http://www.haskins.yale.edu/sr/SR039/SR039_16.pdf</u> (retrieved 18 February 2012).
- Fitzpatrick, Justin and Andrew Nevins (2004) 'Linearization of nested and overlapping precedence in multiple reduplication', in Sudha Arunachalam and Tajana Scheffler (eds.) *Proceedings of the Penn Linguistics Colloquium 29*, University of Pennsylvania Working Papers in Linguistics 10/1, 75-88,

http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevinspublications/fitzpatrick-nevins04 (retrieved 25 September 2010).

Flemming, Edward (2003) 'Speech perception in phonology', <u>http://web.mit.edu/flemming/www/paper/Percphon03.pdf</u> (retrieved 29 July 2008), pre-publication version; published in D. Pisoni and R. Remez (eds.) (2005) *The Handbook of Speech Perception*.

- Flemming, Edward (2004) 'Contrast and perceptual distinctiveness', in Hayes et al. (2004), 232-276.
- Fontaine-De Visscher, Luce (1988) 'La linguistique et les sciences humaines. Comment redéfinir la linguistique, avec Claude Hagège', *Revue Philosophique de Louvain* 86, 378-392, <u>http://www.persee.fr/web/revues/home/prescript/article/phlou_0035-</u>3841 1988 num 86 71 6510 (retrieved 15 February 2012).
- Fougeron, Cécile and Rachid Ridouane (2008) 'On the phonetic implementation of syllabic consonants and vowel-less syllables in Tashlhiyt', MS, published in *Estudios de Fonética Experimental 18*, 139-75, <u>http://hal.archives-ouvertes.fr/docs/00/38/49/27/PDF/Fougeron-Ridouane_2006_.pdf</u> (retrieved 1 December 2011).
- Fourakis, M., Botinis, A. and Niryanakis, E. (2002) = Φουράκης, Μ., Μποτίνης, Α. και Νιργιανάκη, '<u>Χρονικά χαρακτηριστικά των συμφωνικών ακολουθιών [ps], [ts] και [ks]</u>' στην Ελληνική Πρακτικά του 6ου Διεθνούς Συνεδρίου Ελληνικής Γλωσσολογίας, Ρέθυμνο, Κρήτη. <u>http://www.phil.uoa.gr/~abotinis/Publications</u> <u>%202007/ICGL%20(2).pdf</u> (retrieved 29 July 2008).
- Fox, Anthony (1995) Linguistic Reconstruction, Oxford.
- Francis, E. D. (1992) 'The impact of non-Indo-European languages on Greek and Mycenaean', in Edgar C. Polomé and Werner Winters (eds.) *Reconstructing Languages and Cultures*, Trends in Linguistics Studies and Monographs 58, Berlin: Mouton de Gruyter, 469-506.

Frantz, Donald G. (1991) Blackfood Grammar, Toronto: University of Toronto Press.

- Freeman, Frances J. and Tatsujiro Ushijima (1974) 'Laryngeal activity accompanying the moment of stuttering: a preliminary report of EMG investigations', presentation, 87th meeting of the Acoustical Society of America, New York, April 1974; Haskins Laboratories Status Report on Speech Research SR-37/38, 109-116, <u>http://www.haskins.yale.edu/SR/SR037/SR037_10.pdf</u> (retrieved 11 January 2012).
- Freeman, Frances J. and Tatsujiro Ushijima (1975) 'The stuttering larynx: an EMG, fiberoptic study of laryngeal activity accompanying the moment of stuttering', Haskins Laboratories Status Report on Speech Research SR-41, 217-228, http://www.haskins.yale.edu/sr/SR041/SR041_15.pdf (retrieved 18 February 2012).
- Freeman, Frances J. and Tatsujiro Ushijima (1976) 'Laryngeal muscle activity in stuttering', Haskins Laboratories Status Report on Speech Research SR-45/46, 211-236, <u>http://www.haskins.yale.edu/sr/SR045/SR045_15.pdf</u> (retrieved 11 January 2012).
- Frisch, Stefan (1997) *Similarity and Frequency in Phonology*, PhD dissertation, Northwestern University, <u>http://roa.rutgers.edu/files/198-0597/roa-198-frisch-5.pdf</u> (retrieved 29 July 2008).
- Fromkin, Victoria A. (1972) 'Tone features and tone rules', *Studies in African Linguistics* 3/1, 47-76,

http://www.cyberling.elanguage.net/journals/index.php/sal/article/viewFile/918/820 (retrieved 1 February 2012).

- Fujimura, Osamu and Donna Erickson (1999) 'Acoustic phonetics', in William J. Hardcastle and John Laver (eds.) *The Handbook of Phonetic Sciences*, 65-115.
- Fulop, Sean A. and Chris Golston (2008) 'Breathy and whispery voicing in White Hmong', *Proceedings of Meeting on Acoustics*, 155th Meeting of the Acoustical Society of America, Paris, 29 June – 4 July 2008, 1-10, <u>http://scitation.aip.org/getpdf/servlet/GetPDFServlet?</u> <u>filetype=pdf&id=PMARCW000004000001060006000001&idtype=cvips&doi=10.11</u> 21/1.3033931&prog=normal (retrieved 14 January 2012).
- Gabas Jr., Nilson (1998) *Estudo fonológico da língua Karo*. München and Newcastle: LINCOM Europa.
- Gabas Jr., Nilson (1999) A Grammar of Karo (Tupi). Doctoral dissertation, University of California, Santa Barbara.
- Gabbard, Kevin M. (2010) 'A phonological analysis of Somali and the guttural consonants', BA Linguistics Honors Thesis, <u>https://kb.osu.edu/dspace/bitstream/handle/1811/46639/gabbardsomalithesis2010.pdf?</u> sequence=1 (retrieved 31 March 2012).
- Gafos, Adamantios (1996) *The Articulatory Basis of Locality in Phonology*, PhD Dissertation, <u>http://web.jhu.edu/cogsci/people/alumni.html</u> (retrieved 24 August 2009).
- Gafos, Adamantios and Stefan Benus (2006) 'Dynamics of phonological cognition', *Cognitive Science* 30, 1-39, <u>http://sail.usc.edu/~lgoldste/ArtPhon/Papers/Week</u> <u>%2013/06Gafos_color.pdf</u> (retrieved 27 March 2012).
- Gahl, Susanne (1995) 'Consonant gradation as a prosodic constraint', *North East Linguistic Society* 25, 25-47.
- Gamkrelidze, T. V. (1975) 'On the correlation of stops and fricatives in a phonological system,' *Lingua* 35, 231-261.
- Gamkrelidze, T. V. (1978) 'On the correlation of stops and fricatives in a phonological system' in J. H. Greenberg et al. (eds.) *Universals of Human Language, vol. 2, Phonology*, Stanford: Stanford University Press, 9-46.
- Gamkrelize, T. V. (1981) 'Language typology and language universals' in Arbeitman and Bomhard (1981), Part II, 571-609.
- Gamkrelidze, T. V. (1988) 'The Indo-European Glottalic Theory in the Light of Recent Critique,' *Folia Linguistica Historica* 9/1, 3-12
- Gamkrelidze, T. V. (1989) 'Language Typology and Indo-European reconstruction,' in Vennemann (1989), 117-121.
- Gamkrelidze, T. V. (1989) 'Markedness, sound change and linguistic reconstruction', in Olga Mišeska Tomić (ed.) *Markedness in Synchrony and Diachrony*, Trends in Linguistics no. 39, Berlin: Mouton de Gruyter, 87-101.
- Gamkrelidze, T. V. (2003) 'Indo-European and the glottalic theory: in defense of ejectives for proto-Indo-European', in Mary Ruth Wise, Thomas N. Headland

and Ruth M. Brend (eds.) *Language and life: Essays in memory of Kenneth L. Pike*, SIL International and The University of Texas at Arlington Publications in Linguistics, 139, Dallas: SIL International and University of Texas at Arlington, 513-531.

- Gamkrelidze, T. V. (2008) 'The Indo-European Glottalic Theory: a new paradigm in Indo-European comparative linguistics', *Bulletin of the Georgian National Academy of Sciences* 2/1, 140-143, <u>http://science.org.ge/2-1/Gamkrelidze.pdf</u> (retrieved 16 February 2012).
- Gamkrelidze, T. V. and Ivanov, V. V. (1973) 'Sprachtypologie und die Rekonstruktion der gemeinindogermanischen Verschlüsse', *Phonetica* 27, 150-156.
- Gamkrelidze, T. V. and Ivanov, V. V. (1995) *Indo-European and the Indo-Europeans*. Trends in Linguistics, Studies and Monographs 80. Translated by Johanna Nichols. Berlin: Mouton de Gruyter.
- Gao, J.-Y., P. Hallé, K. Honda, S. Maeda, and M. Toda (2011) 'Shanghai slack voice: acoustic and EPGG data', *Proceedings of the International Congress of Phonetic Sciences XVII*, Hong Kong, 17-21 August 2011, 719-722, http://www.icphs2011.hk/resources/OnlineProceedings/RegularSession/Gao,%20J-Y/Gao,%20J-Y/Ddf (retrieved 31 January 2012).
- Garellek, Marc (2010) 'The acoustics of coarticulated non-modal phonation', *Working Papers in Phonetics* 108, Los Angeles: UCLA Berkeley Department of Linguistics, <u>http://escholarship.org/uc/item/8gd9104h</u> (retrieved 11 February 2012).
- Garrett, Andrew (1991) 'Indo-European reconstruction and historical methodologies', Language 67/4, 790-804; review of Vennemann, Theo (ed.) (1989) The New Sound of Indo-European: Essays in Phonological Reconstruction, Berlin: Mouton de Gruyter.
- Garrett, Andrew (1998) 'Adjarian's Law, the glottalic theory, and the position of Armenian', in Benjamin K. Bergen, Madelaine C. Plauché, and Ashlee C. Bailey (eds.) *Proceedings of the Twenty-Fourth Annual Meeting of the Berkeley Linguistics Society, February 14-16, 1998: Special Session on Indo-European Subgrouping and Internal Relations*, Berkeley: Berkeley Linguistics Society, 12-23, http://linguistics.berkeley.edu/~garrett/garrett-1998-bls.pdf (retrieved 17 December 2011).
- Garrett, Andrew (1999) 'A new model of Indo-European subgrouping and dispersal', in Steve S. Chang, Lily Liaw, and Josef Ruppenhofer (eds.) Proceedings of the Twenty-Fifth Annual Meeting of the Berkeley Linguistics Society, February 12-15, 1999, Berkeley: Berkeley Linguistics Society, 146-156,

http://linguistics.berkeley.edu/~garrett/BLS1999.pdf (retrieved 17 December 2011). Garson, James (2010) 'Connectionism', in Edward N. Zalta (ed.) *The Stanford*

- *Encyclopedia of Philosophy (Winter 2010 Edition)*, <u>http://plato.stanford.edu/archives/win2010/entries/connectionism/</u> (retrieved 21 March 2012).
- Gerfen, Chip (2001) 'Nasalized fricatives in Coatzospan Mixtec', *International Journal* of American Linguistics 67/4, 449-466.

- Gierut, Judith A. (2004) 'Enhancement of learning for children with phonological disorders', presentation, Sound to Sense, MIT, 11-13 June 2004, 164-172, <u>http://www.iub.edu/~sndlrng/papers/Gierut%2004c.pdf</u> (retrieved 3 February 2012).
- Gillam, Ronald Bradley, Thomas P. Marquardt, Frederick N. Martin (2010) *Communication Sciences and Disorders: From Science to Clinical Practice*, Sudbury: Jones and Bartlett Learning.
- Gillies, William (1993) 'Scottish Gaelic', in Ball and Fife (1993), 145-227.
- Gnanadesikan, Amalia Elizabeth (1997) *Phonology with ternary scales*, doctoral dissertation, University of Massachusetts Amherst.
- Gobl, Christer and Ailbhe Ní Chasaide (2010) 'Voice source variation and its communicative functions', in William J. Hardcastle, John Laver, and Fiona E. Gibbon (eds.) *The Handbook of Phonetic Sciences*, 378-423.
- Goedemans, Rob (1998) Weightless Segments: A Phonetic and Phonological Study concerning the Metrical Irrelevance of Syllable Onsets, The Hague: Holland Academic Graphics.
- Goldsmith, John (ed.) (1993) *The Last Phonological Rule: Reflections on Constraints and Derivations*, Chicago: University of Chicago Press.
- Goldsmith, John and M. Noske (2006) 'Autosegmental phonology and underspecification theory', in S. Auroux, K. Koerner, H.-J. Niederehe, and K. Versteegh (eds.) *History of the Language Sciences*, Berlin: Walter de Gruyter, <u>http://hum.uchicago.edu/~jagoldsm/Papers/AutosegmentalUnderspecificationWNoske</u> .pdf (retrieved 15 March 2012).
- Goldstein, Louis (1995) 'Lendu consonants and the role of overlapping gestures in sound change', in Connell and Arvaniti (1995), 386-92.
- Goldstein, Louis (2009) 'Consonant clusters', course slides, <u>http://sail.usc.edu/~lgoldste/ArtPhon/Papers/Week%2012/C%20Clusters.pdf</u> (retrieved 23 March 2012).
- Goldstein, Louis, I. Chitoran, and E. Selkirk (2007) 'Syllable structure as coupled oscillator modes: evidence from Georgian vs. Tashlhiyt Berber', in J. Trouvain and W. J. Barry (eds.) *Proceedings of the XVIth International Congress of Phonetic Sciences*, Saarbrücken, 241-244, online version,
 http://sail.usa.edu/.laoldsto/ArtPhon/Papers/Weak%2012/TP. G. roy. all. jup11.pdf

http://sail.usc.edu/~lgoldste/ArtPhon/Papers/Week%2012/TB-G_rev_all_jun11.pdf (retrieved 23 March 2012).

- Goldstein, Louis, H. Nam, E. Saltzman, and I. Chitoran (2008). 'Coupled oscillator planning model of speech timing and syllable structure', in *Proceedings of the 8th Phonetic Conference of China and the International Symposium on Phonetic Frontiers*, Beijing, 18-20 April 2008 (CD-ROM), online version, <u>http://sail.usc.edu/~lgoldste/ArtPhon/Papers/Week%2012/LG_final.pdf</u> (retrieved 23 March 2012).
- Golston, Chris and Wolfgang Kehrein (1998) 'Mazatec onsets and nuclei', *International Journal of American Linguistics* 64/4, 311-337.

- Gonzáles, Hebe. 2003. 'A typology of stops in South American Indian languages', *Proceedings of the Conference on the Indigenous Languages of Latin America I*, 23-25 October 2003, University of Texas at Austin, <u>http://www.ailla.utexas.org/site/cilla1/Gonzales_SA_Languages.pdf</u> (retrieved 30 July 2008).
- Gordon, Matthew (1998) 'The phonetics and phonology of non-modal vowels: a crosslinguistic perspective', *Berkeley Linguistic Society* 24, 93-105, <u>http://www.linguistics.ucsb.edu/faculty/gordon/Nonmodal.pdf</u> (retrieved 29 July 2008).
- Gordon, Matthew (2001) 'Linguistic aspects of voice quality with special reference to Athabaskan', *Proceedings of the 2001 Athabaskan Languages Conference*, 163-178, <u>http://www.linguistics.ucsb.edu/faculty/gordon/athabphonation.pdf</u> (retrieved 30 July 2008).
- Gordon, Matthew (2002) 'A phonetically driven account of syllable weight', *Language* 78/1, 51-80.
- Gordon, Matthew (2007) 'Functionalism in phonology', in de Lacy (ed.) (2007), 61-78.
- Gordon, Matthew and Ayla Applebaum (2006) 'Phonetic structures in Turkish Kabardian', *Journal of the International Phonetic Association* 36/2, 159-186 <u>http://www.linguistics.ucsb.edu/faculty/gordon/kabardianphonetics.pdf</u> (retrieved 29 July 2008).
- Gordon, Matthew and Ayla Applebaum (2010) 'Acoustic correlates of stress in Turkish Kabardian', *Journal of the International Phonetic Association* 40/1, 36-58, <u>http://cakhasa.com/pdf/Acoustsic_Correlates_Kabardian.pdf</u> (retrieved 8 February 2012).
- Gordon, Matthew, Brian Potter, John Dawson, Willem de Reuse, Peter Ladefoged (2001) 'Phonetic structures of Western Apache', *International Journal of American Linguistics* 67/4, 415-48.
- Gordon, Matthew and Peter Ladefoged (2001) 'Phonation types: A cross-linguistic overview', *Journal of Phonetics* 29, 383-406.
- Gordon, W. Terrence (2004) 'Langue and parole', in Carol Sanders (ed.) The Cambridge Companion to Saussure, Cambridge: Cambridge University Press, 76-87.

Goudswaard, Nelleke (2005) *The Begak (Ida'an) language of Sabah*, Utrecht: LOT: Netherlands Graduate School of Linguistics, <u>http://www.lotpublications.nl/publish/articles/001281/bookpart.pdf</u> (retrieved 30 July 2008).

- Grace, George W. (1989) 'On minimal native-speaker competence', *Ethnolinguistic Notes* 3/39, 583-589, <u>http://www2.hawaii.edu/~grace/eln39.html</u> (retrieved 30 December 2011).
- Gregerson, K. J. (1976) 'Tongue-root and register in Mon-Khmer', in P. Jenner, L. Thompson, and S. Starosta (eds.) *Austro-asiatic Studies* vol. 1, Honolulu: University of Hawaii Press, 323-369.

- Green, Antony D. (1997) *The Prosodic Structure of Irish, Scots Gaelic, and Manx*, PhD dissertation, Cornell University, <u>http://roa.rutgers.edu/files/196-0597/196-0597-</u> <u>GREEN-0-0.PDF</u> (retrieved 29 July 2008).
- Green, Antony D. (2003) 'The independence of phonology and morphology: the Celtic mutations', *ZAS Papers in Linguistics* 32: 47–86, <u>http://roa.rutgers.edu/files/652-0404/652-GREEN-0-0.PDF</u> (retrieved 13 December 2011).
- Green, Antony D. (2004) 'Opacity in Tiberian Hebrew: morphology, not phonology', ZAS Papers in Linguistics 37, 37-70, <u>http://www.zas.gwz-</u> <u>berlin.de/fileadmin/material/ZASPiL_Volltexte/zp37/zaspil37-green.pdf</u> (retrieved 25 September 2010).
- Greenberg, Joseph H. (1966) 'Language universals with special reference to feature hierarchies,' The Hague: Mouton.
- Greenberg, Joseph H. (1966b) 'Synchronic and diachronic universals in phonology', *Language* 42, 508-517.
- Greenberg, Joseph H. (1970) 'Some generalizations concerning glottalic consonants, especially implosives,' *International Journal of American Linguistics* 36, 123-145.
- Greenberg, Joseph H. (1975) 'Research on language universals', *Annual Review of Anthropology* 4, 75-94.
- Greenberg, Joseph H. (1978) 'Some generalizations concerning initial and final consonant clusters' in Joseph H. Greenberg (ed.) *Universals of Human Language*, vol. 2, Stanford: Stanford University Press, 243-279.
- Gregerson, Kenneth J. (1976) 'Tongue-root and register in Mon-Khmer,' *Oceanic Linguistics Special Publications*, No. 13, Afroasiatic Studies Part I, 323-369.
- Grégoire, Claire (2003) 'The Bantu languages of the forest', in Nurse and Philippson (2003), 349-370.
- Gurevich, Naomi (2004) Lenition and Contrast: The Functional Consequences of Certain Phonetically Conditioned Sound Changes.
- Haider, Hubert (1985) 'The fallacy of typology', Lingua 65, 1-27.

Hainsworth, J. B. (1968) Review of L. R. Palmer and John Chadwick (1966) Proceedings of the Cambridge Colloquium on Mycenaean Studies, Cambridge: Cambridge University Press, in *The Classical Review* New Series 18/1, 68-71.

- Hajek, John (2010) 'Aspiration in some varieties of Northern Aslian', Oceanic Linguistics 49/2, 359-368.
- Hajek, John and John Bowden (2002) 'A phonological oddity in the Austronesian area: ejectives in Waimoa', *Oceanic Linguistics* 41/1, 222-224.
- Hale, Mark (2007) Historical Linguistics: Theory and Method.
- Hale, Mark, Madelyn Kissock, and Charles Reiss (2006) 'Microvariation, variation, and the features of universal grammar', in press version, <u>http://modlang-phoneticc.concordia.ca/reiss/linguamicro.pdf</u> (retrieved 18 April 2010).
- Hale, Mark and Charles Reiss (1998) 'Formal and empirical arguments concerning phonological acquisition', *Linguistic Inquiry* 29/4, 656-83,

http://people.pwf.cam.ac.uk/bv230/OT/hale-reiss-1998.pdf (retrieved 21 January 2009).

- Hale, Mark and Charles Reiss (2000a) "Substance abuse" and "dysfunctionalism": current trends in phonology', *Linguistic Inquiry* 31, 157-169.
- Hale, Mark and Charles Reiss (2000b) 'Phonology as cognition', Rutgers Optimality Archive MS, <u>http://roa.rutgers.edu/view.php3?id=402</u> (retrieved 5 September 2009).
- Hall, Daniel Currie (2000) 'Infinity limited: constraining a constraint-based theory', University of Toronty phonology generals paper, <u>http://roa.rutgers.edu/files/455-0801/455-0801-HALL-0-0.PDF</u> (retrieved 8 October 2009).
- Hall, Daniel Currie (2010) 'Labial consonants in Mohawk', 18th Manchester Phonology Meeting, 20-22 May 2010, Manchester, <u>http://depot.knaw.nl/8298/1/mfm-handout.pdf</u> (retrieved 4 February 2012).
- Hall, Daniel Currie (2011) 'Contrast', ch. 2 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Hall, Jr., Robert A. (1974) Review in *Language* 50/2, 377-80 of Herbert J. Izzo (1972) *Tuscan and Etruscan: The Problem of Linguistic Substratum Influence in Central Italy.*
- Hall, Tracy Alan (1989) 'Lexical phonology and the distribution of German [ç] and [x]', *Phonology* 6, 1-18.
- Hall, Tracy Alan (2001) 'Introduction: phonological representations and phonetic implementation of distinctive features', in Hall (ed.) (2001), 1-40.
- Hall, Tracy Alan. (ed.) (2001) *Distinctive Feature Theory*, Phonology and Phonetics 2. Berlin: Mouton de Gruyter.
- Hall, Tracy Alan (2007) 'Segmental features', in de Lacy (ed.) (2007), 311-334.
- Hall, Tracy Alan and Silke Hamann (2003) 'Towards a typology of stop assibilation', ZAS Papers in Linguistics 32, 111-136, <u>http://user.phil-fak.uni-</u> duesseldorf.de/~hamann/HallHamann2003.pdf (retrieved 14 March 2012).
- Halle, John (2011) Review of François Dell and Mohammed Elmedlaoui (2008), *Poetic Meter and Musical Form in Tashlhiyt Berber Songs*, MS, Bard College Conservatory of Music, in *Language* 87/1, 181-189.
- Halle, Morris (1973) Review of Ladefoged (1971) in Language 49/4, 926-33.
- Halle, Morris (1995) 'Feature geometry and feature spreading', *Linguistic Inquiry* 26, 1-46.
- Halle, Morris (2009) 'Two comments on "The role of features in phonological inventories", in Raimy and Cairns (2009), 69-73.
- Halle, Morris, Bert Vaux, and Andrew Wolfe (2000) 'On feature spreading and the representation of place of articulation', *Linguistic Inquiry* 31/3, 387-444.
- Halle, Morris and Kenneth N. Stevens (1971) 'A note on laryngeal features', in *Quarterly progress report 101*, Cambridge, Massachussetts: MIT, Research Laboratory of Electronics, 198-213.

- Halle, Morris and Kenneth N. Stevens (2002[1971]) 'A note on laryngeal features', in Morris Halle (ed.) From Memory to Speech and Back: Papers on Phonetics and Phonology, 1954-2002, Berlin: Walter de Gruyter, 45-61.
- Halpert, Claire (2010) 'Place assimilation and its triggers', presentation, 18th Manchester Phonology Meeting, http://web.mit.edu/halpert/www/Site/Papers_files/Manchester2010Handout.pdf

http://web.mit.edu/halpert/www/Site/Papers_files/Manchester2010Handout.pdf (retrieved 20 March 2012).

- Hamann, Silke, Diana Apoussidou, and Paul Boersma (n.d.) 'Modelling the formation of phonotactic restrictions across the mental lexicon', MS, <u>http://www.fon.hum.uva.nl/paul/papers/CLSHamaApouBoers.pdf</u> (retrieved 19 June 2010).
- Hamp, Eric (2003) 'Prehistoric Greek phonological chronology, with some reflections on sibilance, aspiration, and spirancy', *Journal of Greek Linguistics* 4, 65-68.
- Hansson, Gunnar (2004) 'Long-distance voicing agreement: an evolutionary perspective', in M. Ettlinger, N. Fleischer, and M. Park-Doob (eds.) *Proceedings of*

the 30th Annual Meeting of the Berkeley Linguistics Society (2004), Berkeley: Berkeley Linguistics Society, 130-41,

http://faculty.arts.ubc.ca/gohansson/pdf/GH_BLS30_VoicAgr.pdf (retrieved 29 July 2008).

- Harrington, Jonathan (2010a) 'Acoustic phonetics', in William J. Hardcastle, John Laver, and Fiona E. Gibbon (eds.) *The Handbook of Phonetic Sciences*, 81-129.
- Harrington, Jonathan (2010b) Phonetic Analysis of Speech Corpora, Oxford: Blackwell.
- Harris, Jimmy G. (1999) 'States of the glottis for voiceless plosives', in J. J. Ohala, Y. Hasegawa, M. Ohala, D. Granville, and A. C. Bailey (eds.) Proceedings of the 14th International Congress of Phonetic Sciences 3, 2041-2044.
- Harris, Jimmy G. (2001) 'States of the glottis of Thai voiceless stops and affricates', in M. R. Kalaya Tingsabadh and A. S. Abramson (eds.) *Essays in Tai Linguistics*, 3-12, Chulalongkorn University Press,

http://sealang.net/sala/archives/pdf8/harris2001states.pdf (retrieved 1 March 2012).

- Harris, Katherine S. (1981) 'Electromyography as a technique for laryngeal investigation', Haskins Laboratories Status Report on Speech Research SR-66, 1-33, <u>www.haskins.yale.edu/sr/SR066/SR066_01.pdf</u> (retrieved 16 January 2012).
- Harris, Tom (2010) 'Phonation in Sumi nasals', Australian Speech, Science, and Technology Association conference (14-16 December 2010) proceedings, 22-25, <u>http://assta.org/sst/SST-10/SST2010/PDF/AUTHOR/ST100041.PDF</u> (retrieved 14 January 2012).
- Hasler, Juan A. (1961) 'Tetradialectologia Nahua', in A William Cameron Townsend en el vigésimo-quinto aniversario del Instituto Lingüístico de Verano, 455-464.
- Haspelmath, Martin (2006) 'Against markedness (and what to replace it with)', *Journal* of Linguistics 42, 25-70.
- Haspelmath, Martin (2008a) 'Frequency vs. iconicity in explaining grammatical asymmetries', *Cognitive Linguistics* 19/1, 1-33.

- Haspelmath, Martin (2008b) 'Loanword typology: steps toward a systematic crosslinguistic study of lexical borrowability', in Thomas Stolz, Dik Bakker and Rosa Salas Palomo (eds.) *Aspects of Language Contact: New Theoretical, Methodological and Empirical Findings with Special Focus on Romancisation Processes*, Berlin: Mouton de Gruyter, 43-62, <u>http://www.eva.mpg.de/lingua/staff/haspelmath/pdf/LWT.pdf</u> (retrieved 16 Feburary 2012).
- Haspelmath, Martin J. (2002) 'Functional and formal explanation in linguistics', class handouts, Düsseldorf Summer School, co-taught with Frederick J. Newmeyer, <u>http://email.eva.mpg.de/~haspelmt/Haspelmath9Handouts.pdf</u> (retrieved 29 July 2008).
- Haudricourt, A.-G. (1950) 'Les consonnes préglottalisées en Indochine', *Bulletin de la Société de Linguistique de Paris* 46/1 (no. 132), 172-182.
- Haudricourt, A.-G. (1975) 'Les mutations consonantiques (occlusives) en indoeuropéen' in M. J. Moïnfar (ed.) *Mélanges linguistiques offerts à Emile Benveniste*, Paris: Société de Paris, 262-272.
- Haugeland, John (1997) 'What is mind design?' in John Haugeland (ed.) *Mind Design II*, Massachussetts, Cambridge: MIT Press, <u>http://www.cog.jhu.edu/faculty/smolensky/050.313-613/Readings/Haugeland</u> %2097%20Mind%20Design%20chap1.pdf (retrieved 19 June 2010).
- Havers, Wilhelm (1946) *Neuere Literatur zum Sprachtabu*, Sitzungberichte 223/5, Wien: Akademie der Wissenschaften.
- Hay, Jennifer, Paul Warren and Katie Drager (2006) 'Factors influencing speech perception in the context of a merger-in-progress', *Journal of Phonetics* 34/4, 458-484, <u>http://www.ling.canterbury.ac.nz/jen/documents/hay-warren-drager.pdf</u> (retrieved 22 May 2010).
- Hayes, Bruce (1989) 'Compensatory lengthening in moraic phonology', *Linguistic Inquiry* 20/2, 252-306,

http://www.linguistics.ucla.edu/people/hayes/Papers/HayesCompensatoryLengthening 1989Searchable.pdf (retrieved 22 August 2009).

Hayes, Bruce (1992) 'On what to teach the undergraduates: some changing orthodoxies in phonological theory', in Ik-Hwan Lee (ed.) (1995), *Linguistics in the Morning Calm 3*, Seoul, 59-77, prepublication draft
http://www.linguistics.ucla.edu/people/hayes/SECOL/Orthodox.pdf (retrieved 5

<u>http://www.linguistics.ucla.edu/people/hayes/SECOL/Orthodox.pdf</u> (retrieved 5 February 2009).

Hayes, Bruce (1997) 'Phonetically driven phonology: The role of Optimality Theory and Inductive Grounding', MS, UCLA,

http://www.linguistics.ucla.edu/people/hayes/phonet/indgroun.pdf (retrieved 18 January 2009).

Hayes, Bruce (1998) 'On the richness of paradigms, and the insufficiency of underlying representations to account for them', Stanford talk handout, 2 April 1998, <u>http://www.linguistics.ucla.edu/people/hayes/talks/RichnessofParadigms.pdf</u> (retrieved 5 February 2009).

- Hayes, Bruce (1999) 'Phonological restructuring in Yidin^y and its theoretical consequences', in Ben Hermans and Marc Oostendorp (eds.) *The Derivational Residue in Phonological Optimality Theory*, Amsterdam: John Benjamins.
- Hayes, Bruce (2004) 'Phonological acquisition in Optimality Theory: the early stages', ch. 5 in Kager et al. (2004), 158-203.
- Hayes, Bruce, Robert Kirchner, and Donca Steriade (eds.) (2004) *Phonetically Based Phonology*, Cambridge: Cambridge University Press.
- Hayes, Bruce, Kie Zuraw, Péter Siptár, and Zsuzsa Londe (2008) 'Natural and unnatural constraints in Hungarian vowel harmony', MS, <u>http://www.linguistics.ucla.edu/people/hayes/papers/HayesZurawSiptarLondeNatural UnnaturalConstraintsInHungarianSept2008.pdf</u> (retrieved 5 February 2009).
- Hayward, K. M. (1989) 'Review Article: The Indo-European language and the history of its speakers; the theories of Gamkrelidze and Ivanov', *Lingua* 78, 37-86.
- Hayward, K. M. and G. Lindsey (1988) 'The stop consonants of Tonga: a laryngographic study', paper presented at the Colloquium of British Academic Phoneticians, Dublin, April 1988.
- Hayward, K. M. and Muljono (1991) 'The dental/alveolar contrast in Javanese', *Bulletin* of the School of Oriental and African Studies 54, 126-144.
- Headley, Robert K., Jr. (1998) 'Cham evidence for Khmer sound changes', in Thomas, David (ed.) Papers in Southeast Asian Linguistics No. 15: Further Chamic Studies, Canberra: The Australian National University 21-29, <u>http://sealang.net/sala/archives/pdf8/headley1998cham.pdf</u> (retrieved 1 March 2012).
- Heath, Jeff (2007) 'Grammar of Humburi Senni', MS, <u>http://www-</u> personal.umich.edu/~jheath/Hombori_Grammar_entire.pdf (retrieved 20 June 2010).
- Helgason, Pétur (2002) Preaspiration in the Nordic Languages: Synchronic and Diachronic Aspects, PhD dissertation, Stockholm University, <u>http://www.lingfil.uu.se/personal/petur/pdf-filer/helgason_thesis.pdf</u> (retrieved 29 July 2008).
- Helgason, Pétur and Catherine Ringen (2008) 'Voicing and aspiration in Swedish stops', *Journal of Phonetics* 36, 607-628.
- Henke, Eric, Ellen M. Kaisse, and Richard Wright (in press) 'Is the Sonority Sequencing Principle epiphenomenal?', in Stephen George Parker (ed.) *The Sonority Controversy*, Berlin: Mouton de Gruyter.
- Henderson, Eugenie J. A. (1970) 'Acoustic features of certain consonants and consonant clusters in Kabardian', *Bulletin of the School of Oriental and African Studies, University of London*, 33/1, 92-116.
- Henton, Caroline, Peter Ladefoged, and Ian Maddieson (1992) 'Stops in the world's languages', *Phonetica* 49, 65-101.
- Herbert, Robert K. (1986) *Language Universals, Markedness Theory, and Natural Phonetic Processes*, Trends in Linguistics Studies and Monographs 25, Berlin: Mouton de Gruyter.

Heselwood, Barry (1996) 'Glottal states and emphasis in Baghdadi and Cairene Arabic: synchronic and diachronic aspects', University of Durham Center for Middle Eastern and Islamic Studies, Occasional Paper No. 53, www.personal.leeds.ac.uk/~lnpbch/Glottal%20states%20and%20emphasis.pdf

(retrieved 30 July 2008).

- Hickey, Raymond (2004) 'Mergers, near-mergers and phonological interpretation', in Christian J. Kay, Carole Hough and Irené Wotherspoon (eds.) New Perspectives in English Historical Linguistics, Amsterdam: John Benjamins, 125-137, <u>http://www.unidue.de/~lan300/22_Mergers_and_Near_Mergers_%28Hickey%29.pdf</u> (retrieved 27 March 2012).
- Hill, Jane H. (2007) "External evidence" in historical linguistic argumentation: subgrouping in Uto-Aztecan', paper given at the Workshop on Alternative Approaches to Language Classification at LSA Linguistic Institute at Stanford University, July 2007, <u>http://aalc07.psu.edu/papers/HillPaloAlto0707UAsubgroups.pdf</u> (retrieved 29 July 2008).
- Hinnebusch, T. H. (1975) 'A reconstructed chronology of loss: Swahili class 9/10', *Proceedings of the Sixth Conference on African Linguistics*, Ohio State University Working Papers in Linguistics 20, 32-41.
- Hirose, Hajime (1971a) 'An electromyographic study of laryngeal adjustments during speech articulation: a preliminary report', Haskins Laboratories Status Report on Speech Research SR-25/26, 107-116,

http://www.haskins.yale.edu/SR/SR025/SR025_11.pdf (retrieved 13 January 2012).

- Hirose, Hajime (1971b) 'Laryngeal adjustments for vowel devoicing in Japanese: an electromyographic study', Haskins Laboratories Status Report on Speech Research SR-28, 157-165, <u>http://www.haskins.yale.edu/sr/SR028/SR028_06.pdf</u> (retrieved 13 January 2012).
- Hirose, Hajime (1971c) 'Electromyography of the articulatory muscles: current instrumentation and technique', Haskins Laboratories Status Report on Speech Research SR-25/26, 74-86, <u>www.haskins.yale.edu/sr/SR025/SR025_08.pdf</u> (retrieved 16 January 2012).
- Hirose, Hajime (1977) 'Laryngeal adjustments in consonant production', *Phonetica* 34, 289-294.
- Hirose, Hajime (1997) 'Investigating the physiology of laryngeal structures', in William J. Hardcastle and John Laver (eds.) *The Handbook of Phonetic Sciences*, Oxford: Blackwell, 116-136.
- Hirose, Hajime (2010) 'Investigating the physiology of laryngeal structures', in William J. Hardcastle, John Laver, and Fiona E. Gibbon (eds.) *The Handbook of Phonetic Sciences*, 130-152.
- Hirose, Hajime and T. Gay (1972) 'The activity of the intrinsic laryngeal muscles in voicing control: an electromyographic study', *Phonetica* 25, 140-64 (a version of this paper is also available as <u>http://www.haskins.yale.edu/sr/SR028/SR028_04.pdf</u>, retrieved 18 February 2012).

- Hirose, Hajime and T. Gay (1973) 'Laryngeal control in vocal attack: an electromyographic study', *Folia Phoniatrica* 25, 203-213, <u>http://www.haskins.yale.edu/Reprints/HL0141.pdf</u> (similar undated version: <u>http://www.haskins.yale.edu/sr/SR029/SR029_04.pdf</u>) (retrieved 18 February 2012).
- Hirose, Hajime, Charles Y. Lee, and Tatsujiro Ushijima (1973) 'Laryngeal control in Korean stop production', Haskins Laboratories Status Report on Speech Research SR-34, 191-201, <u>http://www.haskins.yale.edu/sr/SR034/SR034_14.pdf</u> (retrieved 13 January 2012).
- Hirose, Hajime, L. Lisker, and A. S. Abramson (1972) 'Physiological aspects of certain laryngeal features in stop production', Haskins Laboratories Status Report on Speech Research SR-31/32, 183-191, <u>http://www.haskins.yale.edu/SR/sr031/SR031_16.pdf</u> (retrieved 18 February 2012).
- Hirose, Hajime and Seiji Niimi (1991) 'The relationship between glottal opening and the transglottal pressure differences during consonant production', in Thomas Baer, Clarence Sasaki, and Katherine S. Harris (eds.) *Laryngeal Function in Phonation and Respiration*, 381-90.
- Hoard, James E. (1978) 'Obstruent voicing in Gitksan: some implications for distinctive feature theory', in Eung-Do Cook and Jonathan Kaye (eds.) *Linguistic Studies of Native Canada*, Vancouver: University of British Columbia Press, 111-119.
- Hock, H. H. (1986) *Principles of Historical Linguistics*. Berlin: Mouton de Gruyter. Hockett, C. F. (1955) *A Manual of Phonology*. Baltimore.
- Holt, D. Eric (2003) *Optimality Theory and Language Change*, Dordrecht: Kluwer Academic Publishers.
- Hoole, Phil and Kiyoshi Honda (2011) 'Automaticity vs. feature-enhancement in the control of segmental F0', in G. Nick Clements and Rachid Ridouane (eds.) *Where Do Phonological Features Come From*?, Amsterdam: John Benjamins, 131-174.
- Holton, Gary (2001) 'Fortis and lenis fricatives in Tanacross Athapaskan', *International Journal of American Linguistics* 67/4, 396-414.
- Hombert, Jean-Marie, John J. Ohala, and William G. Ewan (1979) 'Phonetic explanations for the development of tones', *Language* 55/1, 37-58, <u>http://www.jstor.org/stable/pdfplus/412518.pdf?acceptTC=true</u>, (retrieved 15 February 2012).
- Honda, K. (1996) 'Organization of tongue articulation for vowels', *Journal of Phonetics* 24, 39-52.
- Honda, K. (2008) 'Physiological processes of speech production', in Jacob Benesty, M.Mohan Sondhi, and Yiteng Huang (eds.) Springer Handbook of Speech Processing,Berlin: Springer-Verlag, 7-26.
- Honeybone, Patrick (2004) 'The rise of Optimality Theory in mid-twentieth century London', Linguistic Circle talk, Edinburgh, January 2004, <u>http://www.englang.ed.ac.uk/people/otft.pdf</u> (retrieved 18 January 2009).
- Honeybone, Patrick (2005) 'Diachronic evidence in segmental phonology: the case of obstruent laryngeal specifications', in Marc van Oostendorp and J. van de Weijer

(eds.) *The Internal Organization of Phonological Segments*, Berlin: Mouton de Gruyter, 319-354.

- Hong, Kihwan, Seiji Niimi, and Hajime Hirose (1991) 'Laryngeal adjustments for Korean stops, affricates and fricatives – an electromyographic study', *Annual Bulletin* of the Research Institute of Logopedics and Phoniatrics 25, 17-31, <u>http://www.umin.ac.jp/memorial/rilp-tokyo/R25/R25_017.pdf</u> (retrieved 1 February 2012).
- Hong, Ki Hwan, Hyun Ki Kim, Seiji Niimi (2002) 'Laryngeal Gestures During Stop Production Using High-Speed Digital Images', *Journal of Voice* 16/2, 207-14.
- Hoole, Phillip (2004) 'Abbildungen und Texte zu den Themen Vibrationszyklus, Bernoulli-effekt und myoelastische-aerodynamische Theorie', course handout, <u>http://www.phonetik.uni-</u>

<u>muenchen.de/~hoole/kurse/handouts_phys1/pk_bernoulli_text.pdf</u> (retrieved 9 January 2012).

- Hoole, Phillip (2006a) Experimental Studies of Laryngeal Articulation, Habilitationsschrift, Ludwig-Maximilians-Universität, 'Part I: Electromyographic investigation of laryngeal activity in vowel intrinsic pitch and consonant voicing', <u>http://www.phonetik.uni-muenchen.de/~hoole/pdf/habilemg_chap_all.pdf</u> (retrieved 2 March 2012).
- Hoole, Phillip (2006a) *Experimental Studies of Laryngeal Articulation*, Habilitationsschrift, Ludwig-Maximilians-Universität, 'Part II: Laryngeal-oral coordination in consonant sequences', <u>http://www.phonetik.uni-</u> <u>muenchen.de/~hoole/pdf/habilpgg_chap_all.pdf</u> (retrieved 2 March 2012).
- Hopkins, Alice W. (1987) 'Vowel dominance in Mohawk', *International Journal of American Linguistics* 53/4, 445-459.
- Hopper, Paul J. (1973) 'Glottalized and murmured occlusives in Indo-European,' *Glossa* 7/2, 141-67.
- Hopper, Paul J. (1977a) 'Indo-European consonantism and the New Look,' *Orbis* 26/1, 57-72.
- Hopper, Paul J. (1977b) 'The typology of the Proto-Indo-European segmental inventory', *The Journal of Indo-European Studies* 5/1, 1977, 41-53.
- Hopper, Paul J. (1981) "Decem" and "taihun" languages' in Arbeitman and Bomhard (1981), Part I, 133-142.
- Hopper, Paul J. (1989) 'Remarks on P. Baldi and R. Johnston-Staver, "Historical Italic phonology in typological perspective" in Vennemann (1989).
- Howe, Daryn (2000) *Oowekyala Segmental Phonology*, PhD dissertation, University of British Columbia, <u>http://depts.washington.edu/wll2/files/howe_00_diss.pdf</u> (retrieved 2 April 2010).
- Howe, Darin (2003) *Segmental Phonology*, MS, University of Calgary, <u>http://www.ai.mit.edu/projects/dm/featgeom/howe-segphon-book.pdf</u> (retrieved 10 July 2010).

- Howe, Darin (2004) 'Vocalic dorsality in Revised Articulator Theory', MS, University of Calgary. <u>http://www.ucalgary.ca/dflynn/files/dflynn/Howe2004.pdf</u> (retrieved 29 July 2008).
- Howe, Daryn and Sean Fulop (2005) 'Acoustic features in Athabaskan', MS, <u>http://www.ucalgary.ca/dflynn/files/dflynn/LSA2005.pdf</u> (retrieved 10 December 2011).
- Howe, Darin and Douglas Pulleyblank (2001) 'Patterns and timing of glottalization', *Phonetics in Phonology* 18/1, 45-80.
- Hudgins, C. V. and Stetson, R. H. (1935) 'Voicing of consonants by depression of the larynx', *Archives Néerlandaises de Phonétique Expérimentale* 11, 1-28.
- Huffman, Franklin E. (1976) 'The Register Problem in Fifteen Mon-Khmer Languages,' Oceanic Linguistics Special Publication, No. 13, Austroasiatic Studies Part I, 575-89, <u>http://www.jstor.org/stable/pdfplus/20019172.pdf?acceptTC=true</u> (retrieved 18 July 2011).
- Huld, Martin (1986) 'On the unacceptability of the Indo-European voiced stops as ejectives,' *Indogermanische Forschungen* 91, 67-78.
- Hume, Elizabeth (1996) 'Coronal consonant, front vowel parallels in Maltese', *Natural Language and Linguistic Theory* 14, 165-203.
- Hume, Elizabeth (2003) 'Language-specific markedness: the case of place of articulation', *Studies in Phonetics, Phonology and Morphology* 9, 295-310, prepublication version <u>http://www.ling.ohio-</u> state.edu/~ehume/papers/Hume_LgSpecificMarkedness.pdf (retrieved 29 July 2008).
- Hume, Elizabeth (2004) 'Deconstructing markedness: a predictability-based approach', to appear in *Proceedings of the Berkeley Linguistic Society 2004*, <u>http://www.ling.ohio-state.edu/~ehume/papers/Hume_markedness_BLS30.pdf</u> (retrieved 29 July 2008).
- Hume, Elizabeth and David Odden (1996) 'Reconsidering [consonantal]', *Phonology* 13/3, 345-76.
- Hume, Elizabeth and Georgios Tserdanelis (2002) 'Labial unmarkedness in Sri Lankan Portuguese Creole', *Phonology* 19, 441-58.
- Hunter, Eric J. and Ingo R. Titze (2007) 'Refinements in modeling the passive properties of laryngeal soft tissue', *Journal of Applied Physiology* 103, 206-219, http://jap.physiology.org/content/103/1/206.full.pdf+html (retrieved 24 March 2012).
- Hutters, B. (1985) 'Vocal fold adjustments in aspirated and unaspirated stops in Danish', *Phonetica* 42, 1-24.
- Hyman, Larry (2003) 'Segmental phonology', in Gérard Nurse and Derek Philippson (eds.) *The Bantu Languages*, 42-44.
- Hyman, Larry (2006) 'Phonological theory and fieldwork: is the gap widening?' Paper given at the 14th Manchester Phonology Meeting, 25-27 May, 2006, http://www.siue.edu/~khildeb/Hyman 14.pdf (retrieved 17 October 2009).
- Idsardi, William (1997a) 'Phonological derivations and historical changes in Hebrew spirantization', in Roca, Iggy (ed.), *Derivations and Constraints in Phonology*,

Oxford: Clarendon Press, 367-92.

<u>http://www.ling.umd.edu/~idsardi/papers/1997rocahebrew.pdf</u> (retrieved 9 July 2010). Idsardi, William (1997b) 'Sympathy creates chaos', MS,

http://www.ling.umd.edu/~idsardi/papers/1998sympathy.pdf (retrieved 10 July 2010).

- Idsardi, William (1998) 'Tiberian Hebrew spirantization and phonological derivations', *Linguistic Inquiry* 29/1, 37-73, <u>http://www.ling.umd.edu/~idsardi/papers/1998li.pdf</u> (retrieved 10 July 2010).
- Idsardi, William (2000) 'Clarifying opacity', *The Linguistic Review* 17, 337-50, <u>http://www.ling.umd.edu/~idsardi/papers/2000tlropacity.pdf</u> (retrieved 10 July 2010).
- Idsardi, William (2002) 'Further opacity issues: spontaneous L2 opacity', *Proceedings of the 2002 LSK International Summer Conference*, Linguistic Society of Korea, vol. 2, 259-65, <u>http://www.ling.umd.edu/~idsardi/papers/2002lsk.pdf</u> (retrieved 10 July 2010).
- Ikekeonwu, Clara (1999) 'Igbo', in International Phonetic Association (1999), 108-10.
- Inkelas, Sharon (1995) 'The consequences of optimization for underspecification', *Proceedings of NELS 25*, 287-302.
- International Phonetic Association (ed.) (1999) *Handbook of the International Phonetic Association*, Cambridge: Cambridge University Press.
- Iskarous, Khalil (2010) 'Vowel constrictions are recoverable from formants', *Journal of Phonetics* 38, 375-87.
- Itô, Junko, Armin Mester, and Jaye Padgett (1995) 'Licensing and underspecification in Optimality Theory', *Linguistic Inquiry* 26/4, 571-613.
- Itô, Junko, and Armin Mester (2003) 'Lexical and postlexical phonology in Optimality Theory: evidence from Japanese', MS,

http://people.ucsc.edu/~mester/papers/2003_ito_mester_lexpostlex.pdf (retrieved 6 September 2008), revised version of article of same title in G. Fanselow and C. Féry (eds.) (2003) *Linguistische Berichte. Sonderheft 11: Resolving Constraints in Grammars.*

Itoh, Motonobu, Sumiko Sasanuma, Itaru F. Tatsumi, and Yo Kobayashi (1979) 'Voice onset time characteristics of apraxia of speech', *Annual Bulletin of the Research Institute of Logopedics and Phoniatrics* 13, 123-132, http://www.umin.ac.jp/memorial/rilp-tokyo/R13/R13_123.pdf (retrieved 20 February)

2012).

Itoh, Motonobu, Sumiko Sasanuma, Itaru F. Tatsumi, Shuko Hata, Yoko Fukusako, and Tsutomu Suzuki (1980) 'Voice onset time characteristics of apraxia of speech part II', *Annual Bulletin of the Research Institute of Logopedics and Phoniatrics* 14, 273-284, <u>http://www.umin.ac.jp/memorial/rilp-tokyo/R14/R14_273.pdf</u> (retrieved 20 February 2012).

Iverson, Gregory K. (1983) 'On glottal width features', Lingua 60, 331-339.

Iverson, Gregory K. (1985) 'Grassman's Law in Indo-European' in *Folia Linguistica Historica* 6/2, 203-213.

Iverson, Gregory K. and Sang-Cheol Ahn (2007) 'English voicing in dimensional theory', *Language Sciences* 29, 247-269,

https://pantherfile.uwm.edu/iverson/www/engvce.pdf (retrieved 15 February 2012).

Iverson, Gregory K and Joseph C. Salmons (1995) 'Aspiration and laryngeal representation in Germanic', *Phonology* 12, 369-396, http://www.uwm.edu/~iverson/papers.htm (retrieved 24 March 2012).

Iverson, Gregory K and Joseph C. Salmons (1992) 'The phonology of the Proto-Indo-European root structure constraints', *Lingua* 87, 293-320.

- Iverson, Gregory K and Joseph C. Salmons (1999) 'Laryngeal bias in Germanic', *Linguistische Berichte* 178, 135-151.
- Iverson, Gregory K. and Joseph C. Salmons (2003a) 'Legacy specification in the laryngeal phonology of Dutch', *Journal of Germanic Linguistics* 15/1, 1-26.
- Iverson, Gregory K. and Joseph C. Salmons (2003b) 'Laryngeal enhancement in early Germanic', *Phonology* 20, 43-74 <u>http://www.uwm.edu/~iverson/enhancegmc.pdf</u>, <u>http://wisc.academia.edu/JosephSalmons/Papers/915542/Laryngeal_enhancement_in_early_Germanic</u> (retrieved 16 February 2012).
- Iverson, Gregory K. and Joseph C. Salmons (2008) 'Germanic aspiration: phonetic enhancement and language contact', *Sprachwissenschaft* 33, 257-278, preprint <u>https://pantherfile.uwm.edu/iverson/www/gmcaspsprachwissen.pdf</u> (retrieved 16 February 2012).
- Iverson, Gregory K. and Joseph C. Salmons (2011) 'Final devoicing and final laryngeal neutralization', ch. 69 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Jaeger, Jeri J. and John J. Ohala (1984) 'On the structure of phonetic categories', *Proceedings of the Tenth Annual Meeting of the Berkeley Linguistics Society*, 15-26, <u>http://elanguage.net/journals/index.php/bls/article/viewFile/2353/2315</u> (retrieved 5 December 2011).
- Jaiswal, Sanyukta (2011) Cricothyroid Muscle Activity at Voicing Transitions, PhD dissertation, University of Iowa, <u>http://ir.uiowa.edu/cgi/viewcontent.cgi?</u> article=2374&context=etd (retrieved 13 January 2012).
- Jakobson, Roman (1958) 'Typological studies and their contribution to historical comparative linguistics,' *Proceedings of the 8th International Congress of Linguists*, Oslo: Oslo University Press, 17-25.
- Jakobson, Roman (1962) *Selected Writings*, vol. 1 *Phonological Studies*, 'S-gravenhage: Mouton.
- Jakobson, Roman and Morris Halle (1956) Fundamentals of Language.
- Jakobson, Roman, C. Gunnar M. Fant, and Morris Halle (1954) *Preliminaries to Speech Analysis: The Distinctive Features and Their Correlates*, Cambridge, MIT Press.
- Jamison, Stephanie W. (1991) Review in *Journal of the American Oriental Society*, 111/2, 419-421 of Lubotsky (1988).
- Janda, Richard D. and Brian D. Joseph (1989) 'In further defense of a non-phonological account for Sanskrit root-initial aspiration alternations', *ESCOL '88: Proceedings of*

the Fifth Eastern States Conference on Linguistics, The Ohio State University Department of Linguistics, 246-260, <u>http://www.ling.ohio-</u>

state.edu/~bjoseph/publications/1989aspiration.pdf (retrieved 16 January 2012).

- Jany, Carmen, Matthew Gordon, Carlos M. Nash, and Nobutaka Takara (2007) 'How universal is the sonority hierarchy? A cross-linguistic acoustic study', *Proceedings of the International Congress of Phonetic Sciences* XVI, Saarbrüken, Germany 6-8 August 2007, 1401-1404, <u>http://www.icphs2007.de/conference/Papers/1096/1096.pdf</u> (retrieved 9 December 2011).
- Jacques, Guillaume (2011) 'A panchronic study of aspirated fricatives, with new evidence from Pumi', *Lingua* 121, 1518-1538, http://cnrs.academia.edu/GuillaumeJacques/Papers/1008503/A_panchronic_study_of_aspirated_fricatives_with_new_evidence_from_Pumi (retrieved 9 February 2012).
- Jarkey, Nerida (1987) 'An investigation of two alveolar stop consonants in White Hmong', *Linguistics of the Tibeto-Burman Area* 10/2, 57-60 (partial article), <u>http://sealang.net/sala/archives/pdf4/jarkey1987investigation.pdf</u> (retrieved 16 January 2012).
- Jebbour, Abdelkrim (1999) 'Syllable weight and syllable nuclei in Tachelhit Berber of Tiznit', *Cahiers de Grammaire 24, « Phonologie : théorie et variation »*, 95-116, <u>http://w3.erss.univ-tlse2.fr/textes/publications/CDG/24/CG24-6-Jebbour.pdf</u> (retrieved 1 December 2011).
- Jessen Michael (1998) *Phonetics and Phonology of Tense and Lax Obstruents in German*, Amsterdam: John Benjamins.
- Jessen Michael (2001) 'Phonetic implementation of the distinctive auditory features [voice] and [tense] in stop consonants', in Hall (2001), 237-294.
- Jessen, Michael and Catherine Ringen (2002) 'Laryngeal features in German', *Phonology* 19, 189-218, <u>http://www.uiowa.edu/~linguist/faculty/ringen/laryngeal.pdf</u> (retrieved 25 March 2012).
- Jessen, Michael and Roux, Justus C. (2002) 'Voice quality differences associated with stops and clicks in Xhosa', *Journal of Phonetics* 30, 1-52.
- Jha, Sunil Kumar (1993) 'The Maithili consonants', *Contributions to Nepalese Studies* 20/1, 91-103,
- http://himalaya.socanth.cam.ac.uk/collections/journals/contributions/pdf/CNAS_20_0 1_06.pdf (retrieved 14 January 2012).
- Job, Michael (1989) 'Sound change typology and the "Ejective Model", in Vennemann (1989), 123-26.
- Job, Michael (1995) 'Did PIE have glottalized stops?', *Historische Sprachforschung* 107, 244-261.
- Johannsson, K. F. (1898) 'Anlautendes indogerman. b-', [Kuhns] Zeitschrift für vergleichende Sprachforschung 36 (1898-1902) 343-390.
- Johnson, Keith (2003) *Acoustic and Auditory Phonetics*, second edition, Oxford: Blackwell.

Joseph, Brian D. and Richard D. Janda (1988) 'On the unity of Sanskrit aspiration', Discussion Papers of the Sixth International Phonology Meeting and Third International Morphology Meeting, vol. 1: Phonology, Wiener Linguistiche Gazette Supplement 6, 29-31, <u>http://www.ling.ohio-</u>

state.edu/~bjoseph/publications/1988sktaspiration.pdf (retrieved 16 January 2012).

- Joseph, Frian D. and Rex E. Wallace (1994) 'Proto-Indo-European voiced aspirates in Italic', *Historische Sprachforschung* 107/2, 244-261.
- Jost, Gippert (1998) Review of Gamkrelidze and Ivanov (1995), *Indo-European and the Indo-Europeans. A Reconstruction and Historical Analysis of a Proto-Language and a Proto-Culture*, in *Beitrage zur Namenforschung* 33/1, 39-54.
- Jucquois, G. (1966) 'La structure des racines en indoeuropéen envisagée d'un point de vue statistique, *Linguistic research in Belgium*, Wetteren.
- Jucquois, G. (1977) 'The typology of the Proto-Indo-European segmental inventory, *The Journal of Indo-European studies* 5/1.
- Jun, Jongho (2004) 'Place assimilation', in Hayes et al. (2004), http://ling.snu.ac.kr/jun/work/pbot.pdf (retrieved 30 July 2008).
- Kagaya, R. (1974) 'A fiberscopic and acoustic study of the Korean stops, affricates and fricatives', *Journal of Phonetics* 2, 161-180.
- Kagaya, Ryohei and Hajime Hirose (1975) 'Fiberoptic, electromyographic and acoustic analyses of Hindi stop consonants', *Annual Bulletin of the Research Institute of Logopedics and Phoniatrics* 9, 27-46.
- Kager, René (1999) Optimality Theory, Cambridge: Cambridge University Press.
- Kager, René, Joe Pater, and Wim Zonneveld (eds.) (2004) *Constraints in Phonological Acquisition*, Cambridge: Cambridge University Press.
- Kahigi, Kulikoyela (1988) *Aspects of Sumbwa Diachronic Phonology*. Ph.D. Dissertation, Michigan State University.
- Kahigi, Kulikoyela (n.d.) 'Bantu reconstruction and the stops versus continuants controversy', <u>http://archive.lib.msu.edu/DMC/African</u>
 <u>%20Journals/pdfs/Utafiti/vol9no2/aejp009002002.pdf</u> (retrieved 22 May 2008), with the note, "Another version appeared in Kahigi (1988)."</u>
- Kaisse, Ellen M. (n.d.) 'Lexical phonology: where were we when we left off?', handout, <u>www.web.gc.cuny.edu/dept/lingu/events/phonology_symposium/Kaissehandout.doc</u> (retrieved 6 September 2008).
- Kaisse, Ellen M. (2011) 'The stricture features', ch. 13 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Kaisse, Ellen M. and April McMahon (2011) 'Lexical Phonology and the Lexical Syndrome', ch. 94 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Kallestinova, Elena (2004) 'Voice and aspiration in stops in Turkish', *Folia Linguistica* 38/1, 115-43.

- Kammerzell, Frank (1999) (Besprechungsaufsatz.) 'Glottaltheorie, Typologie, Sprachkontakte und Verwandtschaftsmodelle', *Indogermanische Forschungen* 104, 234-71: review of Gamkrelidze and Ivanov (1995).
- Kang, Yoonjung (2001) 'Loanword phonology', ch. 95 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Kari, James (1979) *Athabaskan Verb Theme Categories: Ahtna*, Alaska Native Language Center Research Papers 2.
- Karttunen, Lauri (2006) 'The insufficiency of paper-and-pencil linguistics: the case of Finnish prosody', <u>http://roa.rutgers.edu/files/818-0406/818-KARTTUNEN-0-0.PDF</u> (retrieved 10 September 2009).
- Kaufman, Terrence (1968) Review of Benjamin Elson's Studies in Ecuadorian Indian Languages: I, in International Journal of American Linguistics 34/2, 153-61.
- Kavitskaya, Darya (2002) *Compensatory lengthening: phonetics, phonology, diachrony*, New York: Routledge.
- Kaye, Alan S. (1997) Review of Masayoshi Shibatani and Theodora Bynon (eds.) 1995, Approaches to Language Typology, Oxford: Oxford University Press, in International Journal of American Linguistics 63/4 (Oct.), 526-32.
- Kaye, Alan S. (ed.) (1997) Phonologies of Asia and Africa, Winona Lake: Eisenbrauns.
- Keating, Patricia A., W. Linker, and M. Huffman (1983) 'Patterns of allophone distribution for voiced and voiceless stops', *Journal of Phonetics* 11, 277-290.
- Keating, Patricia A. (1984a) 'Phonetic and phonological representation of stop voicing', *Language* 60/2, 286-319.
- Keating, Patricia A. (1984b) 'Physiological effects on stop consonant voicing', UCLA Working Papers in Phonetics 59, 29-34.
- Keating, Patricia A. (1985) 'Universal phonetics and the organization of grammars', in Victoria A. Fromkin (ed.), *Phonetic Linguistics: Essays in Honor of Peter Ladefoged*. London: Academic Press, 115-32.
- Keating, Patricia A. (1988) 'A survey of phonological features', Indiana University Linguistics Club, <u>http://www.linguistics.ucla.edu/people/keating/Keating_IULC.pdf</u> (retrieved 6 March 2012).
- Keating, Patricia A. (1991) 'Coronal places of articulation', in Paradis and Prunet (1991), 29-48.
- Keating, Patricia A. (2012) 'Optical phonetics', research-interest description with further resources, <u>http://www.linguistics.ucla.edu/people/keating/keating.htm</u> (retrieved 25 March 2012).
- Keer, Edward (n.d.) 'Icelandic preaspiration and the moraic theory of geminates', http://roa.rutgers.edu/files/312-0499/roa-312-keer-2.pdf (retrieved 29 July 2008).
- Kehrein, Wolfgang (2002) *Phonological Representation and Phonetic Phrasing: Affricates and Laryngeals*. Tübingen: Max Niemeyer Verlag.
- Kehrein, Wolfgang and Chris Golston (2004) 'A prosodic theory of laryngeal contrasts', *Phonology* 21, 325-57.

- Keller, Rudolf Ernst (1961) German Dialects: Phonology and Morphology, with Selected Texts, Google Print (retrieved 10 September 2009).
- Kenstowicz, M. (1994) Phonology in Generative Grammar.
- Kerswill, Paul (1996) 'Children, adolescents and language change', *Language Variation* and Change 8, 177-202.
- Kessler, Brett (n.d.) 'On the phonological nature of the Proto-Indo-European laryngeals', MS, Washington University in St. Louis,

http://artsci.wustl.edu/~bkessler/PIE-laryngeals/ (retrieved 29 July 2008).

- Keyser, S.J. and Stevens, K.N. (2001) 'Enhancement Revisited', in M. Kenstowicz (ed.) *Ken Hale: A Life in Language*, Cambridge: MIT Press, 271-291.
- Keyser, Samuel J. and Kenneth N. Stevens (1994) 'Feature geometry and the vocal tract', *Phonology* 11/2, 207-236.
- Khachaturian, Amalia H. (1983) 'The nature of voiced aspirated stops and affricates in Armenian dialects', *Annual of Armenian Linguistics* 4, Cleveland, Ohio: Cleveland State University, 57-62.
- Khachaturian, Amalia H. (1992) 'Voiced aspirate consonants in the Nor Bayazet dialect of Armenian', in John A. C. Greppin (ed.) *Proceedings of the Fourth International Conference on Armenian Linguistics*, Delmar, New York: Caravan Books, 115-128.
- Kibre, Nicholas J. (1997) *A Model of Mutation in Welsh*, Indiana University Linguistics Club Publications.
- Kildary, Douglas G. (2006) Post on Indo-European root structure constraints in online discussion of 'Indo-European typology and Sanskrit phonology', 21 February, Science groupserv.com, <u>http://www.groupsrv.com/science/post-1277618.html</u> (retrieved 29 July 2008).
- Kiliaan, H. (1911) 'Madoereesche spraakkunst. Eerste stuk: inleiding en klankleer. Semarang, Indonesia: H. A. Benjamins; Semarang-Drukkerij en boekhandel.
- Kim, Chin-W. (1965) 'On the autonomy of the tensity feature in stop classification (with special reference to Korean stops)', *Word* 21, 339-59.
- Kim, Chin-W. (1970) 'A theory of aspiration', Phonetica 21, 107-116.
- Kim, Eun-Sook and Douglas Pulleyblank (2009) 'Glottalization and lenition in Nuuchah-nulth', *Linguistic Inquiry* 40/4, 567-617.
- Kim, Hyunsoon (2005) 'The representation of the three-way laryngeal contrast in Korean consonants', in Marc van Oostendorp and Jeroen Maarten van de Weijer (eds.) *The Internal Organization of Phonological Segments*, Studies in Generative Grammar 77, Berlin: de Gruyter, 287-316.
- Kim, Hyunsoon, K. Honda, and S. Maeda (2005) 'Stroboscopic-cine MRI study of the phasing between the tongue and the larynx in the Korean three-way phonation contrast', *Journal of Phonetics* 33, 1-26.
- Kim, Hyunsoon, Shinji Maeda, Kiyoshi Honda, and Stephane Hans (2010) 'The laryngeal characterization of Korean fricatives: acoustic and aerodynamic data', in Susanne Fuchs, Martine Toda, and Marzena Zygis (eds.) *Turbulent Sounds: An Interdisciplinary Guide*, Berlin: de Gruyter, 143-166.

- Kim, Mi-Ryoung and San Duanmu (2004) "Tense" and "lax" stops in Korean', *Journal* of East Asian Linguistics 13/1, 59-104.
- Kim, Yuni (2002) Phonological features: privative or equipollent? B.A. thesis, Harvard University, <u>http://personalpages.manchester.ac.uk/staff/yuni.kim/Kim-ABthesis.pdf</u> (retrieved 17 April 2010).
- Kingston, John (1985) *The phonetics and phonology of the timing of oral and glottal events*, University of California, Berkeley.
- Kingston, John (2005) 'The phonetics of Athabaskan tonogenesis', in Sharon Hargus and Keren Rice (eds.) *Athabaskan Prosody*, Amsterdam: John Benjamins, 137-184.

Kingston, John (2007) 'The phonetics-phonology interface', in de Lacy (2007), 401-34.

Kingston, John (2009) 'Tonogenesis', MS, <u>http://people.umass.edu/jkingstn/web</u> <u>%20page/research/tonogenesis2.pdf</u> (retrieved 18 February 2012).

Kingston, John, A. Lahiri, and R. L. Diehl (2009) 'Voice', MS, http://people.umass.edu/jkingstn/web

<u>%20page/research/voice_submitted_12_oct_09.pdf</u> (retrieved 8 December 2011).

- Kingston, John and Diehl, R. L. (1994) 'Phonetic knowledge', Language 70, 419-454.
- Kingston, John and Diehl, R. L. (1995) 'Intermediate properties in the perception of distinctive feature values', in Connell and Arvaniti (1995), 7-27.
- Kiparsky, Paul (1965) Phonological Change, PhD dissertation, MIT.
- Kiparsky, Paul (1973a) "'Elsewhere" in phonology', in Stephen R. Anderson and Paul Kiparsky (eds.) *A Festschrift for Morris Halle*, 93-106.
- Kiparsky, Paul (1973b) 'Phonological representations', in Osama Fujimura (ed.), *Three dimensions of linguistic theory*, Tokyo: Tokyo Institute for Advanced Studies of Language, 3-136.
- Kiparsky, Paul (1982) Explanation in Phonology.
- Kiparsky, Paul (1985) 'Some Consequences of Lexical Phonology', *Phonology Yearbook* 2, 82-136.
- Kiparsky, Paul (1995) 'The phonological basis of sound change', in John Goldsmith (ed.) *Handbook of Phonological Theory*, Oxford, Blackwell, 640-70.
- Kiparsky, Paul (2000) 'Opacity and cyclicity', *The Linguistic Review* 17, 351-365, <u>http://www.stanford.edu/~kiparsky/Papers/vdhulst-ritter.pdf</u> (retrieved December 2000).
- Kiparsky, Paul (2003) 'The phonological basis of sound change', 313-42 in Holmes, Janet and Meyerhoff, Miriam (eds.) *The Handbook of Language and Gender*, Oxford: Blackwell, <u>http://cat.inist.fr/?aModele=afficheN&cpsidt=2019662</u> (retrieved 26 July 2008).
- Kiparsky, Paul (2004) 'Universals constrain change; change results in typological generalizations', MS, <u>http://www.stanford.edu/~kiparsky/Papers/cornell.pdf</u> (retrieved 10 May 2010).
- Kiparsky, Paul (2005a) 'The phonological basis of sound change', MS Stanford University, <u>http://www.stanford.edu/~kiparsky/Papers/workshop.pdf</u> (retrieved 10 May 2010).

- Kiparsky, Paul (2005b) 'Grammaticalization as optimization', MS, <u>http://www.stanford.edu/~kiparsky/Papers/yalegrammaticalization.pdf</u> (retrieved 18 February 2012).
- Kiparsky, Paul (2006) 'The Amphichronic Program vs Evolutionary Phonology', MS, Stanford University, <u>http://www.stanford.edu/~kiparsky/Papers/blevins.pdf</u> (retrieved 29 July 2008).
- Kiparsky, Paul (forthcoming) *Opacity and Paradigm Effects*, Stanford: CSLI Monographs.
- Kirchner, Robert (1998) *An Effort-Based Approach to Consonant Lenition*, PhD dissertation, University of California, Los Angeles, <u>http://roa.rutgers.edu/files/276-0898/roa-276-kirchner-2.pdf</u> (retrieved 29 July 2008).
- Kirchner, Robert (2000) 'Geminate inalterability and lenition', *Language* 76/3, 509-45, <u>http://www.jstor.org/stable/pdfplus/417134.pdf</u> (retrieved 29 July 2008).
- Kirkby, James P. (2011) 'Vietnamese (Hanoi Vietnamese)', *Journal of the International Phonetic Association* 41/3, 381-392, <u>http://journals.cambridge.org/download.php?</u> <u>file=%2FIPA</u>

```
%2FIPA41_03%2FS0025100311000181a.pdf&code=9945223d6e43a920ad870197ef0
71dd8 (retrieved 11 February 2012).
```

- Kisseberth, C. W. (1970) 'On the functional unity of phonological rules', *Linguistic Inquiry* 1, 291-306.
- Kitahara, Hisatsugu (2006) 'Some notes on the minimalist program', 3-18 in Cedric Boeckx (ed.) *Minimalist Essays*, Amsterdam: John Benjamins.
- Kitson 1997: 222 'Reconstruction, typology, and the "original homeland" of the Indo-Europeans', in Jacek Fisiak (ed.) *Linguistic Reconstruction and Typology*, 183-256.
- Klatt, D. H. (1985) 'Voice onset time, frication, and aspiration in word-initial consonant clusters', *Journal of Speech and Hearing Research* 18, 686-706.
- Kobayashi, Masato (1999) 'Deocclusion of Sanskrit *dh* after front vowels', paper at the 209th Meeting of the American Oriental Society, 21 March 1999, Baltimore, <u>http://www.aa.tufs.ac.jp/~masatok/pdf/dhaos2.pdf</u> (retrieved 29 July 2008).
- Kobayashi, Masato (2004) *Historical Phonology of Old Indo-Aryan Consonants*, Study of Languages and Cultures of Asia and Africa Monograph Series 42, Tokyo, http://www.hakuoh.ac.jp/~masatok/hpiac 2004.pdf (retrieved 1 March 2012).
- Koenig, Laura L. and Jorge C. Lucero (2008) 'Stop consonant voicing and intraoral pressure contours in women and children', *Journal of the Acoustical Society of America* 123/2, 1077-1088,

http://www.bio.spbu.ru/faculty/departments/vnd/pdf/kurbanova.pdf (retrieved 25 March 2012).

- Kong, Eun Jong, Mary E. Beckman, and Jan Edwards (2011) 'Why are Korean tense stops acquired so early? The role of acoustic properties', *Journal of Phonetics* 39/2, 196-211.
- Kohler, Klaus (1999) 'German', in International Phonetic Association (ed.), 86-89.

- Kortlandt, Frederik (1977) 'Historical laws of Baltic accentuation', *Baltistica* 13, 319-330.
- Kortlandt, Frederik (1978a) 'I.-E. palatovelars before resonants in Balto-Slavic', *Recent developments in historical phonology*, The Hague, 237-243, <u>https://openaccess.leidenuniv.nl/bitstream/1887/1853/1/344_017.pdf</u> (retrieved 28 July 2008).
- Kortlandt, Frederik (1978b) 'Proto-Indo-European Obstruents,' *Indogermanische Forschungen* 83, 107-118.
- Kortlandt, Frederik (1979) 'Three problems of Balto-Slavic phonology', *Zbornik za Filologiju i Lingvistiku* 22/2, 57–63, <u>https://openaccess.leidenuniv.nl/bitstream/1887/1865/1/344_031.pdf</u> (retrieved 28 July 2008).
- Kortlandt, Frederik (1985) 'Proto-Indo-European Glottalic Stops: the Comparative Evidence', *Folia Linguistica Historica* 6, 1985, 183-201, <u>https://openaccess.leidenuniv.nl/bitstream/handle/1887/1886/344_053.pdf?sequence=1</u> (retrieved 27 January 2012).
- Kortlandt, Frederik (1988a) 'Remarks on Winter's Law', Dutch Contributions to the Tenth International Congress of Slavists, Sofia, Linguistics (Studies in Slavic and General Linguistics 11), Amsterdam, 387-96, https://openaccess.leidenuniv.pl/bitstream/handle/1887/1898/344_069.pdf?sequence

https://openaccess.leidenuniv.nl/bitstream/handle/1887/1898/344_069.pdf?sequence=1 (retrieved 27 January 2012).

Kortlandt, Frederik (1988b) 'Vestjysk stød, Icelandic preaspiration, and Proto-Indo-European glottalic stops' in M. Jazayery and W. Winter (eds.) *Languages and Cultures: Studies in Honor of Edgar C. Polomé*, Berlin: Mouton de Gruyter, 353-357.

Kortlandt, Frederik (1989) 'Lachmann's law,' in Vennemann (ed.) (1989), 103-105.

- Kortlandt, Frederik (1993) 'General linguistics and Indo-European reconstruction', MS, <u>http://kortlandt.nl/publications/art130e.pdf</u> (retrieved 11 February 2012).
- Kortlandt, Frederik (2004) 'Glottalization, preaspiration and gemination in English and Scandinavian', <u>http://www.kortlandt.nl/bibliography.html#211</u> (retrieved 29 July 2008).
- Kortlandt, Frederik (2008) 'Balto-Slavic phonological developments', *Baltistica* 43/1, 5-15, <u>http://www.leidykla.eu/fileadmin/Baltistika/43-1/03_Kortlandt.pdf</u> (retrieved 29 July 2008).
- Krämer, Martin (2005) 'How crazy is English *r*-insertion? Or: Can OT account for the 'unnatural'?', ICLaVE 2005 paper, Workshop on Crazy Rules and Lexical Exceptions, Amsterdam, 25 June 2005, <u>http://www.hum.uit.no/a/kraemer/crazy-a-dam05.pdf</u> (retrieved 21 January 2009).
- Krekelberg, B. and J. N. Kok (1992) 'A lateral inhibition neural network that emulates a winner-takes-all algorithm', Technical Report RUU-CS-92-46, Utrecht University, <u>http://www.cs.uu.nl/research/techreps/repo/CS-1992/1992-46.pdf</u> (retrieved 19 June 2010).

Kuipers, A. H. (1960) Phoneme and Morpheme in Kabardian, The Hague: Mouton.

- Kümmel, Martin (2010) 'The distribution of roots ending in IE **ND*', presentation at The Sound of Indo-European 2, Opava, November 2010, <u>http://uni-freiburg.academia.edu/MartinKümmel/Papers/350091/The_distribution_of_IE_roots_in_ND</u> (retrieved 2 January 2012).
- Ladefoged, Peter (2009[1964]) *A Phonetic study of West African Languages*, Cambridge: Cambridge University Press.
- Ladefoged, Peter (1967) *Three Areas of Experimental Phonetics: Stress and Respiratory Activity, the Nature of Vowel Quality, and Units in the Perception and Production of Speech*, London: Oxford University Press.
- Ladefoged, Peter (1971) *Preliminaries to Linguistic Phonetics*, Chicago: University of Chicago Press.
- Ladefoged, Peter (1973) 'The features of the larynx', *Journal of Phonetics* 1973/1, 73-83.
- Ladefoged, Peter (1997) 'Linguistic phonetic descriptions', in William J. Hardcastle and John Laver (eds.) *The Handbook of Phonetic Sciences*, Oxford: Blackwell, 589-618.
- Ladefoged, Peter (2001) *A Course in Phonetics*, fourth edition, Boston: Heinle and Heinle.
- Ladefoged, Peter (2005) 'Features and parameters for different purposes', Linguistic Society of America paper,

http://www.linguistics.ucla.edu/people/ladefoge/PLfeaturesParameters.pdf (retrieved 28 July 2008).

- Ladefoged, Peter (2006) 'Representing linguistic phonetic structure', unfinished MS, <u>http://www.linguistics.ucla.edu/people/ladefoge/PhoneticStructure.pdf</u> (retrieved 31 January 2012).
- Ladefoged, Peter and Gerald Loeb (2009) 'Preliminary studies on respiratory activity in speech', MS,

http://www.linguistics.ucla.edu/people/ladefoge/LadefogedAndLoebRespiration.pdf, stated as posted on Ladefoged's academic web site as of September 2009 (http://www.linguistics.ucla.edu/people/ladefoge/) (retrieved 14 February 2012)

- Ladefoged, Peter and Ian Maddieson (1996) *The Sounds of the World's Languages*. Oxford: Blackwell.
- Lahiri, Aditi and Vincent Evers (1991) 'Palatalization and coronality', in Paradis and Prunet (eds.) (1991), 79-100.
- Lai, Y., C. Huff, J. Sereno, and A. Jongman (2009). 'The raising effect of aspirated prevocalic consonants on F0 in Taiwanese', in J. Brooke, G. Coppola, E. Görgülü, M. Mameni, E. Mileva, S. Morton, and A. Rimrott, (eds.) *Proceedings of the 2nd International Conference on East Asian Linguistics*, Simon Fraser University Working Papers in Linguistics, <u>http://www2.ku.edu/~kuppl/documents/Lai_EtAl.pdf</u> (retrieved 15 February 2012).
- Langendoen, D. Terence (1975) 'The relation of competence to performance', in Doris Aaronson and Robert W. Rieber (eds.) *Developmental Psycholinguistics and Communication Disorders*, Annals of the New York Academy of Sciences vol. 263,

197-200, <u>http://dingo.sbs.arizona.edu/~langendoen/CompetencePerformance.pdf</u> (retrieved 30 December 2011).

- Lass, Roger (1984) *Phonology: An Introduction to Basic Concepts*, Cambridge: Cambridge University Press.
- Laufer, Asher (1991) 'The "glottal fricatives", *Journal of the International Phonetic Association* 21/2, 91-93.
- Laver, John (1994) Principles of Phonetics, Cambridge: Cambridge University Press.
- Lavoie, Lisa M. (1996) 'Consonant Strength: Results of a Data Base Development Project', *Working Papers in the Cornell Phonetics Laboratory* 11, 269-316.
- Lavoie, Lisa M. (2001) Consonant Strength: Phonological Patterns and Phonetic Manifestations.
- Leben, William R. (2011) 'Autosegments', ch. 14 in in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Lee, Sunghwa (2006) 'Post obstruent tensification in Korean', Actes du congrès annuel de l'Association canadienne de linguistique 2006 (Proceedings of the 2006 annual conference of the Canadian Linguistic Association), 1-13, <u>http://westernlinguistics.ca/Publications/CLA2006/Lee.pdf</u> (retrieved 18 February 2012).
- Lehmann, Winfred P (1986) 'Reflexes of PIE d < t" in Dieter Kastovsky and Aleksander Szwedek (eds.) *Linguistics across Historical and Geographical Boundaries: in honor of Jacek Fisiak*, Berlin: Mouton de Gruyter, vol. 1, 483-9.
- Lehmann, Winfred P. (1996) *Theoretical Bases of Indo-European Linguistics*, London: Routledge.
- Lehmann, Winfred P. (2005) *Proto-Indo-European Phonology*, <u>http://www.utexas.edu/cola/centers/lrc/books/piepN.html</u> (retrieved 29 July 2008).
- Lehnert-LeHouillier, H. (2009) 'Phonetic and phonological aspects of liquid devoicing in Thai, Hungarian, and American English stop-liquid sequences', in H. Lehnert-LeHouillier and A. B. Fine (eds.) University of Rochester Working Papers in the Language Sciences 5/1, 49-68,
- http://www.bcs.rochester.edu/cls/s2009v5n1/Lehnert.pdf (retrieved 17 April 2010).
- Leman, Wayne (1981) 'Cheyenne pitch rules', *International Journal of American Linguistics* 47/4, 283-309.
- Lewis, Maureen Warner (1997) Trinidad Yoruba.
- Liberman, Alvin M. and Ignatius G. Mattingly (1989) 'A specialization for speech perception', *Science* 243, 489-94, <u>http://www.haskins.yale.edu/Reprints/HL0659.pdf</u> (retrieved 6 December 2011).
- Lieberstein, H. M. (1971) 'Self-inhibition as a facet of sensory physiology clarifying a critical point in an emerging general theory on the electrical activity of cells', *Mathematical Biosciences* 11/3-4, 365-375.
- Lin, Ying and Jeff Mielke (2008) 'Discovering place and manner features what can be learned from acoustic and articulatory data?' in Joshua Tauberer, Aviad Eilam, and

Laurel MacKenzie (eds.) *Penn Working Papers in Linguistics* 14.1, 241-254, <u>http://repository.upenn.edu/cgi/viewcontent.cgi?article=1035&context=pwpl</u> (retrieved 10 May 2010).

- Lindau, J. (1979) 'The feature expanded', Journal of Phonetics 7, 163-176.
- Lindblom, Björn and Harvey M. Sussman (2012) 'Dissecting coarticulation: how locus equations happen', *Journal of Phonetics* 40, 1-19.
- Lindeman, Fredrik Otto (1987) *Introduction to the 'Laryngeal Theory'*, Oslo: Norwegian University Press.
- Lindqvist, J. (1972) 'A descriptive model of laryngeal articulation in speech', *Speech Transmission Laboratory: Quarterly Progress and Status Report* issue 2-3, pp. 1-9, <u>http://www.speech.kth.se/prod/publications/files/qpsr/1972/1972_13_2-3_001-009.pdf</u> (retrieved 21 January 2012).
- Lindsey, Geoffrey (1992) 'Hausa glottalic consonants: a laryngographic study', *Bulletin* of the School of Oriental and African Studies 55/3, 511-27.
- Lindskoog, John N. and Ruth M. Brend (1962) 'Cayapa phonemics', in Benjamin Elson (ed.) *Ecuadorian Indian Languages: I*, Norman, Oklahoma: Summer Institute of Linguistics of the University of Oklahoma, 31-44, http://www.sil.org/acpub/repository/11430.pdf (retrieved 16 February 2012).
- Lipski, John M. (1976) 'Teaching spoken Spanish', *Hispania* 59/4, 871-877, http://www.jstor.org/stable/pdfplus/340212.pdf (retrieved 29 July 2008).
- Lipski, John M. (2000) Review of Goedemans (1998) in Language 76/3, 723-24.
- Lisker, Leigh (1986) "Voicing" in English: a catalogue of acoustic features signaling /b/ versus /p/ in trochees', *Language and Speech* 29/1, 3-11, http://www.haskins.yale.edu/Reprints/HL0565.pdf (retrieved 18 February 2012).
- Lisker, Leigh and Arthur S. Abramson (1964) 'A cross-language study of voicing in initial stops: acoustical measurements', *Word* 20, 384-422, http://www.haskins.yale.edu/Reprints/HL0053.pdf (retrieved 2 February 2012).
- Lisker, Leigh and Arthur S. Abramson (1968) 'Distinctive features and laryngeal control', Haskins Laboratories, Status Report on Speech Research SR-15, 81-95, <u>http://www.haskins.yale.edu/SR/SR015/SR015_04.pdf</u> (retrieved 3 February 2012).
- Lisker, Leigh and Arthur S. Abramson (1971) 'Distinctive features and laryngeal control', Haskins Laboratories, Status Report on Speech Research SR-27, 133-151, <u>http://www.haskins.yale.edu/sr/SR027/SR027_07.pdf</u> (retrieved 3 February 2012).
- Lisker, Leigh and Thomas Baer (1984) 'Laryngeal management at utterance-internal word boundary in American English', Haskins Laboratories Status Report on Speech Research SR-77/78, 129-136, <u>http://www.haskins.yale.edu/sr/SR077/SR077_10.pdf</u> (retrieved 18 February 2012).
- Liu, Te-hsin (2009) 'Why should markedness constraints be relative? Four cases in tone sandhi directionality', *UC Berkeley Phonology Lab Annual Report*, 81-107, <u>http://linguistics.berkeley.edu/phonlab/annual_report/documents/2009/Liu_Directional ity.pdf</u> (retrieved 17 October 2009).

- Lloret, Maria-Rosa (1992) 'On the representation of ejectives and implosives', in Wolfgang U. Dressler, Martin Prinzhorn, and John R. Rennison (eds.) (1992), 123-33.
- Lobel, Jason William (2010) 'Maranao and Madurese revisited: a follow-up', *Oceanic Linguistics* 49/1, 278-81.
- Lobel, Jason William and William C. Hall (2010), 'Southern Subanen aspiration', *Oceanic Linguistics* 49/2, 319-38.
- Löfqvist, Anders (1980) 'Interarticulator programming in stop production', *Journal of Phonetics* 8, 475-490, <u>http://www.haskins.yale.edu/Reprints/HL0328.pdf</u> (retrieved 2 March 2012).
- Löfqvist, Anders (1995) 'Laryngeal mechanisms and interarticulator timing in voiceless consonant production', in F. Bell-Berti and L. J. Raphael (eds.) *Producing Speech: Contemporary Issues*, New York: AIP Press, 99-116,
- <u>http://www.haskins.yale.edu/Reprints/HL0971.pdf</u> (retrieved 20 February 2012). Löfqvist, Anders and Hirohide Yoshioka (1979) 'Laryngeal activity in Swedish voiceless obstruent clusters', Haskins Laboratories, Status Report on Speech Research SR-59/60, 103-125, <u>http://www.haskins.yale.edu/sr/SR059/SR059_07.pdf</u> (retrieved 11 January 2012).
- Löfqvist, Anders, Thomas Baer, Nancy S. McGarr, and Robin Seider Story (1989) 'The cricothyroid muscle in voicing control', Haskins Laboratories Status Report on Speech Research SR-97/98, 25-40, <u>http://www.haskins.yale.edu/sr/SR097/SR097_02.pdf</u> (retrieved 11 January 2012).
- Lombardi, Linda (1991) *Laryngeal Features and Laryngeal Neutralization*, PhD dissertation, University of Massachusetts at Amherst.
- Lombardi, Linda (1994) *Laryngeal Features and Laryngeal Neutralization*, New York: Garland Press (published version of PhD dissertation).
- Lombardi, Linda (1999) 'Positional faithfulness and voicing assimilation', in McCarthy (2004), 343-64.
- Longacre, Robert. E. (1957) 'Proto-Mixtecan,' International Journal of American Linguistics 23/4, 1-195.
- Lubotsky, Alexander M. (1988) *The System of Nominal Accentuation in Sanskrit and Proto-Indo-European*, Leiden.
- Lubotsky, Alexander M. (1995) 'Reflexes of intervocalic laryngeals in Sanskrit', in Smoczyński, W. (ed.) *Kurylowicz Memorial Volume, Part One*, Cracow, 213-33, <u>http://www.ieed.nl/lubotsky/pdf/sktintervoc.pdf</u> (10 July 2010).
- Lutfi, Robert (2008) 'Human sound source identification', in William A. Yost (ed.) *Auditory Perception of Sound Sources*, New York: Springer, 13-42.
- Mac, Cornelia and Robert Longacre (1960) 'Proto-Mixtec phonology', *International Journal of American Linguistics*, 26/1, 23-40,

http://www.jstor.org/stable/pdfplus/1263729.pdf (retrieved 28 July 2008).

MacAulay, Donald (ed.) (1992) *The Celtic Languages*. Cambridge: Cambridge University Press.

- Machonis, P. A. (1990) *Histoire de la langue française : du latin à l'ancien français*. Lanham, Maryland: University Press of America.
- MacKay, D. G. (1987) *The Organization of Perception and Action: A Theory for Language and Other Cognitive Skills*, Berlin: Springer-Verlag, <u>http://mackay.bol.ucla.edu/publications.html</u> (21 March 2012).
- MacKay, Carolyn J. (1994) 'A sketch of Misantla Totonac Phonology', *International Journal of American Linguistics* 60, 369-419.
- MacWhinney, Brian (2001) 'Emergentist approaches to language', in Joan L. Bybee and Paul Hopper (eds.) *Frequency and the Emergence of Linguistic Structure*, Typological Studies in Language 45, Amsterdam: John Benjamins, 449-470.
- Maddieson, Ian (1984) Patterns of Sounds. Cambridge: Cambridge University Press.
- Maddieson, Ian (1999) 'In search of universals', in *Proceedings of the 14th International Congress of Phonetic Sciences*, San Francisco, 2521-2528, <u>http://www.ims.uni-</u> <u>stuttgart.de/lehre/teaching/2002-SS/SpracherkennungI/maddieson_icphs1999.pdf</u> and <u>http://www.ee.columbia.edu/~dpwe/LabROSA/proceeds/icphs1999/pdfs/2521.pdf</u> (both retrieved 28 July 2008).
- Maddieson, Ian (2001) 'Good timing: Place-dependent voice onset time in ejective stops.' Eurospeech Scandinavia,

http://www.tsi.enst.fr/~chollet/Biblio/Congres/Audio/Eurospeech01/CDROM/papers/p age823.pdf (removed by 29 July 2008; also available at http://www.iscaspeech.org/archive/eurospeech 2001/e01 0823.html, but login required).

- Maddieson, Ian (2003) 'The sounds of the Bantu languages', in Nurse and Philippson (2003), 15-41.
- Maddieson, Ian (2008a) 'Front rounded vowels', in Haspelmath, Martin, Matthew S. Dryer, David Gil, and Bernard Comrie (eds.) *The World Atlas of Language Structures Online*, Munich, ch. 11, <u>http://wals.info/feature/11</u> (retrieved 24 January 2010).
- Maddieson, Ian (2008b) 'Absence of common consonants', in Haspelmath, Martin, Matthew S. Dryer, David Gil, and Bernard Comrie (eds.) *The World Atlas of Language Structures Online*, Munich, ch. 18, <u>http://wals.info/feature/18</u> (retrieved 19 April 2010).
- Maddieson, Ian (2008c) 'Uvular consonants', in Haspelmath, Martin, Matthew S. Dryer, David Gil, and Bernard Comrie (eds.) *The World Atlas of Language Structures Online*, Munich, ch. 6, <u>http://wals.info/feature/description/6</u> (retrieved 21 May 2010).
- Malone, Joseph L. (1993) Tiberian Hebrew Phonology.
- Manaster-Ramer, Alexis (1993) 'On lenition in some northern Uto-Aztecan languages', *International Journal of American Linguistics* 59/3, 334-41.
- Manaster-Ramer, Alexis (1996) 'The search for the sources of the Nahuatl saltillo', abstract only, <u>http://www.indiana.edu/~anthling/v37-1.html#amr</u> (retrieved 24 July 2008); article in *Anthropological Linguistics* 37/1, 1-15.
- Mańczak, Witold (1990) 'Critique des opinions de Gamkrelidze-Ivanov', *Historische Sprachforschung* 103/2, 178-92.

Marshall, M. (1984) 'The dialect of Notre Dame de Sanilhac: a natural generative phonology', *Stanford French and Italian Studies 31*.

Martin, Andrew and Sharon Peperkamp (2011) 'Speech perception and phonology', ch. 98 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.

- Martinet, André (1949) 'La double articulation linguistique', *Travaux du Cercle linguistique de Copenhague* 5, 30-37.
- Martinet, A. (1955) *Economie des changements phonétiques: traité de phonologie diachronique*, Berne: Francke.
- Martinet, André (1957) 'Arbitraire linguistique et double articulation', *Cahiers Ferdinand de Saussure* 15, 105-116.
- Martinet, André (1983) 'Ce que n'est pas la phonologie', *Langue française* 60, 6-13, <u>http://www.persee.fr/web/revues/home/prescript/article/lfr_0023-</u>8368 1983 num 60 1 5171 (retrieved 30 December 2011).
- Martinet, André (1987) Des steppes aux océans: L'indo-européen et les "Indo-Européens", Paris: Payot.
- Martinet, André (1991) 'Thoughts about the Glottalic Theory', Zeitschrift für Phonetik, Sprachwissenschaft und Kommunikationsforschung 44/3, 285-289.
- Martínez-Celdrán, Eugenio (2008) 'Some chimeras of traditional Spanish phonetics', in Laura Colantonni and Jeffrey Steele (eds.) Selected Proceedings of the 3rd Conference on Laboratory Approaches to Spanish Phonology, Somerville: Cascadilla Proceedings Project, 32-46, <u>http://www.lingref.com/cpp/lasp/3/paper1712.pdf</u> (retrieved 28 July 2008).
- Martínez Celdrán, Eugenio and Ana María Fernández Planas (2007) Manual de fonética española: articulaciones y sonidos del español.
- Mascaró, Joan (2011) 'Rule ordering', ch. 74 in in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Matasović, Ranko (2005) 'Toward a relative chronology of the earliest Baltic and Slavic sound changes', *Baltistica* 40/2, 147-57

http://www.leidykla.vu.lt/inetleid/baltistic/40_2/straipsniai/str1.pdf (retrieved 29 July 2008).

- Matasović, Ranko (in progress) *An etymological lexicon of Proto-Celtic* (online), The Indo-European Etymological Dictionary Project, Leiden University, at <u>http://www.ieed.nl/</u> (retrieved 18 July 2008).
- Mathangwane, Joyce (1999) *Ikalanga Phonetics and Phonology*, Stanford: CSLI Publications.
- Matthews, P. H. (1982) *Do Languages Obey General Laws? An inaugural lecture delivered before the University of Cambridge on 17 November 1981,* Cambridge University Press.

- Mayo Foundation for Medical Education and Research (2011) 'Endoscopic sinus surgery', Mayo Clinic, <u>http://www.mayoclinic.com/health/medical/IM01919</u> (retrieved 5 December 2011).
- Mayrhofer, Manfred (1983) 'Sanskrit und die Sprachen Alteuropas Zwei Jahrhunderte des Widerspiels von Entdeckungen und Irrtümern', *Nachrichten der Akademi der Wissenschaften in Göttingen aus dem Jahre 1983: Philologisch-Historische Klasse*, Göttingen: Vandenhoeck & Ruprecht, 121-54.
- Mayrhofer, Manfred (1986-2001) *Etymologisches Wörterbuch des Altindoarischen*, Heidelberg.
- Mayrhofer, Manfred (1986) *Indogermanische Grammatik*, vol. 1, part 2, *Lautlehre*, Heidelberg: Winter.
- McCarthy, John J. (1993) 'A case of surface constraint violation', *Canadian Journal of Linguistics / Revue Canadienne de Linguistique* 38, 169-95.
- McCarthy, John J. (1994) 'The phonetics and phonology of Semitic pharyngeals', *Linguistics Department Faculty Publication Series* 86,

<u>http://scholarworks.umass.edu/linguist_faculty_pubs/86/</u> (retrieved 24 February 2012). McCarthy, John J. (1998) 'Sympathy and phonological opacity', *Phonology* 16, 331-99,

- <u>http://people.umass.edu/jjmccart/sympathy.pdf</u> (retrieved 6 September 2008). McCarthy, John J. (2002) *A Thematic Guide to Optimality Theory*, Cambridge: Cambridge University Press.
- McCarthy, John J. (ed.) (2004) Optimality Theory in Phonology.
- McCarthy, John J. (2005) 'Taking a free ride in morphophonemic learning', *Catalan Journal of Linguistics* 4, 19-56, <u>http://works.bepress.com/john_j_mccarthy/27/</u> (retrieved 25 September 2010).
- McCarthy, John J. (2007) *Hidden Generalizations: Phonological Opacity in Optimality Theory*, London: Equinox.
- McCarthy, John J. (2008a) Doing Optimality Theory.
- McCarthy, John J. (2008b) 'The serial interaction of stress and syncope', *Natural Language and Linguistic Theory* 26, 499-546,

http://works.bepress.com/john_j_mccarthy/31/ (retrieved 27 August 2009)

- McCarthy, John J. (2009a) 'Autosegmental spreading in Optimality Theory', MS, http://works.bepress.com/john j mccarthy/100/ (retrieved 27 August 2009)
- McCarthy, John J. (2009b) 'Harmony in harmonic serialism', MS, http://works.bepress.com/john j mccarthy/61/ (retrieved 27 August 2009)

McCarthy, John J. (2010a) 'Studying Gen', to appear in *Journal of the Phonetic Society* of Japan, <u>http://works.bepress.com/john_j_mccarthy/3/</u> (retrieved 27 August 2009).

McCarthy, John J. (2010b) 'Perceptually grounded faithfulness in Harmonic Serialism', MS, <u>http://works.bepress.com/john j mccarthy/104/</u> (retrieved 18 June 2010).

McCarthy, John J. and Prince, Alan (1993 [revision of]) *Prosodic Morphology: Constraint Interaction and Satisfaction*, <u>http://roa.rutgers.edu/files/482-1201/482-1201-MCCARTHY-0-1.PDF</u> (retrieved December 2008).

- McCarthy, John J. and Prince, Alan (1998) 'Faithfulness and reduplicative identity', <u>http://roa.rutgers.edu/view.php3?id=568</u> (retrieved 16 December 2008).
- McGowan, Richard (1992) 'Tongue-tip trills and vocal tract wall compliance', *Journal* of the Acoustical Society of America 91, 2903–2910.
- McLaughlin, Fiona (2005) 'Voiceless implosives in Seereer-Siin', Journal of the International Phonetic Association 35/2, 201-14.
- McLaughlin, John E. (1989) 'A note on the change of strident to nonstrident in Gosiute Shoshoni', *International Journal of American Linguistics* 55/2, 240-247.
- McLaughlin, John E. (1992) 'A counterintuitive solution in Central Numic phonology', International Journal of American Linguistics 58/2, 158-81.
- McLaughlin, John E. (2000) 'Language boundaries and phonological borrowing in the Central Numic languages', in Casad and Willett (eds.) (2000).
- McLeod, Amy, Megan Baillargeon, Dale Evan Metz, Nicholas Schiavetti, and Robert L. Whitehead (2001) 'Relational invariance for stop place categorization: a direct replication of Sussman, McCaffrey, and Matthews', *Contemporary Issues in Communication Science and Disorders* 28, 98-103,
- http://www.nsslha.org/uploadedFiles/NSSLHA/publications/cicsd/2001FLocusEquations.pdf (retrieved 17 March 2011).
- McMahon, April (2000) Change, Chance, and Optimality.
- McMahon, April (2003) 'Phonology and the Holy Grail', Lingua 113, 103-15.
- Meid, Wolfgang (1989) Das Problem von indogermanische /b/. Innsbrucker Beiträge zur Sprachwissenschaft, Vorträge und kleinere Schriften, 44.
- Meier-Brügger, Michael (2003) *Indo-European Linguistics*, translation by Charles Gertmenian of *Indogermanische Sprachwissenshaft*, with new contributions from Manfred Mayrhofer and Matthias Fritz; Berlin: de Gruyter.
- Melchert, C. (1994) Anatolian Historical Phonology.
- Menn, Lise (2004) 'Saving the baby: making sure that old data survive new theories', in Kager et al. (2004), 54-72.
- Menn, Lise (2010) *Psycholinguistics: Introduction and Applications*, San Diego: Plural Publishing.
- Menn, Lise, Ellen Schmidt, and Brent Nicholas (2009) 'Conspiracy and sabotage in the acquisition of phonology: dense data undermine existing theories, provide scaffolding for a new one', *Language Sciences* 31, 285-304.
- Menn, Lise and Marilyn Vihman (in press) 'Features in child phonology: inherent, emergent, or artefacts of analysis?', to appear in N. Clements and R. Ridouane (eds.) *Where Do Features Come From? The nature and sources of phonological primitives*, <u>http://www.york.ac.uk/media/languageandlinguistics/documents/staff/publications/Me</u> <u>nn&Vihman.pdf</u> (retrieved 3 January 2012).
- Merchant, Nazarré and Bruce Tesar (2005/2008) 'Learning underlying forms by searching restricted lexical subspaces', <u>http://roa.rutgers.edu/files/811-0306/811-MERCHANT-0-0.PDF</u> (retrieved 29 August 2009)..

- Mester, R. Armin and Junko Itô (1989) 'Feature predictability and underspecification: palatal prosody in Japanese mimetics', *Language* 65, 258-93.
- Midtlyng, Patrick J. (2011) 'The effects of speech rate on VOT for initial plosives and click accompaniments in Zulu', in Eyamba G. Bokamba et al. (eds.) *Selected Proceedings of the 40th Annual Conference on African Linguistics*, Somerville, MA: Cascadilla Proceedings Project, 105-118,

http://www.lingref.com/cpp/acal/40/paper2569.pdf (retrieved 28 January 2012).

Mielke, Jeff (2004) *The Emergence of Distinctive Features*, PhD dissertation, The Ohio State University,

http://linguistics.osu.edu/research/publications/dissertations/files/mielke2004.pdf (retrieved 29 July 2008).

Mielke, Jeff (2005) 'Ambivalence and ambiguity in laterals and nasals', *Phonology* 22, 169-203.

Mielke, Jeff (2006) 'Moving beyond innate features: a unified account of natural and unnatural classes', talk handout, North Eastern Linguistic Society, 22 October 2004, <u>http://nels.uconn.edu/handouts/NELS35-Mielke.pdf</u> (retrieved 10 May 2010).

- Mielke, Jeff (2007a) P-base 1.92, <u>http://aix1.uottawa.ca/~jmielke/pbase/index.html</u> (retrieved 3 April 2010).
- Mielke, Jeff (2007b) 'Unnatural classes and the class-pattern connection', University of Ottawa lecture slides,

<u>http://aix1.uottawa.ca/~jmielke/research/mot2007classpattern.pdf</u> (retrieved 10 May 2010).

- Mielke, Jeff (2008a) *The Emergence of Distinctive Features*, Oxford: Oxford University Press.
- Mielke, Jeff (2008b) 'Learning phonological features from different kinds of evidence', talk handout, <u>http://aix1.uottawa.ca/~jmielke/research/BWTL_Mielke_handout.pdf</u> (retrieved 10 May 2010).
- Mielke, Jeff (2009) 'A phonetically-based phonetic similarity metric', Northeastern Linguistic Society 40, Massachussetts Institute of Technology, <u>http://aix1.uottawa.ca/~jmielke/research/NELS_similarity_Mielke.pdf</u> (retrieved 10 May 2010).
- Mielke, Jeff (forthcoming a) 'Phonologization and the typology of feature behavior', submitted for inclusion in Alan Yu (ed.), *Phonologization*, Oxford: Oxford University Press; <u>http://aix1.uottawa.ca/~jmielke/research/mielke_phonologization_article.pdf</u> (retrieved 10 May 2010).
- Mielke, Jeff (forthcoming b) 'The nature of distinctive features and the issue of natural classes'.
- Mikuteit, Simone (2009) A Cross-Language Approach to Voice, Quantity and Aspiration: An East-Bengali and German Production Study, PhD dissertation, Universität Konstanz, http://kops.ub.uni-

konstanz.de/bitstream/handle/urn:nbn:de:bsz:352-opus-78588/Mikuteit_Diss.pdf? sequence=1 (retreived 5 January 2012). Milizia, Paolo (2003) 'Proto-Indo-European nasal infixation rule', paper abstract, UCLA Indo-European Conference 15,

http://www.humnet.ucla.edu/pies/pdfs/IEC/IEC15/Milizia_P_2003.pdf (retrieved 29 July 2008); subsequent version published in *Journal of Indo-European Studies* 32/3-4 (2004), 336-59.

- Miller, Brett A. (2006) 'PIE stops, Voiced Glottalic Theory, and Mon-Khmer', presentation, Cambridge Classics Linguistics Caucus, 24 November 2006.
- Miller, Brett A. (2007) 'Ejectives to plain voiced stops in PIE? Phonetics, typology, and the Glottalic Theory', in Tim Whitmarsh and James Warren (eds.) *Greek and Latin from an Indo-European Perspective, Cambridge Classical Journal Supplementary Volume* 32, 19-33.
- Miller, Brett A. (2008) *Duction, Stricture, and the Diachronic Phonetics-Phonology Interface*, Junior Research Fellowship competition dissertation, Trinity College, Cambridge.
- Miller, Brett A. (in press) 'Sonority and the larynx', in Steve Parker (ed.) *The Sonority Controversy*, Berlin: de Gruyter.
- Miller, D. G. (1974) 'Some problems in formulating aspiration and deaspiration rules in Ancient Greek,' *Glossa* 8/2, 211-32.
- Miller, D. G. (1977a) 'Some theoretical and typological implications of an IE root structure constraint', *The Journal of Indo-European Studies* 5/1, 31-40.
- Miller, D. G. (1977b) 'Was Grassman's Law Reordered in Greek?' [Kuhns] Zeitschrift für vergleichende Sprachforschung 91, 131-158.
- Miller, D. G. (1977c) 'Bartholomae's Law and an IE root structure constraint' in Paul J. Hopper et al. (eds.) *Studies in descriptive and historical linguistics: Festschrift for Winfred P. Lehmann*, Amsterdam: Benjamins, 365-392.
- Miller, George A. (1975) 'Some comments on competence and performance', in Doris Aaronson and Robert W. Rieber (eds.) *Developmental Psycholinguistics and Communication Disorders*, Annals of the New York Academy of Sciences vol. 263, 201-204, <u>http://lingo.stanford.edu/sag/L204/Miller.pdf</u> (retrieved 30 December 2011).
- Miller, Wick R. (1980) 'Preaspirated consonants in Central Numic', in Kathryn Klar, Margaret Langdon, Shirley Silver (eds.) *American Indian and Indoeuropean Studies*, Trends in Linguistics Studies and Monographs 16, The Hague: Mouton, 151-57.
- Miller, Wick R., Dirk Elzinga, and John E. McLaughlin (2005) 'Preaspiration and gemination in Central Numic', *International Journal of American Linguistics* 71:413-44.
- Mohanan, K. P. (1991) 'On the bases of radical underspecification', *Natural Language* and *Linguistic Theory* 9, 285–325.
- Mohanan, K. P., Diana Archangeli and Douglas Pulleyblank (2009) 'The emergence of Optimality Theory', in Linda Uyechi and Lian-Hee Wee (eds.) *Reality exploration* and discovery: Pattern interaction in language and life, 143–158. Stanford: CSLI, <u>http://ling75.arts.ubc.ca/pblk/map_2007May15.pdf</u> (retrieved 16 January 2012).
- Moisik, Scott R., Ewa Czaykowska-Higgins, and John H. Esling (2011) 'The epilaryngeal articulator: a new conceptual tool for understanding lingual-laryngeal contrasts', MS.
- Moisik, Scott R. and John H. Esling (2011) 'The 'whole larynx' approach to laryngeal features', *Proceedings of the International Congress of Phonetic Sciences* XVII, <u>http://www.icphs2011.hk/resources/OnlineProceedings/RegularSession/Moisik/Moisik</u>.pdf (retrieved 9 December 2011).
- Moisik, Scott R., John H. Esling, and Lise Crevier-Buchman (2009) 'High-speed laryngoscopic evaluation of the effect of laryngeal parameter variation on aryepiglottic trilling', poster,

<u>http://www.ncl.ac.uk/linguistics/assets/documents/MoisikEslingBuchman_NewcastleP</u> <u>haryngealsPoster_2009.pdf</u> (retrieved 6 February 2012).

- Mołczanow, Janina (2007) 'The problem of the Russian labial fricatives: why v is different', *Poznań Studies in Contemporary Linguistics* 43/1, 49-74, <u>http://versita.metapress.com/content/t557232106607197/fulltext.pdf</u> (retrieved 24 June 2010).
- Monaka, K. (2001) *Shekgalagari stop consonants: A phonetic and phonological study.* PhD dissertation, University of London.
- Monaka, K. C. (2005) 'Language harmonization and orthography: a challenge to Bakgalagarhi lay translators of the Bible', MS, University of Botswana, <u>http://www.theologyinafrica.com/articles/catagories/files/confpap/t2a9.pdf</u> (retrieved 29 July 2008).
- Monaka, Kems (29 October 2007) 'Qualitative aspects of the articulation of plain and ejective stops: observations from Xhosa and Shekgalagari', talk at the Berkeley Phonetics and Phonology Forum, abstract at the forum site http://socrates.berkeley.edu/~ling298/oldtalksFA2007.html (retrieved 29 July 2008).
- Moosmüller, Sylvia and Catherine Ringen (2004) 'Voice and aspiration in Austrian German plosives', *Folia Linguistica* 38/1-2, 43-62.
- Morén, Bruce (2007) 'Minimalist/Substance-Free Feature Theory: why and how', lecture notes, University of Tromsø, <u>http://www.phil.muni.cz/~docekal/egg/handouts/MorenWeek1aho.pdf</u> (first half),

http://www.phil.muni.cz/~docekal/egg/handouts/MorenWeek1bho.pdf (second half) (retrieved 9 February 2012).

- Morpurgo Davies, Anna (1963) *Mycenaeae Graecitatis Lexicon*, Incunabula Graeca 3, Rome: Edizioni dell'Ateneo.
- Moscati, Sabatino and Anton Spitaler, Edward Ullendorff, and Wolfram von Soden (1964) *An Introduction to the Comparative Grammar of the Semitic Languages: Phonology and Morphology*, Wiesbaden: Otto Harrassowitz.
- Muzale, Henry R. T. (1998) *A Reconstruction of the Proto-Rutara Tense/Aspect System*, PhD dissertation, Memorial University of Newfoundland, http://www.collectionscanada.gc.ca/obj/s4/f2/dsk2/tape15/PQDD 0012/NQ36209.pdf

- Narayanan, Shrikanth, Abeer Alwan, and Katherine Haker (1997) 'Toward articulatoryacoustics models for liquid approximants based on MRI and EPG data. Part I. The laterals', *Journal of the Acoustical Society of America* 101/2, 1064-77.
- Nathan, Geoff (2007) 'Phonology', in Dirk Geeraerts and Hubert Cuyckens (eds.) *The Oxford Handbook of Cognitive Linguistics*, 611-31.
- Nevins, Andrew (2003) Review of Eric Raimy's *The Phonology and Morphology of Reduplication*, in *Language* 78/4, 770-773, <u>http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-publications/raimy00-</u> reviewbynevins (retrieved 25 September 2010).
- Nevins, Andrew (2004) 'Phonological Ambiguity (What UG can and can't do to help the reduplication learner)', in Csirmaz et al. (eds.) *Plato's Problems: Papers on Language Acquisition*, MIT Working Papers in Linguistics 48, 113-126, draft http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-publications/nevins04-reduplearner (retrieved 25 September 2010).
- Nevins, Andrew (2007a) Review of Jonathan Barnes' *Strength and Weakness at the Interface: Positional Neutralization in Phonetics and Phonology*, in *Phonology* 24/3, 461-469, <u>http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-publications/barnes06-reviewbynevins</u> (retrieved 25 September 2010).
- Nevins, Andrew (2007b) Review of Andrea Calabrese's Markedness and Economy in a Derivational Model of Phonology, in Journal of Linguistics 43/1, 223-229, <u>http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-</u> <u>publications/calabrese05-reviewbynevins</u> (retreived 25 September 2010).
- Nevins, Andrew (2008a) Review of Paul de Lacy's *Markedness*, in *Journal of Linguistics*, Volume 44/3, 770-781, http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-publications/delacy06-reviewbynevins (retrieved 25 September 2010).
- Nevins, Andrew (2008b) Review of Tobias Scheer's *A Lateral Theory of Phonology: What is CVCV and why should it be*, in *Lingua* 118/3, 425-434, <u>http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-publications/scheer05-reviewbynevins</u> (retrieved 25 September 2010).
- Nevins, Andrew Nevins and David Braun (2008) 'The role of underlying representations in L2 Brazilian English', in Andrea Calabrese and Leo Wetzels (eds.) *Studies in Loan Phonology*, 181-191, proofs <u>http://www.ucl.ac.uk/psychlangsci/staff/linguistics-</u> <u>staff/nevins-publications/nevins-braun-proofs</u> (retrieved 25 September 2010).
- Nevins, Andrew and Ioana Chitoran (2008) 'Phonological representations and the variable patterning of glides', *Lingua* 118/12, 1979-1997, <u>http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-publications/nevins-chitoran08</u> (retrieved 25 September 2010).
- Nevins, Andrew and Keith Plaster (2008) Review of Paul de Lacy's *Markedness*, in *Journal of Linguistics*, Volume 44/3, 770-781, <u>http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-publications/delacy06-reviewbynevins</u> (retrieved 25 September 2010).

- Nevins, Andrew and Bert Vaux (2007) 'The role of contrast in locality: transparent palatal glides in Kyrghyz', *MIT Working Papers in Linguistics* 54, 275-87, <u>http://www.people.fas.harvard.edu/~nevins/nevins-vaux07.pdf</u> (retrieved 22 August 2009)
- Nevins, Andrew and Bert Vaux (2008) 'Underlying representations that do not minimize grammatical violations', in Sylvia Blaho, Patrik Bye, and Martin Kramer (eds.) *Freedom of Analysis*, 35-61, <u>http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-publications/nevins-vaux08</u> (retrieved 25 September 2010).
- Nicolae, Andreea and Andrew Nevins (2010) 'Underlying laryngeal specifications, fricative alternations, and word-size effects', MS, <u>http://www.ucl.ac.uk/psychlangsci/staff/linguistics-staff/nevins-publications/turkishvsrussian-laryngeal</u> (retrieved 25 September 2010).
- Ní Chasaide, Ailbhe and Christer Gobl (1997) 'Voice source variation', in Hardcastle, William J. and Laver, John (eds.) *The Handbook of Phonetic Sciences*, Oxford: Blackwell, 426-61.
- Ní Chiosáin, Máire (1992) 'Vowel features and underspecification: evidence from Irish', in Dressler, Prinzhorn, and Rennison (1992), 157-64.
- Ní Chiosáin, Máire and Jaye Padgett (1993) 'Inherent VPlace', MS, Linguistics Research Center, University of California at Santa Cruz, http://people.ucsc.edu/~padgett/locker/inherent_vplace.pdf (retrieved 29 July 2008).
- Ní Chiosáin, Máire and Jaye Padgett (2001) 'Markedness, segment realization, and locality in spreading', in Linda Lombardi (ed.), *Segmental Phonology in Optimality Theory*, Cambridge: Cambridge University Press, 118-56.
- Nichols, Johanna (1997) 'Chechen phonology' in Kaye (ed.) (1997), 941-71.
- Nichols, Johanna (2003) 'The Nakh-Daghestanian consonant correspondences,' in Dee Ann Holisky and Kevein Tuite (eds.) *Current Trends in Caucasian, East European and Inner Asian Linguistics: Papers in honor of Howard I. Aronson*, Amsterdam: John Benjamins, 207-64.
- Nihalani, P. (1975) 'Velopharyngeal opening in the formation of voiced stops in Sindhi', *Phonetica* 32/2, 89-102.
- Nihalani, Paroo (2006) 'An interface between phonetics and phonology: evidence from implosives', *Proceedings of the Seventh International Seminar on Speech Production*, 159-66, <u>http://www.cefala.org/issp2006/cdrom/pdfversion/23-nihalani.pdf</u> (retrieved 5 December 2011).
- Nkamigbo, Linda Chinelo (2011) 'Experimental analysis of voicing contrast in Igbo', *Unizik Journal of Arts and Humanities* 12/2, 189-203, <u>http://www.ajol.info/index.php/ujah/article/viewFile/72635/61553</u> (retrieved 6 February 2012).
- Nocedal, Jorge and Stephen J. Wright (1999) Numerical Optimization.
- Noonan, Michael, R. P. Bhulanja, J. M. Chhantyal, William Pagliuca (1999) *Chantyal Dictionary and Texts*, Berlin: Mouton de Gruyter.

- Nothofer, Bernd (1975) *The Reconstruction of Proto-Malayo-Javanic*, Verhandelingen van het Koninklijk Instituut voor Taal-, Land- en Volkenkunde 73, 's-Gravenhage: Nijhoff.
- Nurse, Derek and Gérard Philippson (eds.) (2003) The Bantu Languages.
- Odden, David (1978) 'Further evidence for the feature [grave]', *Linguistic Inquiry* 9, 141-44.
- Odden, David (1992) 'Simplicity of underlying representation as motivation for underspecification', *Papers in Phonology: Working Papers in Linguistics No. 41*, The Ohio State University, 85-100.
- Odden, David (2007) 'The unnatural tonology of Zina Kotoko', in Tomas Riad and Carlos Gussenhoven (eds.) *Tones and Tunes: Typological Studies in Word and Sentence Prosody*, Berlin: Walter de Gruyter, 63-89.
- Odden, David (2011) 'Rules v. constraints', ch. 1 in John Goldsmith, Jason Riggle, and Alan C. L. Yu (eds.) *The Handbook of Phonological Theory*, second edition, Oxford: Wiley-Blackwell,

http://media.wiley.com/product_data/excerpt/82/14051576/1405157682-55.pdf (retrieved 16 February 2012).

Ohala, J. J. (1972) 'How is pitch lowered?', *Journal of the Acoustic Society of America* 52/124, revised manuscript: http://linguistics.berkeley.edu/PhonLab/users/ohala/papers/how-whole.pdf (retrieved

http://linguistics.berkeley.edu/PhonLab/users/ohala/papers/how-whole.pdf (retrieved 24 March 2012).

- Ohala, J. J. (1973) 'The physiology of tone', in Larry H. Hyman (ed.) Consonant Types and Tone, Southern California Occasional Papers in Linguistics 1, 2-14, <u>http://linguistics.berkeley.edu/~ohala/papers/physiology_of_tone.pdf</u> (retrieved 15 February 2012).
- Ohala, J. J. (1981a) 'Articulatory constraints on the cognitive representation of speech', in T. Myers, J. Laver, and J. Anderson (eds.) *The Cognitive Representation of Speech*, Amsterdam: North-Holland Publishing, 111-22.
- Ohala, J. J. (1981b) 'The listener as a source of sound change', in C. S. Masek, R. A. Hendrick, and M. F. Miller (eds.) *Papers from the Parasession on Language and Behavior*, Chicago: Chicago Linguistic Society, 178-203.
- Ohala, J. J. (1983) 'The origin of sound patterns in vocal tract constraints', in P. F. MacNeilage (ed.) *The Production of Speech*, New York: Springer-Verlag, 189-216.
- Ohala, J. J. (1989) 'Sound change is drawn from a pool of synchronic variation', in Leiv Breivik, Egil Jahr, and Ernst Hakon (eds.) *Language Change: Contributions to the Study of Its Causes*, New York: Mouton de Gruyter, 172-98.
- Ohala, J. J. (1990a) 'Respiratory activity in speech', in W. J. Hardcastle and A. Marchal (eds.) *Speech Production and Speech Modelling*, Dordrecht: Kluwer, 23-53.
- Ohala, J. J. (1990b) 'The phonetics and phonology of aspects of assimilation', in John Kingston and Mary E. Beckman (eds.) *Papers in Laboratory Phonology I: Between* the Grammar and Physics of Speech, Cambridge: Cambridge University Press, 258-275.

- Ohala, J. J. (1993) 'The phonetics of sound change', in Charles Jones (ed.) *Historical Linguistics: Problems and Perspectives*, London: Longman, 237-278.
- Ohala, J. J. (1995) 'A probable case of clicks influencing the sound patterns of some European languages', *Phonetica* 52, 160-170,

http://linguistics.berkeley.edu/~ohala/papers/clicks.pdf (retrieved 29 December 2011).

- Ohala, John J. (1997) 'Aerodynamics of phonology', in *Proceedings of the Seoul International Conference on Linguistics*, 20th edition, Seoul: Linguistic Society of Korea, 92-97, <u>http://linguistics.berkeley.edu/~ohala/papers/SEOUL2-aero.pdf</u> (retrieved 24 March 2012).
- Ohala, J. J. (2005) 'Phonetic explanations for sound patterns: implications for grammars of competence', in W. J. Hardcastle and J. M. Beck (eds.) *A Figure of Speech: a Festschrift for John Laver*, London: Erlbaum, 23-38, draft <u>http://linguistics.berkeley.edu/~ohala/papers/laver-pdf.pdf</u> (retrieved 25 February 2012).
- Ohala, J. J. and J. Lorentz (1977) 'The story of [w]: an exercise in the phonetic explanation for sound patterns', *Proceedings of the Annual Meeting of the Berkeley Linguistics Society* 3, 577-599, http://linguistics.berkeley.edu/phonlab/users/ohala/papers/story of w.pdf (retrieved 29)

<u>http://inguistics.berkeley.edu/phoniab/users/onala/papers/story_of_w.pdf</u> (retrieved 29 July 2008).

- Ohala, J. J. and Carol Riordan (1979) 'Passive vocal tract enlargement during voiced stops', in J. Wolf and D. Klatt (eds.) *Speech Communication Papers presented at the 97th Meeting of the Acoustical Society of America*, New York: Acoustical Society of America, 89-92, <u>http://linguistics.berkeley.edu/~ohala/papers/ohala_and_riordan.pdf</u> (retrieved 24 March 2012).
- Ohala, J. J. and Manjari Ohala (1972) 'The problem of aspiration in Hindi phonetics', *Annual Bulletin of the Research Institute of Logopedics and Phoniatrics* 6, 39-46, <u>http://web.archive.org/web/20060906160756/http://linguistics.berkeley.edu/phonlab/u</u> <u>sers/ohala/papers/hindi-aspiration.pdf</u> (retrieved 8 February 2012).
- Ohala, Manjari (1991) 'Phonological features of some Indo-Aryan languages', *Language Sciences* 13/2, 107-24.
- Olson, K. and J. Hajek (2001) 'The geographic and genetic distribution of the labial flap', *SIL Electronic Working Papers* 2001-002.
- Olson, K. and J. Hajek (2003) 'Cross-linguistic insights on the labial flap', *Linguistic Typology* 7, 1-30.
- Operstein, Natalie (2010) *Consonant structure and prevocalization*, Amsterdam: John Benjamins.
- Opgenort, Jean Robert (2004) 'Implosive and preglottalized stops in Kiranti', *Linguistics of the Tibeto-Burman Area* 27/1, 1-27, <u>http://www.opgenort.nl/pdf/publications/Opgenort_2004_Implosive_and_preglottalize</u> <u>d stops in Kiranti.pdf</u> (retrieved 11 February 2012).
- Orie, Olanike Ola and Victoria R. Bricker (2000) 'Placeless and historical laryngeals in Yucatec Maya', *International Journal of American Linguistics* 66/3, 283-317.

- Osborn, Henry (1960) Review of Basilio Ma. de Barral (1957), *Diccionario Guarao-Español, Español-Guarao*, Sociedad de la Sciencias Naturales La Salle, Monografías No. 3, Caracas, in *International Journal of American Linguistics* 26/2, 165-67.
- Osu, Sylvester and George Clements (2005) 'Les nasales en ikwere', MS, <u>http://ed268.univ-paris3.fr/lpp/publications/2005_Osu_Nasales_en_ikwere.pdf</u> (retrieved 13 December 2011).
- Owens, John (1985) *A Grammar of Harar Oromo (Northeastern Ethiopia)*, Hamburg: Buske Verlag.
- Padgett, Jaye (1994) 'Stricture and nasal place assimilation', *Natural Language and Linguistic Theory* 12/3, 465-513.

Padgett, Jaye (1995) Stricture in Feature Geometry, Stanford: CSLI Publications.

Padgett, Jaye (2001) 'The unabridged *Features classes in phonology*', MS, <u>http://people.ucsc.edu/~padgett/locker/fctlong.pdf</u> (retrieved 1 September 2009).

- Padgett, Jaye (2002a) 'Feature classes in phonology', Language 78/1, 81-110.
- Padgett, Jaye (2002b) 'Russian voicing assimilation, final devoicing, and the problem of [v] (*or*, The mouse that squeaked)', MS,

http://people.ucsc.edu/~padgett/locker/newvoice.pdf (retrieved 20 January 2009).

- Padgett, Jaye (2008) 'Glides, vowels, and features', Lingua 118/12, 1937-1955.
- Pandey, Pramod (in press) 'Sounds and their patterns in Indic languages', New Delhi: CUP India.
- Paradis, Carole and Jean-François Prunet (1991) 'Introduction: asymmetry and visibility in consonant articulations', in Paradis and Prunet (eds.) (1991), 1-28.
- Paradis, Carole and Prunet, Jean-François (eds.) (1991) *Phonetics and Phonology*, vol.
 2: *The Special Status of Coronals: Internal and External Evidence*. London: Academic Press.
- Park, Hansang (2002) *Temporal and Spectral Characteristics of Korean Phonation Types*, PhD dissertation, University of Texas at Austin,

http://www.lib.utexas.edu/etd/d/2002/parkh026/parkh026.pdf (retrieved 11 January 2012).

Parker, Stephen George (2002) *Quantifying the sonority hierarchy*, PhD dissertation, University of Massachusetts (Amherst),

http://www.ai.mit.edu/projects/dm/theses/more/parker02.pdf (retrieved 5 December 2011).

- Parker, Stephen George (2008) 'Sound level protrusions as physical correlates of sonority', *Journal of Phonetics* 36, 55-90.
- Parker, Stephen George (2011) 'Sonority', ch. 49 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Pater, Joe (2005) 'Non-convergence in the GLA and variation in the CDA', MS, <u>http://roa.rutgers.edu/files/780-1005/780-PATER-0-0.PDF</u> (retrieved 5 February 2009).

Pater, Joe (2008) 'Serial Harmonic Grammar', colloquium presentation, http://people.umass.edu/pater/pater-SHG-usc.pdf (retrieved 5 December 2008).

Pedersen, H. (1951) 'Die gemeinindoeuropäischen und die vorindoeuropäischen Verschlusslaute', Det Kongelige Danske Videnskabernes Selskab, Historiskfilologiske Meddelelser 32.5. Copenhagen.

Pennington, Mark (2005) *The Phonetics and Phonology of Glottal Manner Features*, PhD dissertation, Indiana University, http://mypage.iu.edu/~mwpennin/GlottalMannerDiss-Pennington.pdf (retrieved 14

http://mypage.iu.edu/~mwpennin/GlottalMannerDiss-Pennington.pdf (retrieved 14 January 2012).

- Pennington, Mark (2011) 'Towards phonetically grounded distinctive features, part I', Indiana University Linguistics Club Working Papers Online 11, <u>https://www.indiana.edu/~iulcwp/pdfs/11-PenningtonA.pdf</u> (retrieved 8 December 2011).
- Penney, J. H. W. (1998) Review of Gamkrelidze and Ivanov (1995), Bulletin of the School of Oriental and African Studies, University of London 61/1, 153-55.
- Perkins, Jeremy (2011) 'Consonant-tone interaction and laryngealization in Thai', presentation, 21st Annual Conference of the Southeast Asian Linguistics Society (SEALS 21), 11-13 May 2011, Kasetsart University, Thailand, <u>http://www.jseals.org/seals21/perkins11consonantd.pdf</u>, <u>http://www.sealsxxi.hum.ku.ac.th/participants.html</u> (retrieved 15 February 2012).
- Petrova, Olga, Rosemary Plapp, Catherine Ringen, and Szilárd Szentgyörgi (2006) 'Voice and aspiration: evidence from Russian, Hungarian, German, Swedish, and Turkish', *The Linguistic Review* 23, 1-35, <u>http://www.uiowa.edu/~linguist/faculty/ringen/tlr.2006.001.pdf</u> (retrieved 26 March

2012).

- Petrova, Olga and Szilárd Szentgyörgyi (2004) '/v/ and voice assimilation in Hungarian and Russian', *Folia Linguistica* 38/1, 87-116.
- Philippson, Gérard and Marie-Laure Montlahuc (2003) 'Kilimanjaro Bantu', in Nurse and Philippson (2003), 475-500.
- Phillips, B. S. (1996) 'Lexical diffusion is not lexical analogy', abstract only, <u>http://cat.inist.fr/?aModele=afficheN&cpsidt=2019662</u> (retrieved 26 July 2008); published in *Word* 49/3, 369-81.
- Phillips, R. L. (1973) 'Mnong vowel variations with initial stops', *Mon-Khmer Studies* 4, 118-25.
- Picard, Marc (1994) Principles and Methods in Historical Phonology: From Proto-Algonquian to Arapaho, Montreal.
- Pierrehumbert, Janet (1990) 'On the value of reductionism and formal explicitness in phonological models: comments on Ohala's paper', in John Kingston and Mary E. Beckman (eds.) Papers in Laboratory Phonology I: Between the Grammar and Physics of Speech, Cambridge: Cambridge University Press, 276-79.
- Pierrehumbert, Janet (2000) 'Exemplar dynamics: word frequency, lenition and contrast', MS, Northwestern University (Evanston, IL), prepublication draft of

Pierrehumbert (2001),

http://faculty.wcas.northwestern.edu/~jbp/publications/exemplar_dynamics.pdf (retrieved 21 February 2012).

- Pierrehumbert, Janet (2001) 'Exemplar dynamics: word frequency, lenition and contrast', in Joan L. Bybee and Paul Hopper (eds.) *Frequency and the Emergence of Linguistic Structure*, Typological Studies in Language 45, Amsterdam: John Benjamins, 137-158.
- Pike, Kenneth L. and Eunice Pike (1947) 'Immediate constituents of Mazatec syllables', *International Journal of American Linguistics* 13, 78-91.
- Pinnow, H.-J. (1979) 'Reflections on the history of the Khmer phonemic system', *Mon-Khmer Studies* 8, 103-30, <u>http://sealang.net/archives/mks/pdf/8:103-130.pdf</u> (retrieved 1 March 2012).
- Pisowicz, Andrzej (1976) *Le développement du consonantisme arménien*, Wroclaw: Wydawnictwo Polskiej Akademii Nauk.
- Pisowicz, Andrzej (1997) 'Consonant shifts in Armenian dialects during the postclassical period revisited', in N. Awde (ed.) Armenian Perspectives. 10th Anniversary Conference of the Association Internationale des Études Arméniennes, Richmond, Surrey: Curzon, 215-230.
- Pisowicz, Andrzej (1998) 'What did Hratchia Adjarian mean by voiced aspirates in the Armenian dialects?', *Annual of Armenian Linguistics* 19, Cleveland, Ohio: Cleveland State University, 43-55, <u>http://cat.inist.fr/?aModele=afficheN&cpsidt=2019335</u> (abstract retrieved 17 December 2011).
- Pittayaporn, Pittayawat (2009) *The Phonology of Proto-Tai*, PhD dissertation, Cornell University.
- Pokorny, Julius (1959) Indogermanisches Etymologisches Wörterbuch, Bern: Francke.
- Poletto, Robert E. (1998) *Topics in Runyankore Phonology*, PhD dissertation, The Ohio State University <u>ftp://ftp.ling.ohio-state.edu/pub/odden/Polettodissertation.pdf</u> (retrieved 29 July 2008).
- Pollard, Carl. 'The Nature of Constraint-Based Grammar', paper given at the Pacific Asia Conference on Language, Information, and Computation at Kyung Hee University, Seoul, Korea, 20 Dec. 1996, <u>http://lingo.stanford.edu/sag/L221a/pollard-96.txt</u> (retrieved 29 July 2008).
- Polomé, Edgar (1971) 'Reflexes of laryngeals in Indo-Iranian with especial reference to the problem of hte voiceless aspirates,' in John M. Weinstock (ed.), Saga og språk: Studies in Language and Literature [in honor of Lee Hollander], Austin: Pemberton Press, 233-51.
- Posner, R. (1966) *The Romance Languages: a Linguistic Introduction*, New York: Anchor Books.
- Prince, Alan and Paul Smolensky (2002[1993]) *Optimality Theory: Constraint Interaction in Generative Grammar*, <u>http://roa.rutgers.edu/files/537-0802/537-0802-9RINCE-0-0.PDF</u> (retrieved 29 July 2008).

- Prokosch, E. (1939 [1966]) *A Comparative Germanic Grammar*, Philadelphia: Linguistic Society of America and University of Pennsylvania Press.
- Pronk, Tijmen and Rick Derksen (2011) *Accent Matters: Papers in Balto-Slavic Accentology*, Amsterdam: Rodopi B.V.
- Proulx, P. M. (2005) 'Initial change in Blackfoot', *Calgary Papers in Linguistics* 26, 1-26, <u>http://www.ucalgary.ca/lingcpl/volume_26/Proulx_paper.pdf</u> (retrieved 24 July 2008).
- Puhvel, Jaan (1991) *Hittite Etymological Dictionary, vol. 3: Words Beginning with H*, Berlin: Mouton.
- Purnell, Thomas (2009) 'Phonetic influence on phonological operations', in Raimy and Cairns (2009), 337-54.
- Raimy, Eric and Charles Cairns (eds.) (2009) Contemporary Views on Architecture and Representation in Phonology, Cambridge, Massachusetts: MIT Press.
- Rasmussen, Jens Elmegard (1999) 'On the status of the aspirated tenues and the Indo-European phonation series', *Selected Papers on Indo-European Linguistics*, vol. I, 216-243.
- Ratelle, Georgia (2009) 'The phonological patterning of glottal consonants', University of Ottawa MA mémoire cited in Mielke (2009).
- Reeve, Matthew (2005) 'The Bernoulli Effect and Vocal Fold Vibration', Milton Keynes: Voicesource Publishing, <u>http://www.voicesource.co.uk/article/151</u> (retrieved 7 May 2008).
- Reid, Lawrence A. (2000) 'Sources of Proto-Oceanic initial prenasalization: the view from outside Oceanic', *Oceanic Linguistics Special Publications*, No. 29, Grammatical Analysis: Morphology, Syntax, and Semantics, 30-45.
- Reiss, Charles (2003) 'Quantification in structural descriptions: attested and unattested patterns', *The Linguistic Review* 20, 305-338, <u>http://www.unice.fr/dsl/egg/constanta10/Reiss/Reiss%20quantification_pub.pdf</u> (retrieved 22 February 2012).
- Rendsburg, G. A. (1997) 'Ancient Hebrew phonology', in Kaye (1997), 65-83, <u>http://jewishstudies.rutgers.edu/index.php?</u> <u>option=com_docman&task=doc_view&gid=93&Itemid=158</u> (retrieved 11 September 2009).
- Renou, L. (1961) Grammaire Sanscrite, Paris: Adrien-Maisonneuve.
- Reynolds, Elinor, Paula West, and John Coleman (2000) 'Proto-Indo-European "laryngeals" were vocalic', *Diachronica* 17/2, 351-87.
- Rice, Keren (1989) A Grammar of Slave, Berlin: Mouton de Gruyter.
- Rice, Keren (1993) 'A reexamination of the feature [sonorant]: the status of 'sonorant obstruents', *Language* 69/2, 308-44.
- Rice, Keren (1994) 'Laryngeal features in Athapaskan languages', *Phonology* 11/1, 107-47.
- Rice, Keren (2002) 'Vowel place contrasts', in Amberber, Mengistu and Collins, Peter (eds.) *Language Universals and Variation*, 239-66.

- Rice, Keren (2004) 'Comments on Andrea Calabrese: Prolegomena to a realistic theory of phonology', handout, Symposium on Phonological Theory: Representations and Architecture, City University of New York; symposium site: <u>www.web.gc.cuny.edu/dept/lingu/events/phonology_symposium/Rice_Cala_handout.</u> <u>doc</u> (all retrieved 28 July 2008).
- Rice, Keren (2005) 'Sequential voicing, postnasal voicing, and Lyman's Law revisited', 25-46 in Jeroen Maarten van de Weijer, Kensuke Nanjo, and Tetsuo Nishihara (eds.) *Voicing in Japanese*.
- Rice, Keren (2007) 'Markedness in phonology', in de Lacy (2007), 79-98.
- Rice, Keren (2009) 'Nuancing markedness: a place for contrast', in Raimy and Cairns (2009), 311-22.
- Rice, Keren and Peter Avery (1991) 'On the relationship between laterality and coronality', in Paradis and Prunet (1991), 101-24.
- Ridouane, Rachid (2003) 'Geminates vs. singleton stops in Berber: an acoustic, fiberscopic and photoglottographic study', *15th ICPHS Barcelona*, 1743-46, <u>http://hal.archives-ouvertes.fr/docs/00/38/49/29/PDF/Ridouane-ICPHS03.pdf</u> (retrieved 17 March 2011).
- Ridouane, Rachid (2006) 'On the feature [spread glottis]', <u>http://lpp.univ-paris3.fr/equipe/rachid_ridouane/Ridouane_Spread.pdf</u> (retrieved 29 July 2008).
- Ridouane, Rachid (2008) 'Syllables without vowels: phonetic and phonological evidence from Tashlhiyt Berber', *Phonology* 25, 321-59.
- Ridouane, Rachid, Nick G. Clements, and Rajesh Khatiwada (forthcoming) 'Languageindependent basis of distinctive features', to appear in J. Goldsmith, E. Hume, and L. Wetzels (eds.) *Tones and Features: in Honor of G. Nick Clements*, prepublication draft, <u>http://ed268.univ-</u>

paris3.fr/lpp/pages/EQUIPE/ridouane/Ridouaneetal_Features.pdf (retrieved 18 February 2012).

- Riggle, Jason (2004) *Generation, Recognition, and Learning in Finite State Optimality Theory*, PhD dissertation, <u>http://hum.uchicago.edu/~jriggle/riggle04_2.pdf</u> (retrieved 18 January 2009).
- Rigsby, Bruce and John Ingram (1990) 'Obstruent voicing and glottalic obstruents in Gitksan', *International Journal of American Linguistics* 56/2, 251-263.
- Ringen, Catherine and Helgason, Pétur (2004) 'Distinctive [voice] does not imply regressive assimilation: evidence from Swedish', *International Journal of English Studies* 4/2, 53-71, <u>http://www.uiowa.edu/~linguist/faculty/ringen/ringenijes.pdf</u> (retrieved 29 July 2008).
- Rix, Helmut, Martin Kümmel, Thomas Zehnder, Reiner Lipp, and Brigitte Schirmer (2001) *Lexikon der indogermanischen Verben: die Wurzeln und ihre Primärstammbildungen*, Wiesbaden: Reichert.

Roberts, John R. (1987) Amele, London: Croom Helm.

- Roberts, John R. (1991) 'Reduplication in Amele', *Papers in Papuan Linguistics* 1, 155-146, <u>http://www.scribd.com/doc/62115794/Reduplication-in-Amele</u> (retrieved 4 February 2012).
- Roberts, John R. (2001) 'Impersonal constructions in Amele', in Alexandra Y. Aikhenvald, R.M.W. Dixon, and Masayuki Onishi (eds.) Non-Canonical Marking of Subjects and Objects, Amsterdam: John Benjamins, 201-250.
- Robins, R. H. (1977) 'Distinctive feature theory', in D. Armstrong and C. H. van Schooneveld (eds.) *Roman Jakobson: Echoes of his Scholarship*, Lisse: The Peter de Ridder Press, 391-402, reprinted in Charles W. Kreidler (ed.) (2001) *Phonology: Critical Concepts*, vol. 2, 53-64.
- Roca, Iggy (1994) Generative Phonology.
- Rood, David (1975) 'The implications of Wichita phonology', Language 51/2, 315-37.
- Rood, David (1976) Wichita Grammar, New York: Garland.
- Rood, David S. and Allan R. Taylor (1996) 'Sketch of Lakota, a Siouan language, part I', *Handbook of North American Indians* 17, 440-482, http://lakxotaiyapi.freecyberzone.com/sk1.htm (retrieved 6 February 2012).
- Rose, Phil (1989) 'Phonetics and phonology of Yang tone: phonation types in Zhenhai', *Cahiers de linguistique: Asie orientale* 18/2, 229-45, http://www.persee.fr/web/revues/home/prescript/article/clao_0153-3320 1989 num 18 2 1304 (retrieved 15 April 2010).
- Rose, Phil (1994) 'Wenzhou tonal acoustics: depressor and register effects in Chinese tonology', Proceedings of the 1994 Australian International Conference on Speech Science and Technology (ASSTA), vol. 1, 138-143, <u>http://www.assta.org/sst/SST-94-Vol-l/cache/SST-94-VOL1-Chapter6-p8.pdf</u> (retrieved 31 January 2012).
- Rose, Sharon and Rachel Walker (2004) 'A typology of consonant agreement as correspondence', *Language* 80, 475-531, <u>http://idiom.ucsd.edu/~rose/ABC.pdf</u>, citation information at <u>http://idiom.ucsd.edu/~rose/indexpubs.html</u> (retrieved 7 May 2008), cf. 2001 version, <u>http://roa.rutgers.edu/files/458-0801/458-0801-ROSE-0-0.PDF</u> (retrieved 27 Mar 2008).
- Rose, Sharon and Rachel Walker (2011) 'Harmony systems', in John A. Goldsmith, Jason Riggle, and Alan Yu (eds.) *The Handbook of Phonological Theory*, ch. 8.
- Rountree, Sharon (1972) 'The phonological structure of stems in Saramaccan', in Joseph E. Grimes (ed.), *Languages of the Guianas*, Summer Institute of Linguistics Publications in Linguistics 35, 22-26

http://www.archive.org/stream/rosettaproject_srm_phon-1 (retrieved 11 August 2011).

- Rousselot, Jean-Pierre (1899) 'Note sur l'évolutions phonétiques', *La Parole: Revue internationale de rhinologie, otologie et laryngologie* 1, 127-136, http://gallica.bnf.fr/ark:/12148/bpt6k57626142/f121.image (retrieved 5 January 2012).
- Ruhlen, M. (1975) 'A guide to the languages of the world', Stanford: Stanford University Language Universals Project.
- Sagey, Elizabeth (1986) *The Representation of Features in Non-Linear Phonology: The Articulator Node Hierarchy.*

Saito, Shuzo (1992) Speech Science and Technology, Tokyo: Ohmsha.

Salffner, Sophie (2004) Learning Ega: A Linguistic Frame of Reference for Primers for Endangered Languages, Magisterarbeit, Bielefeld,

http://emeld.org/school/case/ega/OnAnEgaPrimer.pdf (retrieved 18 February 2012).

- Salmons, Joseph (1993) 'The Glottalic Theory: Survey and Synthesis', *Journal of Indo-European Studies*, Monograph no. 10.
- Salmons, Joseph (2005) 'On the status of the Indo-European labiovelar stops', *Indogermanische Forschungen* 110, pdf from author.
- Samuels, Bridget (2009a) *The Structure of Phonological Theory*, PhD dissertation, Harvard, <u>http://ling.auf.net/lingBuzz/000863</u> (retrieved 24 June 2010).
- Samuels, Bridget (2009b) 'The third factor in phonology', *Biolinguistics* 3/2-3, 355-82, <u>http://biolinguistics.eu/index.php/biolinguistics/article/view/86/107</u> (retrieved 1 September 2011).
- Samuels, Bridget (2009c) 'Primitive operations in phonology', lecture slides, University of Maryland, 20 May 2009, <u>http://www.bridgetsamuels.com/pdfs/UMDslides.pdf</u> (retrieved 1 September 2011).
- Samuels, Bridget (in press) 'A Minimalist Program for phonology', to appear in *The Oxford Handbook of Linguistic Minimalism*,

http://ling.umd.edu/~bridget/pdfs/minimalism-ch.pdf (retrieved 1 September 2011).

- Samuels, Bridget (in press) 'The emergence of phonological forms', in A. M. DiSciullo (ed.) *Towards a Biolinguistic Understanding of Grammar: Essays on the Interfaces*, Amsterdam: John Benjamins.
- Sankoff, Gillian and Hélène Blondeau (2007) 'Language change across the lifespan: /r/ in Montreal French', *Language* 83/3, 560-588.
- Sapir, E. (1992[1930]) Southern Paiute, a Shoshonean Language, in Sapir (1992), 17-314.
- Sapir, E. (1992[1931]) Southern Paiute Dictionary, in Sapir (1992), 557-752 (with an English index by Wick R. Miller, 753-78).
- Sapir, Edward (1917) 'Southern Paiute and Nahuatl: a study in Utoaztekan, part II (concluded)', *American Anthropologist* 17/2, 306-28.
- Sapir, Edward (1938) 'Glottalized continuants in Navaho, Nootka, and Kwakiutl (with a note on Indo-European)', *Language* 14/4, 248-74.
- Sapir, Edward (1992) (William Bright, ed.) *The Collected Works of Edward Sapir*, vol.10. *Southern Paiute and Ute Linguistics and Ethnography*, Mouton de Gruyter: Berlin.

Sawashima, Masayuki (1970) 'Glottal adjustments for English obstruents', paper presented that the meeting of the New York Speech and Hearing Association, May 1970, Haskins Laboratories Status Report on Speech Research SR-21, 187-200, http://www.haskins.yale.edu/SR/SR021/SR021_10.pdf (retrieved 11 January 2012).

Sawashima, Masayuki (1997) 'Phonetic control of the glottal opening', in Shigeru Kiritani (ed.) *Speech Production and Language: In Honor of Osamu Fujimura*, Berlin: de Gruyter, 69-110.

- Scarborough, Rebecca (2005) Class handout, 29 Sept. 2005, 'Introduction to Acoustics', http://www.stanford.edu/class/linguist205/index_files/Handout%202%20-%20Intro %20to%20Acoustics.pdf (retrieved 30 July 2008).
- Scarborough, Rebecca, Patricia Keating, S. Mattys, T. Cho, A. Alwan (2009) 'Optical phonetics and visual perception of lexical and phrasal Stress in English', *Language* and Speech 51/2-3, 135-175,

http://www.linguistics.ucla.edu/people/keating/Scarboroughetal_2009.pdf (retrieved 25 March 2012).

- Schachter, P. (1976) 'An unnatural class of consonants in siSwati', *Studies in African Linguistics* supplement 6, 211-20.
- Schadeberg, Thilo C. (1994) 'Comparative Kadu wordlists', *Afrikanistische Arbeitspapiere* 40, 11-48,

https://www.openaccess.leidenuniv.nl/bitstream/1887/8829/1/5_1234891_049.pdf (retrieved 29 July 2008).

- Schiefer, L. (1986) 'F₀ in the production and perception of breathy stops: evidence for Hindi', *Phonetica* 43, 43-69.
- Schiefer, L. (2004[1992]) 'Trading relations in the perception of stops and their implications for a phonological theory', in Gerard J. Docherty and D. Robert Ladd (eds.) *Gesture, Segment, Prosody*, Papers in Laboratory Phonology vol. 2, Cambridge: Cambridge University Press, 296-313.
- Scholz, Sibyl (2003) The Status of Coronals in Standard American English, doctoral dissertation, Universität zu Köln, <u>http://deposit.ddb.de/cgi-bin/dokserv?</u> <u>idn=972704620&dok_var=d1&dok_ext=pdf&filename=972704620.pdf</u> (retrieved 30 July 2008).
- Schrijver, Peter (1995) Studies in British Celtic Historical Phonology, Amsterdam.
- Schrock, Terrill B. (2011) 'The phonological segments and syllabic structure of Ik', *SIL Electronic Working Papers* 2011-004, <u>http://www.sil.org/silewp/2011/silewp2011-004.pdf</u> (retrieved 11 February 2012).
- Schwartz, Jean-Luc, Louis-Jean Boë, Pierre Badin, Thomas Sawallis (2012) 'Grounding stop-place systems in the perceptuo-motor substance of speech: on the universality of the labial-coronal-velar stop series', *Journal of Phonetics* 40, 20-36.
- Schwink, Frederick W. (1994) *Linguistic Typology, Universality, and the Realism of Reconstruction*, Journal of Indo-European Studies Monograph 12.
- Scripture, Edward Wheeler (1901) *The Elements of Experimental Phonetics*, New York: Charles Scribner's Sons.
- Selkirk, Elisabeth (2004[1992]) 'Comments on chapter 12' [= Schiefer 1992], in Gerard J. Docherty and D. Robert Ladd (eds.) *Gesture, Segment, Prosody*, Papers in
 - Laboratory Phonology vol. 2, Cambridge: Cambridge University Press, 313-321.
- Shadle, Christine H. (1999) 'The aerodynamics of speech', in William J. Hardcastle and John Laver (eds.) *The Handbook of Phonetic Sciences*, 33-64.
- Shadle, Christine H. (2010) 'The aerodynamics of speech', in William J. Hardcastle, John Laver, and Fiona E. Gibbon (eds.) *The Handbook of Phonetic Sciences*, 39-80.

- Shahin, Kimary (2011) 'Pharyngeals', ch. 25 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Shaw, Jason, Adamantios I. Gafos, Philip Hoole, and Chakir Zeroual (2009)'Syllabification in Moroccan Arabic: evidence from patterns of temporal stability in articulation', *Phonology* 26, 187-215.
- Shaw, Patricia A. (1991) 'Consonant harmony systems: the special status of coronal harmony', in Paradis and Prunet (1991), 125-58.
- Sheft, Stanley (2008) 'Envelope processing and sound-source perception', in William A. Yost (ed.) *Auditory Perception of Sound Sources*, New York: Springer, 233-80.
- Shin, Seung-Hoon (n.d.) 'Consonantal weakening and licensing in Optimality Theory', MS, Dongseo University, Kusan,

http://dspace.wul.waseda.ac.jp/dspace/bitstream/2065/12119/1/PACLIC13-203-214.pdf (retrieved 22 May 2008).

Shosted, Ryan K., John J. Ohala, and Ronald Sprouse (2005) 'Analog study of a nasalized fricative', paper, Conference on Turbulences, Zentrum für Allgemeine Sprachwissenschaft, Berlin, 13-14 October 2005, <u>http://www.zas.gwz-berlin.de/turbulences/abstract/Shos2_ab.pdf</u> (retrieved 28 July 2008).

Shrofel, Salina (1980) 'Mutual bleeding: a case from Island Lake Ojibwa', *Toronto Working Papers in Linguistics* 1, page range not given, <u>http://twpl.library.utoronto.ca/index.php/twpl/article/view/6447</u> (retrieved 2 September 2009).

- Sidwell, Paul (2003) *A Handbook of Comparative Bahnaric*, Australia: Pacific Linguistics.
- Sidwell, Paul (2005) 'Proto-Katuic phonology and the sub-grouping of Mon-Khmer languages', in Paul Sidwell (ed.) SEALSXV: papers from the 15th meeting of the Southeast Asian Linguistics Society, Canberra: Pacific Linguistics, 193-204, <u>http://sealang.net/sala/archives/pdf8/sidwell2005proto.pdf</u> (retrieved 18 February 2012).
- Sidwell, Paul (2006) 'Preface' to Shorto, H. L. *A Mon-Khmer Comparative Dictionary*, Canberra: Pacific Linguistics.
- Sidwell, Paul (2007a) 'Katuic', web page, http://web.archive.org/web/20071018114146/http://www.anu.edu.au/~u9907217/langu ages/Katuic.html (retrieved 18 February 2012).
- Sidwell, Paul (2007b) 'West Bahnaric', web page, http://web.archive.org/web/20070809010501/http://www.anu.edu.au/~u9907217/langu ages/Bahnaric-West.html (retrieved 18 February 2012).
- Siebs, Theodor (1904) 'Anlautstudien', [Kuhns] Zeitschrift für vergleichende Sprachforschung 37, 277-324.
- Sihler, Andrew L. (1995) *New Comparative Grammar of Greek and Latin*. Oxford: Oxford University Press.

- Silverman, Daniel (1997) 'Laryngeal complexity in Otomanguean vowels', *Phonology* 14, 235-261, <u>http://www.seedyroad.com/academics/Phonology-14-235-261.pdf</u> (retrieved 31 January 2012).
- Silverman, Daniel (2000) 'Hypotheses concerning the phonetic and functional origins of tone displacement in Zulu', *Studies in African Linguistics* 29/2, 1-32, <u>http://seedyroad.com/academics/StudiesInAfricanLinguistics-29-2-3-32</u> (retrieved 28 January 2012).
- Simpson, Adrian P. and Sarah Neuhauser (2010) 'The persistency of epiphenomenal sound production in foreign accent disguise', abstract of paper for *I*nternational Association for Forensic Phonetics and Acoustics 2010 conference, Department of Phonetics, Trier University, Germany, <u>http://www.personal.uni-</u>

jena.de/~x1siad/papers/simpsonNeuhauser_iafpa10.pdf (retrieved 29 December 2011). Singer, Emily (2009) 'Thought translator knows vowels from consonants', *Technology*

- *Review*, editorial, 21 October 2009, Cambridge, Massachusetts: MIT, <u>http://www.technologyreview.com/blog/editors/24279/</u> (retrieved 13 March 2012). Cites electrocorticographic experiments by Gerwin Schalk et al., linking to Schalk's online profile at the Wadsworth Center (New York State Department of Health), <u>http://www.wadsworth.org/resnres/bios/schalk.htm</u> (retrieved 13 March 2012).
- Smith, Neil (2010) Acquiring Phonology: A Cross-Generational Case Study, Cambridge Studies in Linguistics 124, Cambridge: Cambridge University Press, <u>http://phonphon.pbworks.com/f/Acquiring+Phonology_Neil+Smith.pdf</u> (retrieved 16 February 2012).
- Smith, T. C. and C. E. Jahr (2002) 'Self-inhibition of olbactory bulb neurons', *Natural Neuroscience* 5/8, 760-766, <u>http://www.ncbi.nlm.nih.gov/pubmed/12089528</u> (retrieved 23 March 2012)
- Smolensky, Paul (1996) 'The initial state and "Richness of the Base", Technical Report JHU-CogSci-96-4, John Hopkins University, Department of Cognitive Science, <u>http://roa.rutgers.edu/files/154-1196/roa-154-smolensky-2.pdf</u> (retrieved 14 October 2009).
- Smolensky, Paul (2006) 'On theoretical facts and empirical abstractions', in E. Bakovic, J. Ito, and J. McCarthy (eds.) *Wondering at the Natural Fecundity of Things: Essays in Honor of Alan Prince*, <u>http://escholarship.org/uc/item/1066p928</u> (retrieved 19 June 2010).
- Smolensky, Paul and Géraldine Legendre (eds.) (2006) *The Harmonic Mind* (two volumes), Cambridge, Massachussetts: MIT Press. (Note: since the chapter authors vary, but Smolensky is generally a co-author, all chapters of this work are cited as Smolensky et al. 2006.)
- Snoeren, Natalie D., Pierre A. Hallé, and Juan Segui (2006) 'A voice for the voiceless: production and perception of assimilated stops in French', *Journal of Phonetics* 34/2, 241-68.
- Solé, María-Josep (2002) 'Aerodynamic characteristics of trills and phonological patterning', *Journal of Phonetics* 30, 655–688.

Solé, María-Josep (2007) 'Controlled and mechanical properties in speech: a review of the literature', in Maria-Josep Solé, Patrice Speeter Beddor, and Manjari Ohala (eds.) *Experimental Approaches to Phonology*, Oxford: Oxford University Press, 302-322.

SPE = Chomsky and Halle (1968)

Steed, William (2011) 'Using Qingtian tone sandhi data to refine depressor morphotonemics', *International Congress of Phonetic Sciences* XVII, Hong Kong, 17-21 August 2011, 1894-1897,

http://www.icphs2011.hk/resources/OnlineProceedings/RegularSession/Steed/Steed.p df (retrieved 31 January 2012).

- Steriade, Donca (1993) 'Segments, contours and clusters', in André Crochetière, Jean-Claude Boulanger, and Conrad Ouellon (eds.) *Proceedings of the XVth International Congress of Linguistics*, Sainte-Foy: Les presses de l'université Laval.
- Steriade, Donca (1994) 'Complex onsets as single segments: the Mazateco pattern', in J. Cole and C. Kisseberth (eds.) *Perspectives in Phonology*, Stanford: Center for the Study of Language and Information, 203-291.
- Steriade, Donca (1995) 'Underspecification and markedness', in John Goldsmith (ed.), *The Handbook of Phonological Theory*, Oxford: Blackwell, 114-74.
- Steriade, Donca (1999) 'Paradigm uniformity and the phonetics-phonology boundary', draft to appear in *Papers in Laboratory Phonology 5*, <u>http://www.linguistics.ucla.edu/people/steriade/papers/LabPhon98.pdf</u> (retrieved 30 July 2008).
- Steriade, Donca (2000) 'Directional assymetries in place assimilation: a perceptual account', paper, *ICPHS 2000*,

http://www.linguistics.ucla.edu/people/steriade/papers/ICPHS2000.pdf (retrieved 30 July 2008).

Steriade, Donca (2001) 'Directional asymmetries in place assimilation', in Elizabeth V. Hume and Keith Johnson (eds.) *The Role of Speech Perception in Phonology*, New York: Academic Press, 219-250,

http://www.linguistics.ucla.edu/people/steriade/papers/ICPHS2000.pdf (retrieved 5 December 2011).

Steriade, Donca (2009) 'The phonology of perceptibility effects: the P-map and its consequences for constraint organization', in Kristin Hanson and Sharon Inkelas (eds.) *The Nature of the Word: Studies in Honor of Paul Kiparsky*, Cambridge, Massachusetts: MIT Press, 151-180,

www.linguistics.ucla.edu/people/steriade/papers/P-map_for_phonology.doc (retrieved 5 December 2011).

Stevens, Alan M. (1966) 'The Madurese Reflexes of Proto-Malayopolynesian', *Journal* of the American Oriental Society 86, 147-56,

http://www.jstor.org/stable/pdfplus/596427.pdf (retrieved 29 July 2008).

Stevens, Alan M. (1968) 'Madurese phonology and morphology,' New Haven: American Oriental Society.

- Stevens, Christopher M. (1992) 'The use and abuse of typology in comparative linguistics: an update on the controversy', *Journal of Indo-European Studies* 20/1, 45-58.
- Stevens, Kenneth N. (1988) 'Modes of vocal fold vibration based on a two-section model', in O. Fujimura (ed.), *Vocal Physiology: Voice Production, Mechanisms and Functions*, New York: Raven Press, 357-67.
- Stevens, Kenneth N. (1998) Acoustic Phonetics, Cambridge, Massachusetts: MIT Press.
- Stevens, Kenneth N. (1999) 'Articulatory-acoustic-auditory relationships', in William J. Hardcastle and John Laver (eds.) *The Handbook of Phonetic Sciences*, 462-506.
- Stevens, Kenneth N. and Samuel J. Keyser (1989) 'Primary features and their enhancement in consonants', *Language* 65, 81–106.
- Stevens, Kenneth N. and Samuel J. Keyser (2010) 'Quantal theory, enhancement and overlap', *Journal of Phonetics* 38, 10-19.
- Stevens, Mary and John Hajek (2004a) 'Preaspiration in Sienese Italian and its interaction with stress in /VC:/ sequences', in B. Bel and I. Marlien (eds.) Speech Prosody 2004. Proceedings of the 2nd International Conference on Speech Prosody 2004, 23-26 March 2004, Nara, Japan. SproSIG, 57-60, <u>http://www.iscaspeech.org/archive/sp2004/sp04_057.pdf</u> (retrieved 29 July 2008).
- Stevens, Mary and John Hajek (2004b) 'How pervasive is preaspiration? Investigating sonorant devoicing in Sienese Italian', in S. Cassidy, F. Cox, R. Mannell, and S. Palethorpe (eds.) *Proceedings of the Tenth Australian International Conference on Speech Science and Technology*, Sydney: ASSTA, 334-339, <u>http://www.assta.org/sst/2004/proceedings/papers/sst2004-406.pdf</u> (retrieved 29 July 2008).
- Stevens, Mary and J. Hajek (2005a) 'Raddoppiamento sintattico and word-medial gemination in Italian: Are they the same or are they different? The evidence from spontaneous speech', in R. Gess and E. J. Rubin (eds.) *Theoretical and Experimental Approaches to Romance Linguistics*, Amsterdam: John Benjamins, 257-271.
- Stevens, Mary and J. Hajek (2005b) 'Spirantization of /p t k/ in Siennese Italian and socalled semi-fricatives', presentation, Interspeech 2005 – Eurospeech, Lisbon, 4-8 September 2005, 2893-2896, <u>http://www.isca-</u>

<u>speech.org/archive/interspeech_2005/i05_2893.html</u> (retrieved 18 February 2012). Stevens, Mary and John Hajek (2007) 'Towards a phonetic conspectus of preaspiration:

- acoustic evidence from Sienese Italian', in Trouvain, J. and Barry, W. J. (eds.) *Proceedings of the 16th International Congress of Phonetics Sciences (ICPhS XVI),* Saarbrücken, 429-432, <u>http://www.icphs2007.de/conference/Papers/1319/1319.pdf</u> (retrieved 29 July 2008).
- Stewart, John M. (1989) 'Kwa', in John Bendor-Samuel and Rhonda L. Hartell (eds.) *The Niger-Congo Languages*, Lanham: University Press of America, 216-245.
- Stewart, John M. (1993) 'The second Tano consonant shift and its likeness to Grimm's Law', *Journal of West African Languages* 23/1, 3-39,

http://www.journalofwestafricanlanguages.org/Files/pdf/23-1/JWAL-23-1-Stewart.pdf (retrieved 27 February 2012).

Stoakes, Hywel, Janet Fletcher, and Andy Butcher (2007) 'An acoustic study of Bininj Gun-Wok medial stop consonants', *Proceedings of the International Conference of Phonetic Sciences* XVI, 869-872,

http://www.icphs2007.de/conference/Papers/1413/1413.pdf (retrieved 13 December 2011).

- Strazny, Philipp (2003) 'Depression in Zulu: tonal effects of segmental features', in Jeroen Maarten van de Weijer, Vincent van Heuven, and Harry van der Hulst (eds.) *The Phonological Spectrum: Segmental Structure*, Amsterdam: John Benjamins, 223-240.
- Stuart-Smith, Jane (2004) *Phonetics and Phonology: Sound Change in Italic*, Oxford: Oxford University Press.
- Sumner, Meghan (2009) 'Research interests', author's web page, http://www.stanford.edu/~sumner/research/index.html (retrieved 31 August 2009)
- Sumner, Meghan and A. G. Samuel (2005) 'Perception and representation of regular variation: the case of final /t/', *Journal of Memory and Language* 52, 322-338, <u>http://www.stanford.edu/~sumner/publications/2005_sumner_jml.pdf</u> (retrieved 31 August 2009).
- Sumner, Meghan and A. G. Samuel (2007) 'Lexical inhibition and sublexical facilitation are surprisingly long lasting', *Journal of Experimental Psychology: Learning, Memory, and Cognition* 33/4, 769-90, <u>http://www.stanford.edu/~sumner/publications/2007_sumner_jeplmc.pdf</u> (retrieved 5 September 2009).
- Sundberg, J. and Nordström, P. E. (1976) 'Raised and lowered larynx the effect on vowel formant frequencies', *STL-QPSR* 17/2-3, 35-39, <u>http://www.speech.kth.se/prod/publications/files/qpsr/1976/1976_17_2-3_035-039.pdf</u> (retrieved 30 July 2008).
- Sussman, H. M. and J. Shore (1996) 'Locus equations as phonetic descriptors of consonantal place of articulation', *Perception and Psychophysics* 58/6, 936-946, <u>http://www.mendeley.com/research/locus-equations-phonetic-descriptors-consonantal-place-articulation/</u> (retrieved 18 February 2012).
- Svantesson, Jan-Olof and David House (2006) 'Tone production, tone perception, and Kammu tonogenesis', *Phonology* 23, 309-333.
- Svantesson, Jan-Olof, Anna Tsendina, Anastasia Karlsson, and Vivian Franzén (2005) *The Phonology of Mongolian*, Oxford: Oxford University Press.
- Swiggers, Pierre (1989) 'Latin Phonology and the Glottalic Model,' in Vennemann (1989), 217-19.
- Szemerényi, Oswald (1970 [1989]) *Einführung in die vergleichende Sprachwissenschaft*, Darmstadt: Wissenschaftliche Buchgesellschaft.
- Szemerényi, Oswald (1972) 'Comparative Linguistics' in T. A. Sebeok (ed.), *Current Trends in Linguistics* 9, 119-95.

- Szemerényi, Oswald (1985) 'Recent Developments in Indo-European Linguistics,' *Transactions of the American Philological Society*, Oxford: Blackwell, 1-71.
- Szemerényi, Oswald (1996) *Introduction to Indo-European Linguistics*, translation and expansion of *Einführung in die vergleichende Sprachwissenschaft* fourth edition (1990); Oxford: Clarendon Press.
- Tang, Katrina Elizabeth (2008) The Phonology and Phonetics of Consonant-Tone Interaction, PhD dissertation, UCLA, <u>http://www.linguistics.ucla.edu/general/dissertations/TangPhonologyPhoneticsConson</u> antToneInteraction small.pdf (retrieved 28 January 2012).
- Taylor, C. (1985) Nkore-Kiga, London: Croom Helm.
- Tesar, Bruce (1998) 'Error-driven learning in Optimality Theory via the efficient computation of optimal forms', in Pilar Barbosa, Danny Fox, Paul Hagstrom, Martha Jo McGinnis and David Pesetsky (eds.) *Is the best good enough? Optimality and competition in syntax*, Cambridge, MA, MIT Press, 421-35.
- Tesar, Bruce (2007a) 'Learnability', in de Lacy (2007), 555-74.
- Tesar, Bruce (2007b) 'Output-driven functions: the mathematics of linguistic sound patterns', abstract of talk given at the Mathematics Association of America Seaway Section and New York State Association of Two-Year Colleges, Region I, Fall 2007 Joint Conference, 19-20 October 2007, Monroe Community College, Rochester, New York, http://www.math.binghamton.edu/maa_seaway/Meetings/Meetings/MonroeCC-07b/Fall_2007_Program_with_Abstracts.pdf (retrieved 17 October 2009).
- Tesar, Bruce (2008) 'Output-driven maps', MS, Rutgers University, <u>http://roa.rutgers.edu/files/956-0308/956-TESAR-0-0.PDF</u> (retrieved 29 August 2009).
- Tesar, Bruce (2009) 'Learning phonological grammars for output-driven maps', to appear in *The Proceedings of NELS* 39, <u>http://roa.rutgers.edu/files/1013-0209/1013-TESAR-0-0.PDF</u> (retrieved 29 August 2009).
- Tesar, Bruce and Alan Prince (2003) 'Using phonotactics to learn phonological alternations', <u>http://roa.rutgers.edu/view.php3?id=849</u> (retrieved 27 August 2009).
- Tesar, Bruce, John Alderete, Graham Horwood, Nazarré Merchant, Koichi Nishitani, and Alan Prince (2003) 'Surgery in language learning', in *The Proceedings of WCCFL* 22, 477-490, <u>http://roa.rutgers.edu/files/619-1003/619-TESAR-0-0.PDF</u> (retrieved 29 August 2009).
- Tesar, Bruce and Paul Smolensky (2000) Learnability in Optimality Theory.
- Testen, David (1997) 'Ossetic phonology' in Kaye (1997), 707-732.
- Titze, Ingo R. (1979) 'Variations of pitch and intensity with pre-phonatory laryngeal adjusgtments', in Harry Francis Hollien and Patricia Hollien (eds.) *Current Issues in the Phonetic Sciences*, Current Issues in Linguistc Theory 9, Amsterdam: John Benjamins, 209-216.
- Thelwall, Robin and M. Akram Sa'adeddin (1999) 'Arabic', in International Phonetic Association (1999), 51-54.

- Theraphan, L. Thongkum (1990) 'The interaction between pitch and phonation type in Mon: phonetic implications for a theory of tonogenesis', *Mon-Khmer Studies* 16-17, 11-24, <u>http://sealang.net/archives/mks/pdf/16-17:11-24.pdf</u> (retrieved 24 March 2012).
- Thomas, David D. (1971) Chrau Grammar, University of Hawaii Press.
- Thompson, Becky (2006) 'The pharyngeal hierarchy', *SKY Journal of Linguistics* 19, 229-37, <u>http://www.ling.helsinki.fi/sky/julkaisut/SKY2006/Thompson.pdf</u> (retrieved 20 April 2010).
- Thompson, Rupert J. E. (2005) 'Two notes on Mycenaean', Živa Antika 55, 107-115.
- Thompson, Rupert J. E. (2010) 'Mycenaean Greek', in Egbert J. Bakker (ed.) *A Companion to the Ancient Greek Language*, 189-99.
- Thongkum, T. L. (1987) 'Phonation types in Mon-Khmer languages', University of California, Los Angeles, Working Papers in Phonetics, 76, 29-48.
- Thráinsson , H. (1978) 'On the phonology of Icelandic preaspiration', Nordic Journal of Linguistics 1/1, 3-54.
- Thurgood, Ela (2004) 'Phonation types in Javanese,' Oceanic Linguistics 43/2, 1-295.
- Thurgood, Graham [2008] 'Vietnamese and tonogenesis: revising the model and the analysis', update of paper in *Diachronica* 19/2, 2002, 333-63, <u>http://www.csuchico.edu/~gthurgood/Papers/TonogenesisHarris.pdf</u> (retrieved 29 July 2008).
- Thurneysen, Rudolf (1946) *A Grammar of Old Irish*, revised edition, translated from the German by D. A. Binchy and Osborn Bergin; Dublin: Dublin Institute for Advanced Studies.
- Tiede, Mark K. (1993) 'An MRI-based study of pharyngeal volume contrasts in Akan', Haskins Laboratories Status Report on Speech Research SR-113, 107-130, <u>http://www.haskins.yale.edu/SR/sr113/sr113_08.pdf</u> (retrieved 18 February 2012).
- Topintzi, Nina (2006a) *Moraic Onsets*, PhD thesis, University of London, <u>http://roa.rutgers.edu/files/848-0706/848-TOPINTZI-0-0.PDF</u> (retrieved 23 March 2012).
- Topintzi, Nina (2006b) 'On the existence of moraic onset geminates', *Natural Language* and Linguistic Theory 26/1, 147-184, <u>http://www.uni-</u>
- <u>leipzig.de/~topintzi/papers/topintzi_NLLT%202008.pdf</u> (retrieved 23 March 2012). Topintzi, Nina (2007) 'The relationship between voicing and stress with special focus on
- Karo', talk abstract, Old World Conference in Phonology 4, January 18-21, 2007, <u>http://www.rhodes.aegean.gr/tms/OCP4/Speakers/named-ntopintzi.pdf</u> (retrieved 20 August 2009)
- Trabant, Jürgen (2005) 'Faut-il défendre Saussure contre ses amateurs ? Notes item sur l'étymologie saussurienne', *Langages* 39/159, 111-124, <u>http://www.persee.fr/web/revues/home/prescript/article/lgge_0458-726x_2005_num_39_159_2655</u> (retrieved 30 December 2011).
- Tragut, Jasmine (2009) Armenian: Modern Eastern Armenian, London Oriental and African Language Library 14, Amsterdam: John Benjamins.

- Traill, A., J.S.M. Khumalo, and P. Frijdhon (1987) 'Depressing facts about Zulu', *African Studies* 46/2, 255-274.
- Trigo, Loren (1991) 'On pharynx-larynx interactions', Phonology 8, 113-36.
- Tryon, Darrell T. (ed.) (1994) Comparative Austronesian Dictionary: An Introduction to Austronesian Studies, part 1, fascicle 1, Berlin: Mouton de Gruyter.
- Tsuchida, Ayako, Abigail C. Cohn, and Masanobu Kumada (2000) 'Sonorant devoicing and the phonetic realization of [spread glottis] in English', *Working Papers of the Cornell Phonetics Laboratory* 13, 167-181, <u>http://velar.phonetics.cornell.edu/pdfs/Cohn_Son_Dev_WPCPL.pdf</u> (retrieved 6 Febrary 2012).
- Tuggy, David (2008) 'Common questions about Nahuatl', webpage of the Instituto Lingüístico de Verano, A.C. (Summer Institute of Linguistics in Mexico), <u>http://www.sil.org/mexico/nahuatl/10i-NahuatlQuestions.htm</u> (retrieved 22 May 2008).
- Tuller, Betty, J.A. Scott Kelso, and Katherine S. Harris (1983) 'Further evidence for the role of relative timing in speech: a reply to Barry', Haskins Laboratories Status Report on Speech Research SR-74/75, 187-195,
- http://www.haskins.yale.edu/sr/sr074/SR074_14.pdf (retrieved 6 Febrary 2012). UCLA (2007) The UCLA Phonetics Lab Archive, Los Angeles: UCLA Department of
- Linguistics, <u>http://archive.phonetics.ucla/edu/main2.htm</u>.
- UCLA (2007a) 'Zulu word list 1980 part 1, female speaker from Durban, South Africa, Natal Province dialect', <u>http://archive.phonetics.ucla.edu/Language/ZUL/zul.html</u> retrieved 28 January 2012.
- UCLA (2007b) 'Hindi word list 1981 part 3, male speaker', <u>http://archive.phonetics.ucla.edu/Language/HIN/hin.html</u> (retrieved 26 March 2012).
- Ussishkin, (2011) 'Tier segregation', ch. 105 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Vaissière, J. (1997) 'Phonological use of the larynx: a tutorial', *Larynx* 97, 115-116, <u>http://wenku.baidu.com/view/7164fb3987c24028915fc3e1.html</u> (retrieved 14 January 2012).
- van der Hulst, H. G. (1980) 'Towards a lexical theory of phonological change', in W. Zonneveld and F. Weerman (eds.) *Linguistics in the Netherlands 1977-79*, Dordrecht: Foris, 170-182, <u>http://linguistics.uconn.edu/pdf/012%20-%20Toward%20a%20theory</u> <u>%20of%20phonological%20change.pdf</u> (retrieved 1 March 2012).

van de Laar, Henri M. F. M. (2000) *Description of the Greek Individual Verbal Systems*.

- van Rooy, Bertus and Wissing, Daan (2001) 'Distinctive [voice] implies regressive voicing assimilation', in Hall (2001), 295-334.
- Vaux, Bert (1997) 'The phonology of voiced aspirates in the Armenian dialect of New Julfa', in N. Awde (ed.) Armenian Perspectives. 10th Anniversary Conference of the Association Internationale des Études Arméniennes, Richmond, Surrey: Curzon, 231-248.

- Vaux, Bert (1998a) 'The laryngeal specifications of fricatives', *Linguistic Inquiry* 29/3, 497-511.
- Vaux, Bert (1998b) The Phonology of Armenian, Oxford: Oxford University Press.
- Vaux, Bert (2001) 'Disharmony and derived transparency in Uyghur vowel harmony', MS, Harvard University (revised version of paper with same title published in *NELS* 30), <u>http://www.uyghurdictionary.org/excerpts/uyghur.pdf</u> (retrieved 17 October 2009).
- Vaux, Bert (2002) 'Consonant epenthesis and the problem of unnatural phonology', Yale University Linguistics Colloquium paper, 16 September 2002, http://cambridge.academia.edu/BertVaux/Papers/252895/Consonant_Epenthesis_and_t he Problem of Unnatural Phonology (retrieved 8 December 2011).
- Vaux, Bert (2007) 'Why the phonological component must be serial and rule-based', MS, University of Cambridge, <u>http://people.pwf.cam.ac.uk/bv230/OT/manifesto.pdf</u> (retrieved 6 September 2008).
- Vaux, Bert (2008) 'Why the phonological component must be serial and rule-based', in Bert Vaux and Andrew Nevins (eds.) *Rules, Constraints, and Phonological Phenomena*, Oxford: Oxford University Press, 20-61.
- Vaux, Bert (2009) 'The role of features in a symbolic theory of phonology', in Raimy and Cairns (2009), 75-98.
- Vaux, Bert (2010) 'Feature analysis', in Patrick Hogan (ed.) Cambridge Encyclopedia of the Language Sciences, Cambridge: Cambridge University Press, 304-307.
- Vaux, Bert and Brett Miller (2011) 'The representation of fricatives', ch. 28 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Vaux, Bert and Bridget Samuels (2005) 'Laryngeal markedness and aspiration', *Phonology* 22, 395–436.
- Velleman, Shelley L. and Marilyn M. Vihman (2006) 'Phonological development in infancy and early childhood: implications for theories of language learning', in M. C. Pennington (ed.), *Phonology in Context*, Luton: Macmillan, 25-50, <u>http://www.york.ac.uk/media/languageandlinguistics/documents/staff/publications/Vel</u> <u>leman&Vihman%20ch%202.pdf</u> (retrieved 3 January 2012).
- Vennemann, Theo (1989) 'Indo-European consonant shifts algebra, arithmetic, or normal science?', in Vennemann (ed.) (1989), 231-244.
- Vennemann, Theo (ed.) (1989) *The New Sound of Indo-European: Essays in Phonological Reconstruction*, Berlin: Mouton de Gruyter.
- Vennemann, Theo (2006) 'Grimm's Law and loanwords', *Transactions of the Philological Society* 104, 129-166.
- Vihman, Marilyn M., Rory A. DePaolis, and Tamar Keren-Portnoy (2009) 'Babbling and words: a dynamic systems perspective on phonological development', in E. L. Bavin (ed.) *The Cambridge Handbook of Child Language*, Cambridge: Cambridge University Press, 163-182,

http://www.york.ac.uk/media/languageandlinguistics/documents/staff/publications/Vih

man,%20DePaolis,%20Keren-Portnoy%20CUP%20chapter.pdf (retrieved 3 January 2012).

- Vihman, Marilyn M. and Tamar Keren-Portnoy (in press) 'The role of production practice in lexical and phonological development', *Journal of Child Language*, <u>http://www.york.ac.uk/media/languageandlinguistics/documents/staff/publications/CommentsVihman&Keren-Portnoy.doc</u> (retrieved 3 January 2012).
- Vihman, Marilyn M., Satsuki Nakai, and Rory A. DePaolis (2006) 'Getting the rhythm right: a cross-linguistic study of segmental duration in babbling and first words', in L. Goldstein, D. Whalen and C. Best (eds.) *Laboratory Phonology 8*, Mouton de Gruyter: New York, 341-366,

http://www.york.ac.uk/media/languageandlinguistics/documents/staff/publications/lab phon8_vihman-et-al_v4.pdf (retrieved 3 January 2012).

- Villafaña Dalcher, Christina (2006) 'Revisiting geminate inalterability: evidence from Florentine Italian', presentation, Annual Meeting of the Linguistics Society of America, Albuquerque, <u>http://homepage.mac.com/cvillafana/documents/Geminate</u> <u>%20inalterability%20handout.pdf</u>, <u>http://homepage.mac.com/cvillafana/CV.htm</u> (retrieved 16 February 2012).
- Vine, Brent (1988) 'Review of Gamkrelidze and Ivanov, *Indoevropejskij jazyk i indoevropejcy* (1984)', *Language* 64/2, 396-402.
- Walker, Rachel (2010) 'Nonmyopic harmony and the nature of derivations', *Linguistic Inquiry* 41/1, 169-179.
- Walker, Rachel, Dani Byrd, and Fidèle Mpiranya (2009) 'An articulatory view of Kinyarwanda coronal harmony', *Phonology* 25 (2008), 499-535.
- Walsh, Michael, Travis Wade, Bernd Möbius, and Hinrich Schütze (forthcoming). *Multi-level exemplar theory*, prepublication version, <u>http://www.ims.uni-stuttgart.de/~moebius/papers/walsh_etal_cogsci2010ms.pdf</u> (retrieved 20 March 2012).
- Wannemacher, Mark (2011) A Phonological Overview of the Lacid Language, Linguistics Institute, Payap University, Chiang Mai, Thailand, <u>http://www.ezraland.com/lacid%20phonological%20summary.pdf</u> (retrieved 31 January 2012).
- Warner, Natasha (n.d.) 'Acoustic characteristics of ejectives in Ingush', <u>http://www.asel.udel.edu/icslp/cdrom/vol3/273/a273.pdf</u> (retrieved 30 July 2008).
- Warren, Paul and Jen Hay (forthcoming) 'Using sound change to explore the mental lexicon', to appear in G. Haberman and C. Fletcher-Flinn (eds.) Cognition, language, and development: Perspectives from New Zealand, <u>http://www.ling.canterbury.ac.nz/jen/documents/WarrenHay.pdf</u> (retrieved 22 May 2010).
- Warren, Paul, Jennifer Hay, and Brynmor Thomas (forthcoming) 'The loci of sound change effects in recognition and perception', to appear in *Laboratory Phonology 9*, <u>http://www.ling.canterbury.ac.nz/jen/documents/warren-hay-thomas.pdf</u> (retrieved 22 May 2010).

- Watson, Janet C. E. (2002) *The Phonology and Morphology of Arabic*, Oxford: Oxford University Press.
- Watters, David E. (2006) *Notes on Kusunda: A Language Isolate of Nepal*, Himalayan Linguistics Archive 3, University of California, Santa Barbara, <u>http://www.linguistics.ucsb.edu/HimalayanLinguistics/grammars/2006/HLA03_Watters.pdf</u> (retrieved 29 July 2008).
- Wayland, Ratree and Susan Guion (2002a) 'Perceptual discrimination of Thai tones by naïve and experienced learners of Thai', *SEALS XII: Papers from the 12th Annual Meeting of the Southeast Asian Linguistics Society*, 137-44 http://pacling.anu.edu.au/catalogue/SEALSXII Final.pdf (retrieved 28 July 2008).
- Wayland, Ratree and Susan Guion (2002b) 'Tonogenesis in Khmer', *SEALS XII: Papers from the 12th Annual Meeting of the Southeast Asian Linguistics Society*, 145http://pacling.anu.edu.au/catalogue/SEALSXII Final.pdf (retrieved 28 July 2008).
- Wayland, Ratree and Allard Jongman (2002) 'Registrogenesis in Khmer: a phonetic account', *Mon-Khmer Studies* 32, 101-115, <u>http://sealang.net/sala/archives/pdf8/wayland2002registronesis.pdf</u> (retrieved 24 January 2012).
- Wayland, Ratree and Allard Jongman (2003) 'Acoustic correlates of breathy and clear vowels: the case of Khmer', *Journal of Phonetics* 31, 181-201, <u>http://www2.ku.edu/~kuppl/jongman/Wayland%20%26%20Jongman%2003.pdf</u> (retrieved 16 January 2012).
- Wedekind, F. (1990) 'Glottalization constraints and Ethiopian counter-evidence', *Folia Linguistica* 24, 127-37.
- Wedel, Andrew (2011) 'Self-organization in phonology', ch. 6 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Weijer, Jeroen van der (2004) 'Complex segments: feature geometry and segment inventories', presentation handout, Leipzig Colloquium 2004, <u>http://132.229.7.185/letteren/onderzoek/ulcl/content_docs/faculty/vdweijer/leipzig200</u> <u>4.pdf</u> (retrieved 23 November 2009).
- Weijer, Jeroen van der and Frans Hinskens (2004) 'Segmental markedness and complexity a cross-linguistic study of consonantal inventories', presentation, Generative Linguistics in the Old World (GLOW) conference, 18 April 2004.
- Weiss, Bradley G (2005) 'Development of PIE voiced stops into Italic: evidence from **zC*- medial clusters', paper, Greek and Latin from an Indo-European Perspective conference, University of Cambridge, 8-9 July 2005.
- Weiss, Michael (2009) 'The Cao Bang Theory: some speculations on the prehistory of the PIE stop system', presentation at East Coast Indo-European Conference XXVIII, University of Iceland, Reykjavík, 13 June 2009,

http://conf.ling.cornell.edu/weiss/Cao_Bang_Theory.pptx (retrieved 2 January 2012).

Weitenberg, J. J. S. (2002) 'Aspects of Armenian dialectology', in Johannes Bernardus Berns and Jaap van Marle (eds.) *Present-day Dialectology: Problems and Findings*, Trends in Linguistics Studies and Monographs 137, Berlin: Mouton de Gruyter, 141-158.

- Wescott, Roger W. (1988) 'Derogatory use of the marginal phoneme /b/ in Proto-Indo-European', *Journal of Indo-European Studies* 16, 365-369.
- Westbury, J. R. (1983) 'Enlargement of the supraglottal category and its relation to stop consonant voicing', *Journal of the Acoustical Society of America* 73, 1322-1336.
- Wetzels, W. Leo and Joan Mascaró (2001) 'The typology of voicing and devoicing', *Language* 77/2, 207-44.
- Whalen, D. H. and Andrea G. Levitt (1994) 'The universality of intrinsic F0 in vowels', Haskins Laboratories Status Report on Speech Research SR-117/118, 1-14, <u>http://www.haskins.yale.edu/sr/SR117/SR117_01.pdf</u> (retrieved 18 February 2012).
- Whitney, William Dwight (1913[1896]) *Sanskrit Grammar*, fourth edition reprint, Leipzig: Breitkopk and Härtel, Google Books download.
- Whorf, Benjamin Lee (1937) 'The origin of Aztec *tl*', *American Anthropologist* 39, 265-74.
- Whorf, Benjamin, Campbell, Lee, and Karttunen, Frances (1993) 'Pitch tone and the 'saltillo' in modern and ancient Nahuatl', *International Journal of American Linguistics* 59/2, 165-223, <u>http://www.jstor.org/stable/pdfplus/1265154.pdf</u> (retrieved 24 July 2008).
- Wiese, Richard (2001) 'The phonology of /r/', in Hall (2001).
- Williamson, Kay (1965) *A Grammar of the Kolokuma Dialect of Ijo*, West African Language Monograph Series No. 2, Cambridge: Cambridge University Press.
- Williamson, Kay (1979) 'Consonant distribution in Ijo', in Mohammed Ali Jazayery, Edgar C. Polomé, and Werner Winter (eds.) *Linguistic and Literary Studies in Honor* of Archibald A. Hill, Trends in Linguistics: Studies and Monographs no. 8-10, Lisse: Peter de Ridder Press, 341-53.
- Williamson, Kay (2004) 'Implosives in Mande-Atlantic-Congo', MS, University of Port Harcourt, <u>http://sumale.vjf.cnrs.fr/NC/Public/pdf/williamson1.pdf</u> (retrieved 29 July 2008).
- Willis, Erik W. (2007) 'An acoustic study of the 'pre-aspirated trill' in narrative Cibaeno Dominican Spanish', *Journal of the International Phonetic Association* 37/1, 33-49, <u>http://journals.cambridge.org/action/displayAbstract?fromPage=online&aid=995692</u> (retrieved 29 July 2008).
- Winter, Werner (ed.) (1965) Evidence for Laryngeals, The Hague: Mouton, 190-211.
- Winter, Werner (1978) 'The distribution of short and long vowels in stems of the type Lith. ésti: vèsti : mèsti : and OCS jasti : vesti : mesti in Baltic and Slavic Languages' in J. Fisiak (ed.), Recent Developments in Historical Phonology (The Hague), 431-446.
- Witczak, Krzysztof Tomasz (1993) 'A *b*-series in Linear B', *Kadmos* 32, 162-71, <u>http://www.deepdyve.com/lp/de-gruyter/a-b-series-in-linear-b-MO0KlblBdb</u> (retrieved 18 February 2012).

- Wolf, Matthew A. (2008) Optimal Interleaving: Serial Phonology-Morphology Interaction in a Constraint-Based Model, PhD dissertation, University of Massachusetts at Amherst, ROA, <u>http://roa.rutgers.edu/files/996-1008/996-WOLF-0-0.PDF</u> (retrieved 17 January 2009).
- Wolff, John U. (1997) Review of *Comparative Austronesian Dictionary* in *Language* 73/1, 145-56.
- Woodbury, Hanni (1981) 'The loss of a phoneme', *International Journal of American Linguistics* 47/2,103-120.
- Woodhouse, Robert (1993) 'A reformulation of quasi-Gamkrelidzean occlusive typology and its more fruitful application to Indo-European', *Indogermanische Forschungen* 98, 1-12
- Woodhouse, Robert (1995) 'Proto Indo-European Injective Asperes', *Indogermanische Forschungen* 100, 92-100.
- Woodhouse, Robert (1998) 'Some criticisms of Charles Schleicher's PIE. obstruent chronology', *Indogermanische Forschungen* 103, 61-69.
- Woods, D. L., T. J. Herron, A. D. Cate, X. Kang, and E. W. Yund (2011) 'Phonological processing in human auditory cortical fields', *Frontiers in Human Neuroscience* 5/42, <u>http://www.frontiersin.org/human_neuroscience/10.3389/fnhum.2011.00042/abstract</u> (retrieved 13 March 2012).
- Wright, Richard (2004) 'A review of perceptual cues and cue robustness', in Bruce Hayes, Robert Kirchner, and Donca Steriade (eds.) *Phonetically based phonology*, Cambridge: Cambridge University Press, 34-57, http://www.ucalgary.ca/dflynn/files/dflynn/Wright04.pdf (retrieved 1 December 2011).
- Wright, Jonathan D. (2007) *The Phonetic Contrast of Korean Obstruents*, PhD dissertation, University of Pennsylvania,

http://languagelog.ldc.upenn.edu/myl/Wright083107.pdf (retrieved 25 January 2011).

- Yadav, Ramawatar (1980) 'The influence of aspiration on vowel duration in Maithili', *Contributions to Nepalese Studies* 7/1-2, 1-8, <u>http://himalaya.socanth.cam.ac.uk/collections/journals/contributions/pdf/CNAS_07_0</u> 102 01.pdf (retrieved 14 January 2012).
- Yamane-Tanaka, Noriko (2005) 'The implicational distribution of prenasalized stops in Japanese', 123-56 in Jeroen Maarten van de Weijer, Kensuke Nanjo, and Tetsuo Nishihara (eds.) *Voicing in Japanese*.
- Yip, Moira Jean Winsland (2002) Tone, Cambridge: Cambridge University Press.
- Yip, Moira (2003) 'Variability in feature affiliations through violable constraints: the case of [lateral]', paper, Old World Conference in Phonology I, Leiden University, 9-11 January 2003, <u>http://www.let.leidenuniv.nl/ulcl/events/ocp1/abstracts/yip.pdf</u> (retrieved 5 September 2008).
- Yip, Moira (2011) 'Laterals', ch. 31 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.

- Yoneyama, Kyomo, Mary E. Beckman, and Jan Edwards (2003) 'Phoneme frequencies and acquisition of lingual stops in Japanese', MS, The Ohio State University.
- Yoshioka, Hirohide, Anders Löfqvist, and Hajime Hirose (1979) 'Laryngeal adjustments in the production of consonant clusters and geminates in American English', Haskins Laboratories Status Report on Speech Research SR-59/60, http://www.haskins.vale.edu/sr/SR059/sr059_08.pdf (retrieved 11 February 2012).
- Yoshioka, Hirohide, Anders Löfqvist, and Hajime Hirose (1980) 'Laryngeal adjustments in Japanese voiceless sound production', Haskins Laboratories Status Report on Speech Research SR-63/64, 293-308,

http://www.haskins.yale.edu/sr/SR063/SR063_17.pdf (retrieved 18 February 2012).

- Yu, Alan C. L. (2004) 'Explaining final obstruent voicing in Lezgian: phonetics and history', <u>http://washo.uchicago.edu/pub/Language80.pdf</u> (retrieved 30 July 2008).
- Yu, Alan C. L. (2011) 'Mergers and neutralization', ch. 80 in Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume, and Keren Rice (eds.) *The Blackwell Companion to Phonology*, Oxford: Wiley-Blackwell.
- Yu, Y. and Y. Choe (2006) 'A neural model of the scintillating grid illusion: disinhibition and self-inhibition in early vision', *Neural Computation* 18/3, 521-544, <u>http://www.ncbi.nlm.nih.gov/pubmed/16483406</u> (retrieved 23 March 2012).
- Zhang, Caicai (2009) 'A phonetic study of the syllabic constituents in Lengshuijiang dialect', M.Phil. Thesis, Hong Kong University of Science and Technology, <u>http://repository.ust.hk/dspace/handle/1783.1/6410</u> (retrieved 31 January 2012).
- Zec, Draga (2007) 'The syllable', in de Lacy (2007), 161-94.
- Zeng, Ting (2008) "Voicing"-tone interaction in Xiangxiang Chinese', *Toronto Working Papers in Linguistics* 28, 395-405, <u>http://twpl.library.utoronto.ca/index.php/twpl/article/view/6155</u> (retrieved 13 December 2011).
- Zhou, Xuewen, Yuling Zheng, and Zhenyu Chuai (2010) 'Phonation type of Korean stops', Proceedings of Oriental COCOSDA 2010 (The International Committee for the Co-ordination and Standardization of Speech Databases and Assessment Techniques), paper 6, pp. 1-6, <u>http://desceco.org/O-COCOSDA2010/proceedings/paper_6.pdf</u> (retrieved 31 January 2012).